Feeling Sad, Angry, or Scared: An Investigation of Gender Differences in the Regulation of Specific Emotions, and their Relationship with Variables Associated with Mental Health

Michelle Patrick

University of Strathclyde School of Psychological Sciences and Health Doctor of Philosophy 2021

Author's Declaration

This thesis is the result of the author's original research. It has been composed by the author and has not been previously submitted for examination which has led to the award of a degree.

The copyright of this thesis belongs to the author under the terms of the United Kingdom Copyright Acts as qualified by University of Strathclyde Regulation 3.50. Due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

Signed: M. Patrich

Date: 30th September 2021

Acknowledgements

If someone were to ask me if I'm glad that I decided to do a PhD, I would tell them that it is the hardest thing I've ever done, but I'm so thankful that I chose to do it. It has been such an amazing journey, and has included my hardest days, but also some of my happiest. This experience has forced me to look at the self-doubts that hold me back, and to overcome them, every single day. It has taught me what it means to feel anxious, but to do the thing anyway. I know that this is the beginning of many important things I will do in my life. By far, the best part of the PhD journey has been the extraordinary people I've met along the way.

I would like to thank my supervisors, Susan and Marc, for all of their guidance and support. Susan, thank you for taking a chance on me when I came to you in my second year, in a bit of a shambles. I know for certain that I would not have succeeded if it weren't for your strength and support. Thank you for believing that I could do it, when I was so certain that I couldn't. Marc, thank you for being so kind and accepting, even when I made mistakes, and for your wisdom, advice and support. You are truly one of the kindest and most intelligent people I know, and I feel so lucky that I got to work with you.

Jo and Sinead – thank you for all of your support throughout my first year. Lynn, thank you for being such a supportive panel member during my reviews. Louise, I appreciate you coming on board as my second supervisor for the final part of my PhD, and offering a fresh perspective on my research during my review meetings. Thank you to Helen, David, Ingrid, and all of the staff at the Graduate School for your support.

A special thank you to all of the participants who took part in my research – secondary school pupils, university students, Scottish Government staff, friends and family, and members of the public. Scientific research is built on your kindness.

To all of my fellow PhD students – you are my teammates in this journey, and I would not have been able to do it without you. For the 'old school' PhD students, you guys are the wisest and funniest people I know - I learned so much from you and will continue to do so.

Thank you to all of my wonderful colleagues at the Scottish Government – I'm so lucky to work with such extraordinary people, and I can't thank you enough for your support.

Finally, I want to thank my family and friends for coming on this journey with me – Mum, Dad, Sarah and Robert, Uncle Graeme, Granny, Grandad, all the Mochries, Olivia, Alex, and beautiful Remi, Rachel B, Greig, Kim, Rachel M, Emma, Kirsten, Gail, and Michael.

I also want to thank my puppy Belle, for her cuddles and unconditional love. Dogs are not particularly concerned with whether or not you have a PhD, and for that I'm grateful. Also thank you to my little Daisy, who lay beside me while I wrote many chapters, but sadly wasn't here for the last one.

A special thank you to my nephew Liam for being such a light in the darkest times. You made me laugh so much during the most difficult parts of this journey, and I cherish the time we got to spend together this past year. You are very special, and you and your sister are my favourite people in the whole world. To my new baby niece Anna, I can't wait to get to know you better and have lots of cuddles with you.

Andrew, thank you for all of your love and support. You're so special to me, and I'm excited for all of our future adventures. I know that we're going to do so many amazing things together.

I'd like to dedicate this work to my Mum and Dad - thank you for being there for me, for understanding how hard it's been, and for going through it all with me. Without your love and support, there is no way I would have been able to achieve this.

Table of Contents

Author's Declaration	1
Acknowledgements	2
Table of Contents	4
List of Abbreviations	14
List of Tables	15
List of Figures	17
Abstract	19
Chapter 1: Emotions and Emotion Regulation	21
1.1. Chapter Summary	21
1.2. Emotions	21
1.2.1. The Multi-Component Definition of Emotion	22
1.2.2. Theories of Emotion	24
1.2.2.1. Discrete Emotion Theories	25
1.2.2.2. Dimensional/Constructivist Theories	27
1.2.2.3. Appraisal Theories	. 29
1.2.3. Evidence for Emotion-Specificity	.31
1.2.3.1. Facial Expression Research	32
1.2.3.2. Human Vocalisation Research	32
1.2.3.3. Meta-Analysis of Studies Examining Emotional	
Components	33
1.2.3.4. Bridging the Gap Between Discrete and Dimensional	
Perspectives	33
1.2.3.5. Appraisal Research	35
1.2.3.6. Summary of Evidence for Emotion-Specificity	.36
1.3. Emotion Regulation	37
1.3.1. Definition of Emotion Regulation	37
1.3.2. Theories of Emotion Regulation	41
1.3.2.1. Gross's (1998) Process Model of Emotion Regulation	41
1.3.2.2. Gross's (2015) Extended Process Model of Emotion	
Regulation	42
1.3.2.3. Dual Process Model of Emotion Regulation	44

1.3.2.4. Parkinson and Totterdell's (1999) Taxonomy of Affect
Regulation45
1.3.3. Emotion-Specific Emotion Regulation
1.3.3.1. Research Findings of Emotion-Specific ER 47
1.3.3.2. Why Sadness, Anger, and Fear? 50
1.3.4. Distinction Between Emotion Generation and Emotion Regulation52
1.3.5. Why Is It Important To Investigate Gender Differences in ER? 52
1.4. Conclusion and Next Steps55
Chapter 2: Gender Differences in Emotion Regulation and Mental Health 58
2.1. Chapter Summary
2.2. Gender Differences in Emotion Regulation
2.2.1. Sex and Gender
2.2.2. Gender and Emotion
2.2.3. Gender Differences in Emotion Regulation
2.2.3.1. Gender Differences in Coping
2.2.3.2. Gender Differences in Reappraisal
2.2.3.3. Gender Differences in Suppression
2.2.3.4. Gender Differences in Rumination
2.2.3.5. Gender Differences in Distraction
2.2.3.6. Gender Differences in Engagement and Disengagement
ER
2.2.3.7. Other Gender Differences in Emotion Regulation
2.2.3.8. Gender Differences in Emotion-Specific Emotion
Regulation
2.2.3.9. Gender Differences in Emotion Regulation Summary71
2.3. Gender, Emotion Regulation, and Mental Health71
2.3.1. Definition of Variables Associated with Mental Health
2.3.2. Gender Differences in Variables Associated with Mental Health74
2.3.2.1. Gender Differences in Depression
2.3.2.2. Gender Differences in Self-harm and Suicidal Ideation 75
2.3.3. The Relationship Between Emotion Regulation and Variables
Associated with Mental Health78
2.3.3.1. Reappraisal and Mental Health

2.3.3.2. Suppression and Mental Health	80
2.3.4. Explaining Gender Differences in Depression: Response Styles	
Theory	82
2.3.4.1. Rumination and Mental Health	. 84
2.3.4.2. Distraction and Mental Health	86
2.3.4.3. Summary of Response Styles Theory	87
2.4. Theoretical Basis for Empirical Studies	88
2.5. Conclusion and Next Steps	. 90
Chapter 3: Methodological Approaches to Emotion Regulation and the Methods Use	ed
in the Present Research	. 92
3.1. Chapter Summary	. 92
3.2. The Importance of Measuring Emotion Regulation in an Emotion-Specific	
Manner	.93
3.3. Key Methods for Measuring Emotion Regulation	. 94
3.3.1. Emotion Regulation Frequency	94
3.3.2. Emotion Regulation Effectiveness	98
3.4. The Importance of Measuring Subjective Experience in Emotion Research	. 102
3.4.1. Measuring Physiological Arousal	103
3.4.2. Measuring Emotional Behaviour	. 103
3.4.3. Measuring Subjective Experience	.104
3.4.4. Summary of Measuring Emotional Components	106
3.5. Research Questions and Methods of the Present Research	. 107
3.5.1. Research Questions Used to Guide the Present Research	107
3.5.2. Methods Used in the Present Research	109
3.5.2.1. Study 1: Open-Ended Responses	109
3.5.2.2. Study 2: Regulating Sadness, Anger, and Fear in the Emot	ion
Regulation Task	111
3.6. Conclusion and Next Steps	. 115

Chapter 4: Gender Differences in the Reporting of Emotion Regulation Strategies to	
Regulate Sadness, Anger, and Fear (Study 1)	
4.1. Chapter Summary	
4.2. Introduction	118

4.2.1. Rationale for Study 1	. 118
4.2.2. Research Questions for Study 1	121
4.3. Method	. 121
4.3.1. Participants	.121
4.3.2. Age Groups	123
4.3.3. Procedure	. 124
4.3.4. Materials	. 125
4.3.5. Coding and Analysis	126
4.3.5.1. Coding the Data	126
4.3.5.2. Analytical Approaches for Exploring the Research Questic	ons
of the Present Study	132
4.3.5.2.1. Approach for Testing Emotion-Specific Difference	ces
in Reported ER Strategies	132
4.3.5.2.2. Approach for Testing Gender Differences in	
Individual ER Strategies	133
4.3.5.2.3. Approach for Testing Gender Differences in Wid	er
ER Categories	. 134
4.4. Results	.135
4.4.1. Data Screening and Assumptions	135
4.4.1.1. Missing Data	135
4.4.1.2. Assumptions of Chi-Square Analysis: Cell Counts	138
4.4.2. The ER Strategies Reported for Specific Emotions	138
4.4.3. Gender Differences in Emotion Regulation Strategies	.142
4.4.4. The Relationship Between Gender and Engagement, Disengagemen	t,
and Distraction	.144
4.4.4.1. Main Effect of Gender	144
4.4.4.2. Gender By Age Interaction	144
4.4.5. Summary of Results	147
4.5. Discussion	.147
4.5.1. Results of Study 1	.147
4.5.2. What Are The Key ER Strategies That Are Reported For Sadness,	
Anger, and Fear, and Do These Strategies Vary By Emotion?	149
4.5.3. Are There Gender Differences in the Reporting of Specific ER	
Strategies, and Do These Vary by Emotion?	153

4.5.3.1. Reporting of Strategies Using the Free-Writing Approach.	. 153
4.5.3.2. Social Support	154
4.5.3.3. Expressing or Venting Emotion	154
4.5.3.4. Reappraisal	. 156
4.5.3.5. Listening to Music	157
4.5.4. Are There Gender Differences in Broad Engagement, Disengageme	nt,
and Distraction ER Categories?	.158
4.5.5. Limitations of this Study	162
4.5.6. Implications of Findings	. 165
4.5.6.1. Theoretical Implications	. 165
4.5.6.2. Practical Implications	.167
4.5.7. Directions for Future Research	169
4.6. Conclusion and Next Steps	. 171

Chapter 5: Gender Differences in the Regulation of Sadness, Anger, and Fear due	ring an
Emotion Regulation Task (Study 2a)	174
5.1. Chapter Summary	174
5.2. Introduction	175
5.2.1. Rationale for Study 2a	175
5.2.2. Research Questions for Study 2a	179
5.3. Method	179
5.3.1. Participants	180
5.3.2. Materials	181
5.3.2.1. Session 1: Self-Report Measures	181
5.3.2.1.1. Demographic Information	182
5.3.2.1.2. Measures of Emotion Regulation	182
5.3.2.2. Session 2: Emotion Regulation Task (ERT)	184
5.3.2.2.1. Task Procedure	184
5.3.2.2.2. Emotion Regulation Effectiveness	187
5.3.2.2.3. Task Instructions	187
5.3.2.2.4. Emotional Stimuli	188
5.3.3. Procedure	189
5.3.3.1. Session 1: Online	189
5.3.3.2. Session 2: Experiment	190

5.3.4. Analysis 191
5.3.4.1. Approach for Testing Gender Differences in Emotion-General
ER (SRQ1)191
5.3.4.2. Approach for Testing Whether the Effectiveness of ER
Depends on the Specific Emotion Being Regulated (SRQ2) 191
5.3.4.3. Approach for Testing Gender Differences in the Regulation of
Sadness, Anger, and Fear (SRQ3) 192
5.4. Results
5.4.1. Data Screening and Assumptions 193
5.4.1.1. Missing Data 193
5.4.1.2. Outliers
5.4.1.3. Normality and Linearity 194
5.4.1.4. Homoscedasticity 194
5.4.1.5. Multicollinearity
5.4.1.6. Power
5.4.2. Sample Characteristics: Comparing the Student Participants to the Rest
of the Sample 196
of the Sample 196 5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-
-
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self- Report Questionnaires
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self- Report Questionnaires
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self- Report Questionnaires
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires. 196 5.4.4. Emotion Regulation Task (ERT) 198 5.4.4.1. Emotion Manipulation Check in the ERT. 198 5.4.4.2. Descriptive Statistics: Self-Reported Levels of Emotion
 5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires. 5.4.4. Emotion Regulation Task (ERT) 5.4.4.1. Emotion Manipulation Check in the ERT. 5.4.4.2. Descriptive Statistics: Self-Reported Levels of Emotion During the ERT and Computed ER Effectiveness Scores.
 5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires. 196 5.4.4. Emotion Regulation Task (ERT) 198 5.4.4.1. Emotion Manipulation Check in the ERT. 198 5.4.4.2. Descriptive Statistics: Self-Reported Levels of Emotion 198 5.4.5. Testing Gender Differences in the Regulation of Sadness, Anger, and 198 5.4.5. Testing ANOVA. 201
 5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires. 196 5.4.4. Emotion Regulation Task (ERT) 198 5.4.4. Emotion Regulation Task (ERT) 198 5.4.4.1. Emotion Manipulation Check in the ERT. 198 5.4.4.2. Descriptive Statistics: Self-Reported Levels of Emotion 198 5.4.5. Testing Gender Differences in the Regulation of Sadness, Anger, and 198 5.4.5.1. Model 1: Testing Gender Differences in the Regulation of Sadness. 201 5.4.5.2.1. Model 1: Testing Gender Differences in the Regulation of Sadness. 201
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires. 196 5.4.4. Emotion Regulation Task (ERT) 198 5.4.4.1. Emotion Manipulation Check in the ERT. 198 5.4.4.2. Descriptive Statistics: Self-Reported Levels of Emotion 198 5.4.5. Testing Gender Differences in the Regulation of Sadness, Anger, and 198 5.4.5.1. Model 1: Testing Gender Differences in the Regulation of Sadness. 201 5.4.5.2. Model 2: Testing Gender Differences in the Regulation of 201
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires. 196 5.4.4. Emotion Regulation Task (ERT) 198 5.4.4. Emotion Regulation Task (ERT) 198 5.4.4.1. Emotion Manipulation Check in the ERT. 198 5.4.4.2. Descriptive Statistics: Self-Reported Levels of Emotion 198 5.4.5. Testing Gender Differences in the Regulation of Sadness, Anger, and 198 5.4.5.1. Model 1: Testing Gender Differences in the Regulation of Sadness. 201 5.4.5.2. Model 2: Testing Gender Differences in the Regulation of Anger. 202
5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires. 196 5.4.3. Emotion Regulation Task (ERT) 198 5.4.4. Emotion Regulation Task (ERT) 198 5.4.4. Emotion Manipulation Check in the ERT. 198 5.4.4.2. Descriptive Statistics: Self-Reported Levels of Emotion 198 5.4.5. Testing Gender Differences in the Regulation of Sadness, Anger, and 198 5.4.5.1. Model 1: Testing Gender Differences in the Regulation of 201 5.4.5.2. Model 2: Testing Gender Differences in the Regulation of 201 5.4.5.3. Model 3: Testing Gender Differences in the Regulation of 202

5.5.2. Gender Differences in Emotion-General Emotion Regulation:	
Comparison to the Wider Literature 204	4
5.5.3. Comparing the Effectiveness of Reappraisal and Distraction to Regulate)
Specific Emotions	8
5.5.4. Gender Differences in Reappraisal Effectiveness for Sadness, Anger,	
and Fear During the ERT210	0
5.5.5. Gender Differences in Distraction Effectiveness for Sadness, Anger, and	d
Fear During the ERT 212	3
5.5.6. Limitations of This Study 217	7
5.5.7. Implications of Findings 219	9
5.6. Conclusion and Next Steps	0

Chapter 6: The Regulation of Sadness, Anger, and Fear as Mediators of Gender	
Differences in Variables Associated with Mental Health (Study 2b)	223
6.1. Chapter Summary	
6.2. Introduction	224
6.2.1. Rationale for Study 2b	
6.2.2. Research Questions for Study 2b	229
6.3. Method	
6.3.1. Measures of Variables Associated with Mental Health	229
6.3.1.1. Centre for Epidemiologic Studies Depression	
Scale (CES-D)	229
6.3.1.2. Warwick-Edinburgh Mental Wellbeing	
Scale (WEMWBS)	230
6.3.1.3. Deliberate Self-Harm Inventory (DSHI)	
6.3.1.4. Suicidal Ideation Attributes Scale (SIDAS)	
6.3.2. Analysis	
6.4. Results	235
6.4.1. Data Screening and Assumptions	235
6.4.1.1. Outliers	235
6.4.1.2. Normality and Linearity	235
6.4.1.3. Homoscedasticity	236
6.4.1.4. Multicollinearity	
6.4.1.5. Power	

6.4.2. Descriptive Statistics	237
6.4.2.1. Mental Health Characteristics of the Sample 2	237
6.4.2.2. Gender Differences in the Variables Associated with Mental	l
Health2	238
6.4.2.3. Correlation With Age 2	239
6.4.2.4. Correlation Between the Variables Associated with Mental	
Health 2	239
6.4.2.5. Comparing the Mental Health of Student Participants to the	
Rest of the Sample 2	240
6.4.3. Mediation Analysis2	240
6.4.3.1. Using Mediation Analysis When X Does Not Predict Y2	240
6.4.3.2. Controlling for Covariates in the Mediation Models2	241
6.4.4. Main analysis: Testing the Mediation Models	243
6.4.4.1. Model 1: Gender, Reappraisal, and Depressive Symptoms2	243
6.4.4.2. Model 2: Gender, Distraction, and Depressive Symptoms 2	245
6.4.4.3. Model 3: Gender, Reappraisal, and Suicidal Ideation 2	46
6.4.4.4. Model 4: Gender, Distraction, and Suicidal Ideation 2	247
6.4.4.5. Model 5: Gender, Reappraisal, and Self-Harm Behaviour 2	249
6.4.4.6. Model 6: Gender, Distraction, and Self-Harm Behaviour2	250
6.5. Discussion	252
6.5.1. Results of Study 2b 2	252
6.5.2. The Relationship Between Gender and Variables Associated with	
Mental Health2	252
6.5.2.1. Depressive Symptoms2	253
6.5.2.2. Self-Harm Behaviour and Suicidal Ideation 2	254
6.5.3. The Relationship Between Gender, Distraction of Anger, and	
Depression2	57
6.5.4. The Relationship Between Gender, Distraction of Anger, and	
Suicidal Ideation2	261
6.5.5. Reappraisal of Anger, Self-Harm Behaviour, and Suicidal Ideation2	264
6.5.6. Implications of Findings 2	265
6.5.7. Limitations of This Study 2	266
6.6. Conclusion and Next Steps	267

Chapter 7: General Discussion270	D
7.1. Summary of Research270)
7.2. Emotion Regulation May Differ According to the Specific Emotion Being	
Regulated272	2
7.2.1. Implications for Discrete Emotion Theory	3
7.2.2. Developing Emotion-Specific ER Instruments	3
7.2.3. Conducting Emotion-Specific Research in a More	
Naturalistic Setting 270	5
7.3. Gender Differences in Emotion Regulation Often Depend on the Specific	
Emotion278	8
7.3.1. Gender Differences in Reappraisal 278	8
7.3.2. Gender Differences in Distraction)
7.3.3. Gender Differences in Expressing Sadness and Fear	1
7.3.4. Similarities Between Males and Females	2
7.3.5. Potential Vulnerability of Males	3
7.3.6. The Role of Culture	4
7.3.7. Conducting Emotion-Specific ER Research Amongst Other Groups284	4
7.4. Gender Differences May Exist in Engagement Emotion Regulation280	5
7.4.1. Limitations of the Parkinson & Totterdell Framework	7
7.4.2. Examining Engagement ER in Future Research	7
7.4.3. Limitations of the Adult Sample in Study 1	8
7.5. The Effectiveness of Reappraisal and Distraction May Depend on the Specific	
Emotion Being Regulated	9
7.5.1. Implications for the Process Model of ER	9
7.5.2. Implications for Emotion Theory	2
7.5.3. Implications for Therapeutic Interventions	3
7.6. The Relationships Between Gender, Emotion-Specific ER, and Variables	
Associated With Mental Health	ł
7.6.1. 7.6.1. Distracting From Anger is Associated with Gender Differences in	t
Depressive Symptoms	5
7.6.2. Distracting from Anger is Associated with Gender Differences in	
Suicidal Ideation	7
7.6.3. Why Does Anger Distraction Have a Different Relationship with	
Depressive Symptoms and Suicidal Ideation?	9

7.6.4. Implications for Personalising Mental Health Interventions
7.6.5. Testing Causal Relationships Between Gender, ER, and Variables
Associated with Mental Health Prospectively
7.7. Some Further Methodological Considerations of this Research
7.7.1. Sampling Methodology
7.7.2. Applying the Results of the Emotion Regulation Task to a Real-World
Setting
7.7.3. Range of Specific Emotions
7.7.4. The Role of Positive Emotion
7.8. Conclusion
References
Appendices
Appendix A: Study 1 Information Sheet
Appendix B: Study 1 Consent Form
Appendix C: Study 1 Demographic Questions
Appendix D: Study 1 Dealing with Feelings Questionnaire
Appendix E: Study 1 Velten Mood Induction
Appendix F: Debriefing Sheet
Appendix G: Study 1 ER Strategies Included in Each ER Category
Appendix H: Study 1 Cross-Tabulations used in Chi-Square Analyses
Appendix I: Study 2 Information Sheet402
Appendix J: Study 2 Consent Form410
Appendix K: Study 2 Demographic Questions
Appendix L: Study 2 Emotion-General ER Questionnaires413
Appendix M: Study 2 Variables Associated with Mental Health Questionnaires416
Appendix N: Study 2 Other Questionnaires
Appendix O: Study 2 Velten Mood Induction426
Appendix P: Study 2 Debriefing Sheet427

Abbreviation	Explanation
ABM	Attentional Bias Modification
Ad	Adults
ANCOVA	Analysis of Covariance
ANOVA	Analysis of Variance
ANS	Autonomic Nervous System
BERQ	Behavioural Emotion Regulation Questionnaire
CBT	Cognitive Behavioural Therapy
CERQ	Cognitive Emotion Regulation Questionnaire
CES-D	Centre for Epidemiological Studies Depression Scale
CI	Confidence Interval
DBT	Dialectical Behavioural Therapy
DERS	Difficulties in Emotion Regulation Scale
DSHI	Deliberate Self-Harm Inventory
DV	Dependent Variable
EEG	Electroencephalography
EPM	Extended Process Model of Emotion Regulation
ER	Emotion Regulation
ERI	Emotion Regulation Inventory
EROS	Emotion Regulation of Others and Self Scale
ERP	Event-Related Potential
ERQ	Emotion Regulation Questionnaire
ERT	Emotion Regulation Task
fEMG	Facial Electromyography
fMRI	Functional Magnetic Resonance Imaging
HC3	Heteroscedastic-Consistent Standard Error Estimator
IAPS	International Affective Picture System
IV	Independent Variable
LLP	Late Positive Potential
MCSDS	Marlowe-Crowne Social Desirability Scale
MIP	Mood Induction Procedure
NAPS	Nencki Affective Picture System
NERI	Negative Emotion Regulation Inventory
NSSI	Non-Suicidal Self Injury
OA	Older Adolescents
OR	Odds Ratio
PAQ	Personal Attributes Questionnaire
REQ	Regulation of Emotions Questionnaire
RQ	Research Question
RRS	Ruminative Response Scale
RST	Response Styles Theory
SD	Social Desirability
SES	Socioeconomic Status
SIDAS	Suicidal Ideation Attributes Scale
WEMWBS	Warwick-Edinburgh Mental Well-Being Scale
YA	Younger Adolescents
111	i ounger Aubrescents

List of Tables

Table 1.1. Common ER Strategies in the Literature40
Table 3.1. Popular Self-Report Questionnaires Used to Measure ER
Table 3.2. Overarching Research Questions and the Studies That Address Them
Table 4.1. Mean (M) Age, Standard Deviation (SD) and Sample Size (n) for each Gender and Age Group.122
Table 4.2. Mean (<i>M</i>) Age, Standard Deviation (<i>SD</i>), <i>P</i> -Values, and Direction of Relationships for ANOVA and Post-Hoc Tests for Differences in Age Between the Groups124
Table 4.3. Definitions Used to Code the Data Items into ER Strategies in Study 1128
Table 4.4. Top ER Strategies That Make up the Wider ER Categories of Engagement,Disengagement and Distraction
Table 4.5. Missing Data for Each Variable Broken Down by Age Group, Showing theNumber (n) And Percentage (%) Of Missing Values
Table 4.6. Missing Data for Each Variable Broken Down by Gender, Showing the Number(n) And Percentage (%) Of Missing Values
Table 4.7. Frequency (Freq) Of ER Strategies Reported, and Percentage (%) Of the TotalStrategies Reported for Males and Females, and for Sadness, Anger, and Fear
Table 4.8. Cochran's Q Tests Comparing the Percentage of Participants Reporting Each ERStrategy for Sadness, Anger, and Fear.141
Table 4.9. Results of Chi-Square Analyses Testing the Relationship Between Gender (Males, Females), and Reporting an ER Strategy (Yes, No), for Sadness, Anger, and Fear143
Table 4.10. Results of Chi-Square Analyses Testing the Relationship Between Gender (Males, Females) And Each Category of ER – Engagement, Disengagement, and Distraction (Yes, No) Across All Age Groups, for Sadness, Anger, and Fear
Table 4.11. Results of Chi-Square Analyses Testing the Relationship Between Gender (Males, Females) And Each Category of ER – Engagement, Disengagement, and Distraction (Yes, No), Broken Down by Age Group, for Sadness, Anger, and Fear
Table 4.12. Evidence for Study 1148
Table 5.1. T-Tests and Chi-Square Analyses Comparing Student Scores on Key Variables With Rest of the Sample
Table 5.2. Means (M) And Standard Deviations (SD) Of Habitual ER Reported by Males andFemales
Table 5.3. Means (M) And Standard Deviations (SD) Of Self-Reported Levels of EmotionDuring the No Regulation Condition of the ERT

Table 5.4. Means (M) And Standard Deviations (SD) Of Self-Reported Emotion Levels in theERT for Males and Females.199
Table 5.5. Results of <i>T</i> -Tests Showing Difference in Emotion Levels Between No Regulation Condition and Regulation Condition
Table 5.6. Means (<i>M</i>) And Standard Deviations (<i>SD</i>) Of ER Effectiveness Variables 200
Table 5.7. Evidence for Study 2a205
Table 6.1. Correlation Analyses and T-Tests Showing the Relationship Between SocialDesirability, Age, Gender, and the Variables Associated With Mental Health
Table 6.2. Gender Differences in Depressive Symptoms, Suicidal Ideation, Wellbeing, andSelf-Harm Behaviours (Mean, <i>M</i> , and Standard Deviation, <i>SD</i>)
Table 6.3. T-Tests and Chi-Square Analysis Comparing Student Mental Health Scores WithRest of the Sample
Table 6.4. Variables Used in the Mediation Analyses 243
Table 6.5. Evidence for Study 2b252
Table 7.1. Unique Contribution of This Research to the Emotion Regulation Literature271

List of Figures

Figure 1.1. The Multiple Components of an Emotional Response
Figure 1.2. The Dimensional Approach to Emotion (Adapted From
Russell & L.F. Barrett, 1999)
Figure 1.3. Theoretical Approaches to Emotion (Adapted From Scherer, 2019)31
Figure 1.4. The Relationship Between Different Types of Regulation (Adapted From Gross,
2015)
Figure 1.5. The Process Model of ER (Adapted From McRae & Gross, 2020)42
Figure 1.6. The Extended Process Model of ER (Adapted From Gross, 2015)
Figure 1.7. Dual Process Framework of ER (Adapted From Gyurak et al. 2011)
Figure 1.8. Taxonomy of ER Strategies (Parkinson & Totterdell, 1999)45
Figure 2.1. The Key Tenets of Response Styles Theory
Figure 4.1. The Data Coding Process for Study 1
Figure 5.1. Stages of the Emotion Regulation Task
Figure 5.2. Summary of Study 2 Procedure
Figure 5.3. Bar Chart Showing ER Effectiveness for Males and Females, Using Distraction and Reappraisal to Regulate Sadness During the ERT201
Figure 5.4. Bar Chart Showing ER Effectiveness for Males and Females, Using Distraction and Reappraisal to Regulate Anger During the ERT202
Figure 5.5. Bar Chart Showing ER Effectiveness for Males and Females, Using Distraction and Reappraisal to Regulate Fear During the ERT
Figure 6.1. Mediation Model 1: Mediation Model 1: Gender, Depressive Symptoms, and Reappraisal Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and <i>P</i> -Values of Each Path
Figure 6.2. Mediation Model 2: Gender, Depressive Symptoms, and Distraction Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and <i>P</i> -Values of Each Path
Figure 6.3. Mediation Model 3: Gender, Suicidal Ideation, and Reappraisal Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and <i>P</i> -Values of Each Path
Figure 6.4. Mediation Model 4: Gender, Suicidal Ideation, and Distraction Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and <i>P</i> -Values of Each Path
Figure 6.5. Mediation Model 5: Gender, Self-harm Behaviour, and Reappraisal Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and <i>P</i> -Values of Each Path

Figure 6.6. Mediation Model 6: Gender, Self-harm Behaviour, and Distraction Effectiveness
for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and P-
Values of Each Path251
Figure 7.1. The Process Model of ER as a Framework for Emotion-Specific ER 291

Abstract

Gender differences in the regulation of sadness, anger, and fear, and their relationship with variables associated with mental health, were examined in the present research. Emotion regulation (ER) refers to the different ways that we manage our emotions, and ER has an important impact on many areas of life, including mental health. The majority of research in the field of ER has focused on general affective states, such as negative emotion, stress, or mood. Consequently, it is unclear if ER, and gender differences in ER, differs according to the specific emotion being regulated, such as sadness, anger, or fear.

It is important to investigate gender differences in ER in an emotion-specific manner because there is evidence that males and females are socialised to respond to their emotions in different ways, which may impact how they regulate their emotions. Also, there are prominent gender differences in some variables that are associated with mental health. However, the relationship between gender, ER, and mental health has not been examined in an emotion-specific manner so far. This is problematic, because there is evidence that the relationship between ER and mental health may depend on the specific emotion being regulated.

The aims of the present research were (1) to examine if there are gender differences in the regulation of three specific emotions – sadness, anger, and fear, and (2) to investigate if gender differences in the regulation of sadness, anger, and fear are related to gender differences in variables associated with mental health.

These aims were achieved by conducting two empirical studies. In Study 1, participants completed open-ended questionnaires which asked about the regulation of specific emotions. The responses from these questionnaires were coded into ER strategies to examine the relationship between gender and the use of these strategies to regulate specific emotions. In Study 2, participants completed an Emotion Regulation Task (ERT) which

involved looking at emotional pictures and using ER strategies to regulate the specific emotions that arose. Participants also completed self-report questionnaires measuring variables associated with mental health.

Gender differences in ER were found, but these often depended on the specific emotion being regulated and the ER strategy being used. Also, ER partly explained some of the gender differences in variables associated with mental health, but again, this depended on the specific emotion being regulated, and the ER strategy.

These findings have important implications for theory, because they highlight the lack of a theoretical framework for understanding the regulation of specific emotions. However, it is proposed that existing models can be easily altered to accommodate this emotion-specific paradigm. These findings also have important implications for therapeutic practice, because they show that the emotional context and gender of the individual may impact the use and effectiveness of ER strategies. This may guide tailoring therapeutic interventions such as ER training to the specific emotional context, which may help to increase treatment success.

Overall, the findings of this research highlight the importance of investigating ER in an emotion-specific manner. The key message from this research is that what is known about the regulation of one emotion cannot necessarily be applied to all emotions or negative emotion in general. Therefore, it may be helpful to examine ER in an emotion-specific manner moving forward.

Chapter 1: Emotions and Emotion Regulation

1.1. Chapter Summary

In this chapter, the theoretical literature on emotion and emotion regulation (ER) will be discussed. The nature of emotion is the subject of contentious debate in the field. One area of debate is whether emotions can be organised into discrete categories, such as sadness, anger, and fear, or are more continuous in nature and differ only across broad dimensions, such as arousal and valence. In this chapter, the key theories of emotion and ER will be discussed within the context of this debate, to highlight an area of limitation in the emotion and ER literature – the lack of emotion-specificity in theory and research.

The concept of emotion will be defined, and the key perspectives on emotion will be discussed – discrete emotion theories, dimensional/constructivist theories, and appraisal theories. ER will also be defined, and the main theories of ER will be described – the process model, the extended process model, the dual process model, and Parkinson and Totterdell's taxonomy of affect regulation. This chapter will provide the theoretical framework for the present research.

1.2. Emotions

Emotions fundamentally shape how we experience and interact with the world around us. It is possible for individuals to manage their emotional responses, and to have some level of control over their own emotional experience (McRae & Gross, 2020). The management of emotional responses is known as emotion regulation (ER). The ability to effectively regulate emotions is an essential skill to be able to function adaptively in the world, and the different ways that individuals manage their emotions are related to many important areas of life, including mental health (Cludius et al., 2020).

1.2.1. The Multi-Component Definition of Emotion

Before discussing how emotions are regulated, it is important to clarify what is meant by an emotion. There is no single definition of emotion which is agreed upon in the literature. However, on a broad level, there is a general consensus that emotions are affective states which are composed of three components that arise in response to personally meaningful stimuli (Gross, 2015a; Mauss & Robinson, 2009; Scherer & Moors, 2019). These components, or emotional response systems, which are depicted in Figure 1.1, are: (1) subjective experience, (2) physiological arousal, and (3) behavioural tendencies.

Different theoretical perspectives diverge on which of these components are the most important and the causal relationships between them (Scherer & Moors, 2019). However, most approaches agree that each of these components are involved in an emotional response, either as a basic building block, or as an external manifestation of the response (Coppin & Sander, 2021; Ekman, 1992; Keltner et al., 2019; Kuppens, 2019; LeDoux & Hofmann, 2018; Mauss & Robinson, 2009; Scherer, 2005).

Figure 1.1

The Multiple Components of an Emotional Response



Subjective experience refers to the feeling part of emotion, such as having feelings of joy, sadness or fear. This can be distinguished from physical sensations that may arise in the body, such as physical pain. Subjective experience is an important part of an emotional response, and many regard this as the core component which is at the heart of emotion (Kuppens, 2019; LeDoux & Hofmann, 2018; Lieberman, 2019). Subjective experience is central to our understanding of emotion, and it is the internal aspect of an emotional response (LeDoux & Hofmann, 2018).

Physiological arousal refers to changes in the physical state of the body that arise when an emotional response is occurring, due to activation of the autonomic nervous system (ANS; Kreibig, 2010). Physiological changes that occur during an emotional response include increased heart rate, sweating and pupil dilation (Mauss & Robinson, 2009). However, the ANS does not function exclusively for emotional responding, and can become activated for a variety of reasons, including homeostasis and digestion (Robertson, 2019). This means that although physiological arousal is part of an emotional response, it can also be related to other bodily processes.

Action tendencies (or behavioural tendencies) are the behavioural component of an emotional response. These are the behaviours which are associated with a certain emotion and are likely to occur during the emotional response (Ekman & Oster, 1979; Frijda, 1987; Gross, 2015a). Examples of behavioural tendencies are displaying external acts of violence when feeling angry, or rapidly leaving a threatening situation when experiencing fear (Gray, 1971). Emotional behaviour also includes body language, facial expressions, vocalisations, and gestures (Scherer et al., 2019).

The multi-component definition of emotion is the definition used in the present research. This definition was chosen because it is a broad definition of emotion which most emotion theorists agree on (Coppin & Sander, 2021; Ekman, 1992; Keltner et al., 2019;

Kuppens, 2019; LeDoux & Hofmann, 2018; Mauss & Robinson, 2009; Scherer, 2005). It is consistent with the key theories of emotion (discussed in Section 1.2.2), as well as the main emotion regulation theories (discussed in Section 1.3.2). Furthermore, there is strong empirical evidence that these three components (subjective experience, physiological arousal, behavioural tendencies) are involved in an emotional response (Coppin & Sander, 2021; Mauss & Robinson, 2009), and so this definition helps to guide theory and research, as well as the measurement of emotion and ER. The multi-component definition of emotion is consistent with investigating ER in an emotion-specific manner, by providing guidance for how specific emotions can be conceptualised and measured (e.g., by measuring the subjective experience of specific emotions).

1.2.2. Theories of Emotion

Although it is generally agreed that an emotional response broadly involves the three components discussed in the previous section, there is much debate in the theoretical literature about the nature of emotion. The key theoretical approaches disagree on the extent to which these components are related to one another, and the causal relationships between these components (Adolphs et al., 2019). There is no single, agreed upon definition of emotion in the literature (Gross, 2015a; Kuppens, 2019). In fact, the nature of emotion is the subject of debate in the field of affective sciences as a whole, and emotions remain contested concepts (Adolphs et al., 2019; L. F. Barrett et al., 2018; Cowen & Keltner, 2017, 2018; Ekman, 1992; Gross & L. F. Barrett, 2011).

There are three broad theoretical perspectives in the emotion literature – (1) the discrete emotion perspective (Buck & Powers, 2005; Damasio, 1998; Darwin, 1872; Ekman, 1965, 1992; Ekman & Cordaro, 2011; Izard, 1971; R. Levenson, 1994; Panksepp, 1982; Tomkins, 1962, 1963), (2) dimensional and constructivist approaches (L. F. Barrett, 1998,

2017; James, 1884; Russell, 1980; Russell & L. F. Barrett, 1999; Wundt, 1907; Yik et al., 1999), and (3) appraisal theories (Arnold, 1960a, 1960b; Clore & Ortony, 2008; Frijda, 1986; Lazarus, 1991; Lazarus & Folkman, 1984; Leventhal, 1984; Roseman, 1991; Roseman et al., 1990; Scherer, 1999; Scherer & Moors, 2019; C. A. Smith & Ellsworth, 1985). One of the key areas of debate is whether emotions are organised into distinct categories, or are more continuous in nature and differ only in terms of a small number of dimensions, such as arousal and valence (Adolphs et al., 2019). In the following sections, this concept is referred to as discrete emotions, specific emotions, emotional categories, emotion differentiation and emotion-specificity. Each of the main approaches will be briefly described, including the key tenets of each approach in relation to emotion-specificity.

1.2.2.1. Discrete Emotion Theories. The discrete perspective was perhaps the central narrative on emotions throughout the 20th century (Keltner et al., 2019). According to the discrete approach, emotions can be organised into distinct categories, such as sadness, anger, and fear (Keltner, 2019). The most comprehensive account of discrete emotion theory is Ekman's (1992) Basic Emotion Theory. Ekman (1992) describes emotions as basic, which means that first of all, emotions evolved to allow individuals to respond adaptively to their environment, in order to reach a goal. For example, the experience of fear might signal that a threat is nearby, which would prompt an individual to respond adaptively by running away and therefore achieving the goal of survival. Secondly, the term basic indicates that there are a number of distinct emotions, which differ from one another in important ways, such as in the appraisals that cause the emotion, and in the circumstances that trigger these appraisals, which are known as the antecedents. Ekman (1992) proposed that there are six emotions which can be defined as basic according to these principles – sadness, anger, fear, disgust, enjoyment, and surprise.

The hallmark of the discrete emotion perspective is that each of the basic emotions differ from one another in the emotional components described in Section 1.2.1. Each emotion (such as sadness or fear) is viewed as a distinct unit with unique biological processes, and therefore has a unique pattern of responses (Mauss & Robinson, 2009). For example, discrete emotion theorists propose that specific emotions arise from dedicated neural pathways, discrete areas in the brain that are exclusive to specific emotions and are active solely when this emotion is being experienced (LeDoux, 2000; Panksepp, 1982), which has been supported with evidence from neuroimaging studies (Phan et al., 2002; Vytal & Hamann, 2010).

Further, many discrete emotion theorists also posit that specific emotions have unique patterns of physiological arousal, subjective experience, and emotional behaviours, and that there is coherence across two or more of these response systems (Buck & Powers, 2005; Darwin, 1872; Ekman, 1992; Izard, 1992; Keltner et al., 2019). Support for the discrete approach comes from testing the extent to which there is coherence in these emotional components in an experimental setting. Evidence for the discrete approach will be discussed in Section 1.2.3.

A sub theory of the discrete approach is the functionalist account of emotions. According to the functionalist approach, emotions evolved to allow us to respond adaptively to our environment, and each emotion has a particular function to help us with goal-related situations (Campos et al., 1994; Lench et al., 2015). In other words, each specific emotion represents a different problem we may face in the world, from a survival perspective (Lench et al., 2015). For example, sadness represents the loss of a goal that cannot be recovered, anger comes from obstacles that are preventing a goal, and fear is related to a threat in the environment (Campos et al., 1994). Therefore, these emotions represent unique problems and

encourage us to respond in different ways. It is logical that these specific emotions would thus be regulated differently.

1.2.2.2. Dimensional/Constructivist Theories. The dimensional view is in contrast with the discrete approach. According to the dimensional view, emotions can be captured by a small number of fundamental dimensions (Mauss & Robinson, 2009; Russell & L. F. Barrett, 1999). One of the main dimensional perspectives is the core affect theory, which posits that emotion can be described as a psychophysiological state that is categorised in terms of valence and arousal (Russell & L. F. Barrett, 1999). Valence refers to whether the emotion is experienced as pleasant or unpleasant, whereas arousal is the intensity of an emotion.

This approach allows for different emotions to exist, but these emotions appear continuously on dimensional gradients of arousal and valence, rather than being discrete, as depicted in Figure 1.2. Importantly, emotions which have similar valence and arousal are argued to be similar in each of the emotional response systems, including subjective experience. For example, as seen in Figure 1.2, anger and fear are both negative valence and high arousal emotions, and thus should be experienced in a similar manner. According to the dimensional approach, there is nothing inherent in emotions that makes them fundamentally different from one another, but rather they are systematically related (Russell, 1980). The important differences between emotions are considered to be explained by these dimensions (L. F. Barrett, 1998).

Similarly, the constructivist approach also adopts the idea that emotions can be organised along dimensions, rather than being discrete categories. While the discrete perspective proposes that emotions are biological in nature and are a product of evolution (and so are experienced in the same way across cultures; Ekman, 1992), constructivists argue that emotions as we understand them are a product of social learning (L. F. Barrett, 2017).

Figure 1.2

The Dimensional Approach to Emotion (Adapted From Russell & L. F. Barrett, 1999)



This means that emotion is regarded as being 'soft-wired' (learned through social experience) by constructivists, rather than 'hard-wired' (innate) which is proposed by many discrete theorists (Panksepp, 1982).

According to the constructivist perspective, individuals receive a mass of sensory data from their internal body (e.g., increased heart rate), which is known as interoception (L. F. Barrett, 2017). The brain does not know the cause of this arousal and so must make sense of it, and to do so it draws on previous experience, and interprets this arousal as emotion (Adolphs et al., 2019). The brain uses these past experiences as a reference point for the current sensations, and so if something similar was experienced in the past, then this is categorised as the same emotion as the previous experience. For example, if an individual experienced an affective state in the past which was identified as sadness, a comparable set of physiological sensations in the present moment will be similarly labelled as sadness, and thus the emotion of sadness is learned (Adolphs et al., 2019). This approach builds on the early scientific work on the nature of emotions by James (1884), who proposed that the physiological sensations of emotions precede emotional experience, and individuals construct the subjective experience of emotion through this bodily state.

In sum, the discrete perspective views emotions as originating from biological systems, whereas the constructivist approach understands emotions as being mentally constructed from this raw sensory data (Gross & L. F. Barrett, 2011). Constructivists propose that human inference is required to construct emotion (Russell, 2003), while discrete theorists (and many appraisal theorists, as described in the next section) argue that emotions already exist biologically, and human inference identifies these instances of emotion, but has no role in constructing them (Adolphs et al., 2019). Therefore, according to constructivists, discrete emotions do not exist in the traditional sense of biological units which are products of evolution, but rather are socially and culturally learned.

1.2.2.3. Appraisal Theories. Appraisal theories focus on the cognitive aspect of emotion. According to this perspective, it is not an event as such that causes an emotion, but rather the evaluation and interpretation of the event (Roseman, 1991). The appraisal of the significance of a situation to our own personal wellbeing leads to the emotional response (Lazarus, 1991). For example, if an individual perceives a threat in the environment, this threat appraisal elicits the emotion of fear, which in turn prompts an adaptive response (e.g., running away from the threat).

In terms of emotion-specificity, appraisal theorists propose that specific emotions are differentiated by appraisals (Roseman, 1991; Roseman et al., 1990; Scherer et al., 2001). That is, different types of appraisals elicit specific emotions, even in the same situation. An

appraisal of threat will lead to a fear response, and an appraisal of loss will lead to a sadness response, regardless of the nature of the emotional stimulus. According to this perspective, each of the basic emotions put forward by Ekman (1992; Ekman & Cordaro, 2011) are associated with a different type of appraisal of an emotionally significant situation, which gives rise to these discrete emotions (Roseman & Evdokas, 2004).

In this sense, appraisal theory is largely compatible with the discrete approach in terms of emotion-specificity. Both theories postulate that emotions are discrete, and according to appraisal theory, different appraisals cause the differentiation in specific emotions (Scherer & Moors, 2019). This explains why in some cases, individuals can have different emotions about the same situation, because they are having different appraisals about the situation (Roseman et al., 1990). For example, one person might feel relief after a relationship ends, while another person may feel intense sadness. The difference in emotional response is due to the varying appraisals of the situation by the two individuals. It also explains why the same person may experience different emotions about the same circumstance at different times (Roseman, 1991).

In summary, it is evident that the emotion literature is complex, and many of the main theoretical approaches diverge on important points. A useful diagram which has been adapted from Scherer (2019), shows how the main theories conceptualise emotion and how they relate to one another in a process model. In Figure 1.3, we can see that emotion begins with an emotional situation (an event, behaviour etc.), and this event leads to an appraisal (according to appraisal theories, and some discrete emotion theorists). This appraisal gives rise to an emotional response, and these changes occur across emotional response systems (e.g., physiological responses). Dimensional theories view these changes as having a central representation which can be defined along a small number of dimensions (Scherer, 2019). Similarly, constructivist theories propose that the sensory data from these dimensions are

categorised by the human brain, and it is the social meaning attributed to physiological responses that create the emotion (Adolphs et al., 2019; Scherer, 2019).

Figure 1.3

Theoretical Approaches to Emotion (Adapted from Scherer, 2019)



1.2.3. Evidence for Emotion-Specificity

In this section, the empirical evidence regarding emotion-specificity will be discussed. As outlined in Section 1.2.2.1, according to the discrete perspective, emotions are organised into discrete categories which differ across emotional response systems (i.e., experience, physiology, and behaviour). Thus, evidence for this approach comes from measuring each of the components of an emotional response in an experimental setting, and testing (1) whether there are differences *between* discrete emotions in each of these components, and (2) whether there is correlation *within* discrete emotions in each of these components during an emotional response (e.g., there are coordinated responses in each of these systems for sadness, but the responses differ between sadness and anger).

1.2.3.1. Facial Expression Research. Facial expression research has provided evidence for the discrete approach. In several landmark studies, when individuals were asked to identify an emotion from a facial expression, they were able to do so accurately, with most participants correctly identifying the expressions, and this was consistent across cultures (Ekman & Friesen, 1971; Ekman et al., 1987; Ekman et al., 1969; Izard, 1971). Furthermore, individuals were able to distinguish most of the basic emotions (sadness, anger, fear, disgust, surprise, enjoyment) from one another, although the rate of accuracy in differentiating between surprise and fear was lower (Ekman & Friesen, 1971). These findings support the view that distinct emotions give rise to differential facial expressions, which is a type of emotional behaviour, and that these responses can be meaningfully distinguished from one another.

1.2.3.2. Human Vocalisation Research. Similarly, specific emotions can be identified from human vocalisations. Vocalisations refer to any of the social sounds that humans make, including words, laughter, playful intonation, sarcastic tones, sighs, and singing (Cowen et al., 2018). In a study by Cowen et al. (2018), individuals were asked to identify the emotion represented by brief vocalisations. Participants were able to accurately identify the emotion represented by these vocalisations, both in a laboratory session and in real life, and at least 24 distinct emotions were identified. This finding indicates that specific emotions are represented by these brief vocalisations, which are thought to predate human language (Cowen et al., 2018), thus suggesting that emotions can be organised into discrete categories. Although the number of discrete emotions identified by Cowen et al. (2018) is significantly larger than that proposed by Ekman (1992), this research nonetheless supports the discrete approach.

1.2.3.3. Meta-Analysis of Studies Examining Emotional Components.

Furthermore, hundreds of studies have found differences across each of the emotional response systems for specific emotions (Lench et al., 2011). In a large meta-analysis of 687 studies (Lench et al., 2011), the extent to which specific emotions elicited changes in the three components of emotion – experience, behaviour, and physiology – as well as in cognition and judgment, was investigated. Discrete emotions, on average, differed in the effect they had on these variables, with a medium effect size. Importantly, each of the negative emotions (sadness, anger, and anxiety) were different across emotional response systems, with a small to medium effect size (Lench et al., 2011). This finding indicates that there are unique responses in each of the emotional response systems for discrete emotions, including specific negative emotions.

An important finding to highlight from Lench et al. (2011) was that self-reported subjective experience was different for each specific emotion, and this had a large effect size. This demonstrates that discrete emotions have unique subjective experiences, and this is a robust effect in the literature (Baumgartner et al, 2006a; Baumgartner et al., 2006b; Cowen et al., 2020; Cowen & Keltner, 2017; Dimberg, 1988; Harmon-Jones & Sigelman, 2001; Joseph et al., 2020). Similarly, individuals spontaneously talk about discrete emotions such as excitement, worry, and sadness in focus group settings (Stupnisky et al., 2016). The difference in reported subjective experience lends support to the discrete perspective and highlights the importance of using self-reporting as a method for measuring specific emotions, as this is the key method for measuring subjective experience. This will be detailed further in the discussion of methodological approaches in Chapter 3.

1.2.3.4. Bridging the Gap Between Discrete and Dimensional Perspectives.

Traditionally, the discrete perspective has posited that emotions are organised into mutually exclusive categories. However, recent evidence has provided a more nuanced view on this

debate, and has helped to bridge the gap between discrete and dimensional perspectives. It is notable that in the study by Lench et al. (2011) described above, there was not a perfect correlation between the emotional components, which indicates that there was some variability between the components. Evidence has indicated that although emotions are more accurately described by categorical labels, the boundaries between these categories may not be clearly defined, and can be described as 'fuzzy' rather than hard (Cowen & Keltner, 2017).

In Cowen and Keltner (2017), participants were asked to rate emotional film clips in terms of their subjective experience. Discrete category labels more accurately described emotional states than dimensional labels. However, a greater number of emotions were identified than originally proposed by Ekman (1992), with 27 distinct categories being identified in the study. Further, it was found that the boundaries were not completely distinct between emotions, and in fact there was a gradient between emotions (e.g., the continuous gradient from anxiety to fear to horror), which varied along dimensional lines (such as intensity).

This line of research has been critiqued by constructivist theorists. For example, L. F. Barrett et al. (2018) proposed that the findings of Cowen and Keltner (2017) are inconsistent with other studies which demonstrate that instances of an emotion belonging to the same category (e.g., instances of fear) vary considerably in their features (e.g., feelings, expressions, etc.), and instances of different emotion categories have considerable similarities in their features. That is to say, that there is significant within-category variability, and between-category similarity between different emotions. However, this variability is accommodated by the framework proposed by Cowen and Keltner (2017), which demonstrates that discrete categories can be organised along a gradient, but still remain as categories. They argue that measuring emotion solely along the dimensions of valence and arousal is insufficient to explain the variety of reported categories of emotional experience in

their data (Cowen & Keltner, 2018). There is variability in emotional categories, but nonetheless, categories are the best way to semantically capture the emotional experience reported by participants. This variability represents the gradient between different emotional states that belong to the same category, such as anxiety, fear, and horror.

In sum, there is strong evidence that emotional categories exist, but that the boundaries between them are fuzzy rather than hard. This evidence bridges some of the gaps between discrete and dimensional approaches, and indicates that dimensional aspects (such as valence and arousal) are important, but nevertheless emotions can be more accurately distinguished in terms of emotion categories, especially in terms of subjective experience between these categories.

1.2.3.5. Appraisal Research. Evidence from empirical research on appraisal theory indicates that distinct emotions are elicited by different appraisals of an emotional situation. In Roseman (1991), when participants were asked to read stories in which the appraisals were experimentally manipulated and to report on their perception of the emotion experienced by the person in the story, different appraisals corresponded with specific emotions. For example, when a situation was appraised as being a negative outcome that was caused by another person when a positive outcome had been deserved, this resulted in high levels of anger, indicating that this type of appraisal leads to anger (Roseman, 1991). Specific types of appraisals have also been found to differentiate between boredom and other negative emotions (Van Tilburg & Igou, 2017).

Moreover, in Herrald and Tomaka (2002), 109 adult participants were asked to complete a neutral computer task, while a study confederate attempted to induce anger, shame, or pride within the participants using verbal and non-verbal behaviours. For example, to provoke anger, the confederate made hostile, demeaning remarks and showed little respect to the participant. Subjective experience, appraisals, and physiological responses (cardiac
activity) were some of the outcomes measured at baseline, and after the emotional manipulation. Participants in each of the three emotion conditions (anger, shame, pride) were found to differ in the subjective experience compared to baseline (i.e., when the emotion of anger was elicited, participants reported experiencing more anger than other emotions), and they also varied in the specific appraisals they reported. Some emotions also differed in their physiological responding. For example, anger had higher cardiac reactivity than pride, as measured by pre-ejection period, PEP. These findings support a discrete approach.

These findings are further reinforced by research from Roseman and Evdokas (2004) who found that manipulating appraisals in an experimental setting gave rise to the specific positive emotions of joy, relief, and hope. This lends support to the hypothesis that discrete emotions differ in each of the emotional response systems, particularly subjective experience, and that specific appraisals give rise to these discrete emotions, as postulated by appraisal theory.

1.2.3.6. Summary of Evidence for Emotion-Specificity. In summary, based on the evidence that has emerged in the empirical literature, the following conclusions can be drawn about emotion-specificity. Firstly, there is strong evidence that emotions can be organised into discrete categories, albeit with fuzzy boundaries (Cowen & Keltner, 2017). These specific emotions are differentiated in terms of their subjective experience, physiological arousal and emotional behaviours such as facial expression. Although dimensions of emotion (such as arousal and valence) are important factors of an emotional response, these are not sufficient for distinguishing between different emotions, based on the available evidence (Cowen & Keltner, 2017).

Secondly, the emotional component with the most convincing evidence for emotionspecificity is subjective experience, with hundreds of studies finding that emotions are distinct in subjective experience, and this is a robust finding with a large effect size (Cowen

et al., 2020; Cowen & Keltner, 2017; Joseph et al., 2020; Lench et al., 2011). This highlights the importance of including subjective experience in the study of specific emotions. Further, it is likely that unique appraisals of an emotional situation lead to different emotions, and give rise to these differences in subjective experience, as proposed by appraisal theorists (Roseman, 1991; Roseman et al., 1990).

1.3. Emotion Regulation

1.3.1. Definition of Emotion Regulation

In the previous section, it was established that emotions can be organised into discrete categories, such as sadness, anger, and fear. It is possible for individuals to change these specific emotional responses, which is known as emotion regulation, ER (McRae & Gross, 2020). ER can be defined as the "processes by which individuals influence which emotions they have, when they have them, and how they experience and express these emotions" (Gross, 1998b), which is referred to as the process definition of emotion. The process definition of ER will be used in the present research. This definition of ER was chosen because it derives from one of the key theories of ER, the process model, and this theory is supported with empirical evidence (Gross, 1998a; Hermann et al., 2017; Quinn & Joormann, 2020; Sheppes & Meiran, 2007, 2008; Smoski et al., 2014; Thiruchselvam et al., 2011). This definition conceptualises ER on a temporal basis and so provides a way of understanding how different ER strategies can influence the emotion process. It is consistent with conducting research into ER in an emotion-specific manner, as specific emotions can be conceptualised as different process models. The process definition of ER is also consistent with the multicomponent definition of emotion described in Section 1.2.1, as the different components of an emotional response represent the final stage of the unfolding of an emotional response which is described by the process definition of ER. The process definition of ER is helpful as

it also indicates that individuals can influence which emotions they have but do not always have complete control over their emotional response.

ER may be cognitive in nature, such as changing thought patterns in order to feel better, or behavioural, such as taking part in a fun activity to reduce negative emotions (Parkinson & Totterdell, 1999). ER can also be a deliberate process which involves an intention to feel better, or it can occur subconsciously (Gross, 1998b; Gyurak et al., 2011). ER can be distinguished from the regulation of other affective states, such as affect regulation, coping, and mood regulation. Affect can be regarded as the general umbrella term for all affective states that involve a valuation process (Gross, 2015a), such as mood, emotion, and stress, and thus ER is a subcategory of affect regulation, as depicted in Figure 1.4.

Figure 1.4

The Relationship Between Different Types of Regulation (Adapted From Gross, 2015)



Coping refers to the process of managing stress, which is a physiological response arising from appraisals of challenging environmental demands (Carver et al., 1989). Some distinctions between ER and coping include: (1) ER can be implicit (subconscious) or explicit (deliberate) in nature, whereas coping only involves explicit processes, (2) coping involves the regulation of stress specifically, whereas ER involves regulating emotions in response to a much wider range of circumstances, and (3) ER can involve both intrinsic and extrinsic processes, which means that emotions can be regulated by external factors (e.g., other people), whereas coping involves only intrinsic processes (Compas et al., 2014). ER and coping are independent constructs, although the two can occur simultaneously, for example during a stressful life event (Garnefski et al., 2001).

In terms of the difference between ER and mood regulation, mood and emotion differ in the length of time they tend to be experienced (Gross, 2015a). Emotions occur over a relatively short period of time such as a few seconds, minutes, or hours (Lench et al., 2011; Verduyn & Lavrijsen, 2015), whereas mood can span across longer periods such as days or weeks (Nolen-Hoeksema et al., 1993). They also differ in terms of antecedents, with emotions tending to have a specific trigger event, and mood being more diffuse in nature (Gross, 2015a). This means that ER tends to occur in the moment of encountering the emotion-eliciting stimulus, and is more short-term in nature, whereas mood regulation involves attempting to influence the affective state on a more long-term basis, and is not necessarily in response to a specific stimulus.

Individuals have a repertoire of ER strategies that can be drawn upon during an emotional episode (Dixon-Gordon et al., 2014), and a large number of ER strategies exist. Almost any activity (including cognitive activity) which is undertaken with the intention of influencing emotion can be regarded as a type of ER. Some of the most common ER strategies that are researched in the empirical literature are summarised in Table 1.1. Some attempts have been made to create a taxonomy of ER strategies, such as by Parkinson and Totterdell (1999), and this framework will be discussed in Section 1.3.2.4. We can see from

Parkinson and Totterdell's (1999) taxonomy that there are a large number of strategies available to regulate emotions.

Table 1.1

ER Strategy	Definition
Acceptance	Acknowledging emotions without judgement
Avoidance	An attempt to avoid the emotion or leave the emotional
	situation
Brooding	A type of rumination that is characterised by passively
	focusing on the reasons for the emotion or emotional
	situation
Catastrophising	Focusing on thoughts which emphasise the terror of the
	situation
Cognitive Reappraisal	Changing thoughts about a situation to reduce its
	emotional impact
Distraction	Diverting attention from the situation and doing or
	thinking something that is unrelated
Experiential Suppression	Pushing negative thoughts or feelings out of conscious
	awareness
Expressive Suppression	Not externally expressing an emotion
Passivity	Waiting to see what comes of the situation and not taking
	any action
Positive refocusing	Focusing attention on to pleasant events
Problem Solving	Actively attempting to overcome or solve a problem
Putting into perspective	Thoughts of brushing aside the seriousness of the event
Reflection	A type of rumination characterised by paying attention to
	an emotional situation with curiosity
Rumination	Persistently focusing on the causes and consequences of
	the situation without engaging in problem solving
Social support seeking	Looking to others for comfort
Substitution	Thinking of a happy memory when experiencing a
	negative emotion

Common ER Strategies in the Literature

1.3.2. Theories of Emotion Regulation

There are several theoretical approaches to ER, and the key theories in the ER literature are essentially models of how ER strategies can be categorised. The main approaches to ER tend to complement, rather than contradict, one another. They are not mutually exclusive, but rather they focus on different aspects of the regulation process. Emotions are thought to (1) unfold over time, as outlined by the process model of ER (Gross, 1998b), (2) involve reaching a personally significant goal, as outlined by the extended process model (Gross, 2015b), (3) be regulated by processes that are either implicit or explicit in nature, as posited by the dual process model (Gyurak et al., 2011), and (4) be regulated by both cognitive and behavioural strategies, and strategies that involve engaging or disengaging with a stimulus, as suggested by Parkinson and Totterdell (1999). Each of these perspectives will be discussed in turn.

1.3.2.1. Gross's (1998b) Process Model of ER. Arguably the most influential model of ER is Gross's (1998b) process model. Gross (1998b) developed the process model of ER based on Lazarus's earlier work on threat appraisal and coping, which postulated that the appraisal of a threatening situation triggers the stress response (Lazarus & Alfert, 1964). According to the process model, an emotion unfolds over time, and ER strategies are categorised according to the point at which they intervene in this process. As shown in Figure 1.5, there are four stages of an emotional response. Firstly, there is a situation which triggers the process (1), and the individual must focus attention on to this situation (2). Next, the individual has an appraisal of the situation (e.g., that there is a threat) (3), and this gives rise to activation in each of the emotional response systems described in Section 1.2.1 (4) (McRae & Gross, 2020). The process model is an appraisal model of emotion and is consistent with the broader appraisal approach described in Section 1.2.2.3.

Figure 1.5

The Process Model of ER (Adapted From McRae & Gross, 2020)



1.3.2.2. The Extended Process Model of ER (Gross, 2015b). The process model has evolved in recent years, and the extended process model (EPM) was developed to further thinking around ER (Gross, 2015b). The process model is important to understand because it explains clearly how different ER strategies can intervene in an emotional response, and the effect that this has on the resulting emotion. However, it is also important to discuss the EPM because it relates ER to goal achievement, which was not included in the original process model. The EPM integrates the control theory approach to behaviour into the process model (Carver & Scheier, 1982). It builds on the original process model by conceptualising the emotion generative process as a cycle, rather than on a linear timeline. According to the EPM, emotions arise through a process of valuation, which is deciding that something is 'good for me' or 'bad for me' (Gross, 2015a). The EPM highlights that emotions are regulated in pursuit of achieving goals, e.g., to feel better, and the regulatory process begins when there is a sensed discrepancy between reality and this goal (McRae & Gross, 2020).

Figure 1.6

The Extended Process Model of ER (Adapted From Gross, 2015)



Situation

The valuation process involves four steps, as shown in Figure 1.6. Firstly, the input to the model is the (1) state of the world as it currently exists, which an individual will (2) perceive (or pay attention to). This perception leads to (3) a negative or positive valuation based on how close an individual is to reaching a particular goal. This valuation gives rise to (4) an action which can alter the individual's environment (i.e., the state of the world) which forms the input to a new loop. Thus, emotions arise in a cyclical process, with the output of the first process model (the emotional response) forming the input of the next model (the situation that is being perceived and appraised). The original linear process model maps on to the cyclical depiction of the extended process model, which is shown in Figure 1.6 in bold text (Gross, 2015a).

1.3.2.3. Dual Process Model of ER (Gyurak et al., 2011). As discussed in the previous section, the process model distinguishes ER strategies on the basis of the point at which the strategy intervenes in an emotional response (Gross, 1998b). Alternatively, ER strategies can be differentiated based on whether they are executed deliberately with conscious awareness (explicit strategies) or occur automatically, outside of conscious awareness (implicit strategies), which is proposed by the dual process model (Gyurak et al., 2011). This does not contradict the process model, but rather it focuses on categorising ER on a different level. All of the strategies outlined in the process model can occur either implicitly or explicitly, e.g., cognitive reappraisal may be an automatic strategy that an individual uses naturally without much awareness (Timmer-Murillo et al., 2020), or it can be deliberately used, such as when instructed to during an ER task (Gross, 1998a).

According to the dual process framework, explicit and implicit ER are distinct, yet interrelated processes (Gyurak et al., 2011). As shown in Figure 1.7, explicit and implicit ER are not mutually exclusive categories (as depicted by the dotted outlines), and different ER strategies vary in how implicit or explicit they are over time (Gyurak et al., 2011).

Figure 1.7

Dual Process Framework of ER (Adapted From Gyurak et al. 2011)



1.3.2.4. Parkinson and Totterdell's (1999) Taxonomy of Affect Regulation.

Parkinson and Totterdell's (1999) framework is not a theory of ER per se, but rather is a taxonomy of ER strategies which categorises strategies in a useful way, based on empirical evidence. According to the framework, ER strategies can be distinguished along two high-level dimensions, as shown in Figure 1.8. Firstly, ER strategies can be categorised according to whether they are cognitive or behavioural in nature. A behavioural strategy involves 'doing' something, such as chatting with friends or listening to music. Alternatively, a cognitive strategy involves 'thinking' something, such as changing the meaning of an emotional situation. Some examples have been provided in Figure 1.8, but there are many more instances of cognitive and behavioural ER.

Figure 1.8





Secondly, ER strategies can be categorised according to whether they involve engaging with the emotion or emotional situation or diverting attention away from the emotion or emotional situation. Engagement ER involves paying attention to the emotion, and this includes cognitive strategies such as changing the way a situation is thought about, or behavioural strategies, such as taking action to solve the problem. Alternatively, individuals may divert attention away from the emotion or the situation that caused the emotion, which is known as diversion ER. This can also be cognitive or behavioural in nature. Diversion is disaggregated further in the framework by making a distinction between disengagement and distraction ER (which are both part of diversion ER). While disengaging involves removing attention from the emotion, distraction involves removing attention from the emotion and replacing this with something unrelated to the emotional situation.

This framework is consistent with the process model of ER (Gross, 1998b). The different strategies that arise from the Parkinson and Totterdell (1999) framework fit into the process model at different stages. For example, reappraisal is a cognitive strategy and expressive suppression is a behavioural strategy according to both theories. In terms of the engagement-diversion distinction, any strategies such as rumination or reappraisal that involve paying attention to the emotional situation would be regarded as engagement strategies, whereas strategies that involve diverting attention away from the stimulus such as suppression or distraction would be categorised as diversion strategies. Therefore, they are slightly different categorisation systems but are broadly consistent and refer to many of the same ER strategies.

The value of the Parkinson and Totterdell model is that it provides guidelines for how ER strategies can be categorised, which is supported with empirical evidence (Aldao et al., 2010; Gross & John, 2003; Parkinson & Totterdell, 1999). This means that when conducting ER research, measures which encourage free responses in order to capture a wide range of

strategies can be used, rather than relying on standardised questionnaires which predetermine strategies. These responses can subsequently be categorised using the taxonomy and guidelines provided by this framework, which ensures that the ER strategies recorded are a true reflection of the strategies participants actually use. This method was used in the present research, which is discussed in Section 3.5.2.1. Therefore, the Parkinson and Totterdell framework provides a valuable tool for ensuring accuracy in measuring ER using this novel approach.

Furthermore, one area for which the Parkinson and Totterdell model is particularly useful is that of gender differences. As will be discussed in Chapter 2, there are gender differences in ER. Two patterns that have emerged in the empirical literature are that (1) females tend to use cognitive ER more than males (Garnefski et al., 2004), and (2) females tend to use strategies that involve engaging with their emotions, such as rumination (D. P. Johnson & Whisman, 2013; Nolen-Hoeksema & Aldao, 2011; Zlomke & Hahn, 2010), whereas males are more likely to use strategies that involve diverting attention from emotions, such as avoidance or distraction (Trives et al., 2016; Zimmermann & Iwanski, 2014). Thus, the gendered patterns emerging in the literature mirror the distinctions made by Parkinson and Totterdell (1999) in their framework, and so it may be a useful taxonomy for defining ER strategies using the free response method described, particularly in work investigating gender differences in ER.

1.3.3. Emotion-Specific Emotion Regulation

1.3.3.1. Research Findings of Emotion-Specific Emotion Regulation. Emotions can be organised in terms of discrete categories, such as sadness, anger, and fear (Cowen et al., 2018; Cowen et al., 2020; Cowen & Keltner, 2017; Ekman, 1992; Keltner et al., 2019; Lench et al., 2011). The regulation of these specific emotions is known as emotion-specific

ER (Zimmermann & Iwanski, 2014). Relatively few studies in the literature have focused on the regulation of multiple specific emotions, but rather have focused on the regulation of negative and positive emotion in general (Bomyea et al., 2020; Goldin et al., 2008; Gratz & Roemer, 2004; Gross & John, 2003; Jackson et al., 2000; Ochsner et al., 2002; Richards & Gross, 2000; Sheppes et al., 2014), the stress response (Lazarus & Alfert, 1964; Notarius & Levenson, 1979), or have focused on inducing one emotion in a single study, rather than several emotions (Gross, 1998a; Gross & Levenson, 1993; Hartley & Phelps, 2010; Mauss et al., 2007a; Mauss et al, 2007b; Sheppes & Meiran, 2007; Szasz et al., 2011). This lack of emotion-specific research is possibly due to an underlying assumption that the way that one emotion, or negative emotion in general, is regulated can be extrapolated to other specific emotions.

However, there is empirical evidence that this is not the case. In Zimmermann and Iwanski (2014), the use of several ER strategies to regulate the specific emotions of sadness, anger and fear was measured in a range of age groups (individuals aged 11, 13, 15, 17, 19, 22, 25, 29 and 50 years old) to assess the development of emotion-specific ER from adolescence into adulthood. Two important findings emerged from this study.

Firstly, specific emotions tend to be regulated using different strategies. Specifically, when individuals feel sad, they are more likely to use social support seeking, passivity, and avoidance than when they feel scared or angry. To regulate anger, people are more likely to use dysregulation or dysfunctional rumination. When individuals feel scared, they tend to use suppression and rumination. This is supported with evidence that individuals are likely to regulate specific emotions using different reappraisal tactics, with participants being more likely to use acceptance reappraisals (i.e., 'that's the way life goes') to regulate sadness and changing future consequences through reappraisal (i.e., 'things will turn out better than expected') to regulate fear (Vishkin et al., 2020). These findings indicate that the ER

strategies individuals use may depend on the emotional context, and specific emotions are regulated in different ways.

The second important finding from Zimmermann and Iwanski (2014) is that the development of ER has a different trajectory for specific emotions. For example, the use of expressive suppression to regulate fear increases from adolescence into adulthood, however this is not the case for using expressive suppression to regulate sadness or anger. That is, the use of suppression to regulate sadness or anger does not increase with development. These findings indicate that ER differs between specific emotions, and the development of ER is emotion specific. Therefore, it is incorrect to assume that what is known about the regulation of general negative emotion, or one individual emotion, can be applied to all specific emotions. If general negative emotion had been measured in Zimmermann and Iwanski (2014), these emotion-specific caveats in ER would have been masked.

Although research in this area is limited, some studies have investigated the regulation of multiple specific emotions. The available evidence indicates that the regulation of specific emotions does have a differential impact both on the effectiveness of ER and on other life areas. For example, in a sample of depressed and non-depressed participants, the use of suppression was not effective at reducing sadness levels, but was effective at reducing anxiety for the non-depressed participants (Boland et al., 2019), which shows that the effectiveness of suppression depends on the emotion being regulated. Similarly, when individuals are shown fear-inducing stimuli, there is little difference in the effectiveness of reappraisal and suppression in reducing negative emotion (Olatunji et al., 2017). However, when participants are shown disgust-inducing stimuli, reappraisal is more effective than suppression at reducing negative emotion (Olatunji et al., 2017), which supports the postulation that ER may differ between fear and disgust. In adolescence, reappraisal is more effective than distraction at reducing fear and anxiety, but there was no difference when

regulating other emotions (Theurel & Gentaz, 2018). Therefore, the effectiveness of ER strategies can depend on the emotion being regulated.

The ability to effectively regulate specific emotions is also differentially related to correlates of ER. Amongst adults with post-traumatic stress disorder (PTSD), difficulties in regulating sadness and fear contributed to using self-harm as an ER strategy, but this was not the case for anger (A. Bradley et al., 2019). In children with ADHD, anger dysregulation explained the relationship between negative affect and externalising symptoms (such as aggressive behaviour), but sadness regulation did not account for this relationship (Leaberry et al., 2019). Similarly, in a sample of adults, sadness dysregulation predicted depression and social anxiety but not aggression, whereas anger dysregulation predicted aggressive behaviour, but not depression or social anxiety (Clear et al., 2019). Furthermore, suppression of anxiety has a different effect in depressed people compared to people who are not depressed, but there is no difference between these groups when suppressing sadness (Boland et al., 2019).

These findings demonstrate that what is known about the regulation of one emotion cannot necessarily be applied to all emotions (Vishkin et al., 2020), and regulating specific emotions can have a unique relationship with correlates of ER. Therefore, it is imperative that the ER literature investigates the regulation of emotions in an emotion-specific manner moving forward.

1.3.3.2. Why Sadness, Anger and Fear? In the present research, three specific emotions are focused upon – sadness, anger, and fear. These emotions were chosen for three key reasons. Firstly, they are part of Ekman's (1992) six basic emotions, as described in Section 1.2.2.1, which means that these are key emotional states which can be distinguished from one another (Ekman, 1992; Ekman & Cordaro, 2011). The term basic also denotes that each

emotion has evolved to enable individuals to deal with different circumstances that arise in life, such as loss (sadness), interference with an important goal (anger), or a threat (fear) (Ekman & Cordaro, 2011). In studies which aim to discover how many discrete emotions exist, sadness, anger, and fear consistently emerge as key emotional states (Cowen et al., 2018; Cowen et al., 2020; Cowen & Keltner, 2017). Furthermore, sadness, anger, and fear are amongst the top five most regulated emotions (Gross et al., 2006).

Secondly, in previous research which has examined ER in an emotion-specific manner, sadness, anger, and fear are three of the most commonly-researched emotions (Boland et al., 2019; Bujor & Turliuc, 2020; Endrerud & Vikan, 2007; Perchtold et al., 2019; Vishkin et al., 2020; Wong et al., 2018), although few researchers have examined all three emotions in a single study. Additionally, sadness, anger and fear were examined in Zimmermann and Iwanski (2014), and it was found that the regulation of these emotions may develop independently across adolescence and into adulthood. Therefore, by selecting these emotions for examination in the present research, this provides an opportunity to build on existing knowledge.

Lastly, sadness, anger, and fear are three specific emotions which are strongly associated with clinical disorders, and behaviours associated with these disorders. For example, according to the DSM-V (American Psychiatric Association, 2013), one of the criteria for a diagnosis of clinical depression is prolonged experience of depressed mood, which is comparable to an intense experience of sadness. Dysregulation of sadness and anger have been implicated in depression and social anxiety, and the dysregulation of anger is associated with aggressive behaviour (Clear et al., 2019; Harmon et al., 2019). Similarly, experiencing symptoms of anger is associated with a range of different clinical disorders and behaviours, including major depressive disorder, bipolar disorder, social phobia, generalised anxiety disorder, obsessive compulsive disorder, post-traumatic stress disorder, and alcohol and drug

use disorders (E. L. Barrett et al., 2013). Furthermore, many anxiety disorders are characterised by excessive fear (Craske, 1997), and traditionally anxiety disorders have been examined through models of fear (Shin & Liberzon, 2010). Therefore, by investigating sadness, anger, and fear in the present research, the differential relationships between these specific emotions and variables that are associated with mental health can be assessed.

1.3.4. Distinction Between Emotion Generation and Emotion Regulation

In terms of ER, it is important to note that different theoretical perspectives have different views on whether emotion generation and emotion regulation can be understood as separate processes (Gross & L. F. Barrett, 2011). Discrete emotion theorists view the generation of emotion as coming from more primitive areas of the brain such as the limbic system (Sarkheil et al., 2019), and this is regulated using more evolved parts of the brain such as the prefrontal cortex (Silvers & Moreira, 2019), and therefore these are completely distinct processes. On the other hand, constructivists argue that there is no distinction between emotion generation and emotion regulation, as emotions are being continually constructed from sensory information (Adolphs et al., 2019). In the middle of the continuum is the appraisal approach, which posits that emotion generation and emotion regulation are related processes (Gross & L. F. Barrett, 2011). That is, an emotion can be regulated while it is unfolding (McRae & Gross, 2020).

1.3.5. Why Is It Important to Investigate Gender Differences in Emotion Regulation?

The way that individuals regulate their emotions, and the strategies that they use to do so, can have a significant impact on many different areas of life, such as wellbeing (John & Gross, 2004; McRae et al., 2012b), personal relationships (Gross & John, 2003), income and socioeconomic status (Côté et al., 2010), academic performance (Davis & Levine, 2013; Ivcevic & Brackett, 2014), and physical health (Appleton et al., 2013; 2014). Similarly, there is an abundance of research demonstrating that the way individuals manage their emotions is related to their mental health (Aldao et al., 2016; Aldao & Nolen-Hoeksema, 2010; Aldao & Nolen-Hoeksema, 2012; Aldao et al., 2010; Berking & Wupperman, 2012; Boland et al., 2019; Garnefski & Kraaij, 2006; Hu et al., 2014; Kashdan et al., 2014; McRae et al., 2012b; Westphal et al., 2017).

Accordingly, it is important to examine the factors which are associated with ER because ER is amenable to change (Bomyea et al., 2020; Denny, 2020; Goldin et al., 2012) and so provides an opportunity for improving mental health. ER strategies are learned skills (Young et al., 2019; Zimmermann & Iwanski, 2014), and so individuals have the potential to learn new ways of managing their emotional responses. Training programmes that aim to build participants' skills in using adaptive ER strategies have been shown to improve ER and have a positive impact on mental health (Bomyea et al., 2020; Denny, 2020; Goldin et al., 2012).

Consequently, this means that ER is a potential avenue for improving mental health and increasing wellbeing. ER training is crucial in efforts to tackle poor mental health, and the more that is known about how ER works, the correlates of ER, and for whom ER is most effective, the more these findings can guide therapeutic interventions for poor mental health and clinical disorders. Modifying ER through targeted interventions can aid efforts to reduce the prevalence of clinical disorders (Cludius et al., 2020). ER could also improve sub-clinical symptoms and promote general wellbeing in the population (Gross et al., 2019). Therefore, it is imperative to understand the caveats in ER in order to guide this clinical work

It is helpful to focus on gender differences in particular, because there is evidence that there are gender differences in both ER and mental health. There are gender differences in variables associated with mental health, such as depression, suicidal ideation, and self-harm

behaviours (H. Chen et al., 2019; Knudson et al., 2020; C. S. Lee & Wong, 2020). There is also evidence that ER can explain some of these gender differences in mental health (Nolen-Hoeksema et al., 1999). Therefore, examining these relationships through the lens of gender differences can help to understand them more deeply.

However, a limitation is that most of the previous research in the field has tended to examine the relationships between gender, ER and mental health in a general manner, by examining general affective states such as mood, stress, and negative emotion. An area which is missing is examining these relationships in an emotion-specific manner. It is important to investigate the relationships between gender, ER, and mental health in an emotion-specific manner for two keys reasons. Firstly, males and females may receive socialisation around specific emotions, which may impact how they regulate these specific emotions (L. R. Brody, 1985; Root & Denham, 2010). This means that gender differences in ER may depend on the specific emotion being regulated. Secondly, there is evidence that regulating different emotions, even using the same strategy, may have different relationships with mental health variables (Boland et al., 2019). That is, the regulation of one emotion may have a very different consequence than the regulation of another emotion, even when using the same strategy (Boland et al., 2019). This may indicate that the relationships between gender, ER, and variables associated with mental health may depend on the specific emotional context.

It cannot be assumed that what is known about the regulation of one emotion can be applied to all emotions (Vishkin et al., 2020), and contextual variation in ER may have an impact on its effectiveness (Doré et al., 2016). Tailoring therapies to an individual's needs may help to increase the success of the treatment program for the individual (Z. D. Cohen & DeRubeis, 2018; Fisher & Boswell, 2016; Välimäki & Lantta, 2019). When informing therapeutic practice, it is imperative to understand any factors, and their interaction effects, which may moderate the effectiveness of treatment (Zilcha-Mano, 2019). Two factors which

may have an impact on treatment are (1) the specific emotion being regulated, and (2) the interaction between the gender of the patient and the emotion being regulated.

The present research will build on previous empirical evidence and contribute to the field by examining the relationships between gender, ER, and mental health, in an emotion-specific manner. This may help to understand gender differences in ER, and the relationship between gender, ER, and mental health more deeply, which can guide therapeutic interventions, such as ER training. This information will enhance the development of therapeutic interventions for mental health problems by enabling these programmes to be increasingly tailored to different emotional contexts and thus increase chances of treatment success (Z. D. Cohen & DeRubeis, 2018; Fisher & Boswell, 2016; Zilcha-Mano, 2019).

1.4. Conclusion and Next Steps

In conclusion, the main perspectives in the emotion and ER literatures were discussed in this chapter. One of the key theoretical debates is whether emotions can be organised into discrete categories, or if they differ only on broad dimensions (L. F. Barrett, 1998; Ekman, 1992). As discussed in this chapter, the available evidence indicates that emotions can be grouped into discrete categories that have continuous gradients (Cowen et al., 2018; Cowen & Keltner, 2017). The process model of ER (Gross, 1998b), and the taxonomy of ER strategies developed by Parkinson and Totterdell (1999) provide useful frameworks for studying the regulation of specific emotions. There are a limited number of studies investigating ER in an emotion-specific manner, but those that have done so have found that (1) specific emotions tend to be regulated using different strategies (Vishkin et al., 2020; Zimmermann & Iwanski, 2014), and (2) the regulation of specific emotions is differentially related to correlates of ER (Boland et al., 2019; Clear et al., 2019), and therefore it is useful to research ER in an emotion-specific manner. As will be discussed in Chapter 2, this is

particularly important when examining gender differences in ER, as there is evidence that males and females may (1) experience specific emotions differently, and (2) be socialised to express specific emotions differently.

It was argued in this chapter that ER has an important impact on mental health. It is imperative that empirical studies investigate which ER strategies are effective and for whom, in order to guide therapeutic interventions for poor mental health (Denny, 2020). One prominent pattern that has emerged in the literature is that there are gender differences in variables associated with mental health (Eaton et al., 2012; Girgus & Yang, 2015; Wetherall et al., 2018). Further, it has emerged that there are also gender differences in the way that emotions are regulated (Garnefski et al., 2004; Zimmermann & Iwanski, 2014).

It is possible that ER is a potential mechanism which explains the gender differences in mental health observed in the literature. The research that has been conducted to date demonstrates that ER at least partially contributes to gender differences in some mental health problems (Nolen-Hoeksema et al., 1999). However, a limitation of the existing literature is that very little work has addressed this issue in an emotion-specific manner, and so it is possible that important effects are being masked. Gender differences in ER and variables associated with mental health, and the lack of emotion-specific research in this area, will be discussed in Chapter 2.

KEY POINTS FOR CHAPTER 1

- 1. The empirical evidence indicates that emotions can be organised into discrete categories with 'fuzzy' boundaries, and these emotions differ in their subjective experience.
- 2. There is a lack of research investigating ER in an emotion-specific manner, which is important because the regulation of specific emotions may have different relationships with variables associated with mental health.
- 3. ER is amenable to change, and finding out more about emotion-specific ER can help to guide therapeutic interventions for poor mental health.

QUESTIONS FOR SUBSEQUENT CHAPTERS

- 1. What is the link between gender, ER and mental health? (Chapter 2)
- What are the different methodological approaches to measuring emotions and ER? (Chapter 3)

NEXT STEPS

Now that the theoretical literature has been reviewed, the link between gender, ER and variables associated with mental health will be discussed in Chapter 2. There are gender differences in both ER and mental health, and ER may explain some of the gender differences that are observed in these mental health problems. However, very few studies have investigated gender differences in the regulation of specific emotions and its relationship with mental health problems.

Chapter 2: Gender Differences in Emotion Regulation and Mental Health

2.1. Chapter Summary

In Chapter 1, it was established that ER has an important relationship with a number of life areas, including mental health. ER is amenable to change, and participating in ER training can positively impact mental health and clinical disorders. Research has demonstrated that there are prominent gender differences in ER and variables associated with mental health. It is possible that ER may play a role in these observed gender differences in mental health.

Although this issue has been investigated previously, very little research has examined this link in an emotion-specific manner. In Chapter 1 it was proposed that specific emotions exist, and these emotions differ in their subjective experience. Findings regarding gender differences in the use of certain ER strategies are mixed, and a possible reason for this may be the lack of emotion-specific research in this area. Examining gender differences in the regulation of specific emotions is important, because (1) it is possible that gender differences in ER depend on the emotion being regulated and (2) the regulation of specific emotions may be differentially related to mental health problems.

In this chapter, the relationship between gender, ER and variables associated with mental health (depressive symptoms, self-harm behaviour, and suicidal ideation) will be discussed. Firstly, the concepts of sex and gender will be defined, and gender differences in ER will be reviewed, including how this relationship may change throughout development. Next, the link between gender, ER and mental health will be described within the context of Response Styles Theory (Nolen-Hoeksema, 1987), which provides a theoretical framework for understanding gender differences in depression. Finally, the implications of this work for clinical therapeutic interventions will be discussed.

2.2. Gender Differences in Emotion Regulation

2.2.1. Sex and Gender

The terms 'sex' and 'gender' are often used interchangeably in research, but there is a distinction between these two concepts. Sex is a biological term, which includes differences in anatomy, physiology, genes, and hormones (J. L. Johnson et al., 2009). Sex is usually distinguished along the male/female binary, although sex exists on a continuum, with some people possessing the sexual markers of both males and females (Fausto-Sterling, 2018).

Gender, on the other hand, is a social concept, which is rooted in the culture and history of a society, and so is fluid and can change over time (J. L. Johnson et al., 2009). Gender refers to the socially-prescribed ideals of 'maleness' and 'femaleness', and this is related to the societal structure, such as a patriarchal society (J. L. Johnson et al., 2009). As such, the term 'gender' is usually preferred in research which acknowledges the role of socialisation and learning on behaviour (i.e., 'nurture' rather than 'nature') (Brannon, 2016; J. L. Johnson et al., 2009; Meyers-Levy & Loken, 2015). Many different gender groups exist, including individuals who identify as having no gender. The groups that are focused upon in this chapter are males and females.

The term 'gender' is used throughout this chapter to reflect that the focus of this research is predominantly on individuals who have been socialised as male or female, rather than being related to sex directly. The term sex is rooted in biological processes, which is not the focus of the present research. In order to truly measure sex, you must measure a range of factors, including chromosomal configuration and reproductive physiology (Brannon, 2016). Gender on the other hand, is self-defined and self-reported by adults. Gender is fluid, and as society changes, the way that gender is defined can change too. As gender can be self-reported, gender was measured in the present research.

Gender roles are related to the concept of gender, and refer to the behavioural norms allocated to males and females in a society, including dress code, mannerisms, posture, and occupation (J. L. Johnson et al., 2009). These gender norms often dictate behaviours on an institutional level, such as in the labour market, education systems, and the division of labour in the household (Weber et al., 2019). Another area which may be influenced by gender roles is the way that males and females experience, express, and regulate their emotions.

2.2.2. Gender and Emotion

Emotion-specific gender differences can be observed in many areas of emotion research. Females are more likely to express internalised emotions, such as sadness and fear (Allen & Haccoun, 1976; Chaplin, 2015), whereas males are more likely to express externalising emotions such as anger in certain contexts (Archer, 2004), and this gender effect is found from childhood (Chaplin & Aldao, 2013).

In terms of subjective experience, females tend to report negative emotion as more intense and more arousing than do males (Bodrogi et al., 2020; Gross & Levenson, 1995), particularly in response to emotional stimuli depicting social situations (Marchewka et al., 2014), and to aversive pictures such as threatening animals (M. M. Bradley et al., 2001). Specifically, females often report higher levels of sadness and fear (Blanchard-Fields & Coats, 2008; Fischer et al., 2004; Kring & Gordon, 1998), although this may be related to perceived gender norms, rather than representing an objective difference in emotional experience (Grossman & Wood, 1993). There is less evidence for a gender difference in the self-reported experience of anger (Grossman & Wood, 1993), although some studies have found that males report higher levels of anger than do females (Fischer et al., 2004), and young adolescent females suppress anger more than their male counterparts (Cox et al., 2000).

These gender differences may stem from a combination of biological differences between males and females (Breedlove et al., 2010), and gender socialisation of social norms around appropriate emotional behaviour (L. R. Brody, 1997). Firstly, biological processes may play a role. When a developing foetus is exposed to androgens in the womb, this begins to masculinise the body, including the developing brain, and this process creates sexual differentiation between males and females (Breedlove et al., 2010). Sexual dimorphism of the brain has been observed in humans (Sacher et al., 2013; Seitz et al., 2020), which means that males and females have differences in the size and function of certain brain structures (Breedlove et al., 2010). This sexual dimorphism may impact on the emotional experience of males and females. For example, in adolescence, brain areas which are involved in ER and emotional experience (the amygdala and hippocampus) mature at different rates for males and females, which may result in females experiencing more negative emotions (Frere et al., 2020). Similarly, there is evidence for sexual dimorphism of the amygdala, and this may interact with the presence of hormones to contribute to females having a more intense fear response compared to males (Shansky, 2020). Therefore, these biological processes may contribute to gender differences in emotional experience.

Secondly, there is evidence that males and females are socialised to engage with specific emotions in different ways, with fear and sadness being viewed as more 'feminine', and anger being viewed as a 'masculine' emotion, according to social norms in Western societies (L. R. Brody, 1985; Root & Denham, 2010). Cultural expectations determine the appropriateness of emotional displays for males and females (Root & Denham, 2010). On a societal level, it is generally regarded as less acceptable for females to express anger without experiencing negative consequences (Brescoll & Uhlmann, 2008), and violations of these gender roles can be met with hostility and intolerance (Sanborn-Overby & Powlishta, 2020).

Furthermore, parents play a pivotal role in emotion socialisation, and evidence from parenting research indicates that parents respond in different ways to a child's emotion, based on the child's gender and the specific emotion (Root & Denham, 2010). Parents tend to talk about anger more often with their sons, but talk about sadness more often with their daughters (Fivush, 1989). Parents also express more positive emotions to their daughters, but express more anger towards their sons (Eisenberg et al., 1998). Finally, fathers are more likely to reward their daughters and punish their sons for expressing sadness and fear (Garside & Klimes-Dougan, 2002).

Biological differences and cultural socialisation are intrinsically linked, as gender differences in emotional behaviour which are driven by biological processes, however small, may be amplified and shaped by social learning (Breedlove et al., 2010). Individuals are taught essential social rules about specific emotions, particularly sadness, anger and fear. Thus, it is possible that males and females may regulate their specific emotions in different ways, based on an interaction between biological differences in the experience of emotion, and the rules learned through gender socialisation (Chaplin, 2015; Wager & Ochsner, 2005). However, at the present time, very few studies have examined gender differences in the regulation of specific emotions.

2.2.3. Gender Differences in Emotion Regulation

As defined in Section 1.3.1, ER refers to the processes by which we influence our emotional experience, including which emotions we have, when we have them, and how we experience and express these emotions (Gross, 2015a). In this section, the empirical evidence for gender differences in ER will be reviewed.

2.2.3.1. Gender Differences in Coping. As a concept, ER evolved from the coping literature (Gross, 1998b, 1999), and through research, gender differences in coping have been

found. Females are more likely to use emotion-focused coping (Brougham et al., 2009; Folkman & Lazarus, 1980), whereas males tend to use problem-focused coping (Pearlin & Schooler, 1978). This means that during a stressful situation, females focus on changing the emotional response to the stressor (e.g., venting emotions to others or ruminating), whereas males attempt to change the stressor directly (e.g., through problem solving or planning) (Lazarus & Folkman, 1984), although the findings from the ER literature have been more tentative, as discussed in Section 2.2.3.7.

This greater use of emotion-focused coping may reflect the finding that females tend to use cognitive ER more and rely on a wider variety of ER strategies compared to males, in adolescence (Sanchis-Sanchis et al., 2020) and in adulthood (Garnefski et al., 2004; Nolen-Hoeksema & Aldao, 2011; Thoits, 1991). The gender difference in emotion-focused coping may also point to a tendency for females to be more open to responding to and coping with their emotions, and for males to be less willing to focus on emotions. Females are more likely to pay attention to their emotions during regulation (Thayer et al., 2003), and are more likely to ruminate on an emotional situation (Nolen-Hoeksema & Aldao, 2011; Trives et al., 2016), whereas males are more likely to use avoidance and passivity to regulate negative emotion (Zimmermann & Iwanski, 2014).

2.2.3.2. Gender Differences in Reappraisal. As defined in Section 1.3.2.1,

reappraisal involves changing the meaning of a situation in order to lessen the resulting emotional response, and is a key strategy in Gross's (1998b) process model of ER. Reappraisal is regarded as an adaptive ER strategy, as it is effective at reducing negative emotion in a laboratory session (Gross, 1998a, 2002) and is associated with positive correlates, such as wellbeing (John & Gross, 2004; McRae et al., 2012b).

The findings regarding gender differences in reappraisal are mixed. Some studies have found that when reappraisal frequency is measured through self-report questionnaires,

females are more likely to use reappraisal to regulate their emotions compared to males (R. C. Martin & Dahlen, 2005; Smrtnik-Vitulić & Prosen, 2016; Spaapen et al., 2014), and this is consistent across adolescence (Gullone et al., 2010) and adulthood (Nolen-Hoeksema & Aldao, 2011).

In contrast, one study found that adolescent males reported using reappraisal more than females (Öngen, 2010). Males also report using positive reappraisal more than females (Esmaeilinasaba et al., 2016). Positive reappraisal is a subtype of reappraisal that involves viewing a situation in a positive light. More generally, reappraisal can also involve reinterpreting the meaning of a situation neutrally, such as describing a clinical procedure in unemotional medical terms. This finding may indicate that the specific technique used to reappraise, or the content of the reappraisal, may play a role in moderating gender differences in reappraisal, with males potentially being more likely to use positive appraisals, and females being more likely to use wider neutral appraisals.

Furthermore, some studies have found no gender difference in the use of reappraisal (Garnefski et al., 2004; Gross & John, 2003; Haga et al., 2009; Zlomke & Hahn, 2010). These studies were mostly conducted with undergraduate students, and so may highlight a difference between student and non-student populations, which has been found in psychological research more generally (Gordon et al., 1986).

However, another potential source of these mixed findings in the use of reappraisal may be that the self-report questionnaires used in these studies measure reappraisal in an emotion-general way, and so emotion-specific nuances are missed. For example, the Emotion Regulation Questionnaire (Gross & John, 2003) measures the habitual use of reappraisal to regulate both positive and negative emotion, and no differentiation is made between specific emotions. It may be the case that clearer findings emerge when the use of reappraisal to

regulate specific emotions is examined. This will be discussed further in Chapter 3, which outlines the methodological approach used in the present research to address this issue.

2.2.3.3. Gender Differences in Suppression. Expressive suppression involves keeping the facial expression neutral while experiencing an emotion, and it acts on the emotional response once it has fully developed (Gross, 1998a). Suppression is associated with negative consequences (Gross, 1998a, 2002), and so is generally regarded as a maladaptive ER strategy (Aldao et al., 2010).

Gender differences in suppression have been robustly reported in the ER literature, with males tending to report using suppression more than females (Gross & John, 2003). This has been found within community samples of adults (Spaapen et al., 2014; Zimmermann & Iwanski, 2014), undergraduate students (Gross & John, 2003; Haga et al., 2009; Kwon et al., 2013; Melka et al., 2011), and with children and adolescents (Gullone et al., 2010; Gullone & Taffe, 2012; Larsen et al., 2013; Zimmermann & Iwanski, 2014). One study of adults aged 20-65 years old found no gender difference in suppression (Smrtnik-Vitulić & Prosen, 2016).

However, the gender difference in suppression can change with age. Nolen-Hoeksema and Aldao (2011) found that older females were marginally more likely to use suppression than older males, which indicates that the relationship between gender and suppression may change throughout the lifespan. Overall, males tend to be more likely than females to report using suppression to regulate their emotions (Zimmermann & Iwanski, 2014), although this may be modulated by age (Nolen-Hoeksema & Aldao, 2011).

The gender difference in suppression may also depend on the emotion being regulated. Research has found that females are more likely than males to suppress anger, and this has been found both in adolescents (Cox et al., 2000) and in adults (Kwon et al., 2013). Therefore, it may be possible that females suppress anger more than males, but males tend to use suppression more than females overall. This further highlights the importance of

investigating gender differences in ER in an emotion-specific manner to clarify these mixed findings.

2.2.3.4. Gender Differences in Rumination. Rumination refers to persistently focusing attention onto the causes, meanings, or consequences of an emotion or emotional situation without engaging in adaptive problem solving (Nolen-Hoeksema et al., 1993; Nolen-Hoeksema et al., 2008). Rumination is a maladaptive strategy, and reliance on rumination has been implicated in contributing to depression (Nolen-Hoeksema, 1991), which will be discussed further in Section 2.3.4.1.

There is strong evidence of gender differences in rumination, with females being more likely to ruminate than males. This finding has been consistently reported in research with adults (Ando' et al., 2020; Garnefski et al., 2004; Nolen-Hoeksema & Aldao, 2011; Thomsen et al., 2005; Trives et al., 2016; Zimmermann & Iwanski, 2014), and with undergraduate students (Thayer et al., 2003; Zlomke & Hahn, 2010). In a meta-analysis of over 14,000 adults, females were found to ruminate more than males, including both brooding, which involves self-focused thought that is maladaptive in nature, and reflection, which is a more neutral and adaptive strategy (D. P. Johnson & Whisman, 2013).

Similarly, in adolescence, girls are more likely to ruminate compared to boys (Gomez-Baya et al., 2016; Hilt et al., 2010; Nolen-Hoeksema, 2001; Rood et al., 2009), and this gender difference emerges in children as young as 9 years old (Jose & Brown, 2008; Sanchis-Sanchis et al., 2020). The use of rumination increases for females, but not for males, across adolescence (Gomez-Baya et al., 2016), and gender differences in rumination are consistent across the lifespan (Nolen-Hoeksema & Aldao, 2011; Trives et al., 2016; Zimmermann & Iwanski, 2014). These gender differences in rumination are aligned with the findings from the coping literature discussed in Section 2.2.3.1, that females are more likely to use emotion-focused coping than males, because rumination is an example of emotionfocused coping. The implications of this greater use of rumination amongst females may be a risk factor for the development of clinical disorders, particularly depression, which will be discussed in Section 2.3.4.

2.2.3.5. Gender Differences in Distraction. Distraction is an ER strategy which involves diverting attention away from the emotional situation, and doing or thinking something that is unrelated instead. Distraction is generally regarded as an adaptive strategy (Dörfel et al., 2014; McRae et al., 2010), but this can depend on whether it involves an attitude of acceptance or avoidance (Wolgast & Lundh, 2017). It is common for distraction to be used as a way of coping with difficult life circumstances, particularly through activities such as gaming (Caro & Popovac, 2020). The findings regarding gender differences in distraction are mixed. Amongst adults, males are more likely than females to use distraction when they are feeling sad or depressed (Trives et al., 2016). Conversely, females are more likely to choose to distract (rather than ruminate) from anger, but this was not the case for males (Rusting & Nolen-Hoeksema, 1998).

Similarly, in adolescents aged 12-15, males used distraction in response to a low mood more than females (Gomez-Baya et al., 2016), and adolescent males are more likely to use distraction to cope with stress (Copeland & Hess, 1995). In an undergraduate sample, males were also more likely to use positive refocusing, a type of distraction which involves focusing attention on to positive things which are unrelated to the emotional situation (Esmaeilinasaba et al., 2016).

However, other studies with adults have found that females are more likely to use positive refocusing (Garnefski et al., 2004), although this was a small effect, and some studies have found no gender differences in positive refocusing amongst undergraduate students (Zlomke & Hahn, 2010).

Overall, it is unclear whether gender differences exist in the use of distraction. However, clear gender differences tend to emerge when a specific emotion is focused upon, such as low mood (which is similar to sadness) in Trives et al. (2016), and anger in Rusting and Nolen-Hoeksema (1998). Studies that find no gender differences, such as Zlomke and Hahn (2010), tend to focus on general emotion. It may be the case that males use distraction to regulate feelings of sadness and stress more and females use distraction more for anger, but this gender difference is obscured by examining aggregate emotions. Therefore, a clearer picture regarding gender differences may emerge when distraction is examined in an emotion-specific manner.

2.2.3.6. Gender Differences in Engagement and Disengagement ER. Gender differences in the use of ER strategies such as reappraisal, suppression, rumination, and distraction were reviewed in the previous sections. When examining the literature as a whole, some patterns relating to gender differences in ER can be observed. Females tend to use strategies that involve engaging with their emotions, such as rumination (D. P. Johnson & Whisman, 2013; Nolen-Hoeksema & Aldao, 2011; Zlomke & Hahn, 2010). There is some evidence that males are more likely to use strategies that involve diverting attention or disengaging from their emotions, such as avoidance or distraction (Trives et al., 2016; Zimmermann & Iwanski, 2014). The hypothesis that there are gender differences in engagement and disengagement ER has not been tested in the ER literature, and has not been examined in an emotion-specific manner.

2.2.3.7. Other Gender Differences in Emotion Regulation. The strategies reviewed so far were chosen because they represent some of the key ER strategies researched in the field. However, gender differences have emerged in other ER strategies, although there are some nuances in these gender differences. Generally, females tend to use cognitive strategies more than males, such as catastrophising (Garnefski et al., 2004), or reframing situations by

putting things into perspective (Zlomke & Hahn, 2010). When females take action to regulate their emotions, they are more likely than males to turn to others for social support (Zimmermann & Iwanski, 2014).

In contrast, one study found that males are more likely to focus on planning to regulate negative emotion, which involves thinking about what action to take to deal with the negative event (Esmaeilinasaba et al., 2016). However, compared to males, adolescent females were more likely to use problem solving (Hilt et al., 2010), and adult females were more likely to use active coping to regulate negative emotion (Nolen-Hoeksema & Aldao, 2011), which involves taking action to make a situation better and so is similar to problem solving (Trives et al., 2016) or refocus on planning (Garnefski et al., 2004; Zlomke & Hahn, 2010). Therefore, the findings regarding problem solving are tentative, and it may be the case that gender differences in the use of problem solving are moderated by different factors such as the type of situation (Tamres et al., 2002) or age (Nolen-Hoeksema & Aldao, 2011). As discussed throughout this chapter, it may also be the case that gender differences in problem solving be the case that gender differences in problem solving have be the case that gender differences in problem solving are used by differences in problem solving are solving are moderated by differences in problem solving are the type of situation (Tamres et al., 2002) or age (Nolen-Hoeksema & Aldao, 2011). As

In terms of avoidance, males may be more likely to use ER strategies which involve avoiding their emotional experience, and this gender difference is found across the lifespan (Zimmermann & Iwanski, 2014). However, this may depend on situational factors. One study found that in adolescence, females were more likely to use experiential avoidance overall, although males used avoidance more in response to external stressors, such as frustrating experiences with peers (Xavier et al., 2018). Therefore, the use of avoidance may depend on the context of the emotional situation, particularly for adolescents.

2.2.3.8. Gender Differences in Emotion-Specific Emotion Regulation. Few studies have explored gender differences in the regulation of multiple specific negative emotions.

However, a study by Rivers et al. (2007) investigated gender differences in the regulation of sadness and anger. ER was measured using a qualitative method, which involved participants free-writing about what they did to reduce feelings of anger or sadness while having a conflict with a friend. These text responses were subsequently coded into ER strategies. This method is similar to the approach used in the present research, which will be described in Section 3.5.2.1, although the resulting strategies that emerged from this method were different.

In Rivers et al. (2007), no gender differences were found in the regulation of sadness or anger using the strategies focused upon in the study. However, a limitation of the study by Rivers et al. (2007) is that it did not measure some of the most commonly-researched strategies in the literature, for which gender differences have been found in other studies, such as reappraisal, suppression, rumination or distraction (Gomez-Baya et al., 2016; Nolen-Hoeksema & Aldao, 2011; Zimmermann & Iwanski, 2014). It is possible that gender differences in emotion-specific ER may emerge when these strategies are examined. Further, the ER measured by Rivers et al. (2007) was tied to the context of interpersonal conflict. It may be the case that ER is different in other, more general contexts.

In addition, Sanchis-Sanchis et al. (2020) investigated gender differences in adolescents' use of ER to regulate sadness, anger, and anxiety. Males and females did not differ on emotion-general ER. However, when specific emotions were measured, females used both internal (e.g., reappraisal) and external (e.g., seeking social support) ER more than males to regulate sadness and anxiety. Emotion-specific gender differences in individual strategies were not reported. Therefore, whether gender differences in emotion-specific ER are found may depend on (1) the method used to measure ER (which will be discussed in Chapter 3), and (2) the ER strategies measured in emotion-specific research.

2.2.3.9. Gender Differences in Emotion Regulation Summary. Overall, there is evidence that gender differences in ER exist, with males and females often approaching the regulation of negative emotion in different ways. Females are more likely to use strategies which involve engaging with the emotion or emotional situation, such as rumination (D. P. Johnson & Whisman, 2013), whereas males tend to use strategies which involve disengaging with the emotion or emotional situation, avoidance and suppression (Trives et al., 2016; Zimmermann & Iwanski, 2014).

Furthermore, for some ER strategies, such as reappraisal and distraction, the findings regarding gender differences tend to be mixed. This may be due to the fact that most of the studies that investigate the relationship between gender and ER tend to do so in an emotion-general manner, by measuring an aggregate of negative emotions without distinguishing between specific emotions. When gender differences in ER have been investigated in an emotion-specific manner, such as in Trives et al. (2016), clear gender differences have emerged. Therefore, investigating gender differences in the use of these strategies to regulate specific emotions may bring clarity to the relationship between gender and ER.

2.3. Gender, Emotion Regulation, and Mental Health

2.3.1. Definition of Variables Associated with Mental Health

In this chapter, mental health is defined in a broad sense, which includes (1) subclinical symptomatology measured in a community sample, (2) diagnoses of clinical disorders such as depression, bipolar disorder, and generalised anxiety disorder, (3) behaviours which are linked with poor mental health, such as self-harm and suicidal ideation (O'Connor & Nock, 2014), (4) wellbeing, which is regarded as a marker of mental health (McKay & Andretta, 2017), and (5) stress, which is associated with poor mental health
(Valikhani et al., 2019). In this thesis, depressive symptoms, suicidal ideation, and self-harm behaviours are referred to as variables associated with mental health.

With regards to self-harm, there is an ongoing debate in the literature as to the best way to conceptualise self-harm behaviours. One area of contention relates to whether behaviours should be separated based on their intention. Some researchers distinguish between behaviour such as suicide attempts and non-suicidal self-injury (NSSI), which refers to engaging in self-harm behaviours without the intention of suicide (Andover et al., 2010; Bresin & Schoenleber, 2015; Sornberger et al., 2012). However, as highlighted in Kapur et al. (2013), there is little evidence to support this distinction.

There is a strong association between self-harm and suicide attempts (Duarte et al., 2020). Self-harm is one of the biggest risk factors for future suicide attempts (Andover et al., 2012; Mars et al., 2019; O'Connor et al., 2018). Also, a large proportion of individuals who self-harm report experiencing suicidal ideation while engaging in self-harm behaviours (Klonsky, 2011; Madge et al., 2008). These findings demonstrate that it is difficult to know the intent behind self-harm behaviours with any degree of confidence, and making the distinction between self-harm with and without suicidal intention creates a false dichotomy (Kapur et al., 2013).

For this reason, when referring to the present research, the term self-harm will be used, which is referring to self-harm irrespective of the intention. This is reflected in the measure of self-harm used in the present research (Deliberate Self-Harm Inventory, DSHI; Gratz, 2001), which asks participants both about their experiences of self-harm without suicidal intent (e.g., Have you ever intentionally (i.e., on purpose) cut your wrist, arms, or other area(s) of your body (without intending to kill yourself)?), and their experiences of selfharm without specifying any intention (e.g., Have you ever intentionally (i.e., on purpose)

done anything else to hurt yourself that was not asked about in this questionnaire? If yes, what did you do to hurt yourself?).

Individuals often report a mix of different intentions when engaging in self-harm behaviours (Boergers et al., 1998; Hawton et al., 1982; Hawton et al., 2012; Madge et al., 2008), and so it is likely that the DSHI captures a mix of motives, as many of the questions do not specify intent. Although the authors of the DSHI claim that it measures self-harm without suicidal intent, i.e., NSSI (Gratz, 2001), it is argued that the DSHI is likely to measure self-harm both with and without suicidal intent. In the present research, self-harm is defined as all self-injurious behaviours, regardless of the motive or extent of suicidal intent (Hawton et al., 2012).

In the present research, when referring to studies where this distinction has been made, the language of the study (e.g., NSSI) will be adopted to ensure clarity. This is because the conceptualisation of self-harm in this way will impact on the way it is measured. However, when referring to the present research, the term self-harm (irrespective of suicidal intent) will be adopted.

In addition to self-harm (irrespective of intent), suicidal ideation will also be measured in the present study, which refers to having thoughts about suicide. Suicidal ideation is a central predictor of future suicide risk (O'Connor & Nock, 2014), and so it is important to examine factors which may be related to having suicidal thoughts (such as ER). As will be discussed in this chapter, there are also gender differences in suicidal ideation. Therefore, in the present research self-harm behaviours regardless of intent, and suicidal thoughts, will be measured.

2.3.2. Gender Differences in Variables Associated with Mental Health

It is estimated that almost a billion people suffer from poor mental health worldwide (Dattani et al., 2021). Globally, approximately 280 million people are affected by depression, and more than 700,000 people die by suicide each year (World Health Organisation, 2021b, 2021c). Many gender differences are observed in these mental health problems and clinical disorders. Females often have a higher prevalence of internalising disorders, such as mood and anxiety disorders, whereas males have higher rates of externalising disorders, such as personality disorders (Eaton et al., 2012). Although self-harm behaviours are not a mental health problem, people with a diagnosis of a clinical disorder are much more likely to attempt suicide or die by suicide (Mental Health Foundation, 2016), and so self-harm behaviours are closely related to mental health. Gender differences in depression, self-harm, and suicidal ideation will be discussed in the following sections.

2.3.2.1. Gender Differences in Depression. Females are over twice as likely to experience depression compared to males (H. Chen et al., 2019; Eaton et al., 2012; Girgus & Yang, 2015; Hankin et al., 1998; Kessler, 2003; Kunst et al., 2019; Marcus et al., 2005; Nolen-Hoeksema, 1987, 2001; Nolen-Hoeksema & Aldao, 2011; Nolen-Hoeksema et al., 1999; Van de Velde et al., 2010; Weissman et al., 1996; Wetherall et al., 2020). Females have a higher lifetime prevalence of depression, with almost a quarter (22.9%) of females compared to 13.1% of males experiencing depression at some point during their lifetime (Eaton et al., 2012). Across the lifespan, females are estimated to be twice as likely as males to receive a diagnosis of clinical depression, and females demonstrate twice as many depressive symptoms as do males (Girgus & Yang, 2015).

This higher prevalence of depression amongst females is found across different cultures (H. Chen et al., 2019; Van de Velde et al., 2010; Weissman et al., 1996), age groups (Hankin et al., 1998), and in both community and clinical samples (Marcus et al., 2005;

Nolen-Hoeksema et al., 1999). Females have higher rates of both clinical diagnoses of depression and subclinical symptoms of depression (Nolen-Hoeksema, 2001). Amongst people with a diagnosis of depression, females are more likely than males to have a relapse after 7 months (Kuehner, 1999).

Depression begins to emerge in childhood and increases in prevalence during adolescence (Nolen-Hoeksema, 1994; Salk, Hyde, & Abramson, 2017). Gender differences in depression also emerge at this stage (Nolen-Hoeksema, 1994; Nolen-Hoeksema & Girgus, 1994). In a study by Hankin et al. (1998) the development of depression from adolescence into young adulthood was assessed longitudinally across a 10-year period. Although males initially had slightly higher levels of depression at age 11, females began to have higher levels of depression at age 15, which continued to increase into adulthood (Hankin et al., 1998). This demonstrates that depressive symptoms may develop at a higher rate for females compared to males.

Across adulthood, females consistently have higher rates of depression than males. In a large community sample of 7,485 participants, depressive symptoms were measured in young, middle-aged, and older adults (Leach et al., 2008). Females had higher levels of depression than males in all of the age groups. Nolen-Hoeksema and Aldao (2011) also found that depressive symptoms are consistently higher for females across adulthood. Therefore, there is strong evidence that females tend to have higher rates of depression than males in adolescence and in adulthood.

2.3.2.2. Gender Differences in Self-Harm and Suicidal Ideation. There is also evidence for gender differences in self-harm behaviours (Hawton, 2000). In Scotland, females are more likely to self-harm than males, with 9% of Scottish adult females reporting engaging in self-harming without suicidal intent compared to 5% of males (Knudson et al., 2020). This gender effect is particularly pertinent in adolescence, with adolescent females

being at least three times, and up to six times more likely to report self-harming regardless of the intention, compared to their male counterparts (Hawton et al., 2012; Madge et al., 2008; O'Connor et al., 2009). In a large study of 2008 Scottish school pupils aged 15-16 years old, 5.1% of boys, and 13.6% of girls reported self-harming irrespective of intent, in the past year (O'Connor et al., 2009). In terms of lifetime prevalence, almost 20% of girls in the sample and 6.9% of boys reported a lifetime prevalence of self-harm (O'Connor et al., 2009). Similarly, in the Child and Adolescent Self-Harm in Europe (CASE) study of over 30,000 adolescents across 7 countries, self-harm irrespective of the intention was over twice as common in females as in males (Madge et al., 2008).

In terms of NSSI, which refers to the measurement of self-harm without suicidal intent, females in North America are 1.5 times more likely to engage in NSSI than males (Bresin & Schoenleber, 2015). Females report that they began to engage in NSSI at a younger age (aged 11.57 years on average) compared to males (aged 13.83 years old on average) (Andover et al., 2010). In 2018/2019, 22% of Scottish females aged 16-24 reported self-harming without suicidal intent, compared to 9% of males in this age group (Knudson et al., 2020). This gender difference is most prominent in the youngest age group and decreases across the lifespan (Knudson et al., 2020). Self-harm without suicidal intent tends to decrease with age, and is least prevalent in older age groups (McManus et al., 2019).

When asked whether they had ever attempted to take their own life, 9% of adult females compared to 6% of males in Scotland reported ever attempting suicide by taking an overdose, or through other means (Knudson et al., 2020). Amongst undergraduate students, females are more likely than males to report attempting to take their own life (Sivertsen et al., 2019). Gender differences in self-harm with suicidal intention begin to emerge in adolescence, with adolescent females being twice as likely to report self-harming with

suicidal intention than their male counterparts (C. S. Lee & Wong, 2020; Lewinsohn et al., 2001).

There is also evidence that females are more likely than males to report having thoughts about suicide (Hunt et al., 2006; C. S. Lee & Wong, 2020; Sivertsen et al., 2019; Stephenson et al., 2006). This gender difference may depend on age, as females were found to report higher suicidal ideation than males amongst young adults, but this was not found in older adults (Hunt et al., 2006), suggesting that this gender difference may disappear as people get older. However, in contrast to this, in a study of Scottish young people aged 18-34, males were more likely to have suicidal thoughts but not attempt suicide, whereas females were more likely to attempt suicide (Wetherall et al., 2018). This may indicate that gender differences in suicidal ideation may vary by culture.

Although some studies have found no gender differences in self-harm regardless of intent (Lundh et al., 2007), this may be due to the measure of self-harm used. When different methods of self-harm are distinguished and measured separately, clear gender differences emerge, with females being more likely to use cutting methods (Lundh et al., 2007), and males tending to use hitting or burning behaviour (Andover et al., 2010; Idig-Camuroglu & Gölge, 2018). Further, studies which fail to find gender differences in self-harm tend to be conducted in countries outside of Scotland and the UK, and there is evidence that there are cultural differences in self-harm and suicide rates (Brock et al., 2006; Madge et al., 2008), as well as differences in the measurement of self-harm (Kapur et al., 2013), as highlighted in Section 2.3.1.

However, there is a gender paradox in self-harm behaviour. Compared to males, females are more likely to engage in self-harm behaviour with or without suicidal intent (Hawton et al., 2012; Madge et al., 2008; Miranda-Mendizabal et al., 2019; O'Connor et al., 2009; Wetherall et al., 2018). However, males are more likely to die by suicide (National

Records of Scotland, 2021). In 2020, 71% of the people in Scotland who died by probable suicide were male (National Records of Scotland, 2021). This gender difference is a robust finding (Dougall et al., 2017; Miranda-Mendizabal et al., 2019; P. L. Mok et al., 2012; Stark et al., 2004) that is found across different age groups (S. Lee et al., 2019) and cultures (Jordan & McNiel, 2020; S. Lee et al., 2019; Mościcki, 1994). One theory for this gender paradox is that although females are more likely to attempt suicide, males tend to use more lethal methods of suicide, and so this may explain why death by suicide is more common amongst males (Ajdacic-Gross et al., 2008).

Overall, there are prominent gender differences in variables relating to mental health. Females have higher rates of depression and self-harm behaviour, irrespective of suicidal intent. However, males are more likely to die by suicide. There are many factors which contribute to poor mental health, including biological, social, and psychological influences (Parker, 2019). However, one factor which impacts on mental health is ER, which will be discussed in the next section.

2.3.3. The Relationship Between Emotion Regulation and Variables Associated with Mental Health

The way that emotions are regulated is closely related to mental health, including subclinical symptomatology (Garnefski & Kraaij, 2006; Gonçalves et al., 2019), clinical disorders (Aldao et al., 2010; S. J. Bradley, 1990; Cludius et al., 2020; Dryman & Heimberg, 2018; Fitzgerald et al., 2019; Sheppes et al., 2015), markers of poor mental health such as stress and wellbeing (McRae et al., 2012b; Zahniser & Conley, 2018), and behaviours which are associated with poor mental health, such as self-harm and suicidal ideation (Albanese et al., 2019; Brereton & McGlinchey, 2020). The relationship between reappraisal, suppression and variables associated with mental health will be discussed in the following sections, before

describing Response Styles Theory, which addresses the relationship between rumination, distraction, and depression.

2.3.3.1. Reappraisal and Mental Health. There is a large body of work which demonstrates that reappraisal may have a positive impact on mental health (Aldao et al., 2010). In correlational research, less frequent reappraisal is associated with higher levels of depressive symptoms amongst adults (Aldao et al., 2010; Garnefski & Kraaij, 2006; Hu et al., 2014), adolescents (Garnefski & Kraaij, 2006; Shapero et al., 2019) and undergraduate students (Gross & John, 2003; Haga et al., 2009; R. C. Martin & Dahlen, 2005). More frequent use of reappraisal is also linked with lower levels of social anxiety (Kivity & Huppert, 2019), general anxiety, and stress (R. C. Martin & Dahlen, 2005) in community samples.

These correlational findings are supported by prospective evidence. In undergraduate students, reappraisal predicts lower depressive symptoms, stress and anxiety (Brewer et al., 2016; Zahniser & Conley, 2018), and higher self-efficacy, hope and resilience (which are markers of wellbeing) over the academic year (Brewer et al., 2016). These findings indicate that reappraisal may buffer against the development of internalising symptoms such as anxiety and depression amongst undergraduate students.

There is also evidence that reappraisal is related to experiencing fewer clinical symptoms in diagnoses of clinical disorders. In cross-sectional research, using reappraisal less often is associated with higher depressive symptoms (Garnefski & Kraaij, 2006) and lower wellbeing (Kraiss et al., 2020) amongst people with any clinical diagnosis. Similarly, individuals with clinical depression (Kanske et al., 2012) and bipolar disorder (Kjærstad et al., 2016) are less effective in using reappraisal to downregulate amygdala responses to emotional stimuli compared to control groups.

Longitudinally, amongst individuals diagnosed with bipolar disorder, reappraisal predicted a decrease in depression 12 months later, while controlling for baseline levels of depression (S. L. Johnson et al., 2016). These findings show that reappraisal may be linked with lower symptoms amongst those with a clinical disorder, and prospective evidence may implicate reappraisal as a protective factor against the development of depressive symptoms.

Reappraisal is also associated with behaviours that are related to poor mental health, such as self-harm (T. S. Davis et al., 2014), or having thoughts about suicide (Kudinova et al., 2016). Adults with a history of self-harm regardless of intent were less effective at reducing sadness using reappraisal during an emotion regulation task, compared to people with no history of self-harm, even while controlling for depressive symptoms (T. S. Davis et al., 2014). However, the use of reappraisal has been found to reduce negative emotion amongst individuals who engage in self-harm behaviours (In et al., 2021).

In terms of suicidal ideation, one study used electroencephalography (EEG) to measure the late positive potential (LPP), an event-related potential which represents attention to emotional stimuli (Kudinova et al., 2016). Individuals with a history of suicidal ideation had a higher LPP after being asked to use reappraisal compared to those with no history of suicidal ideation, which demonstrates that the individuals with a history of suicidal ideation may have had more difficulty implementing reappraisal during the task (Kudinova et al., 2016). These findings demonstrate that being less skilled in the use of reappraisal may be associated with higher levels of self-harm and suicidal ideation.

Overall, these findings indicate that frequent and effective use of reappraisal is generally associated with greater mental health, both in clinical and sub-clinical samples, and longitudinal research may implicate reappraisal as a contributing factor to mental health.

2.3.3.2. Suppression and Mental Health. As with reappraisal, suppression has also been linked with mental health in the literature, although suppression is generally regarded as

a maladaptive strategy (Aldao et al., 2010). In correlational research, greater use of suppression is associated with higher depressive symptoms amongst adults (Hu et al., 2014) and undergraduate students (Haga et al., 2009). Using suppression more frequently is also associated with higher anxiety and negative affect, whereas less use of suppression is associated with positive affect and higher life satisfaction (Hu et al., 2014). The link between suppression and mental health is stronger in Western cultures (Hu et al., 2014).

Longitudinally, the use of suppression predicts higher sub-clinical symptoms of depression and anxiety amongst undergraduates across the academic year, while controlling for stress (Zahniser & Conley, 2018). Similarly, in a community sample of adults, greater use of suppression at baseline predicted lower psychological wellbeing 2.5 years later (Kelley et al., 2019). Conversely, one study found that suppression did not predict anxiety and depression longitudinally, although it was associated with lower life satisfaction after 8 months, which is a marker of wellbeing (Brewer et al., 2016).

In terms of self-harm and suicidal ideation, suppression was associated with higher levels of suicidal ideation amongst individuals with a diagnosis of a clinical disorder, while controlling for symptoms of depression (Forkmann et al., 2014). Furthermore, females with a diagnosis of borderline personality disorder who used suppression to reduce feelings of sadness during an experimental task did not report an increase in the urge to self-harm (Svaldi et al., 2012). In the same study, those who used acceptance did have an increase in the urge to self-harm five minutes later (Svaldi et al., 2012), which indicates that suppression may have an adaptive role amongst this group. Many theorists suggest that self-harm may be a technique used by individuals to avoid negative emotion (Chapman et al., 2006; Klonsky, 2007; Slee et al., 2008). Therefore, it may be beneficial to examine if the regulation of specific emotions is differentially associated with self-harm and suicidal ideation.

2.3.4. Explaining Gender Differences in Depression: Response Styles Theory

As described in Section 2.3.2.1, there are prominent gender differences in depression. Response styles theory (RST) (Nolen-Hoeksema, 1987) is a theoretical framework which attempts to explain these gender differences. There are many different theories of why gender differences in depression exist, including gender differences in biological processes (Albert, 2015), gender roles (Rosenfield, 1980), and levels of stress (Nolen-Hoeksema et al., 1999). However, an advantage of RST is that it takes response styles into account in explaining the relationship between gender and depression. Response styles are the different ways that individuals can respond to experiencing a depressed mood, and this is closely related to ER. That is to say, the ER strategies used in this context of a low mood, such as rumination, are a type of response style. We know from the review of the literature in this chapter that (1) there are gender differences in ER, and (2) ER may be a contributor to mental health. Therefore, RST was focused upon in the present research as it is the only theory which takes the role of ER into account when attempting to explain gender differences in the prevalence of depression.

According to RST, the way that individuals respond to a depressed mood can contribute to the onset and maintenance of depression (Nolen-Hoeksema, 1987, 1991; Nolen-Hoeksema et al., 1999; Nolen-Hoeksema et al., 1993; Nolen-Hoeksema et al., 2008). Certain response styles, such as distraction and problem solving, are regarded as adaptive and result in lower depressive symptoms. On the other hand, ruminating in response to a low mood is expected to contribute to and exacerbate depression.

ER can be regarded as a trans-diagnostic process, which means that the impact of ER spans across different clinical disorders (Aldao et al., 2016; Cludius et al., 2020). ER strategies, such as rumination, have been shown to contribute to a wider range of psychopathologies than initially thought, such as anxiety, binge eating, and self-harm (Nolen-

Hoeksema et al., 2008). Other ER strategies such as reappraisal and suppression have also been shown to impact many aspects of mental health and associated behaviours (Aldao et al., 2010). Therefore, it is proposed that the principles of RST can be applied to other variables that are associated with mental health (such as depressive symptoms, self-harm behaviours, and suicidal ideation), and can provide a framework for understanding the relationship between gender, ER and these variables more widely. It is possible that gender differences in ER may contribute to gender differences in variables associated with mental health. Further, no known studies to date have examined this in an emotion-specific manner.

According to RST, females are more likely to ruminate in response to a depressed mood, and this will result in higher rates of depression amongst females (Nolen-Hoeksema et al., 1999). Conversely, males are more likely to use distraction and problem solving in response to a low mood, which will result in lower levels of depression (Nolen-Hoeksema, 1991). This means that regardless of the source of depressive symptoms (whether it is biological or psychological), the approach that males take in responding to these feelings may be more adaptive and result in less depressive symptoms compared to females (Nolen-Hoeksema, 1987).

Therefore, according to RST, gender differences in ER can explain, at least partially, gender differences in rates of depression, which is depicted in Figure 2.1. In addition to response styles, it is also proposed that females experience more chronic strain and have a lower sense of mastery in their lives, and these factors also contribute to higher rates of depression amongst females (Nolen-Hoeksema et al., 1999).

Thus, the key tenets of RST are (1) there are gender differences in rumination and distraction, (2) rumination and distraction are associated with depression in different ways, and (3) gender differences in rumination and distraction can explain gender differences in depression. We know from Section 2.2.3.4 that there is evidence to support the first tenet of

RST. There is strong evidence that females are more likely to use rumination to regulate their emotions compared to males (D. P. Johnson & Whisman, 2013), and this is found consistently across the lifespan (Nolen-Hoeksema & Aldao, 2011; Trives et al., 2016; Zimmermann & Iwanski, 2014). Furthermore, there is some evidence that males are more likely to use distraction, but clearer findings emerge when specific emotions are measured (Trives et al., 2016). In terms of the link between rumination, distraction, and mental health, the evidence will be discussed below.

Figure 2.1





2.3.4.1. Rumination and Mental Health. In correlational research, people who rely on rumination and worrying to regulate their emotions tend to have poorer mental health, and higher levels of clinical symptoms such as depression and anxiety (Dixon-Gordon et al., 2015b). Rumination is implicated in almost all clinical disorders (Ehring & Behar, 2020), and

high levels of rumination are associated with depression (both clinical depression and subclinical symptoms; Watkins, 2008), as well as many other disorders such as social anxiety, eating disorders, insomnia, bipolar disorder, and panic disorder (Ehring & Watkins, 2008). Rumination is also negatively associated with markers of mental health, such as wellbeing (Kraiss et al., 2020). Rumination (particularly rumination which focuses on the symptoms of depression) also predicts depression 4 weeks, 6 months and 3.5 years later, although this was not the case for those with a previous diagnosis of depression (Huffziger et al., 2009).

A study by Peled and Moretti (2010) examined the rumination of sadness and anger in an emotion-specific manner. It was found that using rumination to regulate sadness had a unique relationship with depression, and using rumination to regulate anger had a unique relationship with aggression (Peled & Moretti, 2010). This is particularly important for the present research because it demonstrates that using rumination to regulate specific emotions has a different impact on variables associated with mental health, such as aggression and depression. Thus, it is possible that this is also the case for other ER strategies.

In terms of gender differences, there is also evidence that rumination mediates the relationship between gender and depression, which is in line with RST. In a study by Thayer et al. (2003), gender differences in ER were found to account for gender differences in depression in a sample of undergraduate students. Specifically, females were more likely than males to pay attention to their emotions during regulation. When gender differences in ER were controlled for, the gender difference in depressive symptoms disappeared. This has also been found in adolescents aged 10-17, with rumination explaining the gender difference in depression (Jose & Brown, 2008).

Similarly, in a longitudinal study of adults aged 25 to 75 years old, the effects of experiencing chronic strain, a lower sense of mastery, and the use of rumination on

depression were assessed over a year (Nolen-Hoeksema et al., 1999). Females had higher rumination, chronic strain, and depression, and a lower sense of mastery compared to males. Further, rumination, mastery and chronic strain fully mediated the gender differences in depression. Chronic strain was indirectly related with depression, through rumination (Nolen-Hoeksema et al., 1999). These findings thus provide support for RST and implicate gender differences in rumination as a contributory factor in the observed gender differences in depression.

2.3.4.2. Distraction and Mental Health. Compared to rumination, there are fewer studies considering the relationship between distraction and variables associated with mental health. In correlational research, greater use of distraction is associated with less chance of receiving a diagnosis of depression (Polanco-Roman et al., 2015), less frequent NSSI, and fewer instances of self-harm with suicidal intent (Polanco-Roman et al., 2015). Amongst individuals receiving CBT for clinical depression, individuals who used distraction to regulate their negative emotions were judged by their practitioner as being more engaged in their treatment, and greater distraction predicted more positive feedback from the individuals about the therapy, which is a marker of success of the therapeutic process (Teismann et al., 2012). Distraction has been found to effectively reduce the urge to engage in NSSI after a negative mood induction, amongst individuals who engage in self-harm behaviours (In et al., 2021). Further, intuitively using distraction and positive activities as a way of regulating emotions was associated with less intense suicidal thoughts amongst individuals with a diagnosis of a mood disorder or borderline personality disorder (Stanley et al., 2021).

In longitudinal research, amongst adults with a previous diagnosis of depression, higher distraction at baseline predicted lower levels of subsequent depression 4 weeks, 6 months and 3.5 years later (Huffziger et al., 2009). This is in contrast with the finding that rumination did not predict subsequent depression amongst those with a diagnosis of

depression, but did for those with sub-clinical symptoms (Huffziger et al., 2009). This may indicate that distraction is more strongly related to more severe clinical diagnosis of depression, and so may be a protective factor in clinical depression.

Furthermore, a study by Wolgast and Lundh (2017) highlighted a distinction between different types of distraction and the implications of this for mental health. It is possible to use distraction in two ways – (1) distracting while maintaining an attitude of acceptance towards negative emotions, or (2) distracting in order to avoid the experience of the emotion. Individuals who used distraction paired with acceptance had higher positive emotion, lower negative emotion, and greater wellbeing compared to those who used distraction to avoid negative emotion (Wolgast & Lundh, 2017). Further, people with a diagnosis of depression were less likely to use distraction with acceptance and more likely to use distraction with avoidance, compared to a non-clinical sample (Wolgast & Lundh, 2017). Therefore, there is evidence that distraction is generally adaptive for mental health, with greater use of distraction being associated with lower symptoms, although this depends on whether the distraction involves acceptance or avoidance of the emotion.

2.3.4.3. Summary of Response Styles Theory. To summarise, according to RST, females are more likely to ruminate, and males are more likely to distract in response to a depressed mood, and this results in higher and lower levels of depression respectively. Evidence from the ER literature shows that females do indeed ruminate more (D. P. Johnson & Whisman, 2013), but the evidence regarding distraction is mixed. However, when specific emotions are focused upon, males tend to use distraction more to regulate sadness (Trives et al., 2016), whereas females prefer distraction to rumination when they feel angry, which is not the case for males (Rusting & Nolen-Hoeksema, 1998).

In terms of the impact of rumination and distraction on mental health, rumination tends to have a negative impact on mental health, with greater rumination being associated

with a range of different clinical disorders (Ehring & Watkins, 2008), and predicting depression longitudinally in a community sample (Huffziger et al., 2009). Greater use of distraction tends to predict lower depression over time amongst people with a clinical diagnosis of depression (Huffziger et al., 2009). One study found that rumination (along with mastery and chronic strain) fully explained gender differences in depression (Nolen-Hoeksema et al., 1999). Therefore, there is strong evidence to support the key tenets of RST, and this framework may help to explain the relationship between gender, ER and variables associated with mental health.

2.4. Theoretical Basis for Empirical Studies

Now that the key theories in the emotion and ER literatures have been discussed in Chapter 1, and the relationships between gender, ER, and mental health have been described in the present chapter, in this section the theoretical background for the empirical studies in the present research will be explained. Study 1 involved categorising raw data into ER strategies, and so it was important to choose theories that were clear on how ER strategies are defined and classified. Two theories were used as a basis for Study 1 – the process model of ER, and the Parkinson and Totterdell (1999) taxonomy of affect regulation. These theories were chosen because they each provide a clear framework for categorising ER strategies.

Study 1 involved text responses being coded into individual ER strategies, and wider categories of engagement, disengagement and distraction, in order to examine the relationship between gender and emotion-specific ER. The process model was firstly chosen as a framework for categorising individual ER strategies because it provides guidance on how to distinguish between different ER strategies on a temporal basis. The process model distinguishes between different strategies according to the point at which they intervene in the unfolding of an emotion. This framework was used to categorise the raw data into some

of the key strategies in the field, such as avoidance, reappraisal, suppression, and rumination. In addition to this, other strategies which the process model defines less clearly, such as acceptance, were also used.

The next step was categorising these ER strategies into wider categories of ER. The Parkinson and Totterdell (1999) taxonomy was chosen for these wider categories because it provides clear guidance for how these strategies should be categorised. The Parkinson and Totterdell (1999) framework is empirically-derived, and using this framework ensured that the categorisation process was consistent. It is important to note that the main theories of ER complement, rather than contradict, one another. This means that ER strategies that were categorised according to the process model, were also consistent with the Parkinson and Totterdell (1999) framework.

Study 2 was split into two parts. In Study 2a, gender differences in emotion-specific ER were investigated using an emotion-regulation task. Specifically, how effectively individuals used two ER strategies (cognitive reappraisal and cognitive distraction) was examined. The process model was used as the theoretical basis for Study 2a. The process model was chosen because it provides an explanation for why some ER strategies are more effective than others. Study 2a builds on the findings of the process model by comparing the effectiveness of ER strategies in regulating specific emotions. In other words, Study 2a was testing the process model by examining ER effectiveness in an emotion-specific manner.

In Study 2b, the relationships between gender, emotion-specific ER, and mental health were explored. The key theory in the literature which has focused on these relationships is Response Styles Theory (RST). According to RST, the way that individuals respond to their emotions (i.e., their ER strategies) can have an impact on mental health, and can partly explain gender differences in mental health. Therefore, RST was chosen as a way of understanding the relationships between gender, emotion-specific ER, and mental health.

The present research builds on the findings of RST by examining these relationships in an emotion-specific manner.

2.5. Conclusion and Next Steps

In this chapter, it was established that (1) there are gender differences in ER, (2) ER may contribute to mental health, and (3) there are gender differences in variables associated with mental health, including depression, self-harm, and suicidal ideation. Based on this, it is logical that gender differences in ER may contribute to the gender differences observed in variables associated with mental health. This has been examined within the context of RST, which examines the role of ER in depression, but these principles may apply to other variables that are associated with mental health. No studies to date have examined this relationship in an emotion-specific manner. It is important to do so, because the relationships between gender, ER, and variables associated with mental health may depend on the emotion being regulated. Examining these questions in an emotion-specific manner may provide essential information to guide therapeutic practice by ensuring treatment is tailored to the individual's needs.

Now that the importance of investigating gender differences in ER in an emotionspecific manner has been discussed, the next question to address is how ER is measured. As with many psychological constructs, measuring emotion can be challenging, and measuring specific emotions poses unique challenges. Different methodological approaches to measuring ER can offer different information about how an individual regulates their emotions, such as how often they use a particular strategy, and how effectively they can implement it. As discussed in Chapter 1, specific emotions differ in their subjective experience (Cowen et al., 2020; Cowen & Keltner, 2017; Lench et al., 2011). Therefore, it is crucial to include a measure of subjective experience in any research examining the regulation of specific emotions. In Chapter 3, the methodological approaches to measuring ER will be discussed.

KEY POINTS FOR CHAPTER 2

- There are gender differences in variables associated with mental health. There is also evidence that the use of ER strategies may be associated with some mental health problems.
- 2. Research has found that gender differences in ER can account for the gender differences observed in rates of depression. ER may explain some of the gender differences found in mental health more widely.
- 3. However, no studies to date have examined this in an emotion-specific manner. It is important to do so, because the relationships between gender, ER, and mental health may depend on the specific emotion being regulated.

QUESTIONS FOR SUBSEQUENT CHAPTERS

- 1. How are specific emotions measured in ER research?
- 2. How is the subjective experience of emotion measured in research?
- 3. What do different ER measures tell us about how individuals regulate their emotions?

NEXT STEPS

Now that the ER literature has been reviewed, the next issue to address is how research which measures ER in an emotion-specific manner is conducted. In Chapter 3, the different methodological approaches to measuring ER will be discussed. In particular, this discussion will focus on (1) what different ER constructs tell us about how individuals manage their emotions, (2) the unique challenges around measuring specific emotions, and (3) how subjective experience is measured in emotion research.

Chapter 3: Methodological Approaches to Emotion Regulation and the Methods used in the Present Research

3.1 Chapter Summary

In the previous chapters, it was highlighted that there are gender differences in ER, and ER may explain some of the observed gender differences in mental health. However, no study to date has examined the relationships between gender, ER, and variables associated with mental health in an emotion-specific manner, by measuring specific emotions such as sadness, anger, and fear individually. This is problematic because it is possible that gender differences in ER, and the relationship between ER and variables associated with mental health, may depend on the emotion being regulated.

A challenge in conducting this emotion-specific research is that there are presently few methods available that allow for ER to be measured in an emotion-specific manner. Consequently, methodological approaches must be carefully chosen which allow specific emotions to be measured individually. In Chapter 1, the importance of subjective experience was also discussed. Research evidence indicates that subjective experience is the emotional component that demonstrates the largest difference between specific emotions. Therefore, using a measure of subjective experience is important in emotion-specific ER research.

In this chapter, the different methodological approaches to ER research will be reviewed. These approaches include measuring the frequency and effectiveness of ER using a variety of methods such as self-report questionnaires, and the emotion regulation task (ERT). Next, the importance of measuring subjective experience in emotion-specific research will be discussed. The research questions which drive the present research, and how these will be addressed using the two methodological approaches chosen for the present research, will also be discussed.

3.2. The Importance of Measuring Emotion Regulation in an Emotion-Specific Manner

As discussed in Chapter 2, the majority of ER research relates to how people regulate general emotion, such as measuring the strategies individuals use to regulate negative and positive emotion as a general trait (Gross & John, 2003), or assessing how well an individual can use a particular strategy during an ER task in response to emotionally evocative stimuli, without measuring specific emotions (Gross, 1998a). In comparison, relatively fewer studies have assessed the regulation of specific emotions, such as sadness, anger and fear.

This is problematic for three reasons. Firstly, ER develops in an emotion-specific manner across adolescence (Zimmermann & Iwanski, 2014), and specific emotions tend to be regulated using different strategies (Dixon-Gordon et al., 2015a; Rivers et al., 2007; Vishkin et al., 2020; Zimmermann & Iwanski, 2014). Thus, it cannot be assumed that the way one emotion, or negative emotion in general, is regulated can be extrapolated to other specific emotions. Secondly, there is evidence that males and females may regulate their specific emotions differently (Sanchis-Sanchis et al., 2020), and so gender differences in ER may depend on the emotion being regulated. Thirdly, the regulation of specific emotions may have different consequences, such as on mental health (Boland et al., 2019; Clear et al., 2019). Therefore, in order to fully understand ER, we must distinguish between specific emotions (P. M. Cole, 2014). This presents a valuable opportunity to investigate the emotion-specific nuances which may exist in the relationship between gender, ER, and variables associated with mental health.

However, measuring the regulation of specific emotions presents unique challenges. There are limited emotion-specific self-report instruments available. Few self-report questionnaires measure the regulation of specific emotions, and those that do so are limited in several ways. Furthermore, eliciting specific emotions in an experimental setting can be

challenging, and few studies in the field have done so. However, actions can be taken to address these challenges, which will be discussed in the following sections.

3.3. Key Methods for Measuring Emotion Regulation

The two most common methods for measuring ER are (1) measuring how often a person uses a particular ER strategy (ER frequency), and (2) measuring how skilled an individual is at using a particular ER strategy (ER effectiveness). As with all measures, there are strength and limitations of each approach, and these will be discussed in the following sections.

3.3.1. Emotion Regulation Frequency

ER frequency is how often an individual uses an ER strategy to regulate their emotions. ER frequency is measured through self-report questionnaires, which typically ask participants closed-ended questions regarding the extent to which they use certain ER strategies, often using a Likert scale. Self-report questionnaires of this nature are one of the most common methods of measuring ER. Some of the most widely used questionnaires are described in Table 3.1.

Self-report questionnaires are a useful method for measuring individual differences in the use of ER strategies, and using this method has produced informative findings about the implications of habitual ER on a range of variables, such as relationships and wellbeing (Gross & John, 2003). The use of self-report questionnaires has also highlighted some important emotion-general gender differences in ER, such as that females are more likely to ruminate (Nolen-Hoeksema et al., 1999), and males are more likely to suppress their emotions (Gross & John, 2003).

Table 3.1.

ER Measure	ER strategies	Context	Samples	Emotion- specific?	Gender differences
Behavioural Emotion Regulation Questionnaire (BERQ; Kraaij & Garnefski, 2019)	Seeking distraction, Withdrawal, Actively approaching, Seeking social support, Ignoring the situation	Behavioural ER during stressful life events	Adolescents, Adults	No	Not examined
Cognitive Emotion Regulation Questionnaire (CERQ; Garnefski et al., 2002a)	Self-blame, Acceptance, Rumination, Positive refocusing, Refocus on planning, Positive reappraisal, Putting into perspective, Catastrophising, Other-blame	Cognitive ER during stressful life events	Adolescents, Adults	No	Females use rumination, catastrophising, and positive refocusing more than males (Garnefski et al., 2004).
Difficulties in Emotion Regulation Scale (DERS; Gratz & Roemer, 2004)	Emotion dysregulation: non-acceptance of emotional responses, difficulties with goal- directed behaviour, difficulties with impulse control, lack of emotional awareness, limited access to ER strategies, lack of emotional clarity	Difficulties with ER	Children, Adolescents, Adults	No	Adolescents: females have greater difficulty regulating negative emotions, less access to effective ER, greater non-acceptance of emotions and less emotional clarity (Bender et al., 2012). Adults: males have lower emotional awareness (Gratz & Roemer, 2004)
Emotion Regulation Inventory (ERI; König, 2011)	Controlled expression, Uncontrolled expression, Empathic suppression, Distraction, Reappraisal	Habitual ER in daily life	Adults	No	Not examined
Emotion Regulation of Others and Self Scale (EROS; Niven et al., 2011)	Extrinsic affect-improving, Extrinsic affect-worsening Intrinsic affect-improving, Intrinsic affect-worsening	ER of the self (intrinsic) and others (extrinsic)	Adolescents, Adults, Clinical patients	No	Females use extrinsic affective-improving strategies more, and are less likely to use extrinsic affect- worsening strategies than males (Niven et al., 2011).
Emotion Regulation Questionnaire (ERQ; Gross & John, 2003)	Cognitive reappraisal, Expressive suppression	Gross's (1998) Process Model of ER	Adults (but adaptations are available for other samples)	No	Mixed findings: No gender difference (Malesza, 2019). Males use suppression more than females (Gross & John, 2003).

Popular Self-Report Questionnaires Used to Measure ER

Table 3.1

ER Measure	ER strategies	Context	Samples	Emotion- specific?	Gender differences
FEEL-KJ (Cracco et al., 2015)	Problem Solving, Distraction, Forgetting, Acceptance, Humour Enhancement, Cognitive Problem Solving, Revaluation, Giving Up, Withdrawal, Rumination, Self- Devaluation, Aggressive Actions	ER in children and adolescents	Children, Adolescents	Yes	The use of humour enhancement and cognitive problem solving decreases, and the use of rumination and self-devaluation increases for females across adolescence, but not for males (Cracco et al., 2017).
Negative Emotion Regulation Inventory (NERI; Zimmermann et al., 2008)	Adaptive ER, Social support, Passivity, Avoidance, Expressive suppression, Dysfunctional rumination, Dysregulation	Specific emotional contexts, development of ER	Adolescents, Adults	Yes	Females use social support seeking, and dysfunctional rumination more. Males use passivity, avoidance, and suppression more (Zimmermann & Iwanski, 2014).
Regulation of Emotions Questionnaire (REQ; Phillips & Power, 2007)	Internal- dysfunctional ER, Internal-functional ER, External- dysfunctional ER, External-functional ER	Functional and dysfunctional ER	Adolescents	No	Adolescent females are more likely to use external-functional ER (Kullik & Petermann, 2013).
Ruminative Response Scale (RRS; Treynor et al., 2003)	Rumination	Rumination as a response style to depression, Response Styles Theory	Adolescents, Adults, Clinical patients	No	Females are more likely to ruminate than males (Nolen-Hoeksema et al., 1999).

Popular Self-Report Questionnaires Used to Measure ER (Continued)

However, a limitation of this approach is that many self-report questionnaires do not measure ER in an emotion-specific manner. That is to say, these questionnaires tend to ask individuals to report how they regulate negative emotions in general, or how they regulate their emotions in a certain context such as a stressful life event. For example, the Emotion Regulation Questionnaire (ERQ; Gross & John, 2003), which is perhaps the most popular measure of habitual reappraisal and suppression in the field, asks individuals to report how they attempt to increase positive emotion, or decrease negative emotion, and this is used to calculate separate reappraisal and suppression scores. Thus, the ERQ measures the extent to which an individual uses reappraisal and suppression to regulate emotion generally. In other words, ER is measured as a general trait, rather than a skill which depends on the emotional context.

Emotion-general ER has been linked with a number of important life outcomes (Gross & John, 2003), and so these measures have an important place in the field of ER. However, it is also possible that nuances in the way that individuals regulate specific emotions and the relationship this has with correlates of ER (such as mental health) may be missed when ER is only examined in an emotion-general manner.

To date, there are only two available self-report questionnaires that measure the regulation of specific emotions, the FEEL-KJ (Cracco et al., 2015) and the Negative Emotion Regulation Inventory (NERI) (Zimmermann et al., 2008). The FEEL-KJ measures the use of adaptive and maladaptive ER to regulate anxiety, sadness and anger in children and adolescents. However, no version has yet been validated for use with other samples, such as adults (Cracco et al., 2015). Although the FEEL-KJ measures a wide range of strategies, it does not include some of the most common strategies published in the literature, such as reappraisal and suppression, which are associated with important variables.

Similarly, the NERI asks individuals to report how they respond to situations where they experience joy, sadness, anger or fear (Zimmermann et al., 2008). The NERI is the most comprehensive self-report measure of emotion-specific ER to emerge in the literature, and so represents a step forward for the field of ER. However, as with the FEEL-KJ, the NERI also does not include some of the most well-researched strategies, such as reappraisal, and so is limited as a tool to further explore reappraisal in different emotional contexts.

This limitation reflects a constraint of ER questionnaires generally, which is that only a small number of strategies can be measured in any one session. When also attempting to measure ER in different emotional contexts, this adds an additional layer which can begin to make questionnaires long and cumbersome for participants to complete. Generally, shorter questionnaires have higher response rates (Rolstad et al., 2011). Therefore, there is a gap in the ER literature for a methodological approach which captures the variety of different ways that individuals regulate their specific emotions, and is also short enough to reduce response burden, particularly for younger participants. In Section 3.5.2.1, an approach to measuring emotion-specific ER is presented to address this limitation, which was implemented in the present research. Overall, ER questionnaires are useful tools to use in ER research but have limitations in what they can be used for, particularly if the aim is to measure the regulation of specific emotions.

3.3.2. Emotion Regulation Effectiveness

The second key method for measuring ER is by assessing how skilled an individual is at reducing negative emotion in an experimental setting, after being instructed to use a particular ER strategy, such as reappraisal or suppression. This construct is often referred to as ER effectiveness (Theurel & Gentaz, 2018), ER ability (McRae et al., 2012b), or ER success (McRae, 2013).

It is important to assess how skilled an individual is at reducing their negative emotions using an ER strategy, rather than only measuring how frequently the strategy is used, because there is evidence that skill rather than frequency is more important for life outcomes. In a study of reappraisal amongst community adults by Ford et al. (2017), those who used reappraisal more frequently and more successfully experienced fewer depressive symptoms, but those who used reappraisal more frequently but less successfully experienced more depressive symptoms. This demonstrates that the effectiveness of implementing the ER strategy may be important for how adaptive an ER strategy is.

The most common method of measuring ER effectiveness is a widely used experimental paradigm known as the emotion regulation task (ERT) (Goldin et al., 2008; Gross, 1998a; McRae et al., 2012b). The ERT involves individuals being presented with stimuli which are designed to evoke an emotional response, such as music (Vieillard et al., 2020), pictures (Douw et al., 2020), or film clips (Boland et al., 2019). Participants are instructed to use a particular ER strategy, which is expected to reduce the level of negative emotion induced by the stimuli. Emotion levels are recorded during the task, usually at the end of each trial, or continuously. Emotion can be measured in a number of ways during the task, including self-reported emotions (i.e., subjective experience), physiological responding such as heart rate, and emotional behaviour such as facial expression (including facial electromyography).

In the ERT, participants also view emotional stimuli without any ER instruction on some trials. Levels of emotion during this 'no regulation' condition are compared to emotion levels during the 'regulation' condition, and the reduction in emotion levels is a measure of how effectively the participant used the ER strategy to regulate their emotions (i.e., ER effectiveness). Thus, ER effectiveness refers to an individual's ability to reduce negative

emotion in response to emotional stimuli, after being instructed to use a particular ER strategy.

There are two aspects to using this paradigm to measure ER - (1) emotion must be induced during the experiment, and (2) the emotional response must be measured in some way, to assess the effectiveness of the ER strategy. Consequently, in order to use the ERT to measure ER in an emotion-specific manner, there must be an attempt to (1) elicit specific emotions using carefully chosen stimuli, and (2) measure specific emotions individually.

Although the ERT presents a unique opportunity to measure ER in an emotionspecific manner, very few studies have done so. Previous studies which have used the ERT tend to (1) use stimuli which induce a specific emotion (e.g., disgust), but measure negative emotion in general (Goldin et al., 2008; Gross, 1998a), (2) use stimuli to induce a specific negative emotion, and measure only one specific negative emotion (rather than measuring multiple specific emotions) (Gross & Levenson, 1997; Sheppes & Meiran, 2007; Sullivan & Kahn, 2020), or (3) use stimuli which induce general negative emotion without attempting to induce a specific target emotion, and measure general states such as negative emotion, stress, tension, physiological responding or neural activity (Albanese et al., 2019; Douw et al., 2020; Fitzgerald et al., 2019; Jackson et al., 2000; Lazarus & Alfert, 1964; Notarius & Levenson, 1979; Ochsner et al., 2002; Richards & Gross, 2000; Sheppes et al., 2014). Therefore, the optimal situation for measuring emotion-specific ER is to both elicit and measure specific emotions individually, to assess how ER may differ between emotions. The few studies which have measured multiple specific emotions during the ERT, such as Boland et al. (2019), tend to find emotion-specific differences in ER.

A potential reason for the lack of research examining ER in an emotion-specific manner is the challenge involved in eliciting specific emotions during an experimental task. It cannot be guaranteed that an emotional stimulus will create the desired effect (i.e., the

specific target emotion) within the participant. It is possible that the same emotional stimulus may induce varied emotions across individuals (R. W. Levenson, 2014). People may have different appraisals of the same situation depicted in an emotional stimulus, and these appraisals will lead to different emotional responses (Roseman, 1991). Similarly, an emotional stimulus may prompt mixed emotions within the individual during the ERT (Roseman & Evdokas, 2004).

However, some actions can be taken to address these challenges. Firstly, although it is possible for individuals to have different appraisals about the same situation, it is likely that certain emotional contexts, or antecedent events, will be associated with certain specific emotions (Ekman, 1992). Generally, fear is caused by a perceived threat in the environment, sadness is created by a perceived loss of a goal, and anger comes from the observation that our goals have been prevented by someone or something external to us (Ekman, 1992; Zenses et al., 2020). This means that certain emotional stimuli (such as images of snakes) are likely to elicit a predictable emotional response (such as fear) in the majority of participants during an experiment, although this is not guaranteed for all participants. If emotional stimuli are chosen which represents these themes (e.g., a threatening picture for fear), then this is likely to give rise to a similar emotional responses.

In commonly used databases of emotional stimuli, such as the IAPS (Lang et al., 1993), normative data regarding the emotional responses to different stimuli are provided (Mikels et al., 2005), which helps to guide researchers in selecting stimuli that target specific emotions. Different types of stimuli can be used in the ERT, but pictures are the most effective elicitors of specific emotions (Lench et al., 2011).

The second way to address the challenge of inducing specific emotions in an experimental setting is by recording multiple specific emotions during the ERT. This means

that in any trial of the ERT, all specific emotions of interest are measured, not just the target emotion for that trial. In other words, it should not be assumed that because the target emotion was sadness, that an individual would necessarily experience sadness, and only sadness. Measuring all emotions in each trial allows for a manipulation check of the ERT to be conducted, to ensure that the target emotion has been induced as intended, and to assess the extent of mixed emotions occurring in response to stimuli during the task. This means that the level of success in inducing specific emotions during the ERT can be assessed.

3.4. The Importance of Measuring Subjective Experience in Emotion Research

As discussed in Chapter 1, different theoretical perspectives offer diverse viewpoints as to whether emotions can be regarded as specific. There is evidence that emotions are organised as discrete categories, albeit with fuzzy boundaries between these categories (Cowen et al., 2018; Cowen et al., 2020; Cowen & Keltner, 2017). A prominent finding that has emerged from the emotion literature is that specific emotions tend to differ from one another in how they feel, which is the subjective experience of an emotion (Baumgartner et al., 2006a; Baumgartner et al., 2006b; Cowen et al., 2020; Cowen & Keltner, 2017; Dimberg, 1988; Harmon-Jones & Sigelman, 2001; Joseph et al., 2020; Lench et al., 2011).

Each of the three components that make up an emotional response, as outlined in Section 1.2.1, can be measured during the ERT. The difference in emotion between the 'no regulation' trials and the 'regulation' trials indicates the extent to which an emotional response has changed. However, in terms of measuring specific emotions, the evidence shows that some methods may be more suited to emotion-specificity than others. In the following sections, the measurement of physiological arousal, emotional behaviour and subjective experience will be discussed.

3.4.1. Measuring Physiological Arousal

The autonomic nervous system (ANS) is active during an emotional response (Christopoulos et al., 2019; Kreibig, 2010), and this physiological arousal can be measured during an ERT using a range of tools, such as electrodermal and cardiovascular recording (Mauss & Robinson, 2009). However, the ANS is not exclusively involved in an emotional response, and is in fact implicated in a number of bodily processes, including digestion, attention and homeostasis (Cacioppo et al., 2000; Robertson, 2019). This means that activation of the ANS during the ERT may be due to reasons unrelated to an emotional response.

Furthermore, the evidence around emotion-specificity in physiological responding tends to be mixed (Cacioppo et al., 2000; Kuppens, 2019; Lench et al., 2011; Mauss & Robinson, 2009). Although there is evidence of emotion-specific patterns in ANS activity (Kreibig, 2010), these predominantly tend to show differences between positive and negative emotions (Lench et al., 2011), and there is less evidence for physiological differences between specific negative emotions, particularly sadness, anger and fear (Cacioppo et al., 2000; Ekman et al., 1983).

Furthermore, often multiple physiological measures must be used in a single study in order to observe emotion-specific differences (Collet et al., 1997), which can substantially increase the duration of the experimental session, placing additional burden on participants. Therefore, recording physiological responses may be more suited for distinguishing between positive and negative emotions, and may not be the optimal method for measuring changes in sadness, anger, and fear in the ERT.

3.4.2. Measuring Emotional Behaviour

In terms of emotional behaviour, there is evidence of emotion-specificity in facial expression research (Ekman & Friesen, 1971; Ekman et al., 1987; Ekman et al., 1969; Izard,

1971). However, measures of emotional behaviour are less reliable in differentiating between specific negative emotions during an experimental task, and facial expression does not always directly map on to subjective experience (Lench et al., 2011). Additionally, there are individual differences in the use of expressive suppression (Gross, 1998a; Gross & John, 2003; Peters et al., 2019; Sullivan & Kahn, 2020), and many people use this strategy without conscious awareness (E. A. Butler et al., 2003). Therefore, if a participant naturally suppresses their expression of emotion, this will interfere with using facial behaviour as a measure of emotion.

Similarly, facial behaviour can also be measured using facial electromyography (fEMG), which records muscular activity in the facial muscles associated with emotional expression (Fridlund et al., 1984). Although studies have found that fEMG can reliably differentiate between a positive and a negative emotion (Brown & Schwartz, 1980; Dimberg, 1988; Dimberg & Thunberg, 1998; Magnée et al., 2007), there is little evidence that fEMG can distinguish between specific negative emotions (Brown & Schwartz, 1980). Recent studies have been more successful in identifying emotion-specific patterns using fEMG (Wingenbach et al., 2020), however these results are tentative. Therefore, fEMG may not presently be the most valid measure of specific negative emotions but may be increasingly implemented in emotion-specific research as technology advances.

3.4.3. Measuring Subjective Experience

Subjective experience refers to the internal 'feelings' of an emotion. Subjective experience is the defining element of an emotion (Kuppens, 2019), which is often regarded as the essence of an emotional response (LeDoux & Hofmann, 2018; Lieberman, 2019). An individual's feelings are inextricably tied to their emotions (Keltner et al., 2019), and measuring these feelings can effectively track changes in an emotional response (Lench et al.,

2011). Subjective experience is the component that differentiates an emotion from other affective states such as stress, as well as other bodily states such as physical pain or digestion (Ekman, 1999).

Moreover, subjective experience is the emotional component which has the most marked emotion-specific findings in the literature. In a meta-analysis of over 600 studies, measuring subjective experience differentiated between specific negative emotions, including sadness, anger, and anxiety (Lench et al., 2011). Furthermore, subjective experience was a stronger measure for distinguishing between sadness, anger, and anxiety than physiological or behavioural methods (Lench et al., 2011). Self-reported differences in subjective experience between specific emotions have been reported extensively throughout the literature (Baumgartner et al., 2006a; Baumgartner et al., 2006b; Cowen et al., 2020; Cowen & Keltner, 2017; Dimberg, 1988; Harmon-Jones & Sigelman, 2001; Joseph et al., 2020), and so there is strong evidence that specific emotions such as sadness, anger and fear are experienced differently.

In addition, many studies which have demonstrated associations between ER and variables associated with mental health have used subjective experience as the measure of ER (Aldao et al., 2010; McRae et al., 2012b). For example, ER as measured by self-reported subjective experience has been associated with bipolar disorder (Kjærstad et al., 2016), major depressive disorder (Ellis et al., 2013), and self-harm behaviour (T. S. Davis et al., 2014). Further, when measuring the regulation of specific emotions through self-reported subjective experience, the relationship between ER and clinical disorders such as depression depends on the specific emotion being regulated (Boland et al., 2019).

Subjective experience is measured through self-reporting, usually by using a Likert scale on which an individual records how intensely they are experiencing an emotion (Gross, 1998a). Self-report is widely used in psychological research generally, as it enables the

measurement of internal experiences which cannot be accessed using any other technique (Keltner et al., 2019; LeDoux & Hofmann, 2018; Stone et al., 2000). Self-report also allows for the regulation of specific emotions to be measured using separate scales, which means that an individual's experience of specific emotions can be measured. This makes self-report an extremely valuable measure in emotion-specific research. Similarly, measuring specific emotions on individual self-report scales means that the extent to which the target emotion has been successfully elicited in the ERT can be assessed.

3.4.4. Summary of Measuring Emotional Components

During the ERT, measuring any one of these components of an emotional response can provide useful information about ER. Interesting findings have been found in previous studies, with ER strategies being shown to impact all three emotional components during an experimental task (Boland et al., 2019; Goldin et al., 2008; Gross, 1998a). However, when attempting to conduct ER research in an emotion-specific manner, not all measures are equal. The evidence indicates that subjective experience is the most valid measure of specific emotions (Lench et al., 2011), and emotion-specific differences in physiological arousal and behaviour are less consistent in the field (Keltner et al., 2019; Mauss & Robinson, 2009). Some methods may be more sensitive in detecting the nuances between specific emotions that make them more suitable measures for emotion-specific research. Therefore, although all three methodological approaches can provide useful data about ER, subjective experience is the component which has shown the most marked emotion-specific differences (Lench et al., 2011).

3.5. Research Questions and Methods of the Present Research

3.5.1. Research Questions Used To Guide The Present Research

Based on the review of the research literature in the previous chapters, the following research questions were formulated to guide the present research:

- RQ1. Is ER different for specific emotions?
- RQ2. Are there gender differences in the regulation of specific emotions, and do these gender differences vary across different emotions?
- RQ3. Are there gender differences in engagement and disengagement ER?
- RQ4. Does the effectiveness of reappraisal and distraction depend on the specific emotion being regulated?
- RQ5. Are gender differences in emotion-specific ER related to gender differences in mental health?

Two empirical studies were conducted to address these questions. The research questions which are addressed by each study are shown in Table 3.2.

Relatively few studies have examined whether ER differs across specific emotions, and so RQ1 pertains to whether ER is different for the specific emotions of sadness, anger, and fear. Similarly, very few studies have examined gender differences in ER in an emotionspecific manner, which is important because males and females may receive different cultural messaging about specific emotions. RQ2 is related to whether there are emotion-specific gender differences in ER. A pattern can be observed in the ER literature, which is characterised by females using strategies which involve engagement, and males using strategies which involve disengagement, but this has never been explicitly tested before. RQ3 is relating to whether females are more likely to engage, and males are more likely to disengage with their emotions during ER.
According to Gross's process model of ER, described in Section 1.3.2.1, distraction is a more effective ER strategy than reappraisal, but this does not take into account the role of specific emotions. RQ4 asks whether the effectiveness of reappraisal and distraction depend on the specific emotion being regulated (sadness, anger, or fear). Finally, there is empirical evidence that ER may be related to gender differences in mental health, but this has never been tested in an emotion-specific manner. RQ5 asks whether there is a relationship between gender, emotion-specific ER, and variables associated with mental health (depressive symptoms, self-harm behaviour, and suicidal ideation).

Table 3.2

Overarching Research Questions and the Studies That Address Them

Research question	Study 1	Study 2a	Study 2b
RQ1. Is ER different for specific emotions?	Х		
RQ2. Are there gender differences in the regulation of	Х	Х	
specific emotions, and do these gender differences vary			
across different emotions?			
RQ3. Are there gender differences in engagement and	Х		
disengagement ER?			
RQ4. Does the effectiveness of reappraisal and distraction		Х	
depend on the specific emotion being regulated?			
RQ5. Are gender differences in emotion-specific ER related			Х
to gender differences in mental health?			

3.5.2. Methods Used in the Present Research

In the following sections, the methodological approaches which were chosen to address these research questions will be described.

3.5.2.1. Study 1: Open-Ended Responses. An open-ended questionnaire was chosen for Study 1, to measure emotion-specific ER amongst adolescents and adults. The open-ended questionnaire involved participants being asked to recall a situation during which they experienced a specific emotion – sadness, anger, or fear. Then they were asked to free-write about what they did, if anything, to try to reduce this feeling. People tend to use a small number of ER strategies (Jackson et al., 2000), and so this method is likely to tap into the key strategies that individuals use to regulate specific emotions.

This open-ended approach was chosen for three reasons, and aimed to address the limitations of previous ER measures described in Section 3.3.1. Firstly, there are few self-report questionnaires which measure ER in an emotion-specific manner. Also, these tend to be limited in the number of strategies that can be measured, and they often do not measure some of the most widely researched strategies in the literature, such as reappraisal. By using an open-ended questionnaire, the use of ER could be measured in different emotional contexts (i.e., sadness, anger, and fear) in order to capture the range of strategies that individuals use to regulate specific emotions. This approach is flexible and can be used to measure any number of specific emotions, but sadness, anger, and fear were chosen in the present study, for reasons discussed in Section 1.3.3.2.

Secondly, many of the existing self-report questionnaires have been validated for use with one age group, such as adolescents (Cracco et al., 2015), but cannot be used across age groups. Although the FEEL-KJ can be used with both adolescents and adults, this does not include important ER strategies such as reappraisal. The present study included both adolescent and adult participants in order to examine any interactions between gender and

age. As the open-ended questionnaire was designed for the present study, it could be written in such a way that it was accessible for both adolescents and adults (i.e., written in plain English without jargon), and so was appropriate to use for both age groups.

Thirdly, many of the previous self-report questionnaires are restricted to measuring only a small number of ER strategies. This may not be an accurate reflection of the range of strategies people use to regulate their emotions. The approach chosen for the present research is beneficial because it allows participants to freely report on their ER strategies, which means that strategies that may not be included in previous questionnaires can be captured, and participants have more freedom in reporting their ER.

Finally, as outlined in Section 2.2.3.6, a pattern can be observed in the ER literature which is characterised by males being more likely to regulate their emotions using avoidance (Zimmermann & Iwanski, 2014), and females being more likely to pay attention to emotions during regulation (D. P. Johnson & Whisman, 2013). It is possible that this pattern may differ according to the specific emotion being regulated, but this hypothesis has not been explicitly tested in the literature. Therefore, this open-ended approach can be used to test this hypothesis, by coding the participant responses into broad categories of engagement, disengagement, and distraction ER, using a framework derived from Parkinson and Totterdell's (1999) model of ER strategies, which was described in Section 1.3.2.4.

In summary, this approach allowed for three of the research questions shown in Table 3.2 to be addressed in Study 1 - RQ1, RQ2, and RQ3. RQ1 involved investigating if ER strategies differ across specific emotions, and this was tested in Study 1 by examining if the percentage of people reporting a strategy differed across emotions. This relied on the text data from the open-ended questionnaires being categorised into ER strategies for each emotion. RQ2 was related to whether there are emotion-specific gender differences in ER, and this involved examining the relationship between gender and ER strategies which were

coded from the open-ended questionnaires. RQ3 involved examining whether there are gender differences in engagement or disengagement ER, and this relied on the ER strategies being coded into broader ER categories (engagement, disengagement, and distraction ER) using the Parkinson and Totterdell framework, and then testing for a relationship between gender and these wider categories. Therefore, the open-ended questionnaire used in Study 1 allowed for participant responses to be coded into ER strategies, which enabled these research questions to be addressed.

3.5.2.2. Study 2: Regulating Sadness, Anger, and Fear in the Emotion

Regulation Task. In Study 2, the ERT was used to measure the regulation of sadness, anger, and fear. As outlined in Section 1.3.3.2, these three emotions were chosen because they consistently emerge as key emotions in studies of discrete emotions, they have been researched in previous emotion-specific work, and they have important implications for variables associated with mental health.

From a methodological perspective, a further advantage of measuring sadness, anger, and fear during the ERT is that there are normative data available about the extent to which individuals experience these emotions after viewing the emotional pictures used in this study, which is detailed in this section. This means that images can be carefully chosen which aim to induce these target emotions during the task, which enables emotion-specific ER to be measured. In addition, sadness, anger, and anxiety (which is closely related to fear) have larges differences in their subjective experience (Lench et al., 2011). This means that sadness, anger and fear are ideal candidates for emotion-specific research, because individuals are able to identify which specific emotion they are experiencing and self-report this emotional experience. This is the key method for determining the effectiveness of ER.

The biggest advantage of using the ERT is that it can be modified to allow for ER to be measured in an emotion-specific manner. In the task, general negative emotion is typically

measured by selecting stimuli which elicit a mix of emotions. However, emotional stimuli can be chosen which have been found to predominantly induce a target emotion, according to normative values measured in a sample of participants (Mikels et al., 2005). These carefully chosen pictures are presented in the ERT to attempt to induce specific negative emotions within the participant. This provides the opportunity for the participant to regulate this specific emotion (elicited by the image) and means that the regulation of specific emotions can be measured.

Many different types of emotional stimuli can be used to induce emotion in the ERT, such as emotional film clips, music, autobiographical recall, and situational procedures (Siedlecka & Denson, 2019). However, emotional pictures were chosen for the present research because they reliably induce emotions (Lang et al., 1993; Langer et al., 2020; Mikkelsen et al., 2020; Moodie et al., 2020; Siedlecka & Denson, 2019; Vishkin et al., 2020), and are the most effective stimuli for inducing specific negative emotions in a laboratory setting (Joseph et al., 2020; Lench et al., 2011).

The process for choosing the images to elicit specific emotions was as follows. Images were chosen from two databases which were specially developed for emotion research - the International Affective Picture System (Lang et al., 1997), and the Nencki Affective Picture System (Marchewka et al., 2014). These databases are comprised of static colour images that are frequently used in emotion research. These images depict a wide variety of situations that evoke negative and positive emotions, such as war, surgery, and fighting.

The images selected for the present study were carefully chosen to recreate situations that would commonly be associated with specific emotions. As discussed in Chapter 1, sadness often occurs due to a loss, anger results from interference with an important goal, and fear signals a threat (Ekman & Cordaro, 2011). The sad images thus depict situations of loss,

namely children, animals or elderly people who are injured or ill (where the cause of this injury or illness is not clear), and individuals who have been involved in accidents or are facing tragedy. The images which aim to induce anger contain content of injustices, specifically litter in a green area, Nazi symbolism, people being attacked, a child smoking, an animal which appears to have been abused, and a female who has severe burns. The fear images depict threatening images and are all images of animals in a threatening pose, such as snakes and wolves.

The images were chosen using normative data collected by Mikels et al. (2005) and Riegel et al. (2016). This data contains the self-reported levels of specific emotions reported by a sample of participants after viewing each of the images. The images which have the highest value for the target emotion, and lowest values for inducing other emotions, were selected for use in the ERT. In other words, these normative values were used to choose images which are likely to induce the target emotion. The following criteria were employed to select images for the ERT which induce sadness or fear. As there are fewer images available in these databases which have been found to induce only anger experimentally, a different set of criteria were used for anger, which is described below.

The criteria used to select images for the ERT were:

1. Images must induce one discrete emotion only (i.e., the target emotion)

2. Images in the no regulation and regulation trials must be matched on the level of emotion induced (to within .30 of the mean level of emotion for normative data)

3. Where data are available, images must be rated identically by males and females in terms of the emotional category of the image (i.e., there should be no gender differences in the level of emotion induced by the image).

However, a challenge in selecting images to induce anger during the ERT was that only two anger images met these criteria, and so supplementary images were required.

Consequently, anger pictures which induce mixed emotions, which are known as blended pictures, were permitted. Blended pictures with the highest anger means were chosen, as long as the anger mean was higher than the means for the other emotions. Thus, slightly altered criteria were applied to the anger stimuli, and the stimulus set was supplemented with 7 blended pictures from the IAPS which met these altered criteria.

Therefore, by carefully selecting pictures based on these normative values, the specific emotions of sadness, anger, and fear were aimed to be induced during the ERT. This allowed for participants' regulation of specific emotions to be assessed in the moment. Using the ERT in this emotion-specific manner contributes value to the literature by creating clarity in three key areas. In previous research, these issues have been examined by measuring general negative emotion, and as a consequence, mixed findings have emerged.

Study 2 was split into two parts – Study 2a, and Study 2b. In Study 2a, the ERT was used to address two of the research questions shown in Table 3.2 – RQ2, and RQ4. RQ2 related to whether there are gender differences in ER effectiveness for sadness, anger, and fear. This involved testing whether there were gender differences in the ER effectiveness scores derived from the ERT. RQ4 involved investigating whether the effectiveness of ER (as measured by a reduction in self-reported emotion levels during the ERT) differed according to the specific emotion being regulated. In Study 2b, RQ5 was addressed, which involved investigating whether gender differences in emotion-specific ER are related to gender differences in variables associated with mental health (depressive symptoms, self-harm behaviours, and suicidal ideation). The ER effectiveness scores from the ERT were used to test whether ER effectiveness mediated the relationship between gender and variables associated with mental health.

3.6. Conclusion and Next Steps

In summary, the different methodological approaches to ER research were described in this chapter. The two key approaches in the ER field involve measuring (1) the frequency with which ER strategies are used using self-report questionnaires, and (2) the effectiveness of using ER strategies to reduce emotions during the ERT. Few studies have adopted these methods to examine ER in an emotion-specific manner. An open-ended questionnaire, and a modified version of the ERT, were used to address the research questions of the present research, as they allowed for ER to be measured in an emotion-specific manner. The ERT used the self-reporting of subjective experience to measure emotion, which has demonstrated the largest emotion-specific differences in the literature and is regarded as the core component of an emotional response.

Now that the methodological approaches and research questions have been described, the empirical studies of the present research will be reported in the following chapters. In Chapter 4, Study 1 will be reported, which aimed to investigate if there are gender differences in the regulation of sadness, anger, and fear in an adolescent and adult sample. In Chapter 5, the first part of Study 2 will be described (Study 2a), which investigated whether there are gender differences in the effectiveness of using ER during the ERT to regulate sadness, anger, and fear. In Chapter 6, the second part of Study 2 will be discussed (Study 2b), which examines whether gender differences in the regulation of specific emotions are related to gender differences in depressive symptoms, self-harm behaviours, and suicidal ideation.

KEY POINTS FOR CHAPTER 3

- 1. Many of the previous measures in the field do not allow for ER to be examined in an emotion-specific manner. The emotion-specific measures that are available often miss out important ER strategies, such as reappraisal.
- The largest emotion-specific differences are found in the subjective experience of emotion, and so it is important to include subjective experience in emotion-specific ER research.
- Two complementary measures were chosen for the present research open-ended questionnaires and the ERT - which allow for ER to be measured in an emotionspecific manner.

QUESTIONS FOR SUBSEQUENT CHAPTERS

- 1. Are there gender differences in the reporting of ER to regulate sadness, anger and fear? (Chapter 4)
- 2. Are there gender differences in the effectiveness of using ER to regulate sadness, anger and fear? (Chapter 5)
- 3. Are gender differences in ER related to gender differences in variables associated with mental health? (Chapter 6)

NEXT STEPS

Now that the methodological approaches to measuring ER have been outlined, the empirical studies which test the aims of the present research will be discussed, in Chapter 4 (Study 1), Chapter 5 (Study 2a), and Chapter 6 (Study 2b).

Chapter 4: Gender Differences in the Reporting of Emotion Regulation Strategies to Regulate Sadness, Anger and Fear (Study 1)

4.1. Chapter Summary

In Chapter 2, it was established that although there is evidence for gender differences in ER more generally, there is a lack of research investigating gender differences in the regulation of specific emotions. The aims of Study 1 are therefore: (1) to examine whether the reporting of ER strategies differs across specific emotions, (2) to test if there are emotionspecific gender differences in the reporting of ER strategies to regulate sadness, anger, and fear, and (3) to test if there are gender differences in engagement, disengagement, and distraction ER. To achieve these aims, adolescents and adult participants wrote about the strategies they used to regulate sadness, anger, and fear, and these free text responses were coded into specific ER strategies (e.g., reappraisal), and also into wider categories of ER (i.e., engagement, disengagement, and distraction). Each of these wider categories of ER encompassed several specific ER strategies.

Cochran's Q tests showed that the reporting of almost all of the ER strategies varied across the emotions. Chi-square analyses also demonstrated that there were some gender differences in ER, but these differences depended on (1) the specific emotion, and (2) the ER strategy. A key finding was that gender differences in the use of reappraisal varied by emotion, which brings clarity to an area with mixed findings. These results highlight the importance of conducting ER research in an emotion-specific manner, and indicate that the method used in Study 1 is a valid technique for measuring ER strategies in different emotional contexts. These findings may also have implications for ER training as a treatment for poor mental health.

4.2. Introduction

4.2.1. Rationale for Study 1

As discussed in Section 3.3, many ER researchers tend to rely on the use of closedended self-report questionnaires to gather information about the ways that individuals regulate their emotions (Garnefski et al., 2004; Gross & John, 2003; Kraaij & Garnefski, 2019). A problem with this approach is that an assumption is made in advance about the ER strategies individuals use, and this method is constrained to measuring a relatively small number of ER strategies to limit any response burden to the participants. As a result, it is unclear whether the ER strategies that are commonly researched truly reflect the strategies individuals tend to use in their daily lives. To address this, a methodological approach must be used which allows participants to freely report their ER strategies without the constraints of closed-ended questionnaires and rating scales. Such an approach is used in the present study.

Additionally, much of the research in the field has been restricted to examining the regulation of general negative emotion (Jackson et al., 2000; Lazarus & Alfert, 1964; Notarius & Levenson, 1979; Richards & Gross, 2000; Sheppes et al., 2014), and does not partition out the regulation of specific negative emotions such as sadness, anger, or fear. This is problematic, because the evidence indicates that ER develops in an emotion-specific manner (Zimmermann & Iwanski, 2014), and the regulation of specific emotions may have a different relationship with variables associated with mental health (Boland et al., 2019; Clear et al., 2019), and so it cannot be assumed that what is known about the regulation of one emotion can be applied to all emotions. A small number of studies have addressed this issue by examining ER in an emotion-specific manner (Dixon-Gordon et al., 2015a; Rivers et al., 2007; Zimmermann & Iwanski, 2014), and these studies tend to find that the ER strategies people use are indeed different for specific emotions.

However, there are some limitations of this previous work. Some studies have measured pre-determined ER strategies using self-report questionnaires (Dixon-Gordon et al., 2015a; Zimmermann & Iwanski, 2014). This approach is useful for measuring the most common strategies in the literature, but it is unclear if these strategies would be freely reported in these emotional contexts. In the study by Rivers et al. (2007), ER was measured using an open-ended approach similar to that used in the present study, which allows participants to freely report on their ER; however, this was measured within the context of relationships, rather than a more general context. Further, in the study by Rivers et al. (2007), the data were categorised into quite broad categories that grouped together some of the most commonly researched strategies such as reappraisal, which meant that emotion-specific differences for these strategies could not be examined. The approach used in the present study involves participants writing about their ER strategies when they feel sad, angry, or scared within a more general context, and this will allow all the relevant strategies highlighted as important by the participants to be captured. Therefore, the first aim of the current study is to record the ER strategies that individuals use to regulate sadness, anger, and fear, and to test if these strategies vary across specific emotions.

It has also been established in the previous chapters that there are gender differences in the use of some ER strategies, such as rumination (D. P. Johnson & Whisman, 2013), but mixed findings for others, such as reappraisal and distraction (Esmaeilinasaba et al., 2016; Garnefski et al., 2004; Nolen-Hoeksema & Aldao, 2011; Zlomke & Hahn, 2010). These mixed findings may be the result of previous research being focused on general negative emotion, without examining specific emotions individually. When gender differences in emotion-specific ER have been examined, such as in the studies by Rusting and Nolen-Hoeksema (1998), Cox et al. (2000), and Trives et al. (2016), clearer gender differences emerged. Consequently, if ER strategies were investigated in an emotion-specific manner, it is possible that clearer gender differences may be found. Therefore, the second aim of the present study is to investigate whether there are gender differences in the reporting of ER strategies across specific emotions.

As highlighted in Section 2.2.3.6, a pattern can be observed in the ER literature. Generally speaking, females are more likely to report using some ER strategies that can be classed as engagement strategies, such as rumination (D. P. Johnson & Whisman, 2013). Males are more likely to report using strategies that can be classed as a type of disengagement, such as avoidance or suppression (Zimmermann & Iwanski, 2014). This evidence gives rise to the hypothesis that females may be more likely to engage, and males may be more likely to disengage with their emotions during regulation. However, this hypothesis has not been explicitly tested in the ER literature. As a result, the third aim of this study is to examine whether there are gender differences in engagement, disengagement, and distraction ER, which are broader categories of ER. In the present study, distraction is similar to disengagement, but involves removing attention from the emotional stimulus and replacing this with something unrelated.

This aim will be addressed using an empirically derived framework developed by Parkinson and Totterdell (1999), which was described in Section 1.3.2.4. The Parkinson and Totterdell framework is a taxonomy of ER strategies, which allows for ER strategies to be categorised as types of engagement, disengagement, or distraction, and the framework provides helpful guidelines for doing so. The framework is a valuable tool which is complementary to the method of the present study, as it provides a structure for classifying individual ER strategies into engagement and disengagement categories, as well as a distraction category. In the present study, gender differences in both individual ER strategies,

as well as these wider categories of engagement, disengagement, and distraction, will be tested.

4.2.2. Research Questions for Study 1

Study 1 was guided by three specific research questions (SRQs):

- SRQ1. What are the key ER strategies that are reported for sadness, anger, and fear, and do these strategies vary by emotion?
- SRQ2. Are there gender differences in the reporting of specific ER strategies to regulate sadness, anger, and fear, and do these gender differences vary by emotion?
- SRQ3. Are there gender differences in broad engagement, disengagement, and distraction ER categories?

4.3. Method

4.3.1. Participants

Four hundred and sixty-eight participants took part in the study (199 males and 269 females). Adolescent participants aged 12-17 were recruited from secondary schools in North Lanarkshire, and young adult participants aged 18-30 were recruited online. The demographic characteristics of the sample are presented in Table 4.1.

The inclusion criteria were that participants must (1) be able to read and write in English, (2) be aged 12-30 (which includes both adolescent and young adult participants), and (3) identify as either male or female. It was important that participants were proficient in English to ensure that they were able to understand what was being asked of them in the study. The age range of 12-30 was chosen because an aim of the present study was to examine if gender differences varied across different age groups. In order to control the effect of age, specific adolescent and young adult age groups were chosen and individuals outside of these groups were excluded. The rationale for these age groups is provided in Section 4.3.2 below.

Furthermore, a key aim of the present study was to examine gender differences in the reporting of ER strategies, and so males and females were chosen for these groups. A requirement of this research was to have well-defined gender categories, and so individuals not identifying as male or female were excluded. However, although males and females were focused upon in the present study, it is acknowledged that gender is broader than this binary, and an important direction for future research in this area is to examine ER processes amongst other gender-identification groups. Although data were originally collected from 476 participants, data from seven adult participants were removed due to the individuals being over 30 years old, and one participant's data was removed who did not identify as male or female.

The sample was mostly white (96.37%) and attending school or university (94.47%) at the time of data collection. In terms of the highest level of education attained by the adult sample (which is regarded as a proxy for socio-economic status; Maksimović et al., 2008), 2.41% had completed secondary school, 2.41% had completed technical training level education, 39.76% were current undergraduates, 26.51% had completed university, and 28.92% had completed post-graduate study. Socio-economic status in the adolescent group was measured using the Family Affluence Scale (Boyce et al., 2006). In this group, 3.88% were low, 66.67% were medium, and 29.46% were regarded as being in the high socio-economic group.

Table 4.1

Mean (M) Age, Standard Deviation (SD) and Sample Size (n) for Each Gender and Age Group

	Male			Female				
	n	M	SD	п	M	SD		
Young adolescents	131	12.98	0.73	137	12.89	0.77	_	
Older adolescents	45	15.49	0.76	64	15.56	0.66		
Adults	19	25.74	3.97	67	23.39	3.55	1	
Total	195	14.80	4.00	268	16.15	4.71		

4.3.2. Age Groups

The sample was split into three groups according to age. These groups were (1) young adolescents (12 to 14 years old), older adolescents (15 to 17 years old) and young adults (18 to 30 years old). Young adults were recruited in this study (as opposed to older adults or adults in general) as this group are relatively neglected in the ER literature (Rawana et al., 2014). Organising the sample in this way allowed for analyses to be conducted which examined gender differences in ER across different age groups, while maximising the cell counts for each group. This is important because ER tends to change across adolescence and into adulthood (Garnefski & Kraaij, 2006; Zimmermann & Iwanski, 2014), and there is often an interaction between gender and age in the use of ER strategies (Garnefski & Kraaij, 2006; Nolen-Hoeksema & Aldao, 2011).

The World Health Organisation (2021a) defines adolescence as ranging from 10 to 19 years old, but there is no standard classification system for organising adolescents into developmental subgroups based on their age, and different studies have used a variety of categories (Cox et al., 2000; Garnefski & Kraaij, 2006; Garnefski et al., 2002b; Larsen et al., 2013; Zimmermann & Iwanski, 2014). The groups used in the present study (younger adolescents aged 12-14, and older adolescents aged 15-17) were chosen based on empirical evidence demonstrating that generally speaking, there tend to be changes in emotional development across these age groups (B. Casey et al., 2017; Esnaola et al., 2017; Ross et al., 2019; Young et al., 2019), and ER increases across these groups (Garnefski & Kraaij, 2006). Medical research often uses 15 years old as a cut off point for the older adolescent period (Bleyer & Albritton, 2003). It would also be useful to split the adolescent sample into three age groups (younger, middle, and older adolescents) in order to examine any variations in gender differences across adolescence in more detail; however, the cell counts in the cross-

tabulations did not allow for disaggregation at this level, and so younger and older adolescent groups were examined in the present study.

To ensure that the age groups were homogenous enough to conduct the statistical analyses, a one-way analysis of variance (ANOVA) was conducted on the age groups to test the hypothesis that the mean age of each group was statistically significant from the other age groups. This test was significant, F(2,460) = 1285.30, p < .001. Post-hoc *t*-tests with Bonferroni correction revealed that the mean age of each group significantly differed from the other groups. The means, standard deviations, *p*-values and direction of results for the post-hoc tests are shown in Table 4.2.

4.3.3. Procedure

The present study was granted ethical approval by the University of Strathclyde Ethics Committee (reference number: UEC15/70). The study took place either in secondary school classrooms (for the adolescents) or online (for the adults). Participants received an information sheet (Appendix A) at the beginning of the study which contained details about what the study involved. Participants were then free to decide whether or not they would like to take part in the study. All participants provided informed consent on either a paper

Table 4.2

Mean (M) Age, Standard Deviation (SD), P-Values, and Direction of Relationships for ANOVA and Post-Hoc Tests for Differences in Age Between the Groups

Age group	M	SD	
Young adolescents	12.93	.75	
Older adolescents	15.53	.70	
Adults	23.91	3.76	
Post-hoc comparison	p	Direction	
YA v OA	<.001	OA > YA	
YA v Ad	<.001	Ad > YA	
OA v Ad	<.001	Ad > OA	

Note. YA refers to younger adolescents, OA is older adolescents, and Ad is the adult age group.

(adolescents) or online (adults) consent form (Appendix B), and for participants under 16 years old, opt-out parental consent was also obtained using a paper consent form. Once consent was obtained, participants self-reported their age, gender, whether or not they could read and write in English, and SES (Appendix C). SES was measured using the Family Affluence Scale (Boyce et al., 2006) for adolescents, and by asking about the highest level of education achieved (which can be an indication of SES; World Health Organisation, 1988) for the adult participants. Information about ER strategies was captured qualitatively using open-ended questionnaires, which are described in the next section. The full questionnaire is included in Appendix D. The adolescent participants completed the questionnaire in pen and paper format in school during class time, and the adults completed the questionnaire online via the survey platform Qualtrics.

To reverse any potential negative affect brought about during the study, a modified Velten (1968) Positive Mood Induction Procedure (MIP) was included at the end of the study (Appendix E). This method involves participants reading positive sentences and has consistently been associated with an increase in positive mood (Teasdale & Russell, 1983). Once the participant responses had been collected, the participants were provided with a debriefing sheet which contained additional details about the purpose of the study, researcher contact information, and sources of additional support for participants (Appendix F).

4.3.4. Materials

A novel questionnaire was designed for the purposes of the present study, to allow for the self-reported use of different ER strategies to be measured in an emotion-specific manner. Participants were firstly asked to provide demographic information on their (1) age (What is your age?), (2) gender (What is your gender? Male or Female), (3) whether they could read and write in English (Can you read and write in English?). Participants were invited to freewrite about a recent situation during which they experienced sadness, anger, or fear, and then to write about what they did to try to reduce this emotion (i.e., their ER strategies). Participants were encouraged to report any other things they do to reduce sadness, anger, or fear.

The questionnaire consisted of three questions for each emotion:

- Think about a time recently when you felt sad/angry/scared, and you did something to try to make yourself feel less sad/angry/scared. Describe what happened to make you feel sad/angry/scared in the box below.
- 2. When you felt sad/angry/scared during this time, what did you do to try to make yourself feel less sad/angry/scared?
- 3. Can you think of any other things you have done in the past to make yourself feel less sad/angry/scared?

4.3.5. Coding and Analysis

4.3.5.1. Coding the Data. Existing methods of conducting research on ER offer few options to measure ER in an emotion-specific manner, and they often measure only a small number of strategies and miss out important strategies. The approach developed for Study 1 aimed to address these limitations. This approach involved collating the text responses from the open-ended questionnaire and then processing and coding these into ER strategies. These ER strategies were then coded into wider categories of ER using a framework developed by Parkinson and Totterdell (1999), which allows for the engagement hypothesis (i.e., that females are more likely to engage with their emotions during regulation) to be tested. A bottom-up, data-driven approach was adopted in this process, which involved categorisation being guided by the content of the data, and the researcher's knowledge of the ER literature. Individual strategies were focused upon initially, rather than focusing on categories derived

from a theoretical framework, to ensure that the richness in the data was captured. The danger in coding the data items into broad categories too early is that the caveats of individual strategies could be lost. Therefore, a data-driven approach was chosen to ensure that that rich information available in the data was represented in the coding process. This process involved four stages, which are described below.

Stage 1: Preparation of the data. Each participant's text data was entered into and collated on a Microsoft Excel spreadsheet. Raw data items were identified based on the smallest meaningful action that could be regarded as a way to regulate an emotion. For example, the response 'I went out with my friends and played football', was separated into two data items – (1) went out with friends, and (2) played football. The aim of this stage was to prepare the data for coding into ER strategies.

Stage 2: Coding of ER strategies. The data were then coded into individual ER strategies, based on the content of the data items, and the researcher's knowledge of ER strategies from the theoretical and empirical literature. The purpose of this stage was to build a data corpus of ER strategies, and to capture the diversity of strategies used within the sample. If a participant reported an ER strategy at least once this was coded as '1', and if they did not report an ER strategy it was coded as '0'. This produced a range of ER strategies which were dichotomous variables (Yes, No).

This stage yielded 132 unique ER strategies and included some of the most commonly reported strategies in the literature, including acceptance, avoidance, distraction, reappraisal, rumination, expressive suppression, and seeking social support (although some of these were reported in low numbers). The definitions used to code some of the most frequently reported ER strategies are found in Table 4.3.

Table 4.3

ER strategy	Definition/data items included
Avoid the situation	Ignoring or avoiding the emotional situation
Creative/relaxing activity	Taking part in an activity with the intent of relaxing or
	doing something creative (as opposed to distracting)
Distraction	Thinking or doing something unrelated to the emotional
	situation, or when participants explicitly reported that they
	distracted themselves (without specifying how this was
	achieved)
Experiential avoidance	Trying not to feel the emotion
Hostility to others	Taking the emotion out on others (e.g., shouting at
	someone, being violent towards them)
Leave emotional situation	Leaving the situation that triggered the emotion
Listen to music	Listening to music
Memories	Thinking about happy memories
Problem solving (action)	Taking action to solve the problem that triggered the
	emotion (this is distinguished from problem solving
	cognitively by strategizing or thinking differently).
Reappraisal	Changing one's thoughts about the situation, reframing the
	situation, changing your mindset, having happy thoughts
Removing attention	Removing the attention from the situation, e.g., by closing
	your eyes
Rumination	Thinking about the situation in detail, dwelling on it,
	thinking about the causes and consequences of the situation
Self-talk	Talking to yourself in a positive way, reassuring yourself,
	encouraging yourself
Social support	Turning to others for comfort, talking about the situation
	with others, spending time with others
Sports/exercise	Taking part in a physical activity such as sports or exercise
Take time out	Taking a break, spending time on your own, resting, taking
	time out
TV/movies	Watching a TV or movie
Venting/expressing emotion	Outwardly expressing emotion, ranting to others, crying,
	hitting/punching things, letting out emotion

Definitions Used to Code the Data Items into ER Strategies in Study 1

Stage 3: Coding of wider ER categories. These specific ER strategies were then coded into three broader ER categories – engagement, disengagement, and distraction. First, this involved deciding whether a strategy was an engagement strategy or a diversion strategy. A diversion strategy could then be categorised as a type of disengagement or distraction. Of the 132 unique strategies identified, 52 (39%) were classed as engagement, 19 (14%) were disengagement strategies, and 39 (30%) were categorised as distraction. Some strategies were judged to encompass aspects of both engagement and disengagement, and so these were coded as 'multipurpose' and were excluded from further analysis. This applied to 22 (17%) of the ER strategies. The following guidance, which was adapted from Parkinson and Totterdell (1999), was used to code the ER strategies into these wider ER categories:

1. Is the strategy intended as a way of addressing or avoiding the problem or related affect? Engagement strategies involve sustained attention to, or work on, the problem or affect. Diversion strategies involve redirecting cognition or action away from the problem or affect.

2. If the strategy is a diversion strategy, is the strategy simply avoidant (disengagement), or does it explicitly involve actively thinking about (or doing) something else in order to divert attention from the concern (distraction)? Disengagement strategies include avoiding thinking about the problem or affect and avoiding the problematic situation.

As with the specific ER strategy stage, if a participant reported using a type of ER at least once then this was coded as '1', and if they did not report a type of ER, it was coded as '0'. This stage yielded the dichotomous variables of engagement, disengagement, and distraction ER. The top ER strategies included in each wider category are found in Table 4.4. Almost half of the category of engagement was made up of expressing/venting emotions and social support. Over half (65%) of the disengagement category consisted of experiential avoidance (trying not to feel the emotion), leaving the situation (leaving a situation that you

are in when experiencing the emotion), and avoiding the situation (not attending the emotional situation in the first place). Around a third (35%) of the distraction category was made up of listening to music and watching TV or movies. A full list of ER strategies included in each category is found in Appendix G.

Stage 4: Reliability analysis. Reliability analyses were conducted to test the reliability of the coding process, both for the individual ER strategies, and for categorisation into the wider engagement, distraction, and disengagement categories. Two coders - the author and an undergraduate student who was assisting with the coding - completed the coding process independently, and interrater agreement was tested using Cohen's κ . Cohen's κ is considered a conservative test of interrater agreement as it does not take into account chance agreement.

Table 4.4

ER strategies	% of ER category
Engagement	
Express/vent emotion	24.48%
Social support	17.39%
Problem solving (action)	10.31%
Reappraisal	9.98%
Self-talk	5.19%
<u>Disengagement</u>	
Experiential avoidance	33.24%
Leave emotional situation	21.97%
Avoid situation	10.14%
Hostility to others	7.61%
Go to room	5.63%
Distraction	
Listen to music	18.29%
TV/movies	16.27%
Distraction	11.34%
Creative/relaxing activity	10.21%
Sports/exercise	10.21%

Top ER Strategies That Make up the Wider ER Categories of Engagement, Disengagement and Distraction

Cohen (1960) suggested that κ could be interpreted as follows: ≤ 0 is no agreement, 0.01 - 0.20 is none to slight agreement, 0.21 - 0.40 is fair agreement, 0.41 - 0.60 is moderate agreement, 0.61 - 0.80 is substantial agreement, and 0.81 - 1.00 is almost perfect agreement.

The first reliability analysis, which tested the extent to which the two independent coders categorised the ER strategies (such as social support, venting/expressing emotion etc.) in the same way, showed that Cohen's κ was $\kappa = .67$, which represents substantial agreement between the coders, according to Cohen's (1960) classification. The second analysis tested the agreement between the two coders that the ER strategies should be allocated into the categories of (1) engagement, (2) disengagement, or (3) distraction ER. Cohen's κ was $\kappa = .59$, which is considered a moderate level of agreement according to Cohen's (1960) original interpretation of κ . Although this is an acceptable level of interrater agreement by Cohen's standards, it should be noted that this is generally on the lower end of what is considered acceptable (M. L. McHugh, 2012). Therefore, there was substantial agreement between the coders for the ER strategies, and there was moderate agreement for the wider ER categories.

The stages of coding the data for the present study are summarised in Figure 4.1. Figure 4.1





4.3.5.2. Analytical Approaches for Exploring the Research Questions of the

Present Study. Three analytical approaches were implemented to explore the research questions of the present study. Cochran's Q tests were used to investigate whether reported ER strategies vary by emotion (SRQ1). Chi-square analyses on ER strategies were used to explore whether there are emotion-specific gender differences in reported ER (SRQ2). Chi-square tests on the wider ER categories were used to test there are gender differences in engagement, disengagement, and distraction (SRQ3). In the following sections, the analytical approaches that were used in the present study will be described in more detail.

4.3.5.2.1. Approach for Testing Emotion-Specific Differences in Reported ER

Strategies. In order to explore the key ER strategies that are used to regulate sadness, anger, and fear, and to test whether the key strategies vary according to the specific emotion being regulated, some of the most frequently reported strategies were selected to examine the percentage of individuals who reported using this strategy, and to compare these percentages between the specific emotions. Any number of strategies could be chosen for this analysis. However, the top 15 strategies for each emotion were chosen because this includes approximately 75% of the total recorded responses. The final 25% of responses included between 65-68 of the remaining ER strategies, depending on the emotion. It should be noted however that the extent to which the top reported strategies vary between specific emotions will depend on the number of top strategies that are examined (e.g., comparing the top 10 strategies will show a slightly different result than comparing the top 15 strategies), although the general pattern is the same.

In addition, Cochran's Q tests were used to assess whether the proportion of the sample reporting a particular strategy varied between the emotions (e.g., was the percentage of participants reporting social support significantly different for sadness, anger, and fear?). Cochran's Q is used to compare dichotomous dependent variables between three or more

related groups (Laerd Statistics, 2017). As a large number of strategies (132) were reported in total, it was not practical to run inferential analyses on every reported ER strategy. Therefore, a selection of ER strategies was chosen for further analysis using Cochran's Q.

The first selection criterion was that if an ER strategy appeared in the top 5 reported strategies for any emotion, then this strategy would be tested using Cochran's Q. This yielded a list of 10 strategies – social support, sports/exercise, listening to music, watching TV/movies, expressing/venting emotion, hostility to others, taking time out, problem solving (action), reappraisal, and self-talk. In addition, the second criterion was that most commonly researched strategies in the field were also chosen for analysis in order to compare the results of the present research to the wider literature. These 5 strategies were distraction, avoiding the situation, experiential avoidance, passivity, and rumination. Therefore, 15 strategies in total were analysed using Cochran's Q tests, as reported in Section 4.4.2.

4.3.5.2.2. Approach for Testing Gender Differences in Individual Emotion

Regulation Strategies. Due to the exploratory nature of this study, participants were invited to spontaneously report on their ER without any prompting. Consequently, this meant that although the overall sample size for Study 1 was healthy, the sample size for the reporting of some strategies was too small to enable chi-square analysis to be conducted. For this reason, gender differences could not be tested for the majority of ER strategies reported. However, as the top five reported strategies for each emotion had a sufficient sample size, the relationship between gender and each of these strategies was analysed using chi-square analysis. Also, if a strategies (such as acceptance, reappraisal, distraction, suppression, rumination, self-harm, or avoidance) and was reported by a sufficient number of people, gender differences in the use of this strategy were also tested. At this stage, the analyses were not disaggregated by age

group because doing so resulted in low cell counts which meant that the chi-square analysis could not be conducted. The results for this analysis are presented in Section 4.4.3.

4.3.5.2.3. Approach for Testing Gender Differences in Wider ER Categories.

Gender differences in engagement, disengagement, and distraction were examined, which involved testing the main effects of gender and age first, and then conducting analyses to test for an interaction between gender and age on reported ER strategies.

1. Firstly, the main effect of gender was tested by conducting a series of chi-square analyses examining gender (males, females) as the independent variable (IV), and reported ER strategy for each emotion (Yes, No) as the dependent variable (DV), with age groups collapsed for these analyses (i.e., all age groups were analysed together). This stage involved running the following nine chi-square analyses, the results of which are reported in Section 4.4.4.1: (1) gender x engagement for sadness, (2) gender x engagement for anger, (3) gender x engagement for fear, (4) gender x disengagement for sadness, (5) gender x disengagement for anger, (6) gender x disengagement for fear, (7) gender x distraction for sadness, (8) gender x distraction for anger, and (9) gender x distraction for fear.

2. Chi-square analyses for the relationship between gender and reported ER in each age group were conducted. This tested the interaction between gender and age (i.e., if gender differences in reported ER strategies varied across age groups), and these are reported in Section 4.4.4.2.

As the chi-square tests involve multiple comparisons, the Benjamini-Hochberg correction (Benjamini & Hochberg, 1995) was used to adjust the alphas to reduce the risk of a Type I error occurring in the analysis (i.e., that a significant effect is reported that does not exist). The Benjamini-Hochberg correction involves controlling the expected proportion of falsely rejected hypotheses (Benjamini & Hochberg, 1995). Using this method, the *p*-values for each test are ranked in order from lowest to highest, and each alpha is multiplied by the

number of comparisons divided by the position of the *p*-value in the list, to provide the correct alpha for each *p*-value (Jafari & Ansari-Pour, 2019). The Benjamini-Hochberg approach was preferred over more conservative methods (such as the Bonferroni correction) as it also controls for the risk of Type II error (i.e., that a significant effect is not reported when it exists) (S.-Y. Chen, Feng, & Yi, 2017; Jafari & Ansari-Pour, 2019), which means that it has more power than other methods (Benjamini & Hochberg, 1995). It is appropriate to use Benjamini-Hochberg corrections when there are a large number of hypotheses being tested (S.Y. Chen et al., 2017), which is the case in the present study.

4.4. Results

4.4.1. Data Screening and Assumptions

4.4.1.1. Missing Data. The data were analysed for patterns in the missing values. Missing values meant that the participant had left a question blank in the questionnaire (e.g., the sadness question) which meant that no data was available to subsequently be categorised as an ER strategy for this emotion. Missing data variables were computed by categorising each participant as having either (1) a missing value or (2) no missing value for each question in the questionnaire. Chi-square analyses were conducted to test if there was a relationship between (1) age (young adolescents, older adolescents, adults) or (2) gender (males, females) and whether or not there were missing values (missing, not missing).

Overall, the responses for some emotions had a larger proportion of missing data than others. There were approximately 4% missing values in the sadness data, 11% missing in the anger data, and 23% missing in the fear data. One potential reason for this difference in the reporting of strategies is that there may be fewer opportunities for experiencing fear in everyday life compared to the other emotions, and so participants may have fewer examples to report the ER strategies for this question.

Another reason for this finding may be the presence of an order effect. During the procedure, the questions were presented in a particular order – the sadness question appeared first, followed by the anger question, and then the fear question. The pattern in the missing values may be due to presenting the questions in this order for all participants. In the wider literature, there is often more missing data in later questions due to participant fatigue when questions have not been counterbalanced (Egleston et al., 2011), and questionnaires that use open-ended questions tend to produce more missing data than those with closed questions (Reja et al., 2003).

Table 4.5 shows the percentage of missing data for gender, sadness, anger, and fear, across the different age groups, and the results of the chi-square analyses testing the relationship between these variables and whether or not there was missing data. The percentage of missing data in the sadness question appears to differ across some of the age groups, with older adolescents having higher levels of missing values compared to adults. The older adolescents were the group with the highest percentage of missing values. The

Table 4.5

Study 1	Ŋ	ΥA	(DA	I	4d	Т	otal			Pos	t-hoc t	ests
Variables	п	%	n	%	n	%	n	%	χ^2	p	Y-O	Y-A	O- A
Age	-	-	-	-	-	-	5	1.07	-	-	-	-	-
Gender	0	0.00	0	0.00	0	0.00	0	0.00	-	-	-	-	-
Sadness data	12	4.48	9	8.26	0	0.00	21	4.49	7.57	.02	.15	.04	.01*
Anger data	29	10.82	15	13.76	4	4.65	50	10.68	4.44	.11	-	-	-
Fear data	66	24.63	27	24.77	14	16.28	108	23.08	2.78	.25	-	-	-

Missing Data for Each Variable Broken Down by Age Group, Showing the Number (n) And Percentage (%) Of Missing Values

Note. The abbreviations used in this table are as follows:

YA is the Young Adolescent age group, OA are Older Adolescents, Ad are Adults, χ^2 is the chi-square value, *p* is the *p*-value, Y-O refers to the follow up chi-square analysis comparing the Young Adolescent and Older Adolescent age groups, Y-A is Young Adolescents v Adults, and O-A is Older Adolescents v Adults. Significant values are in bold. Follow up tests were corrected for multiple comparisons using the Benjamini-Hochberg method. For the follow up tests, the symbol *** denotes significance at the *p* < .001 level, ** is *p* < .01 level, and * is *p* < .05 level after correcting for multiple comparisons. missing data may be due to these younger participants becoming more easily fatigued than adults when responding, or having more difficulty recalling their ER strategies, although this does not explain why the younger adolescents did not have higher levels of missing values for this question.

Similarly, Table 4.6 shows the percentage of missing data for age, sadness, anger, and fear, across the gender groups, and the results of the chi-square analyses testing for patterns in the missing data. There were gender differences in the proportion of missing values. Specifically, males had a higher percentage of missing data than females, for all three emotions questions. This may indicate that males had more difficulty recalling their ER strategies or were less able to verbalise these strategies. The implications of these findings are that any gender differences in reported ER should be interpreted with caution in the main analysis.

Table 4.6

Missing Data for Each Variable Broken Down by Gender, Showing the Number (n) And Percentage (%) Of Missing Values

	M	ales	Fen	nales	Te	otal	·		
Study 1 Variables	n	%	n	%	n	%	χ^2	p	
Age	4	2.01	1	0.37	5	1.07	-	-	
Gender	-	-	-	-	0	0.00	-	-	
Sadness data	16	8.04	5	1.86	21	4.49	10.20	.001	
Anger data	30	15.08	20	7.43	50	10.68	7.00	.08	
Fear data	67	33.67	41	15.24	108	23.08	21.88	<.001	

Note. χ^2 refers to the chi-square value, p is the p-value. Significant values are in bold.

To address the missing data problem, listwise deletion was used. Listwise deletion involves excluding any cases with missing data from analysis. It is appropriate to use listwise deletion when loss of power is tolerable (Curley et al., 2019), as is the case with the present data set. Missing data is more problematic in studies with smaller samples (Tabachnick & Fidell, 2014), and since the present study has a substantial sample size, it was judged that the missing data is not a serious violation of assumptions. Therefore, missing values were addressed using listwise deletion.

4.4.1.2. Assumption of Chi-Square Analysis: Cell Counts. In order to run chisquare analysis, all cells in the cross-tabulation must have an expected cell count that is greater than 5 (Laerd Statistics, 2016). Most of the cross-tabulations met this assumption. However, three of the cross-tabulations had at least 1 cell with an expected cell count of less than 5. For these tests, Fisher's exact test was reported, which provides a more accurate *p*value when cell counts are low (Laerd Statistics, 2016), and these tests are indicated by the 'a' symbol.

4.4.2. The ER Strategies Reported for Specific Emotions

Across all emotions, 3,377 counts of ER strategies were reported in total (which includes the same strategy being reported more than once by a participant). Females provided more responses than males, with 67% of all responses being reported by females, and 32% being reported by males. Interestingly, participants reported a large number of unique ER strategies – 132 different ER strategies were reported in total. Females also reported more unique strategies than males. Across all emotions, females reported 116 unique strategies, and males reported 84 unique strategies. There was no single strategy that dominated the data, although the top reported strategy (seeking social support) made up around a fifth of recorded responses (see Table 4.7).

The 15 most frequently reported strategies for each emotion are shown in Table 4.7. This table shows the number of times a strategy was reported (and so includes strategies that were reported by the same individual multiple times). The top strategy reported for sadness and fear (for both males and females) was seeking social support.

Table 4.7

			ales	Fem Freq	nales %	Total		
Emotion	ER strategy					Freq	%	
<u>Sadness</u>	Social support	75	15.79	201	21.34	278	19.54	
	Sports/exercise	54	11.37	66	5.94	110	7.73	
	Listen to music	21	4.84	59	7.01	89	6.25	
	TV/movies	28	5.89	56	5.73	83	5.83	
	Express/vent emotion	11	2.32	54	6.26	71	4.99	
	Creative/relaxing activity	29	4.00	41	4.14	58	4.08	
	Experiential avoidance	15	6.11	39	2.65	54	3.79	
	Reappraisal	23	2.74	39	4.14	52	3.65	
	Distraction	19	3.58	35	3.72	52	3.65	
	Memories	13	2.53	34	3.61	46	3.23	
	Comfort eating	3	0.63	41	4.35	44	3.09	
	Problem solving (action)	21	4.42	21	2.23	42	2.95	
	Take time out	15	3.16	25	2.65	40	2.81	
	Sleep	15	3.16	23	2.44	39	2.74	
	Video games	26	5.47	7	0.74	34	2.39	
A = 0 = = =								
<u>Anger</u>	Express/vent emotion	65 26	17.86	121	16.13	187	16.73	
	Social support	26	7.14	98 51	13.07	124	11.09	
	Sports/exercise	33	9.07	61	8.13	94 5 5	8.41	
	Hostility to others	15	4.12	40	5.33	55	4.92	
	Take time out	21	5.77	31	4.13	52	4.65	
	Listen to music	11	3.02	37	4.93	48	4.29	
	Leave emotional situation	8	2.20	29	3.87	37	3.31	
	TV/movies	19	5.22	14	1.87	34	3.04	
	Creative/relaxing activity	9	2.47	23	3.07	32	2.86	
	Problem solving (action)	12	3.30	19	2.53	31	2.77	
	Reappraisal	11	3.02	19	2.53	30	2.68	
	Experiential avoidance	15	4.12	15	2.00	30	2.68	
	Breathing techniques	7	1.92	20	2.67	27	2.42	
	Sleep	13	3.57	11	1.47	25	2.24	
	Self-calming	5	1.37	17	2.27	22	1.97	
Fear	Social support	95	12.60	31	16.70	126	15.42	
	Problem solving (action)	52	11.79	29	9.14	81	9.91	
	Reappraisal	51	3.66	9	8.96	60	7.34	
	Self-talk	30	4.47	11	5.27	41	5.02	
	Express/vent emotion	35	2.03	5	6.15	40	4.90	
	Leave emotional situation	21	6.10	15	3.69	36	4.41	
	Experiential avoidance	23	4.47	11	4.04	34	4.16	
	Distraction	23	2.44	6	4.04	29	3.55	
	TV/movies	16	4.88	12	2.81	28	3.43	
	Listen to music	20	2.44	6	3.51	26	3.18	
	Seek comfort	5	2.03	20	3.51	25 25	3.06	
	Creative/relaxing activity	6	2.44	13	2.28	23 21	2.57	
	Breathing techniques	6	2.44	13	2.26	20	2.37	
	Attentional deployment	8	3.25	14	1.93	20 19	2.33	
	Passivity	11	3.23 4.47	8	1.93	19	2.33	

Frequency (Freq) Of ER Strategies Reported, and Percentage (%) Of the Total Strategies Reported for Males and Females, and for Sadness, Anger, and Fear

However, the top strategy reported for anger (for both males and females) was venting or expressing emotion (although social support was still high for anger). Social support, listening to music, watching TV/movies, expressing/venting emotion, creative/relaxing activities, experiential avoidance, reappraisal, and problem solving appeared in the top 15 strategies for all three emotions. Distraction was in the top 15 strategies for sadness and fear, but not for anger.

Thinking of happy memories, comfort eating, and playing video games were reported in the top 15 for sadness, but not for anger or fear. Hostility to others and self-calming were reported in the top 15 strategies for anger, but not for sadness or fear. Finally, self-talk, seeking comfort, attentional deployment, and passivity were reported in the top 15 for fear, but not for sadness or anger.

Eleven out of the fifteen top strategies reported for sadness and anger were the same, and ten out of fifteen strategies were the same for anger and fear. Sadness and fear shared nine out of fifteen strategies. This means that sadness and anger shared 73% of their top 15 strategies (with 27% being different), sadness and fear shared 60% of their top strategies (40% were different), and anger and fear shared 67% of their top strategies (33% were different).

In order to have a sufficient sample size to run Cochran's Q tests, the sample size (n) must be equal to or greater than 4, and nk (the sample size, n, multiplied by the number of related samples, k) must be equal to or greater than 24 (Laerd Statistics, 2017). Also, responses where scores are the same for related groups are subtracted when calculating the sample size (e.g., cases that scored 'No' across all emotions). All of the ER strategies in the present study, except suppression, met this assumption. Suppression was reported in numbers too low to enable further analysis (n = 3).

The results of the Cochran's Q tests are shown in Table 4.8. For all ER strategies, with the exception of passivity, there were significant differences in the percentage of participants reporting this strategy for each specific emotion. In other words, the reported use of ER strategies varied between emotions. Follow up tests were conducted using Dunn's test with Bonferroni correction (Laerd Statistics, 2017).

Table 4.8

Cochran's Q Tests Comparing the Percentage of Participants Reporting Each ER Strategy for Sadness, Anger, and Fear

	Sad	lness	Anger		Fear		Cochr	an's Q	Post-hoc tests		
ER Strategy	п	%	п	%	п	%	Q	р	S-A	S-F	A-F
Avoid situation	4	0.85	14	2.99	16	3.42	7.52	.02	.10	.03	1.00
Distraction	46	9.83	18	3.85	26	5.56	16.64	<.001	<.001	.01	.77
Avoidance	50	10.68	0	0.00	34	7.26	49.52	<.001	<.001	.08	<.001
Hostility	8	1.71	51	10.90	2	0.43	72.64	<.001	<.001	1.00	<.001
Listen to music	86	18.38	47	10.04	25	5.34	53.51	<.001	<.001	<.001	.03
Passivity	12	2.56	10	2.14	16	3.42	1.65	.44	-	-	-
Problem solving	38	8.12	28	5.98	70	14.96	23.87	<.001	.80	.001	<.001
Reappraisal	47	10.04	25	5.34	57	12.18	15.46	<.001	.03	.69	<.001
Rumination	7	1.50	2	0.43	12	2.56	7.14	.03	.54	.54	.02
Self-talk	16	3.42	8	1.71	36	7.69	26.00	<.001	.47	.001	<.001
Social Support	192	41.03	98	20.94	107	22.86	71.74	<.001	<.001	<.001	1.00
Sports/exercise	86	18.38	76	16.24	4	0.85	90.29	<.001	.87	<.001	<.001
Take time out	38	8.12	48	10.26	9	1.92	30.78	<.001	.51	<.001	<.001
TV/movies	64	13.68	31	6.62	26	5.56	26.93	<.001	<.001	<.001	1.00
Venting	60	12.82	130	27.78	37	7.91	84.81	<.001	<.001	.09	<.001

Note. The abbreviations in this table are as follows – n refers to the number of participants who reported that strategy, % is the percentage of the sample that reported that ER strategy, S-A is the post-hoc test comparing sadness and anger, S-F is the post-hoc test comparing sadness and fear, A-F is the post-hoc test comparing anger and fear. Significant values are in bold. Post-hoc tests were carried out using Dunn's test with Bonferroni correction for multiple comparisons.

In terms of the strategies people use when they feel sad, more people reported using distraction, listening to music, watching TV or movies, or seeking social support when they feel sad compared to when they feel angry or scared.

In relation to anger, more people report that they express or vent their emotions, or are hostile to others when they feel angry, compared to when they feel sad or scared. More participants also reported listening to music more when they feel angry than when they feel scared. Similarly, a higher percentage of participants reported using sports or exercise, or taking to time out, to regulate sadness or anger compared to fear.

When individuals feel scared, they are more likely to report problem solving by taking action or talking to themselves (in a reassuring way) than when they feel sad or angry. A higher proportion of participants reported avoiding the situation when they feel scared compared to sad. More people reported ruminating when they feel scared compared to angry. Also, people were more likely to use reappraisal and experiential avoidance to regulate sadness or fear than anger. There were no differences in the use of passivity between the emotions, which was the only strategy that did not have emotion-specific differences.

4.4.3. Gender Differences in Emotion Regulation Strategies

For strategies that were reported by a substantial number of participants (i.e., they yielded a cell count of over 5 in the cross-tabulation), chi-square analyses were conducted to test for gender differences in the reporting of these strategies to regulate sadness, anger, and fear, across all age groups.

As shown in Table 4.9, more females than males reported seeking social support when they felt sad, angry, or scared. Females were also more likely than males to report listening to music, but this was found only for sadness (this was not tested for anger or fear as the cell counts were too low to run the chi-square analysis with sufficient power).

Table 4.9

						% of	group	
						-	ting ER	
					6 CI	stra		
ER strategy	χ^2	р	OR	L	U	Males	Females	Power
<u>Sadness</u>								
Social support	20.20	<.001***	2.41	1.64	3.56	29.15	49.81	.97
Sports/exercise	1.72	.19	-	-	-	21.11	16.36	.27
Listen to music	14.13	<.001***	2.70	1.59	4.60	10.55	24.16	.90
TV/movies	2.01	.16	-	-	-	11.06	15.61	.29
Venting/expressing	16.48	<.001***	3.81	1.92	7.53	5.53	18.22	.93
Reappraisal	2.41	.12	-	-	-	7.54	11.90	.22
Distraction	4.25	.04	-	-	-	6.53	12.27	.32
Avoidance	1.28	.26	-	-	-	12.56	9.29	.29
Anger							- <u> </u>	
Venting/expressing	4.60	.03	-	-	-	22.61	31.60	.46
Social support	20.43	<.001***	3.17	1.89	5.31	11.06	28.25	.98
Sports/exercise	1.82	.18	-	-	-	13.57	18.22	.16
Hostility to others	5.32	.02	-	-	-	7.04	13.75	.50
Take time out	.02	.90	-	-	-	10.05	10.41	.05
Reappraisal	.02	.88	-	-	-	5.53	5.20	.07
Fear				-			<u>.</u>	
Social support	16.96	<.001***	2.70	1.66	4.37	13.57	29.74	.95
Problem solving	1.56	.21	-	-	-	12.56	16.73	.16
Reappraisal	18.98	<.001***	4.59	2.19	9.59	4.52	17.84	.95
Self-talk	4.90	.03	-	-	-	4.52	10.04	.39
Venting/expressing	13.83	<.001***	5.24	2.00	13.70	2.51	11.90	.90
Removing attention	.03	.87	-	-	-	4.02	3.72	.04
Avoid the situation	.17	.68	-	-	-	3.02	3.72	.07
Rumination	.004	.95	_	-	_	2.51	2.60	.05

Results of Chi-Square Analyses Testing the Relationship Between Gender (Males, Females), and Reporting an ER Strategy (Yes, No), for Sadness, Anger, and Fear

Note. The abbreviations used in this table are as follows:

 χ^2 is the chi-square value, *p* is the *p*-value, OR refers to the odds ratio, 95% CI are the 95% confidence intervals, L is the lower confidence interval, U is the upper confidence interval, sample size (*n*) for each chi-square test was *n* = 468, degrees of freedom (*df*) for each chi-square were *df* = 1, alphas (α) were adjusted for multiple comparisons using Benjamini-Hochberg correction, symbol *** denotes significance at the *p* < .001 level, ** is *p* < .01 level, and * is *p* < .05 level after correcting for multiple comparisons, significant values are in bold, power refers to the actual power of the test which was tested post-hoc using G*Power version 3.1 143
In terms of emotion-specific findings, compared to males, more females reported expressing or venting emotions when feeling sad or scared, but this gender difference was not found for anger. Furthermore, females were more likely to report using reappraisal when they felt scared, but this gender difference did not emerge for sadness or anger. For the tests that were significant, the odds ratios show that females were over twice as likely to report these strategies.

4.4.4. The Relationship Between Gender and Engagement, Disengagement, and Distraction

A series of chi-square analyses were conducted to test the relationship between gender and the wider ER categories (engagement, disengagement, and distraction), and these analyses were run in three stages to test (1) the main effect of gender (Section 4.4.4.1), and (2) the gender by age interaction (Section 4.4.4.2). The cross-tabulations used in these chisquare analyses are found in Appendix H.

4.4.4.1. Main Effect of Gender. In the first stage, the relationship between gender and the reported ER categories was examined with the age groups collapsed. The results of the chi-square analyses for this stage are shown in Table 4.10.

Across all age groups, there was a relationship between gender and the use of engagement ER to regulate sadness, anger, and fear. The odds ratios demonstrate that females were around twice as likely as males to report using engagement to regulate sadness, anger, and fear.

4.4.4.2. Gender by Age Interaction. In the previous two stages, significant effects emerged for the use of engagement to regulate sadness, anger, and fear, and the use of disengagement and distraction to regulate sadness. Chi-square tests for each of these

Table 4.10

Results of Chi-Square Analyses Testing the Relationship Between Gender (Males, Females) And Each Category of ER – Engagement, Disengagement, and Distraction (Yes, No) Across All Age Groups, for Sadness, Anger, and Fear

						95	5% CI	% of group reporting strategies		
ER type	Emotion	n	χ^2	р	OR	L	U	Μ	F	Power
Engage	Sadness	447	20.40	<.001***	2.42	1.64	3.57	44.81	66.29	.95
	Anger	418	7.93	.005*	1.80	1.19	2.70	57.65	70.97	.64
	Fear	358	16.20	<.001***	2.48	1.58	3.87	50.00	71.24	.92
Disengage	Sadness	446	.83	.36	-	-	-	16.39	13.31	.13
	Anger	418	2.10	.15	-	-	-	25.29	31.85	.22
	Fear	357	.003	.96	-	-	-	30.08	30.36	.05
Distract	Sadness	446	.46	.50	-	-	-	59.56	62.74	.09
	Anger	419	.93	.33	-	-	-	37.06	41.77	.13
	Fear	355	.97	.33	-	-	-	23.48	28.25	.08

Note. n is the sample size, χ^2 is the chi-square value, *p* is the *p*-value, OR refers to the odds ratio, 95% CI are the 95% confidence intervals, L is the lower confidence interval, U is the upper confidence interval. Degrees of freedom (*df*) for each chi-square test were *df* = 1, alphas (α) were adjusted for multiple comparisons using Benjamini-Hochberg correction. The symbol *** denotes significance at the *p* < .001 level, ** is *p* < .01 level, and * is *p* < .05 level after correcting for multiple comparisons. Significant values are in bold. Power refers to the actual power of the test which was testing post-hoc using G*Power version 3.1

variables were conducted, disaggregated by age, to test if gender differences varied across

different age groups. The results of this analysis are shown in Table 4.11.

There were gender differences in the use of engagement to regulate sadness and fear

in the young adolescent group, but not in the older adolescent or adult age groups.

Specifically, females were over twice as likely to report using engagement to regulate sadness

and fear than males. There were no gender differences in the adult group, for distraction or

disengagement, or for anger.

Table 4.11

Results of Chi-Square Analyses Testing the Relationship Between Gender (Males, Females) And Each Category of ER – Engagement, Disengagement, and Distraction (Yes, No), Broken Down by Age Group, for Sadness, Anger, and Fear

						95% CI		% of group reporting ER strategies		
ER strategy	Group	n	χ^2	р	OR	L	U	Μ	F	Power
<u>Sadness</u>										
Engagement	YA	256	10.03	.002*	2.25	1.36	3.72	46.28	65.93	.69
	OA	100	6.28	.01	2.88	1.24	6.66	33.33	59.02	.48
	Ad	86	3.40	.07	-	-	-	52.63	74.63	.30
Disengagement	YA	255	.15	.70	-	-	-	19.83	17.91	.06
	OA	100	.33	.68 ^a	-	-	-	7.69	4.92	.08
	Ad	86	.20	.70 ^a	-	-	-	15.79	11.94	.08
Distraction	YA	255	1.04	.31	-	-	-	57.85	51.49	.09
	OA	100	.85	.36	-	-	-	56.41	65.57	.12
	Ad	86	.59	.73 ^a	-	-	-	89.47	82.09	.10
Anger										
Engagement	YA	239	4.02	.05	-	-	-	53.98	66.67	31
	OA	94	.02	.88	-	-	-	59.46	57.89	.05
	Ad	82	4.10	.06 ^a	-	-	-	72.22	90.63	.30
Disengagement	YA	239	2.11	.15	_	-	-	24.78	33.33	.17
8-8-	OA	94	.87	.35	-	_	-	24.32	33.33	.08
	Ad	82	.18	.67	-	-	-	33.33	28.13	.05
Distraction	YA	240	.18	.67	-	-	-	38.05	35.43	.98
	OA	94	3.28	.07	-	-	-	27.03	45.61	.29
	Ad	82	.17	.68	-	-	-	55.56	50.00	.06
<u>Fear</u>	-	-		•	-					
Engagement	YA	200	15.95	<.001***	3.22	1.80	5.76	40.66	68.81	.90
	OA	82	.01	.94	-	-	-	64.00	64.91	.05
	Ad	72	.13	.71 ^a	-	-	-	76.92	81.36	.29
Disengagement	YA	201	1.27	.26	-	-	-	35.87	28.44	.12
	OA	80	.62	.43	-	-	-	24.00	32.73	.06
	Ad	72	3.19	.10 ^a	-	-	-	7.69	32.20	.13
Distraction	YA	199	.10	.75	-	-	-	23.08	25.00	.05
	OA	80	.28	.60	-	-	-	20.00	25.45	.05
	Ad	72	.04	1.00 ^a	-	-	-	38.46	35.59	.06

Note. The abbreviations used in this table are as follows:

n is the sample size, χ^2 is the chi-square value, p is the *p*-value, OR refers to the odds ratio, 95% CI are the 95% confidence intervals, L is the lower confidence interval, U is the upper confidence interval, M is males, F is females. The degrees of freedom (*df*) for each chi-square were *df* = 1

The alphas (α) were adjusted for multiple comparisons using Benjamini-Hochberg correction. The symbol *** denotes significance at the *p* < .001 level, ** is *p* < .01 level, and * is *p* < .05 level after correcting for multiple comparisons. The symbol ^a denotes that Fisher's exact test was reported due to at least one cell having an expected count of <5 Significant values are in bold.

Power refers to the actual power of the test which was tested post-hoc using G*Power version 3.1

4.4.5. Summary of Results

A total of 132 unique ER strategies were reported, and females reported more strategies than males. Seeking social support was the top strategy reported for sadness and fear, and venting or expressing emotion was the top strategy for anger. Emotion-specific differences emerged in the percentage of participants who reported distraction, hostility, listening to music, problem solving, reappraisal, self-talk, social support, sports/exercise, taking time out, watching TV/movies, or venting/expressing emotion. No emotion-specific differences emerged in the reporting of passivity.

In terms of gender differences, compared to males, females were more likely to report seeking social support when they feel sad, angry, or scared, listening to music when they feel sad, expressing or venting emotions when they feel sad or scared, and using reappraisal when they feel scared. There was no evidence that males were more likely to use any strategy more than females.

In relation to the wider ER categories, females were more likely than males to use engagement to regulate sadness, anger, and fear, when this was examined across all age groups. When age groups were tested separately, the gender differences in engagement remained for sadness and fear in the young adolescent group only. There were no gender differences in disengagement or distraction.

4.5. Discussion

4.5.1. Results of Study 1

Some gender differences in the reporting of ER strategies emerged in Study 1, however this depended on (1) the emotion being regulated, and (2) the ER strategy, which is summarised in Table 4.12.

Table 4.12

Evidence for Study 1

Research Questions	Evidence
SRQ1. What are the key ER strategies that are reported for sadness, anger, and fear, and do these strategies vary by emotion?	 Although there was overlap in the top strategies, some of the strategies for each emotion were unique. Sadness and anger shared 73% of their most frequently reported strategies, sadness and fear had 60% of the same strategies, and anger and fear shared 67% of their top strategies. For most of the ER strategies, the percentage of participants reporting the strategy varied by emotion.
SRQ2. Are there gender differences in the reporting of specific ER strategies to regulate sadness, anger, and fear, and do these gender differences vary by emotion?	 Gender differences emerged for social support, listening to music, expressing emotions, and reappraisal. Some (but not all) of these gender differences in ER varied across emotions. Females were more likely than males to report using reappraisal to regulate fear, but this gender difference did not appear for sadness or anger Compared to males, females were more likely to vent or express their emotions when they felt sad or scared, but this gender difference did not emerge for anger
SRQ3. Are there gender differences in broad engagement, disengagement, and distraction ER categories?	 Gender differences emerged for engagement, although these were driven by seeking social support and expressing/venting emotion (which made up the majority of the engagement category). There were no gender differences for disengagement or distraction

4.5.2. What Are The Key ER Strategies That Are Reported for Sadness, Anger, and Fear, and Do These Strategies Vary By Emotion?

Compared to emotion-general research, there are relatively few studies which examine the variety of ER strategies individuals use to regulate specific emotions. In the present study, between 27-40% (depending on the emotion) of the most frequently reported ER strategies were unique for specific emotions. This indicates that there is some overlap in the main strategies reported to regulate specific emotions, but there are also strategies that are more likely to be used for certain specific emotions over others.

In previous research, often pre-determined ER strategies are measured using selfreport questionnaires. This approach is limited because (1) it is unclear if there are variations in the ER used to regulate specific emotions when participants freely report on this, and (2) strategies which are deemed important by the participant may not be captured. The Study 1 approach aimed to address these limitations. Individuals appear to use different ER strategies based on the specific emotion that is being regulated (Dixon-Gordon et al., 2015a; Rivers et al., 2007; Vishkin et al., 2020; Zimmermann & Iwanski, 2014). The results of the present study were consistent with these previous findings, as all of the ER strategies that were tested (with the exception of passivity) varied across the emotions. This result indicates that there are emotion-specific differences in the use of ER.

Specifically, individuals were more likely to seek social support, avoid the experience of the emotion, and use distraction techniques such as listening to music, watching TV or movies, or engaging in sports and exercise when they felt sad. Sadness is often thought to serve a social function by eliciting sympathy and support from others after a loss (Balsters et al., 2013; Lench et al., 2015), and so it is logical that individuals are more likely to seek social support in this specific emotional context. This finding that individuals use social

support more for sadness compared to other emotions was also reported in Zimmermann and Iwanski (2014), which was measured using a closed-ended questionnaire.

In the present study, two types of avoidance were measured – avoidance of the emotional experience (experiential avoidance) and avoiding the emotional situation (avoid the situation). Avoiding the situation was reported more for fear, but experiential avoidance was reported more for sadness. There is evidence that individuals may try to avoid the experience of sadness (Dixon-Gordon et al., 2015a; Zimmermann & Iwanski, 2014), which mirrors the findings of the present research, and may suggest that sadness is viewed as being a less tolerable emotion than others. When people feel scared or angry, they may be more likely to leave or avoid the situation, which was found in the present study and in Rivers et al. (2007). Similarly, individuals in the current study were more likely to take action to solve the problem when they were afraid. This may reflect that anger and fear are more action-oriented emotions than sadness, that prompt us into action in situations which may pose a threat to our survival (fear) or block the attainment of a desired goal (anger) (Lench et al., 2016). In contrast, sadness is associated with reduced activity, as this is thought to reflect the loss of a goal, with no possibility of restoring it (Lench et al., 2016). Thus, it is important that the distinction is made in ER research between experiential avoidance and physically avoiding a situation, as the use of these strategies demonstrated an emotion-specific pattern.

In terms of reappraisal, individuals were more likely to use reappraisal when they feel sad or scared, but less likely to use it for regulating anger. This result mirrors that of Dixon-Gordon et al. (2015a), in which reappraisal was more likely to be used for sadness and anxiety than anger. People tend to regulate low intensity emotions using reappraisal, but regulate higher intensity negative emotions using distraction, which intervenes in the emotional response at an earlier stage (Sheppes et al., 2011). Anger is often understood to be an intense, high-energy emotion compared to sadness which is characterised by reduced

physiological activity, seeking comfort and support rather than taking action (Lench et al., 2016). Furthermore, individuals may feel more justified in their anger experience, particularly if it is caused by a perceived wrongdoing, and so they may be less motivated to change it cognitively using reappraisal. The use of reappraisal to regulate fear may depend on whether or not there is an imminent threat in the environment. It may be the case that individuals will take action if there is an immediate threat to wellbeing, but will use cognitive strategies such as reappraisal to reduce the negative emotion if the threat is imagined or not present.

When individuals feel angry, they are more likely to vent or express their emotions or behave in a hostile fashion towards others. This is consistent with the findings of Rivers et al. (2007), who found that people are more likely to verbally express their anger (compared to sadness), and with Zimmermann and Iwanski (2014), who found that individuals dysregulate their anger more than sadness, such as by blaming others for their emotions. This is consistent with the conceptualisation of anger as being triggered by a perceived injustice which can be remedied with the removal of the goal obstacle (Lench et al., 2016), prompting an action-oriented approach to regulating anger, such as venting emotions or expressing hostility to others.

An additional and unexpected finding that emerged in the current study is that many of the most commonly-researched strategies in the field were reported in relatively low numbers. For example, few people spontaneously reported expressive suppression, mindfulness strategies, or rumination. This may indicate that individuals may not have awareness of their own regulation to be able to accurately report on these ER strategies, particularly unconscious strategies. However, it is also possible that these ER strategies are perhaps not as commonly employed in ER as previously thought. In the present research, behavioural strategies such as seeking social support, expressing emotion, and taking action to solve the problem, rather than cognitive strategies, were more frequently reported to

regulate emotions. One potential reason for this finding may be that individuals have more awareness of their behavioural strategies, and less awareness of any cognitive strategies they may use to regulate their emotions. This reflects the dual process model of ER (described in Section 1.3.2.3), in which it is proposed that many ER strategies occur automatically and outside of conscious awareness (Gyurak et al., 2011).

Another explanation for the high number of reported behavioural strategies is that it is likely that people are taught by caregivers to 'do' things when they feel bad, such as seeking social support or solving the problem, rather than being taught to cognitively regulate their emotions such as by using reappraisal (Calkins & Hill, 2007). Thus, it is logical that behavioural strategies were more frequently reported by laypeople in the present study, as these are more explicitly taught. In recent years, more researchers have begun to investigate behavioural ER strategies (Kraaij & Garnefski, 2019), and based on the findings of the present study, this appears to be an important direction for future research.

Two of the most commonly researched strategies are reappraisal and suppression. When ER first began to emerge as a subject of research, these strategies were initially focused upon by Gross and colleagues for theoretical reasons. As mentioned in Section 1.3.2.1, in Gross's (1998b) process model, there are two broad stages of ER - antecedentfocused ER and response-focused ER, and reappraisal and suppression are examples of these stages. Gross's early research strategy involved focusing on a small number of well-defined strategies, and these were strategies that are commonly used in everyday life, lend themselves to both experimental and individual differences research, and are examples of antecedent and response-focused strategies (Gross et al., 2006).

It is unclear how Gross and colleagues initially determined that reappraisal and suppression were commonly used strategies, and thus worthy of further investigation. However, when individuals are asked if they use these strategies (e.g., using the ERQ), some

people do report using reappraisal and suppression in their day to day lives, although there are individual differences in the frequency of use (Gross & John, 2003). However, there is little evidence that these cognitive strategies are necessarily more commonly used than other strategies, such as the behavioural strategies identified in the present study. As previously mentioned, it may also be the case that participants are less able to describe cognitive strategies such as reappraisal and suppression, and/or may use these strategies unconsciously, which would also lead to these strategies not emerging in the data for the present study.

4.5.3. Are There Gender Differences in the Reporting of Specific ER Strategies, and Do These Vary By Emotion?

4.5.3.1. Reporting of Strategies Using the Free-Writing Approach. Females reported more ER strategies than males, which is in line with previous research (Goubet & Chrysikou, 2019; Rivers et al., 2007; Sanchis-Sanchis et al., 2020). These findings may indicate that females have a wider repertoire of ER strategies and may attempt to regulate emotions several times using multiple strategies during an emotion-eliciting event.

Interestingly, Goubet and Chrysikou (2019) and Rivers et al. (2007) used a similar methodological approach to that used in the present study. Although Goubet and Chrysikou (2019) included many common ER strategies in their study, this was not emotion-specific, and was relating to particular scenarios such as academic life, relationships, and health. In relation to gender differences, they found that females were more likely than males to use social support and emotional expression in the academic and romantic situations, which is consistent with the present study. In the study by Rivers et al. (2007), the regulation of sadness and anger was examined within the context of close personal relationships. Although the approach for collecting data was similar, the resulting ER framework in Rivers et al. (2007) differed from that used in the present study, because strategies were grouped into

broad categories which did not distinguish between some of the most common strategies in the literature (such as reappraisal), and so the findings cannot be compared in this regard. Nonetheless, the use of these similar approaches highlights this method as a promising tool for future research, particularly emotion-specific ER studies, and demonstrates how this approach can be used flexibly to meet the research aims of the study.

4.5.3.2. Social Support. Relative to males, females were more likely to report seeking social support when they feel sad, angry, or scared. Females being more likely than males to seek social support from others (particularly emotional support) is a common finding (Barbee et al., 1993; Day & Livingstone, 2003; Kurdek, 1987; Liddon et al., 2018), and so this result is consistent with the wider literature. In Zimmermann and Iwanski (2014), females were more likely to seek social support when they experience negative emotion in general. Although the self-report questionnaire used by Zimmermann and Iwanski (2014) was emotion-specific (i.e., it measured the regulation of sadness, anger, and fear individually), the researchers did not report gender differences in ER in an emotion-specific manner (i.e., all emotions were combined into a single scale). The findings of the present study build on previous knowledge by showing that females are more likely to seek social support for sadness, anger, and fear (i.e., there are no emotion-specific gender differences for this strategy).

4.5.3.3. Expressing or Venting Emotion. In the present study, the relationship between gender and expressing/venting emotions varied according to the emotion being regulated. Females were more likely than males to report that they expressed or vented their emotions (e.g., ranting to others, crying, hitting/punching things) when they felt sad or scared, but this gender difference did not appear for anger.

In Rivers et al. (2007), which used a similar method, females were more likely than males to report verbal expression of emotions to regulate both sadness and anger. The finding

that females are more likely to use verbal expression for sadness is in line with the present study, but the finding for anger is inconsistent with the present study. These differences may be due to the definition of expression. In Rivers et al. (2007), verbal expression refers to telling others how you feel, whereas in the present study expression also included non-verbal behaviours such as crying. Also, Rivers et al. (2007) asked people to describe an emotional situation in which they were sad or angry at a close friend, whereas in the present study participants were asked about a more general emotional situation, and so this difference might also account for some variation between Rivers et al. (2007) and the present study.

The gender difference in expressing/venting emotion is in line with studies from the emotional expression literature which show that females are more likely than males to express their emotions (L. R. Brody & Hall, 1993; Kring & Gordon, 1998), particularly sadness and fear (Allen & Haccoun, 1976; Chaplin, 2015; Chaplin & Aldao, 2013). Generally, in emotional expression research, emotional expressiveness tends to be measured using a self-report questionnaire (Allen & Haccoun, 1976), or in an experimental setting, by eliciting emotion using stimuli (e.g., emotional pictures, films) and measuring participants' outward expressions of emotion such as facial, vocal, and postural expressions, which are coded by trained observers (Kring & Gordon, 1998). These methodologies differ from that used in the present study, which relied on participants recalling situations where they had used venting or emotional expression (e.g., crying or ranting to another person) as a way of regulating their emotions and reporting on this behaviour. Although the methodologies used are different, the results for sadness and fear found in the present study mirror the wider expression literature and suggest that females may be more likely than males to express feeling sad or scared, and to use expressing or venting as a way of regulating these emotions.

The findings for anger are more complex. Some studies have shown that males express anger and overt aggression more than females (Archer, 2004; Chaplin & Aldao,

2013), even in the absence of gender differences in reported levels of anger (Archer, 2004), and females suppress their anger more (Cox et al., 2000; Kwon et al., 2013). On the other hand, females are more emotionally expressive towards anger-inducing stimuli during experimental tasks (Kring & Gordon, 1998; Schwartz et al., 1980; Wagner et al., 1993; Zuckerman et al., 1975). These findings may indicate that other factors, such as the context of the situation (e.g., whether a female feels safe to express her feelings), play a role in whether or not females express feelings of anger (Jack, 2001). They may also point to the hypothesis that females are more emotionally expressive of anger in a laboratory setting, but perhaps have learned to dampen this response by using suppression to align with societal expectations of display rules around anger (Chaplin, 2015).

In the present study, there was not enough evidence to support the hypothesis that gender differences exist in the use of expression/venting to regulate anger. However, it should also be noted that the chi-square test for anger was somewhat less powered (.46) than that for sadness and fear, and so this test may have been less sensitive in detecting a gender difference if one exists (i.e., type II error). As previously noted, the sample size for adults, particularly adult males, was limited, and so a larger sample size amongst this age group may yield different findings. In sum, no gender differences in using venting or expressing emotion to regulate anger were found in the present study.

4.5.3.4. Reappraisal. Reappraisal is arguably the most researched strategy in the field of ER (Aldao et al., 2010; Gross, 1998a, 2002). In the present study, females reported using reappraisal more than males, but this gender difference appeared for fear only (not for sadness or anger). This result indicates that when individuals are experiencing fear, females are more likely than males to attempt to change their thoughts about the situation in order to reduce their fear. Interestingly, a higher percentage of males reported engaging in problem solving than reappraisal when they feel afraid, whereas females were equally likely to

reappraisal or problem solve. This highlights that females may be more likely than males to engage in cognitive strategies to cope with their fear, whereas males prefer behavioural strategies such as taking action during fearful situations. These findings are consistent with evidence that females tend to use cognitive ER more than males (Garnefski et al., 2004) and males are more likely to use problem-focused coping whereas females use emotion-focused coping more (Brougham et al., 2009; Folkman & Lazarus, 1980; Pearlin & Schooler, 1978).

Previous findings in the literature regarding gender differences in reappraisal have been mixed. Using self-report questionnaires, previous studies indicate that females report using reappraisal more (R. C. Martin & Dahlen, 2005; Nolen-Hoeksema & Aldao, 2011; Smrtnik-Vitulić & Prosen, 2016; Spaapen et al., 2014), whereas others show that males report using reappraisal more (Esmaeilinasaba et al., 2016; Öngen, 2010). A potential reason for these mixed findings may be that previous research has failed to measure ER in an emotionspecific manner. In the present study, when specific emotions were measured separately, clearer gender difference emerged. That is to say, there were gender differences in reappraisal for fear (with females being more likely to use reappraisal), but not for sadness or anger. In other words, gender differences in the reported use of reappraisal depended on the emotion being regulated. This finding brings clarity to an area which has previously had complex and mixed findings. This finding also lends support to the value of investigating gender differences in the regulation of specific emotions, and suggests that other researchers may benefit from conducting ER research in an emotion-specific manner moving forward.

4.5.3.5. Listening to Music. Some studies have recognised listening to music as a way of regulating emotions and coping with stress (Cook et al., 2019; Kurdek, 1987; Miranda & Claes, 2009; Silverman, 2020). However, it is not a commonly studied strategy, and consequently there are few studies examining gender differences in this ER strategy. In the present study, listening to music was reported more by females (compared to males), and this

gender difference was found for sadness only (this was not tested for anger or fear due to low cell counts in some of the cells of the cross-tabulations).

This finding is consistent with previous research showing that females are more likely than males to listen to music to regulate emotions more generally (Cook et al., 2019; Kurdek, 1987). However, the results of the present research add value to the literature and build on these previous findings by (1) highlighting the importance of listening to music as one of the most common strategies reported to regulate emotions, and so accordingly should receive more attention in the ER literature, and (2) elucidating that gender differences in listening to music occur for sadness specifically.

4.5.4. Are There Gender Differences in Broad Engagement, Disengagement, and Distraction ER Categories?

Compared to males, females were more likely to report using engagement strategies to regulate sadness, anger, and fear. When gender differences were tested in each age group separately, gender differences in engagement remained only for sadness and fear, and only for adolescents. Contrary to what was expected, there were no gender differences in disengagement or distraction.

The engagement category mostly consisted of expressing/venting emotion and social support, and so gender differences in engagement were driven by these strategies. However, as previously noted, it is possible that expressing/venting emotion is not an ER strategy as such, but is rather a type of emotional behaviour, and the reporting of expressing/venting emotion may reflect a lack of awareness on the part of the individual as to how they regulate their emotional responses. This lack of awareness is consistent with previous studies, which indicate that ER is often an unconscious process (Gyurak et al., 2011; Timmer-Murillo et al., 2020).

This research question (SRQ3) was largely developed on the basis of observed gender differences in rumination in the literature (D. P. Johnson & Whisman, 2013), which was conceptualised in this research as a type of engagement. However, we can see from the breakdown of ER strategies in the engagement category (see Table 4.4), that rumination did not appear as a top contributor to engagement, and in fact rumination made up only 2% of the engagement category. However, as noted in Section 4.5.3, individuals may not intuitively have an understanding or awareness of using a cognitive ER strategy such as rumination, and so may be less likely to report using this compared to an action-oriented strategy such as seeking social support. Nevertheless, it cannot be inferred that females use engagement as an ER strategy more generally, but rather females are more likely to report using specific strategies such as seeking social support, and these were interpreted as examples of engagement by the researcher, using the chosen framework for this study.

It is possible that this framework, which grouped individual ER strategies into wider categories, was not an appropriate method to sufficiently explore this research question. Instead, examining gender differences in ER at the level of individual ER strategies, as reported in Section 4.4.3, may be a more accurate test of whether females are more likely to use strategies classed as engagement. This approach may be more appropriate to test this hypothesis because it may be the case that females are more likely to use some types of engagement (e.g., rumination) but not others (e.g., reappraisal), and as demonstrated in the present study, this may also depend on the specific emotion. Therefore, the wider categories may be too general to capture these gender differences.

There was no evidence that males were more likely to use any strategy more than females. Females used engagement more than males and they use other strategies just as much as males. This suggests that females appear to have more ER strategies available to them, and these additional ER strategies can be classed as 'engagement' ER. However, as

reported in Section 4.4.1.1, there were gender differences in the missing values for all of the emotion questions, with males having a higher percentage of missing values than females for the sadness, anger, and fear questions (Table 4.6). This may indicate that males were perhaps less able to verbalise the strategies that they use to regulate their emotions, or may have had less awareness of these strategies to be able to accurately report them. Therefore, it is possible that males have more difficulty reporting their ER strategies, and were at a disadvantage with the open-ended approach to data collection.

It was surprising that no gender differences emerged amongst the adult group, as the findings from previous studies show gender differences in strategies such as avoidance (Zimmermann & Iwanski, 2014) and distraction (Trives et al., 2016) amongst adults. However, a limitation of the present study was that the sample size for the adult group was small, and there was a particularly low number of adult males (n = 19). This low sample size meant that some of the analyses for the adult age groups were quite severely underpowered, and so it would be unlikely to find a gender difference using these tests, even if one did exist (i.e., type II error may have occurred).

A further consideration is the possibility that the categories used to classify the ER strategies were not valid. The agreement between the independent coders for these categories was $\kappa = .59$, which is considered a moderate level of agreement, according to Cohen's (1960) original interpretation. However, there is still substantial room for improvement, and this value demonstrates that there was still a degree of disagreement in how the ER strategies were categorised. The framework used in the present research was empirically derived (Parkinson & Totterdell, 1999), however it may be the case that these categories were not an accurate representation of different types of ER. This issue is relevant to all of the results regarding the wider categories – engagement, disengagement, and distraction.

Contrary to what was expected, no gender differences emerged in distraction or disengagement. This finding is surprising considering that gender differences in distraction (Copeland & Hess, 1995; Gomez-Baya et al., 2016; Trives et al., 2016), suppression (Gross & John, 2003; Spaapen et al., 2014), and avoidance (Zimmermann & Iwanski, 2014) have emerged in the wider literature. The lack of gender difference may be due to the definition of the wider ER categories (see Table 4.4.).

The distraction category consisted of different activities to distract from the emotion such as listening to music, watching TV or movies, and sports/exercise, as well as a general distraction strategy. We know from the results of the specific ER strategies (reported in Section 4.4.3), that there are gender differences in some of these activities but not others, and so it is possible that grouping them together into one broad category masks these gender differences. Similarly, disengagement consists of a mix of experiential avoidance (i.e., trying not to feel the emotion) and behaviours such as avoiding the situation. It may be that this category is too broad to detect gender differences that involve disengaging with the emotion.

Furthermore, most of the gender differences in the literature emerged amongst adult participants (Esmaeilinasaba et al., 2016; Garnefski et al., 2004; Smrtnik-Vitulić & Prosen, 2016; Trives et al., 2016), and as already discussed, the limited sample size in the adult age group and the resulting lack of power in the analysis may have led to potential gender differences being undetected.

In sum, although statistically significant gender differences were found in the use of engagement, these were driven by social support seeking and venting/expressing emotion, and so there was only partial support that females are more likely to use engagement when they are regulating emotions. Furthermore, there were no gender differences in the use of disengagement or distraction, possibly due to low sample size or the definition of these categories. Thus, there was no evidence to suggest that males are more likely to disengage or

distract from their emotions. In hindsight, it is possible that grouping ER strategies into wider categories of engagement, disengagement, and distraction may have not been sufficient to test this hypothesis. Instead, it may be more appropriate to examine gender differences at the level of the individual ER strategy (such as rumination, and reappraisal), as was reported in Section 4.4.3.

4.5.5. Limitations of this Study

There are two broad categories of limitations of this study which should be noted – the methodological approach, and the generalisability of the findings. The methodological approach used in the present study relies on individuals being able to accurately recall their past ER strategies. There is evidence from previous research that this method is valuable for measuring ER (Gross & John, 2003; Rivers et al., 2007; Zimmermann & Iwanski, 2014). However, it is possible that individuals may not always be able to accurately recall their ER, and may instead rely on their general beliefs about emotions (Robinson & Clore, 2002). Nonetheless, the vast majority of ER research relies on participants self-reporting either their ER strategies or their levels of emotion, and many important findings have emerged using this approach.

Similarly, the method also depends on individuals having an awareness of their ER processes and emotional experiences. In other words, people must have an understanding of when they are experiencing an emotion, what emotion it is, and what they did when they experienced that emotion. The high percentage of participants reporting on their emotional behaviour (e.g., emotional expression) rather than a regulation strategy may indicate that people have limited insight into this process. Some strategies which are salient in the ER literature, such as suppression, were reported in low numbers in the present study using this participant-led approach. The absence of these ER strategies may not have been reported by

participants because they are unconscious strategies that the individual is not familiar with, or is aware of using.

This difficulty in identifying emotions and ER may be a problem for the younger participants in particular, who may not yet have developed the emotional awareness to be capable of reflecting and reporting on their ER strategies in detail. It is notable that the older adolescents had more missing values than adults, suggesting that they may have had more difficulty to report on their ER strategies. The higher percentage of missing values amongst males (compared to females) also suggests that males may be less able to verbalise their ER strategies or were less aware of them. Nevertheless, despite these missing values, the majority of participants in all groups were able to provide insightful answers to the ER questions, suggesting that these issues probably only affect a minority of participants.

Furthermore, although the overall sample size in the present study was healthy (n = 468), the methodological approach relied on individuals spontaneously reporting on ER strategies without prompting. This means that if no participants reported using a particular strategy, then there would be no data for that strategy. Consequently, the number of people reporting some strategies was quite low, and resulted in low cell counts in the cross-tabulations, and unequal cell sizes. This resulted in low power for some of the analyses, and meant that some strategies could not be analysed using inferential analysis. The non-significant analyses with low power should therefore be interpreted with caution, as there is an increased risk of Type II error, i.e., failing to find an effect when one exists (Laerd Statistics, 2016).

A similar issue relates to the group sizes. There was a small number of participants in the adult male group, which suggests that there may be self-selection bias amongst this group which could limit the generalisability of the findings. The main reason for the underrepresentation of males in the adult age group is that fewer males expressed interest in taking

part in the study. This reflects a wider problem with sampling in social research as a whole, particularly research in psychological sciences. Males are much less likely than females to take part in research (Burg et al., 1997; Dunn et al., 2004; Hille et al., 2005; Markanday et al., 2013; Porter & Whitcomb, 2005; Wild et al., 2001), particularly studies regarding mental health (Woodall et al., 2010). Males are also less likely than females to participate in online surveys (G. Smith, 2008), which was the method for collecting data from adults in the present study. This may be a particular challenge for research which involves emotions or feelings. As previously discussed, males often receive messages that any sort of expression of emotion is not perceived as 'masculine' and is generally regarded as less acceptable for males (Chaplin, 2015; Root & Denham, 2010). Anecdotally, some males explicitly expressed to the author during the course of this study that they did not wish to take part because of the research topic.

It is important that future studies examining gender take additional measures to ensure that males are recruited in adequate numbers to enable equal group sizes between males and females. This may involve recruiting males only for a period of time during the recruitment phase, and targeting male-dominated environments, such as traditionally male industries or university courses. The way studies are advertised may result in self-selection bias (Sutton & Edlund, 2019), and so it may be prudent to present the research as examining emotions amongst a range of other variables, to mask the 'emotional' aspect of the study in order to increase the attractiveness of the study to males.

A related issue is the potential influence of cultural expectations on the reporting of ER strategies. According to cultural norms, it is generally regarded as less acceptable for males to express feelings of sadness and fear, and for females to express anger (Root & Denham, 2010). Therefore, these beliefs may influence the reporting of emotional expression, and this may result in males feeling less able to report that they express sadness or fear, and

females being less likely to report expressing anger, which would reflect the findings of the present study, at least for sadness and fear.

It should also be noted that the majority of the strategies reported are regarded as adaptive and socially acceptable strategies, such as seeking social support, sports/exercise, listening to music, problem solving etc. It is possible that social desirability (i.e., that the participant wants to present themselves in a favourable light to the researcher) may limit the reporting of perceived maladaptive strategies, which the participant may believe is less acceptable to report. Therefore, there is a risk that the use of maladaptive strategies is underrepresented in the present research.

With regards to how generalisable the results are, a relatively narrow age range was chosen for the present study in order to examine if gender differences vary across these age groups, and to control the impact on age on the results. However, it may be the case that these findings cannot necessarily be generalised to people of all ages, such as older adults. Similarly, convenience sampling is often used in psychological research to utilise the availability of student participants on university campuses (Richmond et al., 2015). The sample used in the present study mostly consisted of secondary school pupils and university students, and so caution should be exercised when generalising the findings to the wider population. However, there is evidence that ER amongst undergraduate students is similar to that of the general community (Preece et al., 2019), which lends support to the generalisability of the present study.

4.5.6. Implications of Findings

4.5.6.1. Theoretical Implications. The present study is one of the first to examine gender differences in the reporting of ER strategies to regulate specific emotions. These findings have important implications for the field of ER. First of all, this research highlights

that what we know about the regulation of one negative emotion cannot necessarily be applied to all negative emotions. None of the key theories of ER (discussed in Section 1.3.2), such as the process model (Gross, 1998b), the dual process framework (Gyurak et al., 2011), or the Parkinson and Totterdell (1999) taxonomy take into account the divergent characteristics of specific emotions in their theoretical frameworks of ER. The field may benefit from the development of emotion-specific theory based on the empirical evidence of the present study.

Furthermore, gender differences in ER have been shown to depend on the specific emotion being regulated. For example, the present study was the first to show that gender differences in the use of reappraisal vary by emotion. This brings clarity to an area that was plagued with contradictory findings, and shows that this emotion-specific approach may provide valuable information about gender differences in the use of other strategies. Therefore, in order to gain a clear understanding of gender differences in ER (and the impact that these relationships may have with outcomes of ER), future research may benefit from being conducted in an emotion-specific manner.

A related point is that there is a lack of available instruments to measure emotionspecific ER. The present research highlights the methodological approach used in the present study as a valid method of measuring the regulation of specific emotions. The flexibility of the approach also means that it could be used to research ER in different contexts or for different emotional intensities, as in studies by Goubet and Chrysikou (2019) and Rivers et al. (2007). However, a caveat of this approach is that it requires a large sample size to enable analysis to be conducted on the resulting ER strategies, as the cell counts in the crosstabulations are directed by the strategies that participants report (i.e., it cannot necessarily be used to target a particular strategy for analysis, because participants may not report this in great numbers). Nonetheless, this approach has shown to be a valuable method of measuring

emotion-specific ER, and has provided unique information regarding gender differences to the ER literature.

The present study also revealed strategies which may be under-researched as potential areas for growth in the field. In particular, some of the most commonly reported strategies in the present study such as social support, sports/exercise, listening to music, and even expressing or venting emotion (as a way of regulating emotions) do not receive as much attention in the field, whereas cognitive strategies which were not widely reported in the present study, may receive disproportionate attention in the literature, and so this may be an important avenue for future research. Expressive suppression, one of the most frequently studied strategies (Gross & John, 2003; Gross & Levenson, 1993; Kelley et al., 2019; Peters et al., 2019; Rogier et al., 2019; Sullivan & Kahn, 2020; Webb et al., 2012), was reported by very few people in the present study. As of yet, it is unclear if behavioural strategies have an adaptive or maladaptive impact on life outcomes over time, such as on mental health. Thus, the present research has highlighted these strategies as an interesting area for exploration in future research.

4.5.6.2. Practical Implications. The findings of the present study also have implications for therapeutic practice. The way that individuals regulate their emotions is known to have an important impact on variables associated with mental health (Garnefski & Kraaij, 2006; Gross & John, 2003; Hu et al., 2014; Kudinova et al., 2016; Zahniser & Conley, 2018). Therapies to tackle poor mental health are moving towards becoming more personalised to the individual (Denny, 2020), and ER training is becoming a more popular method for tackling mental health problems (Gratz et al., 2015; Kiosses et al., 2018; LeBlanc et al., 2017; Morris et al., 2015; Ranney et al., 2017).

Consequently, it is important to gather data on group differences in ER strategy use, such as gender differences. The benefits of this approach are twofold. Firstly, it highlights

areas which may be problematic for individuals from a certain group. For example, less than 10% of males reported using reappraisal to regulate sadness, anger or fear, which demonstrates that the use of this adaptive cognitive strategy may be being underutilised amongst this group and so males may need to receive more training and support around this strategy.

Secondly, it may uncover strategies which are more likely to already be used by members of a certain group. For example, half of females reported that they seek social support when they feel sad. This demonstrates that many females are already sufficient in reaching out for support when required (although there is still room for improvement), and so females may need less support in this area, but may benefit more from focusing on other strategies. Therefore, the findings of the present study may provide valuable information to practitioners which guides tailoring treatments for mental health problems and clinical disorders to an individual's needs.

A further finding from the present study is that the majority of strategies that individuals reported were behavioural (e.g., seeking social support, listening to music). This highlights that cognitive strategies (such as reappraisal) may be underutilised in the general population. There is a vast evidence base for the benefits of using cognitive strategies for mental health (T. S. Davis et al., 2014; Garnefski & Kraaij, 2006; Gratz et al., 2015; Gross & John, 2003; Hu et al., 2014; Huffziger et al., 2009; Kiosses et al., 2018; Kudinova et al., 2016; LeBlanc et al., 2020; LeBlanc et al., 2017; Morris et al., 2015; Polanco-Roman et al., 2015; Ranney et al., 2017; Zahniser & Conley, 2018).

The lack of reporting on cognitive strategies reveals a potential risk factor for the development of mental health problems. It is clear that many individuals do not have the cognitive skills to cope with intense emotions, which may begin to manifest as poor mental health over time (Brewer et al., 2016; S. L. Johnson et al., 2016; Zahniser & Conley, 2018).

Therefore, it is important that therapies focus on training individuals with these cognitive skills that can help to improve mental health amongst individuals struggling with mental health problems and clinical disorders. Additionally, promoting the cultivation of these skills within the education system may also have benefits for the mental health of the wider population (Macklem, 2011; R. E. Martin & Ochsner, 2016).

Females used social support more than males. As social support is generally regarded as an adaptive strategy with many benefits for mental health (Harandi et al., 2017), this finding indicates that males may be less likely to get the support that they need when they are experiencing intense emotions. There is evidence that the cultural messages males receive about what it means to ask for help (e.g., that it is a sign of weakness), prevent them from seeking social support when they are depressed (Barbee et al., 1993; B. P. Cole & Ingram, 2020), and this may account for the low proportion of males reporting this strategy in the present study (14-29%, depending on the emotion). This finding highlights a potential risk factor for males, and may reflect the higher rates of dying by suicide that are observed amongst males (Dougall et al., 2017; Miranda-Mendizabal et al., 2019; National Records of Scotland, 2021; P. L. Mok et al., 2012; Stark et al., 2004). Therefore, it is important to examine the reasons why males are less likely to reach out for support and investigate the impact that this may have on their mental health and wellbeing, in order to address any barriers males face in seeking support.

4.5.7. Directions for Future Research

As discussed in Chapter 3, different methods can be used to investigate ER, and these approaches provide different information about ER. The approach used in the present study represents a departure from previous methods (Gross & John, 2003). A benefit of adopting this approach is that it allowed for emotion-specific variations in the reporting of ER to be

assessed, for the first time, across a wide range of strategies. As discussed in this chapter, this method yielded some interesting findings, including gender and emotion-specific differences in the reporting of ER strategies. These findings therefore bring us closer to understanding the relationship between gender and the regulation of specific emotions.

However, there are also other approaches that can be used to investigate ER. Despite the unique contribution of this study, the method used does not allow for the effectiveness of ER strategies to be assessed. It is important to examine not only whether there are gender differences in the reported use of ER, but also if males and females differ in how successfully they can implement ER strategies to reduce an emotion. This is important, because ER effectiveness is associated with a number of affective, cognitive, and social consequences (John & Gross, 2004).

To measure ER effectiveness, an experimental paradigm can be used, which involves participants attempting to reduce emotions that are induced using emotional stimuli, and reporting on their emotional experience. This approach has been used in many studies in the field (Goldin et al., 2008; Gross, 1998a; Gross & Levenson, 1997; McRae et al., 2012b; Richards & Gross, 2000; Sheppes & Meiran, 2007; Sheppes et al., 2014; Sullivan & Kahn, 2020). The change in levels of emotion after using the strategy provides an indication of how successfully the individual used the ER strategy, which is their ER effectiveness. Now that the findings of Study 1 have demonstrated there are emotion-specific gender differences in the reporting of some ER strategies, and these can vary across specific emotions, it is important to investigate whether there are emotion-specific gender differences in ER effectiveness.

In addition, an important next step for this research is to examine the potential relationship these gender differences may have with factors which are known to be related to ER, such as mental health. There are gender differences in variables associated with mental

health, such as depression (H. Chen et al., 2019; Girgus & Yang, 2015), and self-harm regardless of the intention (Hawton et al., 2012; Knudson et al., 2020; Madge et al., 2008; O'Connor et al., 2009). Further, ER is associated with these variables (T. S. Davis et al., 2014; Garnefski & Kraaij, 2006; Polanco-Roman et al., 2015). Therefore, it is possible that ER may play a role in gender differences in mental health problems. No research to date has investigated these relationships in an emotion-specific manner. Investigating gender differences in the effectiveness of regulating sadness, anger, and fear, and examining the relationship between emotion-specific ER and gender differences in variables associated with mental health will be addressed in Study 2, which is reported in Chapter 5 (Study 2a) and Chapter 6 (Study 2b).

4.6. Conclusion and Next Steps

The aims of Study 1 were (1) to measure the key strategies used to regulate specific emotions and test whether these varied across emotions, (2) to examine gender differences in the reporting of ER to regulate sadness, anger, and fear, and (3) to test whether emotion-specific gender differences exist in the use of engagement, disengagement, and distraction.

There was partial support for the hypotheses of the present study. Gender differences in the reporting of some ER strategies emerged, but this depended on (1) the specific emotion being regulated, (2) the age group being examined, and (3) the ER strategy. Of note, gender differences emerged in the reporting of social support, listening to music, venting/expressing emotion, and reappraisal. Importantly, gender differences in reappraisal emerged for fear, but not for sadness or anger, which demonstrates that gender differences in the use of reappraisal are indeed emotion-specific. This finding brings clarity to a literature which has been plagued with mixed findings. Support for the engagement hypothesis was limited. The most important message to take away from this study is that what is known about the regulation of one specific emotion cannot necessarily be applied to all emotions. The reporting of some ER strategies, and gender differences in the reporting of some ER strategies, differed according to the specific emotion. This variation in the use of ER indicates that specific emotions are regulated in different ways, and gender interacts with the specific emotion. The findings of the present study highlight the importance of moving towards a discrete emotions framework by conducting future ER research, particularly around gender differences, in an emotion-specific manner.

The methodological approach in the present study allowed for the reporting of ER strategies to be assessed but does not measure the effectiveness of ER. As ER effectiveness is an important skill for many areas of life, and few previous studies in the literature have examined ER effectiveness in an emotion-specific manner, in Study 2, the effectiveness of regulating sadness, anger, and fear during an emotion regulation task will be examined.

Furthermore, it emerged in the present study that behavioural strategies, rather than cognitive strategies, were more commonly reported by participants. This is interesting, because the majority of strategies that are investigated in the field of ER are cognitive strategies, such as reappraisal, suppression, distraction, and rumination. It is possible that participants may be less able to report on these cognitive strategies. As cognitive strategies have been highlighted as being important in the field, these will be examined in Study 2. The first part of Study 2 (Study 2a) will be reported in Chapter 5.

KEY POINTS FOR CHAPTER 4

- The aims of Study 1 were (1) to measure the key strategies used to regulate specific emotions and test whether these varied across emotions, (2) to examine gender differences in the reporting of ER to regulate sadness, anger, and fear, and (3) to test whether emotion-specific gender differences exist in the use of engagement, disengagement, and distraction
- 2. The reporting of ER strategies often varied across emotions, and gender differences in the reporting of some (but not all) ER strategies were emotion-specific.
- 3. There was some evidence to suggest females are more likely than males to use engagement ER.
- 4. This research demonstrates the importance of examining ER in an emotionspecific manner, and highlights the methodological approach used in this study as a valuable measure of emotion-specific ER for future research.

QUESTIONS FOR SUBSEQUENT CHAPTERS

1. Are there gender differences in how effectively individuals can regulate sadness, anger, and fear? (Chapter 5)

2. Are gender differences in the regulation of sadness, anger, and fear related to gender differences in variables associated with mental health? (Chapter 6)

NEXT STEPS

Study 1 demonstrated that there are gender differences in the reporting of some ER strategies. However, this does not provide information about how effectively males and females use these ER strategies to regulate their specific emotions.

Study 2 is in two parts. The aim of Study 2a (Chapter 5) is to investigate if there are gender differences in the effective regulation of sadness, anger and fear, using two specific ER strategies – reappraisal and distraction – during an emotion regulation task. In Study 2b (Chapter 6), the relationships between gender, emotion-specific ER, and variables associated with mental health will be explored.

Chapter 5: Gender Differences in the Regulation of Sadness, Anger, and Fear during an Emotion Regulation Task (Study 2a)

5.1. Chapter Summary

The results of Study 1 demonstrated that the ER strategies people report using, and gender differences in the reporting of ER, often depend on the specific emotion. However, it remains unclear whether there are gender differences in how effectively individuals regulate specific emotions. This is important, because being able to effectively use an ER strategy is associated with several significant life outcomes. To date, no study has examined gender differences in ER by systematically comparing the effectiveness of ER strategies across specific emotions, which is the aim of the present study.

An emotion regulation task (ERT) was used to examine the regulation of sadness, anger, and fear. The ERT allows for ER to be measured in the moment and does not rely on participants having to recall previous ER strategies. Measuring ER in this way provides a different lens in understanding gender differences in emotion-specific ER.

Study 2 is split into two parts. The data for Study 2 were collected at one time-point, but two separate sets of analyses were conducted to test separate aims. In Study 2, emotiongeneral ER, depressive symptoms, self-harm behaviour, suicidal ideation, and wellbeing were measured using online self-report questionnaires, and the ERT was used to assess emotionspecific ER. In Study 2a (described in the present chapter), gender differences in the regulation of specific emotions using reappraisal and distraction were tested using analyses of variance (ANOVA). In Study 2b (reported in Chapter 6), mediation analyses were used to examine whether the regulation of specific emotions using reappraisal and distraction mediates the relationship between gender and self-harm behaviours, suicidal ideation and depressive symptoms. As exploring gender differences amongst the adult age group was limited in Study 1 due to a low sample size, adults were recruited in Study 2, and correlation analyses were conducted to examine if age was related to the study variables.

5.2. Introduction

5.2.1. Rationale for Study 2a

As discussed in the preceding chapters, previous research has examined gender differences in ER in an emotion-general manner. An important objective of all empirical research is to test the robustness of evidence by examining whether the findings of previous studies can be replicated (Lamal, 1990). For example, testing whether prominent gender differences, such as males using suppression more than females to regulate general emotion (Gross & John, 2003; Spaapen et al., 2014; Zimmermann & Iwanski, 2014), are found in the sample of the present study. Therefore, the first aim of the current study is to test whether previous findings regarding gender differences in emotion-general reappraisal, distraction, suppression, and rumination can be replicated, using standard and commonly used self-report questionnaires from the literature.

The Study 1 method resulted in the ER strategies that participants were aware of using being recorded. However, as mentioned in Section 4.5.3, the strategies that were reported by participants were different from the strategies identified in the ER literature as being important. The lay view captured in Study 1 was that the majority of ER strategies were behavioural, such as seeking social support, sports/exercise, and listening to music. However, psychologists who study ER have proposed that cognitive strategies, such as reappraisal, suppression, distraction, and rumination are important strategies that people use frequently to regulate emotions in their daily lives (Aldao et al., 2010; Gross & John, 2003; Webb et al., 2012). Their views are based on empirical evidence that these cognitive strategies are important in ER (Gross, 1998a; Gross & John, 2003). On the basis of these experts' views, reappraisal, distraction, suppression, and rumination were chosen to examine in the present study in an emotion-general manner using questionnaires, and reappraisal and distraction were investigated in an emotion-specific manner experimentally. Although there is evidence

that these strategies are used by individuals to regulate their emotions (as measured by selfreport questionnaires such as the ERQ), it remains unclear whether these strategies are used more so than others, such as the behavioural strategies that emerged in Study 1.

The Study 1 approach does not provide information about how effectively individuals regulate their emotions, and whether gender differences exist in ER effectiveness. This is an important aim for ER research, because ER effectiveness is associated with a number of outcomes, such as mental and physical health (Denny, 2020; Ford et al., 2017; Quinn & Joormann, 2020), quality of social relationships (E. A. Butler et al., 2003; Ivcevic & Eggers, 2021), positive behaviour in children (Kao et al., 2020), levels of self-control (Lawyer & Jenks, 2020), academic achievement (Ivcevic & Eggers, 2021), and positive work outcomes (Chandra et al., 2020).

A useful method for capturing ER effectiveness is by measuring the change in selfreported emotion levels after an individual uses an ER strategy during an emotion regulation task (ERT). The ERT has been used in many previous ER studies (Boland et al., 2019; Goldin et al., 2008; Gross, 1998a; Gross & Levenson, 1997). An individual's ability to regulate their emotions in the moment is measured during the ERT, and this method does not rely on the participant's ability to recall past ER strategies, or even to have an awareness of their ER processes. In the present study, the ERT will be adapted to measure the regulation of specific emotions, which may produce valuable and unique emotion-specific findings.

Two ER strategies which are commonly implemented during the ERT are reappraisal and distraction. These strategies were focused upon in the present research specifically because (1) there tend to be mixed findings regarding gender differences, and so including these in the present research offers an opportunity to find out if examining this in an emotionspecific manner brings clearer findings, (2) these are two of the most widely studied strategies in the literature and so findings can be related to the wider literature, (3) they are

both adaptive strategies associated with positive outcomes and so are beneficial for participants to implement in an experimental setting, and (4) both strategies have clinical relevance as they are associated with mental health variables.

Reappraisal and distraction tend to have mixed findings in the literature regarding gender differences in their use (Esmaeilinasaba et al., 2016; Garnefski et al., 2004; Nolen-Hoeksema & Aldao, 2011; Trives et al., 2016). A reason for these mixed findings may be that previous studies have failed to examine these strategies in an emotion-specific manner. When gender differences in the use of distraction to regulate a low mood (which is similar to sadness) was examined, clearer gender differences emerged (Trives et al., 2016). However, very few studies have examined gender differences in reappraisal or distraction in an emotion-specific manner.

Reappraisal and distraction have both been shown to effectively reduce negative emotion in an experimental setting (Gross, 1998a; Quinn & Joormann, 2020; Smoski et al., 2014). However, according to Gross's process model, which was described in Section 1.3.2.1, distraction is a more effective strategy than reappraisal for reducing negative emotion, as it intervenes earlier in the emotion generative process (Gross, 1998a). This has been supported with some empirical evidence (Sheppes et al., 2009; Sheppes & Meiran, 2007, 2008; Smoski et al., 2014; Thiruchselvam et al., 2011). Although this has been researched more generally, it remains unclear whether the effectiveness of these strategies differ according to the specific emotion being regulated. Research suggests that reappraisal is used to regulate low intensity emotions, whereas distraction is preferred for high intensity emotions (Sheppes et al., 2011). Sadness is often thought of as a low intensity emotion that is associated with reduced physiological activity (Lench et al., 2016). Based on this, it is possible that reappraisal may be more effective at reducing sadness, and distraction may be

more effective at reducing anger and fear. The second aim of this study is to investigate if the effectiveness of reappraisal and distraction depends on the specific emotion being regulated.

In any research examining gender and emotion, it is important to have an awareness of the cultural messaging around specific emotions, and how this may impact on ER processes. In relation to gender differences in ER, adhering to gender norms may influence the way that males and females regulate their specific emotions using reappraisal and distraction. There are many factors which influence the way people regulate their emotions. However, one of the factors which may contribute to gender differences in ER is the way that males and females are socialised to understand emotional behaviour.

When regulating emotions, people are motivated to avoid 'gender inappropriate' expressions of emotion (Timmers et al., 1998). Fear and sadness are often labelled as 'feminine' emotions, whereas anger is regarded as a 'masculine' emotion (L. R. Brody, 1985; Root & Denham, 2010), and so adherence to these gender norms may influence the way that males and females regulate their specific emotions. Males may be more motivated to avoid expressing sadness and fear (Timmers et al., 1998), whereas females may be more motivated to avoid expressing anger (Brescoll & Uhlmann, 2008). This is supported with evidence that shows that males are less likely than females to express sadness and fear (Chaplin, 2015; Chaplin & Aldao, 2013), whereas females are less likely to express anger, at least in certain contexts (Archer, 2004).

The motivation to avoid expressing these specific emotions may impact on the ER process, and may contribute to gender differences in the strategies used to regulate these particular emotions. Distraction is often used to regulate more intense emotions, whereas reappraisal is used to regulate lower intensity emotions (Sheppes et al., 2011; Van Bockstaele et al., 2019). As distraction appears earlier in the emotion generative process, this strategy may be used (rather than reappraisal) for emotions that are undesirable to experience or

express, by stopping the emotion before it gathers momentum. Reappraisal involves more engagement with the emotion, and so may be used to regulate more 'acceptable' emotions. Thus, males may be more likely to use distraction to regulate sadness and fear, as they are more highly motivated to avoid these emotions based on societal expectations, whereas females may be more likely to use reappraisal for sadness and fear, as there is less urgency to avoid expressing this emotion. There is less evidence for gender differences in anger (Grossman & Wood, 1993), and so no gender differences are expected in the regulation of anger. The third aim of the present study is to examine if there are emotion-specific gender differences in ER effectiveness.

5.2.2. Research Questions for Study 2a

On the basis of the literature reviewed in previous chapters, the following specific research questions (SRQs) were formulated for Study 2a.

- SRQ1. In line with the wider literature, can gender differences in the reported use of reappraisal, suppression, distraction, and rumination to regulate general emotion be replicated in the present study?
- SRQ2. Does the effectiveness of reappraisal and distraction depend on the specific emotion being regulated?
- SRQ3. Are there gender differences in the effectiveness of regulating sadness, anger, and fear using reappraisal and distraction, and do these gender differences vary according to the specific emotion and ER strategy?

5.3. Method

In Study 2, participants took part in two sessions, which are detailed in Section 5.3.3. Participants firstly completed self-report questionnaires online, and then were invited to
participate in the ERT in person, approximately one week later. The details of these sessions are described in the following sections.

Please note that the data for Study 2 were split into two parts – Study 2a and Study 2b – to examine separate research questions. In Study 2a, data on reappraisal, suppression, rumination, and distraction were collected using self-report measures, and emotion-specific ER effectiveness was examined using the ERT. These measures will be described in this chapter. In Study 2b, self-report measures were used to collect data on depressive symptoms, psychological wellbeing, self-harm behaviours, and suicidal ideation; however, these measures are not discussed here, but are reported in Chapter 6.

5.3.1. Participants

The participants for Study 2 were recruited from three key sources – (1) the undergraduate and postgraduate student communities at the University of Strathclyde in Glasgow, (2) government analysts from the Scottish Government, and (3) the wider community. The reason for government analysts taking part in the study was because the researcher worked at the Scottish Government during the data collection period and so this organisation represented an important source of participants for the present study. Two hundred and sixty-five participants took part in the online session, and of these participants, 77% also participated in the experimental session, leaving a final sample of 203 participants (104 males and 99 females) who completed the study in its entirety. The age of participants ranged from 17 - 72 years old, and the mean age of the sample was 25.60 (*SD* = 9.63).

Of the participants who completed both parts of the study, 82% were students, 15% were government analysts, and 3% were from the wider community. The participants in the study were highly educated, with 57% of the sample having completed university, 6% having completed an intermediate qualification between secondary level and university (such as technical training), and 36% of the sample having completed secondary education (which

reflects the fact that many of the participants were undergraduate students at the time of participation). In terms of ethnic background, 89% of the sample were White, 6% were of Mixed ethnic background, and 5% were Asian, Black, or belonged to an 'Other' ethnic group (these categories were combined due to low numbers in these groups).

The inclusion criteria for Study 2 were that participants (1) must identify as either male or female, and (2) must be able to read and write in English. The exclusion criterion was that participants must not have a clinical diagnosis of depression. A key aim of the present research was to examine gender differences in emotion-specific ER, and so it was important to have well-defined gender categories. Further, participants were required to speak English to ensure that all participants had a clear understanding of the task. Often there are marked differences in ER between individuals with and without a clinical diagnosis of depression (Kanske et al., 2012; Kjærstad et al., 2016). Therefore, in order to control for any effects that a clinical diagnosis of depression would have on the results, and to decrease the heterogeneity in the sample, only individuals without a diagnosis of depression took part in the study.

5.3.2. Materials

5.3.2.1. Session 1: Self-Report Measures. Participants completed the self-report measures online, on the survey platform Qualtrics. The measures included in the present study were the Emotion Regulation Questionnaire (ERQ; Gross & John, 2003) to measure reappraisal and suppression, the Ruminative Response Scale (RRS; Nolen-Hoeksema & Morrow, 1991) to measure rumination, the brief COPE scale (Carver, 1997) to measure distraction, the Centre for Epidemiologic Studies Depression Scale (CES-D; Radloff, 1977) to measure depressive symptoms, the Warwick-Edinburgh Mental Well-Being Scale (WEMWBS; Tennant et al., 2007) to measure wellbeing, the Deliberate Self-Harm Inventory (DSHI; Gratz, 2001) to measure self-harm behaviours, and the Suicidal Ideation Attributes

Scale (SIDAS; van Spijker et al., 2014) to measure suicidal ideation. The questionnaires that were used to measure variables associated with mental health (CES-D, WEMWBS, DSHI, SIDAS) are not discussed here, but are reported in Section 6.3.1.

5.3.2.1.1. Demographic Information. Participants were asked questions to collect data on their demographic information during the online session. Age was measured using the question 'How old are you (Please type your answer into the box)'. Sex and gender were measured using the questions 'What is your sex? (Please type your answer into the box)' and 'What gender do you identify as? (Please type your answer into the box)'. All participants reported their gender as being consistent with their biological sex. Gender was measured on a male/female binary, and this variable was used to examine gender differences in the present research by comparing the data for males and females. Ethnic background was measured based on the guidelines provided by the Office for National Statistics (2021) for measuring ethnic groups in the UK. Educational attainment was measured by asking the question 'What is the highest level of education you have completed? University or College or Equivalent, Intermediate between Secondary Level and University (e.g., Technical Training), Secondary School, Primary School only or less)', as this can provide an indication of the socio-economic distribution of the sample. Finally, to determine the proportion of students in the sample, participants were asked 'Are you currently a student? Yes, No'.

5.3.2.1.2. Measures of Emotion Regulation. Although the main measure of ER was the regulation of specific emotions during the ERT, some standardised questionnaires which measure the habitual regulation of general emotion were included to assess if the present study replicated previous findings in the field. The ERQ, RRS, and brief COPE were used to measure emotion-general ER in the present study.

Emotion Regulation Questionnaire (ERQ; Gross & John, 2003). The ERQ is a 10item scale which measures individual differences in the habitual use of two ER strategies to

regulate general positive and negative emotion – cognitive reappraisal and expressive suppression. The ERQ measures the frequency with which participants use these strategies in their daily life by recording the extent to which participants agree with a number of statements on a 7-point Likert scale which ranges from 1 (strongly disagree) to 7 (strongly agree). An example of an item on the ERQ is 'When I want to feel less negative emotion (such as sadness or anger), I change what I'm thinking about'. The ERQ is a leading measure of ER and has demonstrated robust psychometric properties. The ERQ has high internal consistency, with alphas ranging from $\alpha = .75$ to .82 for the reappraisal scale, and from $\alpha =$.68 – .76 for the suppression scale across samples (Gross & John, 2003). In the present study, the Cronbach's alphas were $\alpha = .84$ for reappraisal and $\alpha = .74$ for suppression, which demonstrates acceptable levels of reliability (DeVellis, 2016). The ERQ also shows strong convergent and discriminant validity (Gross & John, 2003).

Ruminative Response Scale (Nolen-Hoeksema & Morrow, 1991). The RRS is a 22item scale which measures the tendency to use rumination as a way of responding to a depressed mood. The RRS measures the extent to which participants agree with a number of statements describing how people respond when they are feeling sad or depressed. An example of an item from the RRS is 'I won't be able to do my job if I don't snap out of this', and this is measured on a 4-point Likert scale ranging from 1 (almost never) to 4 (almost always).

The RRS has been extensively used in depression research and has led to many insightful findings regarding the relationship between depression and rumination. The RRS has demonstrated high internal consistency, with Cronbach's α ranging from .88 to .92 (Bagby et al., 1999; Just & Alloy, 1997; S. A. Nolan et al., 1998; Nolen-Hoeksema & Davis, 1999; Nolen-Hoeksema et al., 1999; Nolen-Hoeksema & Morrow, 1991; Nolen-Hoeksema et al., 1994), and test-retest reliability (r = .75) across 18 months is high (Nolen-Hoeksema &

Davis, 1999). In the present study, the rumination scale had a Cronbach's alpha of $\alpha = .90$ which demonstrates excellent internal consistency (DeVellis, 2016).

Brief COPE (Carver, 1997). The brief COPE is an abbreviated 28-item version of the COPE inventory (Carver et al., 1989), which measures the different ways that people cope in response to stress, including active coping, denial, and substance use. Participants indicate the extent to which they agree with statements such as 'I've been using alcohol or drugs to make myself feel better' on a 4-point Likert scale which ranges from 1 (I haven't been doing this at all) to 4 (I've been doing this a lot). The subscale of interest in the present study was the distraction subscale, which measures the tendency to use distraction as a coping strategy. The full brief COPE scale was administered to ensure that no response bias was introduced by presenting the distraction subscale alone. The distraction subscale has been found to have good internal consistency (Cronbach's $\alpha = .71$) in previous research (Carver, 1997). However, in the present study, the internal consistency of the distraction subscale was poor, $\alpha = .41$, as generally an alpha of .70 or over is regarded as acceptable (Clark & Watson, 2016; DeVellis, 2016). As the subscale showed poor reliability, it was not included in the analysis.

5.3.2.2. Session 2: Emotion Regulation Task (ERT). ER was measured in the present study using the emotion regulation task (ERT), which is a commonly used method in ER research (e.g., Goldin et al., 2008; Gross, 1998; Gross & Levenson, 1997). The ERT allows for ER effectiveness to be assessed by measuring self-reported levels of emotion during the task. The task was developed for the present study using the E-Studio component of E-Prime application suit, a world-leading software application for developing behavioural experiments.

5.3.2.2.1. *Task Procedure.* In the ERT, participants were shown emotion-eliciting pictures. These pictures were selected with the aim of inducing three specific emotions –

sadness, anger or fear. The procedure for choosing these pictures was described in Section 3.5.2.2.

There were three conditions in the ERT – the no regulation condition, the reappraisal condition, and the distraction condition. In the no regulation condition, participants viewed the emotional picture without any instructions to use ER (i.e., look at the image as you normally would). In the reappraisal condition, participants were instructed to use reappraisal in response to the picture (i.e., change the meaning of the situation in the picture). In the distraction condition, participants were asked to use distraction in response to the picture (i.e., think about something unrelated). Immediately after viewing each picture, the participants were asked to rate their levels of sadness, anger, and fear on three independent 9-point Likert scales. These Likert scales represented the level of subjective experience for each specific emotion, and were used to calculate the ER effectiveness scores, which is described in more detail in Section 5.3.2.2.2 below.

Each picture in the ERT had a target emotion and was chosen with the aim of eliciting that target emotion within the participant. The ERT consisted of 27 trials, of which 9 were sadness trials, 9 were anger trials, and 9 were fear trials. As the experiment used a within-subjects design, each participant took part in all of the conditions of the experiment. The number of trials in the ERT were chosen with the aim of maximising the number of trials for each condition, while balancing this with reducing participant fatigue, and this is similar to the number of trials used in previous studies (e.g., Goldin et al., 2008; Gross, 1998; Gross & Levenson, 1997). Each trial consisted of 5 screens, as depicted in Figure 5.1. The participant was presented with a fixation point to focus their attention onto the screen, which appeared for 3 seconds. The instruction screen then appeared for 5 seconds, followed by the emotional picture, which appeared for 20 seconds.

The next screen allowed the participant to self-report their levels of sadness, anger, and fear. The timing of this screen was participant-controlled, which meant that the participant pressed a button to move on to the next screen when they were ready. A final screen appeared asking the participant to briefly describe the regulation strategy they used during the trial. The purpose of this stage was to gather evidence that the participant made a genuine attempt to implement the regulation strategy during the task. All participants described their ER strategies sufficiently at this stage, which indicates that all of the participants were engaged with the task. Sufficient engagement with the task was demonstrated by a detailed and relevant description of how the strategy was implemented, which all participants provided. The participant then pressed a button to move on when they were ready to begin the next trial. The trials appeared in a randomised order for each participant to avoid any order effects. Although trials appeared in a random order, each instruction (i.e., no regulation, reappraisal, or distraction) appeared with the same picture for **Figure 5.1**

Stages of the Emotion Regulation Task



This represents one trial in the ERT. Each trial was presented in a randomised order.

Time

every participant. In other words, the instructions were tied to the same pictures for all participants. The ERT took around 30 minutes to complete.

5.3.2.2.2. Emotion Regulation Effectiveness. Levels of emotion (i.e., subjective experience) were recorded during the ERT using three independent 9-point Likert scales for sadness, anger, and fear. The Likert scales ranged from 1 (not at all sad/angry/scared) to 9 (extremely sad/angry/scared). This means that a lower score reflects feeling less emotion, and a higher score represents higher levels of emotion. ER effectiveness scores were computed using these self-reported emotion levels, which is a method that has been used with success in several studies (Goldin et al., 2008; Gross, 1998a; Gross & Levenson, 1997; Sheppes & Meiran, 2007; Sullivan & Kahn, 2020).

On trials during which participants were instructed to look at the image as they usually would (no regulation condition), the level of emotion reported after viewing the emotional picture was regarded as the level of emotion participants experience when they are not instructed to use ER. On trials during which participants were instructed to use reappraisal or distraction (the regulation conditions), it is assumed that the level of emotion reflects the use of this ER strategy. Thus, the difference in self-reported levels of emotion between the no regulation and regulation trials was a measure of how well participants used the ER strategies to regulate their emotions, which is known as ER effectiveness in the present study. ER was calculated using the following formula:

Level of emotion in the no regulation condition – level of emotion in the regulation condition = ER effectiveness. This means that a higher score represents higher ER effectiveness (i.e., a larger reduction in emotion).

5.3.2.2.3. *Task Instructions.* The following instructions appeared at the beginning of the experiment:

'Thank you for taking part in this study. In this experiment, you will be shown pictures, and you will be asked to use two different strategies to try to change how you interact with these pictures.

One of these strategies involves changing the meaning of the situation in the picture so that you see it as more positive or neutral. So, for example, if the picture shows an individual who seems upset, you might tell yourself that it will all work out for the best or that it's not a big deal.

The other strategy involves thinking of something else in order to divert your attention away from whatever is happening in the picture. So, if something upsetting is happening in the picture, you will try to think about something different. At the end of each trial, you will be asked to report how sad, angry and scared you feel on a 9-point scale. It is important that you report how you actually feel, not how you think you should feel. Please try to give an honest account of how you feel during the experiment. Please let the researcher know if you have any questions before continuing.'

The specific instructions that were presented before each trial were as follows. For the no regulation trials, participants were instructed to 'Please look at the picture as you normally would'. For the reappraisal trials, participants were instructed to 'Please change the meaning of the situation in the picture'. For the distraction trials, participants were instructed to 'Please think of something unrelated to what is happening in the picture'.

5.3.2.2.4. *Emotional Stimuli.* Using images as emotional stimuli has been shown to be the most effective elicitor of discrete emotions (Lench et al., 2011). As described in Section 3.5.2.2, emotional images were carefully chosen based on normative values of emotion levels elicited by the pictures (Mikels et al., 2005; Riegel et al., 2016) with the aim of inducing the specific emotions of sadness, anger, and fear. The following images from the

IAPS (Lang et al., 1993) were used in the ERT: 6313, 6360, 6550, 6540, and the following images from the NAPS (Marchewka et al., 2014) were used in the ERT: People_127_h, People_075_v, Faces_032_h, Landscapes_139_h, Landscapes_026_h, Faces_031_v, People_226_h, People_040_h, Faces_283_h, People_003_h, People_122_h, Animals_053_h, Animals_013_h, Animals_060_h, Animals_076_v, Animals_011_h, Animals_030_h, Animals_006_v, Animals_007_h, Animals_004_v, Animals_055_h, Animals_022_h, Animals_144_h. The mean self-reported levels of each specific emotion induced by these images during the task are reported in Section 5.4.4.1.

5.3.3. Procedure

Ethical approval for this study was obtained from the University of Strathclyde Ethics Committee (reference number: UEC17/58), and all participants provided informed consent before taking part. There were two parts to the present study – an online session during which self-report data on ER and variables associated with mental health were collected, and an experimental session which took place in person, during which participants completed the ERT. The data were collected confidentially, and participant information was kept physically under lock and key, and digitally on an encrypted drive. The procedure for Study 2 is summarised in Figure 5.2.

5.3.3.1. Session 1: Online. The online session took place on the survey platform Qualtrics. Participants firstly read an information sheet (Appendix I) before deciding whether or not to take part. All participants provided informed consent by signing a consent form (Appendix J). If participants wished to take part, they provided information regarding their age, sex, gender, ethnic background, educational background, and student status (Appendix K), and then completed self-report measures of ER (Appendix L), variables associated with mental health (Appendix M), and a measure of gender roles and social desirability (which

Figure 5.2



were not included in the final analysis) (Appendix N). The measures were presented in two blocks, which were counterbalanced across the sample to reduce any bias that could be introduced into the data due to participant fatigue.

The measures in block A were presented in the following order – ERQ, RRS, brief COPE, and for block B – WEMWBS, CES-D, DSHI, SIDAS, and MCSDS, and the measures were always presented in the same order within each block. The gender roles and social desirability questions were administered after the self-harm and suicidal ideation scales, as these have a neutral focus and may help to distract from any negative mood induced when answering the self-harm and suicidal ideation questions. On completion of the measures, the participant was invited to provide their name and email address if they wished to take part in the next stage of the study, and participants were thanked for their time.

5.3.3.2. Session 2: Experiment. The second part of the study took place approximately one week later. On arrival at the laboratory, participants received an information sheet which contained details about what was entailed during the ERT. If the

participant wished to proceed, they were asked to sign a consent form. The participant then completed the ERT, which is described in Section 5.3.2.2. Following the task, participants completed a Velten (1968) positive mood induction procedure (Appendix O), which involved reading positively worded sentences. The aim of this procedure was to reverse any negative affect induced by the ERT and it has been shown to be effective in increasing positive affect (Teasdale & Russell, 1983). Following this, participants were provided with a debriefing sheet (Appendix P), which contained additional details about the study, contact details for the investigators, and sources of support if required.

5.3.4. Analysis

5.3.4.1. Approach for Testing Gender Differences in Emotion-General ER

(**SRQ1**). In the present study, it was hypothesised that there would be gender differences in the self-reporting of rumination, suppression, and distraction, and there would be no gender differences in reappraisal. However, as distraction did not demonstrate an adequate level of internal consistency (reported in Section 5.3.2.1), it was excluded from the analysis. Gender differences in the reporting of emotion-general reappraisal, suppression and rumination were tested using independent *t*-tests, which is reported in Section 5.4.3. Independent *t*-tests were chosen because these demonstrate if there is a significant difference between the mean levels of strategy use between males and females, and so is a good indicator of gender differences in ER.

5.3.4.2. Approach for Testing Whether the Effectiveness of ER Depends on the Specific Emotion Being Regulated (SRQ2). In the ERT, self-reported levels of sadness, anger, and fear were measured in a no regulation condition, a reappraisal condition, and a distraction condition. ER effectiveness scores were computed as the difference between self-reported emotion in the no regulation condition, and the regulation conditions, which

represent the ability to reduce emotions using these ER strategies. Three 2 x 2 mixed ANOVAs were used to test for gender differences in ER effectiveness, which are reported in Section 5.4.5 below. Each ANOVA was used to test a specific emotion – sadness, anger, or fear. The independent variables were Gender (Male, Female), and ER Strategy (Reappraisal, Distraction), the dependent variable was the change in emotion (sadness/anger/fear), or ER effectiveness. The effectiveness of reappraisal and distraction were tested by examining the main effects of the ANOVAs.

5.3.4.3. Approach for Testing. Gender Differences in the Regulation of Sadness, Anger, and Fear (SRQ3).

Gender differences would be demonstrated by finding a main effect of gender (which shows that one group have a larger change in emotion, regardless of the strategy used), or a gender by ER strategy interaction (which demonstrates that one group were more effective at reducing emotions using a certain strategy). If a significant interaction is found, this will be followed up with simple main effects (adjusted for multiple comparisons) to examine where the significant differences lie.

To explore whether these gender differences vary across specific emotions, the results across all the ANOVAs must be examined, because each ANOVA is conducted on a specific emotion. It is expected that the direction of the results may differ across each of the emotions, or some findings may be significant for one emotion but not for others. For example, females may be more effective than males at reducing sadness using reappraisal, but this gender difference may not be found for anger.

5.4. Results

5.4.1. Data Screening and Assumptions

5.4.1.1. Missing Data. Two cases (less than 1% of the total sample) were missing emotional data due to computer error in the ERT during data collection. Listwise deletion was used for cases with missing data, which involves excluding cases with any missing data from the analysis, and results in all statistics being computed from the same complete sample of data. This is the standard method used in ANOVA and is appropriate when there are no patterns in the missing data and removing cases results in no (or little) loss of power (Sweet & Grace-Martin, 2011), which was the case in the present study.

5.4.1.2. Outliers. When performing an analysis on grouped data, such as ANOVA, outliers are identified separately within each group (Tabachnick & Fidell, 2014), and so males and females were screened independently for outliers. The data were tested for univariate outliers (i.e., cases which have unusually high scores on one variable) using standardised z-scores, with z-scores over 3.29 being regarded as an outlier (Tabachnick & Fidell, 2014). Six univariate outliers were found in total. One univariate outlier was found for change in sadness using distraction (male), and there was one univariate outlier for change in anger using reappraisal (male). Two outliers were found for the change in fear using distraction (one male, one female). Two outliers were also found for the change in fear using reappraisal (both female), and one for the change in fear using distraction (female). All the univariate outliers, except the outliers for the change in fear using distraction, were caused by an unusually large reduction in emotion. The outlier for the change in fear using distraction was caused by an unusually large increase in emotion.

The data were also checked for multivariate outliers (i.e., cases which have an unusual pattern across the repeated measures variables) using Mahalanobis distance, Cook's values and Leverage values. The ER effectiveness variables (sadness reappraisal effectiveness,

sadness distraction effectiveness, anger reappraisal effectiveness, anger distraction effectiveness, fear reappraisal effectiveness, fear distraction effectiveness) were tested for outliers. Any cases which had unusual values for two out of the three of these criteria (Mahalanobis distance, Cook's values or Leverage values) were regarded as a multivariate outlier and were omitted from the analysis (Tabachnick & Fidell, 2014). Four multivariate outliers (all female) were found for the sadness variables. For the anger variables, 5 multivariate outliers were identified (2 male, 3 female), and 9 were found in the fear variables (2 male, 7 female). The removal of outliers did not impact on the significance of the results.

5.4.1.3. Normality and Linearity. The normality and linearity of residuals were assessed through visual inspection of normal PP-plots and a histogram of standardised residuals. The residuals for the sadness data were normally distributed. However, there was evidence of slight positive skew in the anger and fear data. Some positive skew is to be expected due to the nature of psychological variables, as there is often a pile up of lower scores (Cain et al., 2017). In terms of linearly, inspection of the plots showed that each of the variables appeared to be linearly related.

Non-normal data are only problematic when there is severe skewness and the sample size is small (Hayes, 2017), which was not the case in this study. Three options for analysis were explored – (1) using a standard parametric ANOVA, (2) using a robust ANOVA, and (3) transforming the data using square root transformation. All three approaches produced similar results, with no difference in the significance or direction of the results between these approaches. As the robust ANOVA results in loss of cases, and the transformation of the data impacts the interpretation of the findings, the results from the standard ANOVA are reported.

5.4.1.4. Homoscedasticity. An assumption of ANOVA is that the error variances (i.e., residuals) in each group should be equal, which is known as homoscedasticity (Field, 2017). Homoscedasticity was tested in two ways: (1) by plotting the standardised residuals

against the standardised predicted values for each model, and (2) by conducting Levene's test of equality of error variances. Visual inspection of the scatterplots suggested no evidence of heteroscedasticity in the anger or fear data. Further, Levene's test was non-significant for the anger and fear variables.

There was some evidence of heteroscedasticity in the sadness data, with Levene's test being significant for the change in sadness using reappraisal (p = .01) and the change in sadness using distraction (p = .01). However, ANOVA is reasonably robust to violations of this assumption, provided that group sizes are approximately equal (Stevens, 2012), which is the case in the present study.

5.4.1.5. Multicollinearity. Multicollinearity refers to perfect or near-perfect correlation between variables (e.g., r = .99), which can prevent multivariate analyses from running correctly (Tabachnick & Fidell, 2014). A correlation analysis between each of the within-subjects variables was conducted to test for multicollinearity for each ANOVA model. None of the variables exhibited perfect (or near-perfect) correlation, with the highest correlation between variables being r = .67, and so there was no evidence of multicollinearity in the data.

5.4.1.6. Power. Statistical power is the ability of a test to find an effect if one exists (Field, 2017). An a priori power analysis was conducted using the software G*Power Version 3.1.9.2 (Faul et al., 2007) to determine the sample size required for the ANOVA to be adequately powered. This analysis found that a sample size of n = 70 was required for the ANOVA to find a small to medium effect size at 0.80 power. Therefore, this analysis is sufficiently powered to find an effect if one exists.

5.4.2. Sample Characteristics: Comparing the Student Participants to the Rest of the Sample

As noted in Section 5.3.1, the sample in the present study mostly consisted of students (82%) as well as government analysts (15%) and members of the community (3%). Analyses were run on the key study variables to determine if there were differences between the student participants and the rest of the sample (i.e., the government analysts and community members). As shown in Table 5.1, the student group were younger and had a lower percentage of females than the rest of the sample. There were no differences between the students and the rest of the sample relating to ER effectiveness, emotion-general reappraisal, or suppression. However, students had a higher level of rumination than the rest of the sample. Similar analyses were run on the variables associated with mental health for Study 2b and these results are reported in Section 6.4.2.5.

5.4.3. Emotion-General ER: Gender Differences in ER as Measured by Self-Report Questionnaires

Habitual, emotion-general ER was measured using standardised measures, to assess if the sample demonstrated gender differences in ER which reflect patterns found in the wider literature. Gender differences in the use of reappraisal, suppression, and rumination were tested using independent *t*-tests.

Table 5.1

T-Tests and Chi-Square Analyses Comparing Student Scores on Key Variables With Rest of the Sample

	Students		Rest of	sample		
Variable	Μ	SD	Μ	SD	t	р
Age	23.45	8.31	35.27	9.27	-7.14	<.001
Reappraisal Sadness	04	.89	.16	1.01	-1.24	.22
Distraction Sadness	.52	.95	.77	.97	-1.44	.15
Reappraisal Anger	1.00	.94	1.10	1.06	57	.57
Distraction Anger	1.26	1.01	1.22	.96	.18	.85
Reappraisal Fear	.21	.77	.20	.72	.07	.95
Distraction Fear	.09	.76	.05	.84	.23	.82
Reappraisal (emotion-general)	4.64	1.24	4.90	1.07	-1.16	.25
Suppression (emotion-general)	3.74	1.32	3.42	1.24	1.34	.18
Rumination (emotion-general)	47.27	12.09	42.59	10.03	2.19	.03
Social desirability	6.78	2.72	6.68	2.86	.22	.42
	% M	% F	% M	% F	χ^2	р
Gender	54.82	45.18	35.14	64.86	4.69	.03

Note. Significant values at p < .05 are shown in bold. All tests are two-tailed. *t* refers to the *t*-value, χ^2 is the chi-square value, and *p* is the *p*-value, % M is the percentage of Males, % F is the percentage of Females.

As shown in Table 5.2, there were no gender differences in emotion-general

reappraisal or rumination. However, there were gender differences in the use of suppression,

with males reporting significantly higher levels of suppression than females.

Table 5.2

Means (M) And Standard Deviations (SD) Of Habitual ER Reported by Males and Females

	Ma	ales	Fem	nales	To	otal	
ER Strategy	M	SD	М	SD	М	SD	<i>p</i> value
Reappraisal	4.53	1.24	4.86	1.17	4.69	1.21	.06
Suppression	4.14	1.20	3.19	1.25	3.68	1.31	<.001
Rumination	46.73	12.25	46.08	11.48	46.41	11.86	.70

Note. Significant differences shown in bold.

5.4.4. Emotion Regulation Task (ERT)

5.4.4.1. Emotion Manipulation Check in the ERT. A challenge in all emotion research is successfully inducing the intended emotion during an experiment. To determine if the emotion manipulation in the ERT was successful, the mean emotion scores in the no regulation condition were observed for each target emotion. As shown in Table 5.3, each target emotion elicited the corresponding self-reported emotion more highly than the other two emotions, which demonstrates that the emotional pictures induced the target emotion as intended. One-way analyses of variance confirmed that the emotional stimuli induced the target emotion more highly than the other emotions.

It should be noted that each target emotion also induced the other emotions at a lower level, which indicates that emotional pictures tend to produce mixed emotions in an experimental setting. However, only the self-reported levels of emotion consistent with the target emotion were included in the main analyses.

Table 5.3

	Level of	f Sadness	Level o	of Anger	Level	of Fear	-
Target emotion	M	SD	M	SD	M	SD	p
Sadness	4.98	2.21	3.01	2.59	1.10	1.82	<.001
Anger	3.44	2.45	4.69	2.45	1.14	1.86	<.001
Fear	.62	1.29	.49	1.18	1.94	2.28	<=.002

Means (M) And Standard Deviations (SD) Of Self-Reported Levels of Emotion During the No Regulation Condition of the ERT

Note. The highest mean values appear in bold.

5.4.4.2. Descriptive Statistics: Self-Reported Levels of Emotion During the ERT

and Computed ER Effectiveness Scores. The mean levels of emotion reported by males and females in each condition of the ERT are shown in Table 5.4. Independent *t*-tests were conducted to test for gender differences in levels of sadness, anger, and fear in the no regulation condition. These *t*-tests indicated that females had higher self-reported levels of

Table 5.4

	Ma	les	Fem	ales
_	М	SD	М	SD
Sadness – no regulation	2.80	1.35	3.24	1.37
Sadness - reappraisal	2.89	1.35	3.15	1.50
Sadness - distraction	2.49	1.33	2.41	1.32
Anger – no regulation	2.61	1.42	2.87	1.36
Anger - reappraisal	1.61	1.31	1.82	1.43
Anger - distraction	1.58	1.23	1.38	1.10
Fear – no regulation	1.20	1.11	1.60	1.40
Fear - reappraisal	1.14	1.15	1.24	1.30
Fear – distraction	1.31	1.28	1.32	1.27

Means (M) And Standard Deviations (SD) Of Self-Reported Emotion Levels in the ERT for Males and Females

sadness, t(199) = -2.27, p = .02, and fear, t(183.35) = -2.21, p = .03, than males in the no regulation condition. There were no gender differences in self-reported anger in the no regulation condition, t(199) = -1.32, p = .19.

For each ER strategy, *t*-tests were conducted to determine if the strategy reduced emotion levels as expected during the ERT. The results are shown in Table 5.5. As expected, distraction reduced sadness and anger, and reappraisal reduced anger and fear. However, reappraisal did not reduce sadness, and distraction did not reduce fear during the task.

ER effectiveness variables were computed from these scores, by subtracting the mean level of emotion in the regulation (reappraisal or distraction) condition from the mean level of emotion from the no regulation condition. This means that a higher score indicates higher ER effectiveness (i.e., a greater reduction in emotion). This resulted in six ER effectiveness variables – reappraisal effectiveness for sadness, distraction effectiveness for sadness, reappraisal effectiveness for anger, distraction effectiveness for anger, reappraisal effectiveness for fear, and distraction effectiveness for fear. The means and standard deviations of these variables are shown in Table 5.6. These variables were included in the main analysis.

Table 5.5

Results of T-Tests Showing Difference in Emotion Levels Between No Regulation Condition and Regulation Condition

	No regulation		Reapp	Reappraisal		ction
Condition	M	SD	М	SD	M	SD
Sadness	3.01	1.37	3.02	1.43	2.45	1.32
Anger	2.73	1.39	1.71	1.37	1.48	1.17
Fear	1.39	1.27	1.19	1.22	1.31	1.27
	<i>t</i> -test r	esult				
Comparison	t	р	Result			
Sad – no reg v dist	8.35	<.001	Distractio	on reduced	l sadness.	
Sad – no reg v reap	08	.94	Reapprais	al did not i	reduce sad	ness.
Anger – no reg v dist	17.80	<.001	Distractio	on reduced	l anger.	
Anger – no reg v reap	15.03	<.001	Reapprai	sal reduce	ed anger.	
Fear – no reg v dist	1.47	.14	Distraction	n did not r	educe fear	
Fear – no reg v reap	3.87	<.001	Reapprai	sal reduce	ed fear.	

Note. 'No reg' refers to the no regulation condition, 'dist' refers to the distraction condition, and 'reap' refers to the reappraisal condition in the ERT. Significant values appear in bold.

Table 5.6

Means (M) And Standard Deviations (SD) Of ER Effectiveness Variables

	Males		Females		Total	
	M	SD	М	SD	М	SD
Sadness reappraisal	09	.80	.09	1.02	.00	.91
Sadness distraction	.31	.81	.83	1.03	.56	.96
Anger reappraisal	1.00	.95	1.04	.98	1.02	.96
Anger distraction	1.03	.97	1.49	.97	1.25	1.00
Fear reappraisal	.07	.66	.36	.83	.21	.76
Fear distraction	11	.65	.28	.84	.08	.77

Note. Positive values represent a reduction in emotion (i.e., more effective ER), and negative values represent an increase in emotion after implementing the regulation strategy (i.e., less effective ER).

5.4.5. Testing Gender Differences in the Regulation of Sadness, Anger, and Fear Using ANOVA

5.4.5.1. Model 1: Testing Gender Differences in the Regulation of Sadness. A 2 x

2 ANOVA was used to test for gender differences in the regulation of sadness using distraction and reappraisal. The IVs were Gender (Males, Females; between subjects) and ER Strategy (Reappraisal, Distraction; within subjects). The DV was ER effectiveness for sadness. The mean ER effectiveness values are shown in Table 5.6, and the results are depicted in Figure 5.3.

The main effect of ER strategy was significant, F(1,195) = 121.65, p < .001, $\eta_p^2 = .38$. Distraction had a larger reduction in sadness than reappraisal. In fact, reappraisal did not show a reduction in sadness during the ERT. The main effect of gender was also significant,

Figure 5.3

Bar Chart Showing ER Effectiveness for Males and Females, Using Distraction and Reappraisal to Regulate Sadness During the ERT



F(1,195) = 7.94, p = .01, $\eta_p^2 = .04$., with females having a larger reduction in sadness than males.

There was a statistically significant interaction between gender and ER strategy on the change in sadness, F(1, 195) = 8.87, p = .003, $\eta_p^2 = .04$. Simple main effects were run to determine the nature of this interaction, which involved testing for gender differences at each level of ER strategy (reappraisal and distraction). There were no gender differences in the change in sadness using reappraisal, F(1, 195) = 2.04, p = .16, $\eta_p^2 = .01$. However, compared to males, females had a larger reduction in sadness using distraction, F(1, 195) = 14.34, p < .001, $\eta_p^2 = .07$.

5.4.5.2. Model 2: Testing Gender Differences in the Regulation of Anger. In this 2 x 2 ANOVA, the IVs were Gender (Male, Female; Between subjects), and ER Strategy (Reappraisal, Distraction; Within subjects), and the DV was change in anger. The results are shown in Figure 5.4.

Figure 5.4

Bar Chart Showing ER Effectiveness for Males and Females, Using Distraction and Reappraisal to Regulate Anger During the ERT



The main effect of ER strategy was significant, F(1, 194) = 16.84, p < .001, $\eta_p^2 = .08$. Both reappraisal and distraction reduced anger during the task, but there was a larger reduction in anger from distraction compared to reappraisal. The main effect of gender was also significant, F(1, 194) = 5.83, p = .02, $\eta_p^2 = .03$, with females having a larger reduction in anger than males during the task.

The interaction between gender and ER strategy on the change in anger was significant, F(1, 194) = 10.74, p = .001, $\eta_p^2 = .05$. The simple main effect for ER strategy showed that there were no gender differences in the change in anger using reappraisal, F(1, 194) = .75, p = .39, $\eta_p^2 = .004$. However, females had a larger reduction in anger using distraction than males, F(1, 194) = 12.17, p = .001, $\eta_p^2 = .06$.

5.4.5.3. Model 3: Testing Gender Differences in the Regulation of Fear. The third 2 x 2 ANOVA model tested for gender differences in the regulation of fear using reappraisal and distraction. The IVs were Gender and ER Strategy, and the DV was the change in fear levels. The results are depicted in Figure 5.5.

The main effect of ER strategy was significant, F(1, 190) = 7.03, p = .01, $\eta_p^2 = .04$. Reappraisal reduced fear, but distraction did not. The main effect of gender was also significant, F(1, 190) = 11.35, p = .001, $\eta_p^2 = .06$, with females having a larger reduction in fear than males using both reappraisal and distraction. The gender by ER strategy interaction was non-significant, F(1, 190) = .43, p = .51, $\eta_p^2 = .002$.

Figure 5.5

Bar Chart Showing ER Effectiveness for Males and Females, Using Distraction and Reappraisal to Regulate Fear During the ERT



5.5. Discussion

5.5.1. Results of Study 2a

There was partial support for the hypotheses of Study 2a, although the direction of some of the results were not as expected. There were gender differences in ER effectiveness for sadness, anger, and fear. However, these gender differences depended on the ER strategy, and the specific emotion being regulated. The results are summarised in Table 5.7.

5.5.2. Gender Differences in Emotion-General ER: Comparison to the Wider Literature

Gender differences in ER were examined in an emotion-general manner to test if the results of the present study replicated those found in the wider literature. Males reported

Table 5.7

Evidence for Study 2a

Research Questions	Evidence
SRQ1. In line with the wider literature, can gender differences in the reported use of reappraisal, suppression, distraction, and rumination to regulate general emotion be replicated in the present study?	 Males used suppression more than females. No gender differences were found for reappraisal or rumination. Distraction was not tested due to low reliability.
SRQ2. Does the effectiveness of reappraisal and distraction depend on the specific emotion being regulated?	 Reappraisal was effective at reducing anger and fear, but not sadness. Distraction was effective at reducing sadness and anger, but not fear.
SRQ3. Are there gender differences in the effectiveness of regulating sadness, anger, and fear using reappraisal and distraction, and do these gender differences vary according to the specific emotion and ER strategy?	 Compared to males, females had a larger reduction in sadness and anger using distraction. Compared to males, females had a larger reduction in fear using reappraisal and distraction.

using suppression more than females to regulate their emotions, a result which is consistently reported amongst adults (Spaapen et al., 2014; Zimmermann & Iwanski, 2014), undergraduate students (Gross & John, 2003; Haga et al., 2009; Kwon et al., 2013; Melka et al., 2011), and children and adolescents (Gullone et al., 2010; Gullone & Taffe, 2012; Larsen et al., 2013; Zimmermann & Iwanski, 2014).

In contrast, no gender differences were found in the use of reappraisal or rumination to regulate general emotion. Some studies have failed to find a gender difference in reappraisal (Garnefski et al., 2004; Gross & John, 2003; Haga et al., 2009; Zlomke & Hahn, 2010). A possible reason for this lack of gender difference may be due to ER being measured in an emotion-general manner, as when strategies such as distraction are examined for specific emotions, clearer gender differences emerge (Trives et al., 2016). For this reason, ER was also measured in an emotion-specific manner using the ERT in the present study.

However, a robust finding which is inconsistent with the results of the present study is that females tend to ruminate more than males, and this has been found with adolescents (Gomez-Baya et al., 2016; Hilt et al., 2010; Nolen-Hoeksema, 2001; Rood et al., 2009), community adults (Garnefski et al., 2004; Thomsen et al., 2005; Zimmermann & Iwanski, 2014), and students (Ando' et al., 2020), and so the lack of gender difference in rumination found in Study 2a is not in line with the wider literature.

A potential reason for this finding may be the characteristics of the sample in Study 2. The sample consisted of a high proportion (82%) of students. There is empirical evidence that undergraduate males are at a heightened risk for engaging in rumination (Gladstone & Koenig, 1994; Grant et al., 2002; R. Nolan & Willson, 1994), which would result in a smaller gap between males and females in the use of rumination and may eliminate the gender difference entirely. In the present study, students were more likely to report ruminating compared to the rest of the sample (see Section 5.4.2), and so it is possible that the male students were more prone to using rumination, which resulted in the gender difference in rumination disappearing. Research using the same measure of rumination as the present study, the Ruminative Response Scale (RRS; Nolen-Hoeksema & Morrow, 1991) has also found no gender differences in rumination amongst undergraduates (Roelofs et al., 2006).

However, other studies have found that amongst undergraduates, females are more likely than males to ruminate (Ando' et al., 2020). The reason for this discrepancy may be due to the measure of rumination that was used. The measure used in the present study, the RRS, asks about rumination within the context of depression. Rumination is associated with depression (Kovács et al., 2020; Liu et al., 2020), and rumination has been found to mediate

gender differences in depression (Jose & Brown, 2008; Nolen-Hoeksema et al., 1999; Thayer et al., 2003).

The studies which have found gender differences in rumination amongst undergraduate students, such as Ando' et al. (2020), tend to use measures which ask about rumination in general, not within the context of depression, such as the Rumination and Reflection Questionnaire (RRQ; Trapnell & Campbell, 1999). On the other hand, the studies which do not find gender differences, such as Roelofs et al. (2006) and the present study, used measures which ask about rumination within the context of depression or sadness. It is possible that these measures are also tapping into levels of depressive symptoms. Therefore, it may be the case that student males are actually more depressed than males in the wider community, and this eliminates both the gender difference in depression, and the gender difference in rumination. Depressive symptoms were measured in the second part of Study 2 and are reported in Section 6.4.2.

It should also be noted that reporting lower rumination was associated with higher social desirability, and so it is possible that individuals may have been less likely to report using rumination if they were concerned about presenting themselves in a favourable light. In Study 1, participants largely reported their use of adaptive strategies, and so this is in line with the idea that people may be less inclined to report using maladaptive strategies if they are concerned about self-presentation. However, there was no difference in social desirability between the students and the rest of the sample, which means that social desirability decreasing the likelihood of reporting rumination was an issue across the whole sample, not just for students.

5.5.3. Comparing the Effectiveness of Reappraisal and Distraction to Regulate Specific Emotions

An important finding that emerged from the present research is that the effectiveness of reappraisal and distraction depended on the specific emotion being regulated. It was hypothesised that distraction would be more effective for anger and fear and reappraisal would be more effective for sadness, as previous research has found that distraction is used for more intense emotions and reappraisal is used for lower intensity emotions (Sheppes et al., 2011), and anger and fear are often thought of as more intense emotions than sadness (Lench et al., 2016).

Reappraisal emerged as a more effective strategy than distraction for reducing fear. However, distraction was more effective than reappraisal for reducing sadness and anger (although this also interacted with gender, with females being more effective than males at reducing sadness and anger using distraction). Therefore, these results were not entirely in the expected direction. However, this demonstrates that the effectiveness of an ER strategy depends on the specific emotion being regulated. In other words, how effective a strategy is at reducing an emotion depends on the emotional context.

Most of the previous research has examined the effectiveness of reappraisal and distraction in an emotion-general manner. These previous studies have found that reappraisal and distraction are both effective at reducing negative emotion (Gross, 1998a; Hermann et al., 2017; Quinn & Joormann, 2020; Smoski et al., 2014; Thiruchselvam et al., 2011), although distraction may be more effective than reappraisal at reducing general negative emotion (Smoski et al., 2014).

Reappraisal effectively reduced anger and fear, but not sadness in the present study. Research has shown that reappraisal is only effective at reducing sadness when it is implemented early in the emotion-generative process (Sheppes & Meiran, 2007). Consequently, the reason for reappraisal failing to successfully reduce sadness in the present study may be that participants only used reappraisal when the emotional response had fully developed during the ERT, which may have reduced its effectiveness.

Distraction effectively reduced sadness and anger, but not fear. Distraction is helpful when used with high intensity emotions as it diverts attention away from the emotional stimulus before the emotional response has had the chance to fully develop (Sheppes et al., 2011). Research has shown that individuals have an attentional bias towards fearful stimuli (Van Bockstaele et al., 2014), particularly images of animals (Lipp & Derakshan, 2005; Mogg & Bradley, 2006), which were used in the present study. It may be the case that the fearful images used in the ERT grabbed the participants' attention, and thus made it more difficult to divert attention away from the images. This would result in distraction being a less effective strategy for reducing fear. Therefore, the present study is the first to demonstrate that the effectiveness of reappraisal and distraction depends on the specific emotion being regulated, with distraction being more effective for sadness and anger, and reappraisal being more effective for fear. This shows that the emotional context within which an ER strategy is used has an impact on its effectiveness.

The findings of the present study have important implications for the ER literature. According to Gross's (1998b) Process model of ER, distraction is a more effective strategy than reappraisal at reducing negative emotion because it appears earlier in the timeline of an unfolding emotional response. However, the process model does not take into account that the effectiveness of an ER strategy may differ according to the emotion being regulated. This highlights an important avenue for future development of this theory, to take into account emotion-specific differences that may occur in the process model.

The process model has the potential to provide a theoretical framework for emotionspecific ER, although no theorists have discussed the model in this way so far. This

approach allows for specific emotions to be regulated as part of separate process models. As outlined earlier in Section 1.2.2.3, appraisals are theorised to cause emotion differentiation (Scherer & Moors, 2019), which means that different appraisals give rise to specific emotions. An individual may have an appraisal that produces the feeling of sadness, and this can be regulated using a specific strategy such as reappraisal in one process model. On the other hand, a different appraisal of the same situation could elicit feelings of anger, which could be regulated using suppression in a different process model. This also explains the phenomenon of mixed emotions during an emotionally evocative situation (Roseman & Evdokas, 2004). Therefore, the results of the present research highlight the importance of examining ER in an emotion-specific manner. The process model is a potential framework for understanding and comparing the regulation of specific emotions.

5.5.4. Gender Differences in Reappraisal Effectiveness for Sadness, Anger, and Fear During the ERT

The findings regarding gender differences in reappraisal effectiveness for sadness, anger, and fear will be discussed in this section. It was expected that females would be more effective at using reappraisal to regulate sadness than males, as evidenced by a larger reduction in sadness using reappraisal during the ERT. However, this was not the case, as there were no gender differences in reappraisal effectiveness for sadness. It was also expected that females would be more effective than males at reducing fear using reappraisal. This was found to be the case in the present study, with females having higher reappraisal effectiveness for fear than males (i.e., a larger reduction in fear using reappraisal), and so there was partial support for this hypothesis.

There are very few studies which have examined gender differences in ER using the ERT, as many of the studies which adopt this paradigm focus on one gender to increase

homogeneity in the sample, such as all females (Goldin et al., 2008; Gross & Levenson, 1997; McRae et al., 2010), or all males (Lazarus & Alfert, 1964; Notarius & Levenson, 1979), while others fail to test for gender differences (Boland et al., 2019; Gross, 1998a; Jackson et al., 2000; Richards & Gross, 2000; Sheppes & Meiran, 2007; Sheppes et al., 2011; Sheppes et al., 2014). Similarly, there are no studies to date which have examined the regulation of multiple specific emotions using the ERT.

In terms of general negative emotion, a study by McRae et al. (2008) examined gender differences in reappraisal during the ERT to reduce emotion in response to negative pictures, which is similar to the method used in the present research. They used two measures of ER – self-reported negative emotion (i.e., subjective experience) which was also used in the present study, and activation of neural areas that are associated with ER, using functional magnetic resonance imaging (fMRI). In McRae et al. (2008), it was found that males and females had similar emotional reactivity to the negative images. This means that males and females self-reported similar levels of negative emotion while viewing the negative images during the no regulation condition. This contrasts with the findings of the present study, which found that females reported higher levels of sadness and fear in response to the sadness and fear images (i.e., the images that were chosen to target these emotions), but there were no gender differences in self-reported anger levels in response to the anger images.

This discrepancy in gender differences in emotional reactivity (i.e., self-reported emotion levels in response to the negative images in the no regulation condition) between McRae et al. (2008) and the present study could be due to two reasons. Firstly, it may reflect the fact that McRae et al. (2008) used general negative images with a variety of themes, whereas in the present study, images were carefully chosen to induce the specific emotions of sadness, anger, and fear, by ensuring these images included content that had been shown to induce these specific emotions (Mikels et al., 2005). It may be the case that gender

differences in emotional reactivity do exist when this is examined for specific emotions individually. However, it is also possible that gender differences in the reporting of sadness and fear during the no regulation condition is actually influenced by individuals conforming to gender stereotypes about specific emotions, i.e., that sadness and fear are more 'feminine' emotions, that are more acceptable for females than males to experience and express (Root & Denham, 2010). The cultural messaging around sadness and fear would be less likely to be captured using the general negative emotion scale used by McRae et al. (2008).

In terms of ER, McRae et al. (2008) found no gender differences in the reduction of negative emotion using reappraisal during the task. In other words, there were no gender differences in reappraisal effectiveness. This finding may indicate that males and females have similar skills in regulating their general negative emotion using reappraisal. However, the findings of the present study build on the findings of McRae et al. (2008), and give nuance to these results, by demonstrating that gender differences in ER effectiveness depend on the specific emotion.

In the present study, females were more effective than males at reducing fear using reappraisal, but there were no gender differences in reappraisal effectiveness for sadness or anger. Therefore, it may be the case that McRae et al. (2008) failed to find a gender difference in ER effectiveness during the ERT because this was not measured in an emotion-specific manner (i.e., by carefully choosing stimuli to induce the target emotion, and using separate scales to measure specific emotions). When this is examined in an emotion-specific manner, gender differences emerge.

Further, the finding that females are more effective at reducing fear using reappraisal is consistent with the findings of Study 1. In Study 1, females reported that they were more likely to use reappraisal to regulate fear compared to males, but this gender difference was not found for sadness or anger. The findings of Study 1 and Study 2 together demonstrate that

compared to males, not only do females use reappraisal more frequently when they are experiencing fear, but they are also more effective at doing so.

Although there are no studies examining gender differences in the use of reappraisal in the context of fear, previous research has examined the relationship between gender and fear more generally. Females tend to report experiencing greater fear and have a higher risk of developing anxiety disorders than males (McLean & Anderson, 2009). The finding that females are more effective than males at reducing their fear response may reflect the fact that females have had more experience with this emotion and therefore have had more opportunity to develop skills in reducing fear. Females may also have a more severe threat appraisal than males (Kucharska, 2017; Tamres et al., 2002), which means that females interpret events in a more negative light, which may result in females experiencing a larger fear response (Campbell & Muncer, 2017). Reappraisal involves changing the initial appraisal of an emotional stimulus (Gross, 1998a), and so if females are instructed to use reappraisal during the ERT, this may result in an initial threat appraisal being changed to something more neutral, which would result in females' larger fear response decreasing to comparable levels to males, which was found in the present study.

5.5.5. Gender Differences in Distraction Effectiveness for Sadness, Anger, and Fear During the ERT

The findings regarding gender differences in distraction effectiveness for sadness, anger, and fear will be discussed in this section. It was expected that males would be more effective at using distraction to reduce sadness, as evidenced by a larger reduction in sadness using distraction. A gender difference did emerge, but this was in the opposite direction as predicted, with females having a larger reduction in sadness using distraction (i.e., higher distraction effectiveness for sadness). No gender differences in the regulation of anger were expected, however, it emerged that females were also more effective than males at reducing anger using distraction. Therefore, compared to males, females had higher distraction effectiveness for both sadness and anger.

The findings of the present study, that females are more effective at reducing sadness using distraction, are in contrast with findings in the wider literature that males are more likely to report using distraction when they feel sad. In Trives et al. (2016), undergraduate students aged 18-29 and older adults aged 50-76 were asked to report on their ER when they feel discouraged, sad, or depressed using the Response Styles Questionnaire (RSQ). Males were found to report using distraction more than females when they felt sad (Trives et al., 2016). This has also been found amongst adolescents. In Gomez-Baya et al. (2016), adolescents aged 12-15 were asked to report on their response styles when they feel sad, using the Children's Response Styles Scale (CRRS). Males reported using distraction more than females when they feel sad, and the use of distraction for sadness decreased for females over time (Gomez-Baya et al., 2016).

These findings appear to be in contrast to the findings of the present study. However, it is important to highlight the difference between cognitive distraction and behavioural distraction. In the present study, cognitive distraction was measured, which involves cognitively removing attention from the emotional stimulus by thinking about something unrelated. On the other hand, behavioural distraction involves doing something unrelated to the emotional situation (e.g., going out with friends to distract yourself). In the studies which found that males were more likely to distract, a mix of behavioural and cognitive distraction were measured using the RSQ and the CRSS. In the present study, where females were more effective at reducing sadness using distraction, only cognitive distraction was measured. It may be the case that males are more likely to behaviourally distract, whereas females are

more likely to cognitively distract. This is in line with evidence that females use more cognitive ER strategies than males (Garnefski et al., 2004).

It is also important to draw attention to the difference between ER effectiveness and ER frequency. As discussed in Section 3.3, ER frequency is how often a strategy is used, but it is not necessarily interchangeable with how skilfully a strategy is used. It is possible that males use distraction (particularly behavioural distraction) more often than females, but females are more skilled than males at reducing feelings of sadness using distraction (particularly using cognitive distraction). Further, there is evidence that females generally report experiencing more negative affect in their daily lives compared to males (Thomsen et al., 2005) and experience more negative life events (Nolen-Hoeksema, 2001), and thus may be more practiced at using ER strategies such as distraction to reduce negative emotions.

For anger, a similar pattern was found than that for sadness. Compared to males, females had a larger reduction in anger using distraction, which suggests that females may be more effective than males at reducing anger using distraction. There are very few studies examining gender differences in the use of distraction as an ER strategy for anger. In a study by Rusting and Nolen-Hoeksema (1998), when given the choice during an experiment, females were more likely to choose to distract themselves (rather than ruminate) when they felt angry, but this was not the case for males. This may indicate that females are more likely than males to try to downplay or avoid feelings of anger, which is in line with gender socialisation theories which state that anger is seen as a less acceptable emotion for females to express (Brescoll & Uhlmann, 2008; Root & Denham, 2010). If females are more likely to use distraction to regulate anger in order to avoid this feeling, then this would be consistent with females being more effective at reducing anger using distraction, which was found in the present research.
However, this finding is in contrast with a study by Cox et al. (2000), who measured anger expression styles amongst children and adolescents aged approximately 9-14 years old. No gender differences in anger distraction were found, which may indicate that males and females are equally likely to distract from feelings of anger in childhood. Although no gender differences in distraction emerged, gender differences in other aspects of anger were found in the study by Cox et al. (2000). Females scored higher on anger-in (denying, suppressing, or internalising anger) and anger control (attempts to maintain discipline over one's anger) compared to their male counterparts, and males scored higher on anger-out (outward expression of anger). This demonstrates that even from as young as 9 years old, females are suppressing and trying to maintain control over their anger response. It may be the case that females learn distraction as an ER strategy as they get older, in order to maintain control over anger and avoid anger expression, which is consistent with females being more effective at reducing anger using distraction.

Females were also more effective than males at reducing fear using distraction. A robust finding in the literature is that females report experiencing more fear than males (Coreia et al., 2017; C. P. Li, 2018; McLean & Anderson, 2009; McLenon & Rogers, 2019; Yıldırım et al., 2021), and this was mirrored in the present study by females reporting higher levels of fear in the no regulation condition compared to males. There is also evidence that females are more likely than males to have an attentional bias towards fearful stimuli (Catuzzi & Beck, 2014; Conway et al., 2007; Goos & Silverman, 2002; McClure, 2000; Tan et al., 2011), which means that females find it more difficult to remove attention from stimuli which is perceived as threatening. In other words, females pay more attention to perceived threats in their environment. In the ERT, the participants were instructed to remove their attention from the emotional stimulus and think of something unrelated. It is possible that females were able to follow this instruction and remove their attention from the emotional

stimulus, which resulted in a reduction in self-reported fear. As females had a large fear response to the fearful images, this meant that females had a larger reduction in fear, i.e., higher ER effectiveness. Therefore, although females have an attentional bias towards fearful stimuli, it is possible that females were also able to successfully remove attention from the emotional stimuli when instructed to during the task, which is in line with their higher ER effectiveness using distraction to reduce fear.

5.5.6. Limitations of this Study

A number of assumptions are made when conducting research with the ERT. It is assumed that a reduction in negative emotion during the ERT represents an effective use of ER (ER effectiveness), a skill which an individual may possess in their daily lives. When computing ER effectiveness scores, the difference between the level of emotion in the 'no regulation' condition and the 'regulation' condition is thought to represent a reduction in emotion using the ER strategy. However, it is possible that participants may not report their emotions accurately, or may have different levels of emotions elicited by the images in different conditions of the task, which would result in the ER effectiveness scores being less accurate. Although this is a potential limitation of using this paradigm, ER effectiveness scores have been used in several studies with much success (Albanese et al., 2019; Douw et al., 2020; Fitzgerald et al., 2019; Goldin et al., 2008; Gross, 1998a; Gross & Levenson, 1997; Jackson et al., 2000; Lazarus & Alfert, 1964; Notarius & Levenson, 1979; Ochsner et al., 2002; Richards & Gross, 2000; Sheppes et al., 2014; Sheppes & Meiran, 2007; Sullivan & Kahn, 2020).

During the ERT, is also assumed that participants are engaged with the task and that they execute the ER strategy as instructed. In the present research, this limitation was addressed by asking participants to briefly describe how they implemented the strategy

during the task. These text responses were reviewed by the researcher and judged to demonstrate sufficient engagement with the task. Nevertheless, it cannot be completely guaranteed that the participant has fully followed the task instructions.

It should also be noted that in the ERT, each emotional picture appeared with the same instructions (no regulation, reappraisal, or distraction) for all participants. Although this has the advantage of ensuring that the task was consistent across all the participants, it also poses the risk that if a certain picture perhaps was not a strong elicitor of a certain emotion, then this may affect the ER effectiveness score as this picture was shown for this condition for all of the participants. For example, if one of the pictures shown in the no regulation condition was not a strong elicitor of sadness, then this may result in the sadness score being low in the no regulation condition which would lead to the ER effectiveness score being low (even if that individual was skilled in that ER strategy). Although this is a possibility using this method, the emotion scores in each condition were averaged across a number of trials (i.e., a number of different images), which reduces the chance of this issue occurring.

Furthermore, this study used a repeated measures design, which means that all participants took part in all conditions of the ERT. The trials were presented in a randomised order for different target emotions, and for reappraisal and distraction. This design poses the risk that there was contamination between the trials, such as participants still experiencing residual emotion from a previous trial. However, it is assumed that during the ERT, the emotional response subsides during the final stage of each trial which involved the participant briefly describing the strategy they used (see Figure 5.1), and during the beginning of the next trial during the fixation point screen. Also, when the emotional image is presented in the next trial, it is expected that a new emotional response will occur. Another approach would be to include a temporal aspect to the experiment, which would involve measuring baseline levels of emotion before the image and regulation instructions are presented. This may be

beneficial as it would potentially allow for gender differences in baseline levels of emotion to be examined. Also, using a between-subjects design in which participants only take part in one condition may reduce this risk of contamination, although this approach would require a much higher sample size, and would increase variance from individual differences.

The present study used cross-sectional data, which means that all the data were collected at one time-point. The ability of participants to regulate their emotions in the short term was examined. This means that no claims can be made about the causal relationships between variables, the impact of ER over time, or habitual ER.

5.5.7. Implications of Findings

The present study is the first to examine gender differences in the effectiveness of regulating multiple specific emotions (sadness, anger, and fear) using distraction and reappraisal. The findings from the present research thus contribute important novel knowledge to the literature. Firstly, it demonstrates that the relationship between gender and ER depends on the specific emotion being regulated. For example, a similar pattern of gender differences in ER emerged for sadness and anger, however the pattern for fear was different. Therefore, this highlights the importance of investigating gender differences in ER in an emotion-specific manner. It cannot be assumed that the findings regarding one specific emotion, or general negative emotion, can be extrapolated to all emotions.

Secondly, some strategies seem to be better suited to regulating certain emotions over others, with the effectiveness of reappraisal and distraction being dependent on the specific emotion. Specifically, distraction was more effective at reducing sadness and anger during the ERT (although this interacted with gender), whereas reappraisal was more effective at reducing fear and anger. The effectiveness of a strategy to reduce negative emotion in the short-term has been associated with other correlates of ER, such as wellbeing, depression,

and bipolar disorder (Kanske et al., 2012; Kjærstad et al., 2016; McRae et al., 2012b). This means that if these strategies are effective at regulating emotions before they become maladaptive responses, then this can have a positive impact on a range of outcomes.

Importantly, the results from this study show that the effectiveness of ER strategies depends on both (1) gender, and (2) the emotion being regulated. This has important implications for therapeutic practices and may implicate reappraisal and distraction in efforts to promote mental health and wellbeing. Specifically, this may help to guide more individualised treatments for mood and anxiety disorders, depending on whether the patient is experiencing more sadness-based problems, such as depression, or more fear-based problems, such as anxiety disorders or phobias. If an individual is presenting with a more sadness-based problem such as depression, or an anger-based problem such as aggressive behaviour, then distraction may be more effective at reducing these symptoms (although this is particularly relevant for females). This includes therapies that feature distraction at the centre of their teachings, such as dialectical behavioural therapy. On the other hand, if people have more fear-based problems such as anxiety or phobias, then therapies which focus on reappraisal techniques may be better suited to treating this, such as cognitive behavioural therapy.

The main consideration that arises from these findings is that therapies for clinical disorders should be personalised and tailored to an individual's specific needs, particularly along the lines of gender, and the specific type of emotional problem that the individual is presenting with. The relationship between gender, emotion-specific ER, and variables associated with mental health will be explored further in Chapter 6.

5.6. Conclusion and Next Steps

To summarise, the aims of Study 2a were (1) to examine if there were gender differences in emotion-general ER, in line with the wider literature, and (2) to test if there are

gender differences in the effectiveness of regulating sadness, anger and fear using reappraisal and distraction.

There was partial support for the hypotheses of the present study. The effectiveness of ER was dependent on the specific emotion being regulated – reappraisal was effective at reducing anger and fear, but not sadness, and distraction was effective at reducing sadness and anger, but not fear. Gender differences also emerged in the regulation of sadness, anger, and fear, although the patterns differed according to the specific emotion being regulated. Females were more effective than males at reducing sadness and anger using distraction. Females were also more effective than males at reducing fear using both distraction and reappraisal. These findings show that the relationship between gender and ER depends on the specific emotion being regulated, and so it is imperative for future research to investigate gender differences in ER in an emotion-specific manner. These findings also have implications for personalising mental health therapies.

The results of the present study demonstrate that certain strategies may be more effective for certain emotions, and females may be more effective at reducing some emotions compared to males. However, ER effectiveness is not necessarily interchangeable with how adaptive an ER strategy is. Effectiveness refers to the ability of a strategy to reduce negative emotion. On the other hand, adaptiveness refers to the relationship an ER strategy has with other variables, such as variables associated with mental health. It is possible that a strategy may be effective for reducing a negative emotion but does not have an adaptive relationship with other variables (e.g., it has a maladaptive relationship with mental health).

As discussed in Chapter 2, there are gender differences in variables associated with mental health, such as depression, self-harm behaviours, and suicidal ideation. ER is also known to play a role in mental health. Some studies have examined whether ER can explain gender differences in variables associated with mental health, however no study to date has

examined this in an emotion-specific manner. In the second part of Study 2, which is reported in Chapter 6, the data from the ERT will be used to examine if the regulation of specific emotions is related to gender differences in depressive symptoms, self-harm behaviours, and suicidal ideation.

KEY POINTS FOR CHAPTER 5

- 1. The aims of Study 2a were (1) to examine if there are gender differences in emotion-general ER, in line with the wider literature, and (2) to test if there are gender differences in the effectiveness of regulating sadness, anger and fear using reappraisal and distraction.
- 2. Emotion-specific findings emerged. Females were more effective than males at reducing sadness and anger using distraction. For fear, females were more effective than males at reducing fear overall, using both reappraisal and distraction.
- 3. Distraction may be more effective for reducing sadness and anger (although this also depends on gender), whereas reappraisal may be a more effective strategy for fear and anger.
- 4. The results highlight the importance of examining ER in an emotion-specific manner, and demonstrate that the effectiveness of ER may depend on gender and the specific emotion being regulated.

QUESTIONS FOR SUBSEQUENT CHAPTERS

1. Is the regulation of sadness, anger, and fear related to gender differences in the variables associated with mental health?

NEXT STEPS

In Study 1 and Study 2a, some emotion-specific gender differences in ER emerged. We know from previous research that there are gender differences in mental health problems, and ER may mediate the relationship between gender and mental health. However, no studies to date have examined these relationships in an emotion-specific manner. In Study 2b, the relationships between gender, emotionspecific ER, and mental health, will be examined. Study 2b will be reported in Chapter 6.

Chapter 6: The Regulation of Sadness, Anger, and Fear as Mediators of Gender Differences in Variables Associated with Mental Health (Study 2b)

6.1. Chapter Summary

Study 2a is reported in Chapter 5. The second part of Study 2, Study 2b, is described in the present chapter. The findings from Study 1 and Study 2a demonstrate that gender differences are often (but not always) emotion-specific. ER has an important impact on mental health, and there are prominent gender differences in variables associated with mental health, so it is possible that the way males and females regulate their emotions may play a role in gender differences in variables associated with mental health. Some researchers have addressed this question within the context of Response Styles Theory, but no studies to date have examined this in an emotion-specific manner. The aims of Study 2b were (1) to test if there are gender differences in depressive symptoms, self-harm behaviours, and suicidal ideation, in line with the wider literature, and (2) to examine if the regulation of specific emotions using reappraisal and distraction are related to gender differences in depressive symptoms, self-harm behaviours, and suicidal ideation.

To address these aims, participants completed self-report questionnaires measuring variables associated with mental health and took part in the Emotion Regulation Task (ERT). In the ERT, the change in self-reported levels of sadness, anger and fear after using reappraisal or distraction was measured (i.e., ER effectiveness). Mediation analyses were conducted to explore the relationships between gender, emotion-specific ER, and variables associated with mental health. Females were more effective at reducing anger using distraction, and this was associated with higher depressive symptoms and lower suicidal ideation, and so the Study 2b hypotheses were partially supported. These findings indicate that emotion-specific ER may be related to gender differences in variables associated with mental health, and this has implications for theory and therapeutic practice.

6.2. Introduction

6.2.1. Rationale for Study 2b

Few studies have examined gender differences in ER in an emotion-specific manner. In Study 1, gender differences in the reporting of some (but not all) ER strategies were found to vary across specific emotions. In Study 2a, females were more effective than males at reducing sadness and anger using distraction, and females were also more effective than males at reducing fear using both distraction and reappraisal, which demonstrates that gender differences in ER are often emotion-specific. An important next step for this research is to examine if these emotion-specific findings also have implications for variables associated with mental health.

As discussed in Chapter 2, ER may have an impact on variables associated with mental health (Aldao et al., 2010; Zahniser & Conley, 2018). Reappraisal and distraction are generally considered to be adaptive strategies which have a positive association with mental health and wellbeing (Huffziger et al., 2009; Zahniser & Conley, 2018). There are also prominent gender differences in variables associated with mental health. Compared to males, females tend to have higher levels of depression, self-harm regardless of intent, and suicidal ideation (Barzilay et al., 2019; H. Chen et al., 2019; Hawton et al., 2012; Knudson et al., 2020), but more males die by suicide (National Records of Scotland, 2021). Thus, it is possible that ER plays a role in these gender differences in variables associated with mental health. In the present research, the relationships between gender, ER, and variables associated with mental health will be examined in an emotion-specific manner.

As a first step of this research, gender differences in depressive symptoms, self-harm behaviours, and suicidal ideation were tested, to examine if the gender differences observed in the wider literature were also found in the sample of the present study. Therefore, the first aim of the present study is to examine if the gender differences in variables associated with mental health found in the wider literature can be replicated in the present study.

Depressive symptoms, self-harm behaviours, and suicidal ideation were chosen as the variables for the present research for two key reasons. First of all, depression and suicide represent two of the biggest global challenges in public health. Almost 280 million people are affected by depression worldwide, and 700,000 people die by suicide each year (World Health Organisation, 2021b, 2021c). These are serious mental health problems that affect a large number of people. We know that ER can have a positive impact on mental health, and there is evidence that ER can improve outcomes for people with depression and who struggle with suicidal behaviour (Cludius et al., 2020; Dryman & Heimberg, 2018). Depressive symptoms were chosen as a measure of depression, and self-harm and suicidal ideation are two of the biggest risk factors for eventual suicide.

The second reason is that there are prominent gender differences in depressive symptoms, suicidal ideation, and self-harm. Females have higher rates of depression, and are more likely to self-harm and attempt suicide (Knudson et al., 2020; Wetherall et al., 2020), although more males die by suicide (National Records of Scotland, 2021). There is evidence that ER contributes to these mental health variables (Cludius et al., 2020; Dryman & Heimberg, 2018). Therefore, examining gender differences in ER represents an opportunity to better understand the variables which may be associated with depression, suicidal ideation and self-harm.

As discussed in Section 2.3.4, some research has examined ER as a mediator of gender differences in variables associated with mental health, within the context of Response Styles Theory (RST; Nolen-Hoeksema, 1987). These studies have found that rumination explains some of the gender differences found in depression (L. D. Butler & Nolen-Hoeksema, 1994; Gomez-Baya et al., 2016; Hilt et al., 2010; Nolen-Hoeksema, 1991; Nolen-

Hoeksema et al., 1999; Nolen-Hoeksema et al., 1993; Nolen-Hoeksema, 1995; Trives et al., 2016). Therefore, this research demonstrates that there is a relationship between gender, ER, and mental health, and this may be extended to include other ER strategies (i.e., reappraisal and distraction), and other variables associated with mental health (i.e., depression, self-harm behaviour, and suicidal ideation). ER has been found to predict mental health outcomes prospectively (Brewer et al., 2016; Huffziger et al., 2009; S. L. Johnson et al., 2016; Kelley et al., 2019; Nolen-Hoeksema et al., 1999; Zahniser & Conley, 2018).

In the present research, the relationships between gender, emotion-specific ER, and variables associated with mental health (depressive symptoms, self-harm behaviours, and suicidal ideation) will be investigated in an exploratory manner. This study is cross-sectional, which means that the associations between these variables, rather than causal relationships, are being explored. Although mediation analysis is used to understand these relationships, no claim is being made about the causal direction of these relationships. The present study is exploratory in nature.

No research to date has examined the relationships between gender, emotion-specific ER, and mental health in this manner. The regulation of specific emotions has been found to have a differential relationship with variables associated with mental health (Boland et al., 2019; A. Bradley et al., 2019; Clear et al., 2019; du Pont et al., 2018; Leaberry et al., 2019; Peled & Moretti, 2010). Therefore, it is also possible that the regulation of specific emotions may have a different relationship with gender, and the variables associated with mental health.

In the present study, reappraisal and distraction were focused upon. These two strategies in particular were chosen because (1) evidence regarding gender differences in reappraisal and distraction tend to be mixed, and a reason for this may be that previous research has not investigated this in an emotion-specific manner, (2) they are two of the most

widely research strategies in the literature, (3) they both derive from Gross's process model of ER, (4) they are adaptive strategies which are associated with positive outcomes, and (5) they have clinical relevance in that they are associated with mental health variables.

Reappraisal and distraction are both effective at reducing negative emotion (Gross, 1998a; Quinn & Joormann, 2020; Smoski et al., 2014), and are adaptive strategies which are associated with positive outcomes (Aldao et al., 2010; In et al., 2021; Polanco-Roman et al., 2015), and so it is important to understand if there are gender differences in these strategies, as they can contribute to mental health and wellbeing (Brewer et al., 2016; Huffziger et al., 2009; Zahniser & Conley, 2018). According to the process model, distraction is thought to be more effective than reappraisal (Gross, 1998a). However, the findings of Study 2a show that this depends on the specific emotion being regulated. Distraction emerged as being more effective for regulating sadness and anger (but not fear), and reappraisal was more effective for regulating anger and fear (but not sadness). This means that the effectiveness of these strategies was dependent on the specific emotional context. Consequently, it is important to also examine if gender differences in reappraisal and distraction have a different relationship with variables associated with mental health depending on the specific emotion being regulated, which will be investigated in the present study.

Further, no studies to date have compared these two ER strategies (reappraisal and distraction) across three variables associated with mental health (depressive symptoms, self-harm behaviours, and suicidal ideation). An examination of the relationship between gender, ER, and variables associated with mental health in an emotion-specific manner will contribute to the wider literature by testing if the regulation of specific emotions (1) contributes to gender differences in variables associated with mental health and (2) has a differential impact on gender differences in variables associated with mental health. Both depression and suicide are serious public health issues (World Health Organisation, 2021b,

2021c), and so it is crucial to examine the factors which may contribute to these issues. Therefore, the second aim of the present study is to examine if the regulation of sadness, anger, and fear using reappraisal and distraction mediates the relationship between gender and depressive symptoms, self-harm behaviours, and suicidal ideation.

In terms of measuring the key variables of the present study, self-report questionnaires measuring depressive symptoms are a common method of measuring subclinical depression in ER research (Aldao et al., 2010; Garnefski et al., 2004; Gross & John, 2003; Nolen-Hoeksema & Aldao, 2011), and are correlated with clinical diagnoses (Wang et al., 2017). Furthermore, wellbeing tends to be negatively correlated with depressive symptoms and poor mental health (Tennant et al., 2007). Thus, a measure of psychological wellbeing was included in the present study as an additional tool to gauge the levels of general wellbeing in the sample, and to test if there were gender differences in wellbeing.

Suicide cannot be directly observed using a cross-sectional design, and so suicidal ideation and self-harm behaviours were measured in the present study using self-report measures. Suicidal ideation, or having thoughts about suicide, is an important predictor of future suicide risk (O'Connor & Nock, 2014). Self-harm behaviours are also related to suicide. Individuals often report having thoughts about suicide while engaging in self-harm behaviours (Klonsky, 2011), and self-harm irrespective of intent is a risk factor for future suicide (Andover et al., 2012)

In terms of gender differences, females report engaging in self-harm behaviours more than males, whether this is measured without suicidal intent (Bresin & Schoenleber, 2015; Knudson et al., 2020), or includes suicidal intent (Knudson et al., 2020; Wetherall et al., 2018). However, more males die by suicide (National Records of Scotland, 2021). Therefore, measuring self-harm and suicidal ideation may provide insight into suicide risk, and so it is

important to examine the factors which may contribute to these behaviours, such as emotionspecific ER.

6.2.2. Research Questions for Study 2b

There were two specific research questions (SRQs) for Study 2b:

- SRQ1. Are there gender differences in the reporting of depressive symptoms, selfharm behaviours, and suicidal ideation?
- SRQ2. Is there a relationship between gender, emotion-specific ER effectiveness and variables associated with mental health?

6.3. Method

As study 2 was in two parts, information about the participants (Section 5.3.1), materials (Section 5.3.2), and procedure (5.3.3) can be found in Section 5.3. In this section, only the information that is unique to Study 2b will be reported.

6.3.1. Measures of Variables Associated with Mental Health

The following self-report instruments were used to measure variables associated with mental health – Centre for Epidemiologic Studies Depression Scale (CES-D; Radloff, 1977) to measure depressive symptoms, Warwick-Edinburgh Mental Well-Being Scale (WEMWBS; Tennant et al., 2007) to measure psychological wellbeing, the Deliberate Self-Harm Inventory (DSHI; Gratz, 2001) to measure self-harm behaviours, and the Suicidal Ideation Attributes Scale (SIDAS; van Spijker et al., 2014) to measure suicidal ideation.

6.3.1.1. Centre for Epidemiologic Studies Depression Scale. The CES-D is a 20item scale which measures depressive symptoms in a non-clinical population. Examples of items on the CES-D include 'I did not feel like eating; my appetite was poor' and 'I thought my life had been a failure'. Participants indicate the extent to which they agree with these statements on a 4-point Likert scale ranging from 'rarely or none of the time (less than 1 day)' to 'most or all of the time (5-7 days)'. The CES-D is not a diagnostic measure, which means that it cannot provide a diagnosis of depression and obtaining a high score on this measure is not necessarily indicative of depression. However, the CES-D has been found to be a useful screening measure of depressive symptoms in the general population and has high internal consistency (Cronbach's $\alpha = .85 - .90$) and adequate test-retest reliability (r = .40 and higher; Radloff, 1997). The CES-D had a Cronbach's alpha of $\alpha = .90$ in the present study, which demonstrates excellent internal consistency (DeVellis, 2016).

6.3.1.2. Warwick-Edinburgh Mental Wellbeing Scale. The WEMWBS is a 14-item measure of psychological wellbeing. The WEMWBS measures the extent to which participants agree with positively worded statements, such as 'I've been feeling optimistic about the future' and 'I've been feeling relaxed' on a 5-point Likert scale that ranges from 'none of the time' to 'all of the time'. The scale has been found to have high internal consistency (Cronbach's $\alpha = .89 - .91$) and high test-retest reliability across one week (r = .83; Tennant et al., 2007). The WEMWBS demonstrated excellent internal consistency in the present study, with a Cronbach's alpha of $\alpha = .90$. Wellbeing was not an variable of interest in the present research, but this was included to gauge any gender differences in general mental health within the sample, and also to see if there was a difference in psychological wellbeing between the students and the rest of the sample.

6.3.1.3. Deliberate Self-Harm Inventory. The DSHI is a 17-item measure of self-harm behaviours which enquires about various aspects of self-harm, including frequency, severity, duration, and type of self-harming behaviour. As described in Section 2.3.1, there is an ongoing debate as to the best way to conceptualise self-harm behaviours. Some researchers distinguish between self-harm behaviours without suicidal intent (i.e., NSSI) and

behaviours with suicidal intent (i.e., suicide attempts). However, the evidence base for this distinction is weak (Kapur et al., 2013). In the present research, self-harm refers to self-injury regardless of the motive or extent of suicidal intent, as there are prominent gender differences in self-harm both with (Knudson et al., 2020; Wetherall et al., 2018) and without (Bresin & Schoenleber, 2015; Knudson et al., 2020) suicidal intention.

Although the authors of the DSHI claim that it measures self-harm behaviours without suicidal intent (Gratz, 2001), in actuality it likely measures a mix of intentions. In the DSHI, some questions ask about self-harm behaviours without suicidal intent, e.g. 'Have you ever intentionally (i.e., on purpose) cut your wrist, arms, or other area(s) of your body (without intending to kill yourself)'. Other questions do not specify intent, e.g. 'Have you ever intentionally (i.e., on purpose) done anything else to hurt yourself that was not asked about in this questionnaire? If yes, what did you do to hurt yourself?'. Therefore, it is likely that participants will report self-harm behaviours that had a mix of intentions, which is often the case with self-harm behaviour (Boergers et al., 1998; Hawton et al., 1982; Hawton et al., 2012; Madge et al., 2008).

The DSHI was chosen for the present study because it asks about a range of different self-harm behaviours, such as cutting, burning, biting, and hitting. Some of these behaviours are used more by females (e.g., cutting), and others are used more by males (e.g., burning) (Andover et al., 2010; Idig-Camuroglu & Gölge, 2018; Lundh et al., 2007). Therefore, using this measure is beneficial for the present study as it captures methods used by both males and females and so may reduce any gender bias that could emerge from using a measure of self-harm that focussed on a method that was used more by one gender (e.g., cutting).

In the DSHI, Participants answer 'yes' or 'no' to these questions asking about selfharm behaviour. If the participant answers 'yes', then there are 5 follow-up questions: 'How old were you when you first did this?', 'How many times have you done this?', 'When was the last time you did this?', 'How many years have you been doing this?' and 'Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment'. The DSHI has been found to have high internal consistency (Cronbach's $\alpha = .82$; Gratz, 2001) and adequate test-retest reliability across 2-4 weeks (r = .68; Gratz, 2001). In the present study, the DSHI had an acceptable internal consistency of $\alpha = .71$ (DeVellis, 2016).

6.3.1.4. Suicidal Ideation Attributes Scale. In addition to self-harm behaviours, suicidal ideation was also measured in the present study. The SIDAS is a 5-item scale which measures the extent to which individuals have thoughts about suicide, and this scale has been validated for online use (van Spijker et al, 2014). The SIDAS not only measures the presence of suicidal thoughts, but also assesses the severity of these thoughts. Participants report the extent to which they have had suicidal thoughts over the past month using independent 10-point Likert scales that range from (for example) 'Never' to 'Always'. If the participant answered 'No' to the first question ('In the past month, how often have you had thoughts about suicidal ideation over the past month. The SIDAS has been found to have high internal consistency (Cronbach's $\alpha = .91$; van Spijker et al., 2014) and good convergent validity with other measures of suicidal ideation (r = .82; van Spijker et al., 2014). In the present study, SIDAS had a Cronbach's alpha of $\alpha = .79$, which is an acceptable level of internal consistency (DeVellis, 2016).

6.3.2. Analysis

The hypotheses of the present study were tested using mediation analysis, which uses linear regression. Mediation analysis is a useful tool in psychological research, which can be utilised in a number of ways, including testing correlational relationships between variables, and testing potential causal relationships between variables. As the present study is crosssectional in nature, which means that all of the measurements were taken simultaneously (i.e., there was no temporal ordering of variables), mediation analysis was used to examine correlational relationships between the study variables, not to establish causality. Namely, it was used to test whether gender, emotion-specific ER, and variables associated with mental health are related to one another. Therefore, no claim is being made that any of these variables have a causal impact on any other variables. Mediation analysis was used as a tool to understand the relationships between the variables more deeply. However, the present study can be seen as the first step in testing potential causality between the variables. Specifically, that gender differences in emotion-specific ER may contribute to gender differences in depressive symptoms, self-harm behaviours, and suicidal ideation. However, these results would have to be confirmed in a separate, longitudinal study.

The process outlined by Hayes (2017) was implemented to test for a mediation effect in the present study. This approach is in contrast to the causal steps method described by Baron and Kenny (1986). In the causal steps approach, each component of the model (e.g., the $X \rightarrow M$ path, the $M \rightarrow Y$ path) must be tested separately, using individual null-hypothesis tests, and each test must be significant before it can be concluded that there is a mediation effect. However, according to Hayes (2009), it is only the indirect effect which must be significant for a mediation effect to have occurred. The indirect effect, *ab*, refers to the product of the *a* path ($X \rightarrow M$) and the *b* path ($M \rightarrow Y$). If testing the indirect effect shows that the indirect effect is significantly different from zero, then it can be concluded that a mediation effect has occurred. In other words, X has impacted M, which in turn has impacted Y.

Hayes' approach to mediation has an advantage over the causal steps approach because it relies only on one significance test (which reduces the chance of Type II error). It is also not contingent on the total effect of X on Y being significant. Other methods, such as

the causal steps approach, assert that X must have a significant relationship with Y from the outset (Baron & Kenny, 1986). If it does not, then one cannot continue to test for a mediation effect using this approach. However, it is possible for X to exert a mediation effect on Y through M, even if a relationship between X and Y (i.e., that the total effect is different from zero) cannot be established through hypothesis testing (Hayes, 2017). This is because the total effect of X on Y is partitioned into the indirect effect ($X \rightarrow M \rightarrow Y$), and the direct effect ($X \rightarrow Y$, while controlling for M). It is possible for these paths to have different signs (e.g., one can be positive, and one can be negative), so they can essentially cancel one another out. This would result in the total effect not being significantly different from zero (which is the first criterion that must be fulfilled in the causal steps approach). In other words, the indirect effect (which tells us if there has been a mediation) can be different from zero, even if the total effect is not (Hayes, 2017). In the present study, this would mean that a mediation effect could be tested for even if there were no gender differences in depressive symptoms, suicidal ideation or self-harm behaviour.

Furthermore, when conducting analyses, statistical inference can be tested in two ways. Firstly, a null hypothesis test can be conducted (e.g., the null hypothesis that there is no relationship between two variables) and the *p*-value shows the probability of finding that result by chance. A second method is to construct confidence intervals around an estimate and these intervals provide information about the margin of error of an estimate. If the confidence intervals do not contain zero, which means that it is unlikely the estimate is zero, then the null hypothesis can be rejected (with 95% confidence).

In the present study, the total effect and direct effects were tested using null hypothesis testing of the regression coefficients, using a *p*-value of <.05 as the alpha. The indirect effect (which demonstrates if mediation has occurred) was tested using confidence intervals, as described by Hayes (2017). This method involves generating 95% percentile

bootstrapped confidence intervals around the regression coefficient for the indirect effects (for each potential mediator) and testing the null hypothesis that the indirect effect is equal to zero. If the confidence interval does not include zero this means that the indirect effect is unlikely to be zero (with 95% confidence), and so the null hypothesis can be rejected, and this finding indicates that there is a mediation effect.

Bootstrapped confidence intervals were used rather than other methods such as the Sobel test, because bootstrapped confidence intervals make no assumptions about the sampling distribution of the indirect effect (Hayes, 2017), and simulation studies have found that bootstrapped confidence intervals have more power and are more accurate than other estimation methods (Hayes & Scharkow, 2013; MacKinnon, Lockwood, & Williams, 2004).

6.4. Results

6.4.1. Data Screening and Assumptions

6.4.1.1. Outliers. The data were screened for outliers by calculating Mahalanobis Distance, Cook's Values and Leverage Values for each of the final mediation models (Tabachnick & Fidell, 2014). Any cases with values over the cut-off level was removed from further analysis. Information about the specific number of outliers removed is provided in the section for each mediation model below.

6.4.1.2. Normality and Linearity. An assumption of linear regression is that residuals must be normally distributed. This assumption was tested by inspecting a histogram and probability (P-P) plot of standardised residuals for each model. Visual inspection of these charts indicated some minor issues with normality. That is to say, there was some evidence of slight positive skew for all of the mediation models.

According to Hayes (2017), this assumption is one of the least important in linear regression, and violation of normality is only problematic when residuals are severely skewed

and the sample size is small, which is not the case in the present research. Further, PROCESS is generally robust to violations of normality (Hayes, 2017). Therefore, no further action was taken to correct the normality of residuals.

In linear regression, the relationship between continuous variables must be linear. To test this assumption, the data for each pair of variables were displayed on scatterplots of the predictor variable plotted against the outcome variable. Visual inspection of these scatterplots confirmed that the relationship between each pair of variables was indeed linear, and so this assumption of linear regression was met.

6.4.1.3. Homoscedasticity. An assumption of linear regression is that the residuals (or errors) must be evenly spaced across the regression line, which is known as homoscedasticity (Field, 2017). This assumption was tested by plotting the standardised residuals against the standardised predicted values for each mediation model. Visual inspection of these scatterplots indicated that there was homoscedasticity in the data for most of the models. There was some evidence of heteroscedasticity in the mediation models for suicidal ideation, which has been highlighted in the individual sections for those models (Section 6.4.4.3 and 6.4.4.4). When evidence of heteroscedasticity was found, a heteroscedastic-consistent standard error estimator, HC3, was used in PROCESS to compute a standard error which is robust to violations of this assumption (Hayes, 2017).

6.4.1.4. Multi-collinearity. A correlation analysis between each of the predictor variables was conducted to ensure there was no multi-collinearity in the data. The variables were not highly correlated and so there was no evidence of multi-collinearity in the data.

6.4.1.5. Power. An a priori power analysis was conducted using the software G*Power Version 3.1.9.2 (Faul et al., 2007) to determine the sample size required for the mediation analyses to be adequately powered. This analysis found that a sample size of n =

143 was needed to find a small-medium effect at 0.80 power. As the sample size in the

present study was n = 203, this means that the mediation analyses were adequately powered.

Table 6.1

Correlation Analyses and T-Tests Showing the Relationship Between Age, Gender, and the Variables Associated With Mental Health

	Age		-	essive ptoms	Suicidal ideation	
Variable	r	р	r	р	r	р
Depressive symptoms	19	.01	-	-	-	-
Suicidal ideation	16	.03	.56	.001	-	-
	t	р	t	р	t	р
Gender	98	.33	-	-	-	-
Self-harm behaviour	3.46	<.001	-3.50	<.001	-3.32	.001

Note. r is the Pearson correlation coefficient, p is the p-value, t refers to the t-value in the t-tests, Age is the age of the sample. Significant values at p < .05 are shown in bold. All tests are two-tailed.

6.4.2. Descriptive Statistics

6.4.2.1. Mental Health Characteristics of the Sample. The questionnaires relating to variables associated with mental health provided an indication of the general mental health of the study sample. With regards to depressive symptoms, the mean depression score was M = 15.77 (SD = 10.11), and 43% of the sample scored 16 or higher in the CES-D, which is considered depressed (with 57% of the sample falling below this cut off for depression). This prevalence is comparable to other studies which have used the CES-D to measure depressive symptoms amongst adults (Flett, Besser, & Hewitt, 2014), and undergraduate students (Horgan, Kelly, Goodwin, & Behan, 2018), but the prevalence is high compared to studies which use other self-report measures of depression (M. Green & Benzeval, 2011).

In terms of self-harm behaviour, 31% of the sample reported a lifetime prevalence of self-harm, which is relatively high compared to other studies with young adults (O'Connor et al., 2018) and undergraduate students (Sivertsen et al., 2019), although is similar to other studies which use the DSHI with a community sample of young adults (Latimer, Meade, & Tennant, 2013).

The mean score for suicidal ideation was M = 3.00 (SD = 6.37), and 35% of the sample reported having suicidal thoughts (which is indicative of risk of self-harm with suicidal intention), while 3% of the sample reported severe ideation (a score of 21 or over) which indicates a high risk of self-harm with suicidal intention (Van Spijker et al., 2014). 65% of the sample reported no suicidal ideation. This prevalence is similar to other studies which use the SIDAS to measure suicidal ideation amongst adults (K. Mok et al., 2020; Van Spijker et al., 2014).

In terms of wellbeing, 19% of participants reported probable depression, 16% reported possible depression, 59% had average wellbeing, and 6% had high psychological wellbeing. The median wellbeing score was M = 48.00, which is similar to other studies conducted with community and student samples (Tennant et al., 2007). Therefore, the mental health scores found in the present research are comparable to studies which use similar measures, but may be slightly higher than studies which use alternative measures in the literature.

6.4.2.2. Gender Differences in the Variables Associated with Mental Health. *T*-tests were used to test for gender differences in depression, suicidal ideation, and psychological wellbeing. As self-harm behaviour was a dichotomous variable, chi-square analysis was used to test for gender differences in self-harm. The results of these analyses are presented in Table 6.2. There were no significant differences between males and females in any of the variables associated with mental health.

Table 6.2

	Males		Females		Total		Range			
	M	SD	M	SD	M	SD	Min	Max	t	p
Depressive symptoms	16.88	10.51	14.61	9.59	15.77	10.11	0.00	51.00	1.61	.11
Suicidal ideation	3.17	6.08	2.83	6.69	3.00	6.37	0.00	36.00	.39	.70
Wellbeing	46.57	9.14	48.82	7.72	47.67	8.53	14.00	70.00	-1.90	.06
	% Y	% N	% Y	% N	% Y	% N			χ^2	р
Self-harm behaviour	36.54	63.46	24.24	75.76	30.54	69.46	-	-	3.62	.06

Gender Differences in Depressive Symptoms, Suicidal Ideation, Wellbeing, and Self-Harm Behaviours (Mean, M, and Standard Deviation, SD)

Note. α =.05, *t* refers to the *t*-value in the *t*-test, χ^2 is the chi-square value, *p* is the *p*-value, range refers to the range of scores on each measure, the self-harm variable shows frequency data. % Y is the percentage of the group who reported self-harm behaviours, and % N is the percentage of the group who did not report self-harm behaviours. *p*-values are two-tailed.

6.4.2.3. Correlation with Age. Correlation analyses and *t*-tests were used to test if the key variables were associated with age, and the results are shown in Table 6.1. Age was significantly correlated with depressive symptoms, suicidal ideation, and self-harm behaviour. Specifically, depression and suicidal ideation decreased with age, and participants who engaged in self-harm were younger.

As reported in Section 5.4.6.3, age was also significantly correlated with sadness reappraisal, sadness distraction, and anger reappraisal, but not anger distraction, fear reappraisal, fear distraction, or gender. As age was correlated with several of the study variables, age was included as a covariate in the mediation analysis.

6.4.2.4. Correlation Between the Variables Associated with Mental Health. It is known in the literature that there are correlations between depression, self-harm behaviour, and suicidal ideation (De Beurs et al., 2020; Gilchrist & Sadler, 2019; Mars et al., 2019), and so this was also tested in the present research using correlation analysis and *t*-tests. The results of these tests are shown in Table 6.1. Depressive symptoms were positively correlated with suicidal ideation, and individuals who engaged in self-harm behaviours had significantly

higher levels of depressive symptoms, and suicidal ideation. Thus, there were significant correlations between the variables associated with mental health.

6.4.2.5. Comparing the Mental Health of Student Participants to the Rest of the

Sample. Analyses were run on the variables associated with mental health to determine if there were differences between the student participants and the rest of the sample (government analysts and community members). The results are shown in Table 6.3. Students had higher levels of depressive symptoms and suicidal ideation and were more likely to report self-harm behaviours than the rest of the sample. However, there was no difference in wellbeing between the students and the rest of the sample.

Table 6.3

T-Tests and Chi-Square Analysis Comparing Student Mental Health Scores With Rest of the Sample

	Students		Rest of	st of sample To		otal Ra		nge		
	M	SD	M	SD	M	SD	Min	Max	t	р
Depressive symptoms	16.78	10.16	11.27	8.67	15.77	10.11	0.00	51.00	3.38	.001
Suicidal ideation	3.46	6.64	.97	4.49	3.00	6.37	0.00	36.00	2.76	.01
Wellbeing	47.30	8.83	49.32	6.89	47.67	8.53	14.00	70.00	-1.31	.19
	% Y	% N	% Y	% N	% Y	% N			χ^2	р
Self-harm behaviour	34.94	65.06	10.81	89.19	30.54	69.46	-	-	8.30	.004

Note. Significant values at p < .05 are shown in bold. All tests are two-tailed. % Y is the percentage of the group who reported self-harm behaviours, and % N is the percentage of the group who did not report self-harm behaviours.

6.4.3. Mediation Analysis

6.4.3.1. Using Mediation Analysis When X Does Not Predict Y. In terms of the

implications of the lack of gender differences in the variables associated with mental health (reported in 6.4.2.3) in conducting mediation analysis, a logical assumption to make is that X must first have a significant effect on Y for a third variable M to mediate this effect. However, contemporary mediation theorists suggest that this is not the case, and there is a consensus amongst these theorists that finding a total effect of X on Y should not be a prerequisite for testing a mediation effect (Bollen, 1989; Cerin & MacKinnon, 2009; Hayes, 2009, 2017; LeBreton, Wu, & Bing, 2009; MacKinnon, 2008; Rucker, Preacher, Tormala, & Petty, 2011; Shrout & Bolger, 2002; Zhao, Lynch Jr, & Chen, 2010).

As discussed in Section 6.3.2, it is possible for X to influence Y through M, even if the total effect of X on Y is found to be not significant using a hypothesis test (Hayes, 2017). Thus, the best test of a mediation effect is by testing the indirect effect itself, regardless of the total effect of X on Y (Hayes, 2017). As the Hayes (2017) method of mediation was used, this means that a mediation analysis could be run by testing the indirect effect, even in the absence of the total effect of gender (X) on depressive symptoms, suicidal ideation, or selfharm behaviour (Y) being significant, which was the case in the present study.

6.4.3.2. Controlling for Covariates in the Mediation Models. Three outcome variables were tested in the mediation models – depressive symptoms, suicidal ideation, and self-harm behaviour. In terms of covariance, there is evidence that these variables are correlated with one another to a degree. Depression is positively correlated with suicidal ideation and self-harm behaviour irrespective of suicidal intent (De Beurs et al., 2020; Gilchrist & Sadler, 2019; Hafferty et al., 2019; Kanwal & Aslam, 2015; Kessler, Berglund, Borges, Nock, & Wang, 2005; O'Connor, Rasmussen, & Hawton, 2010; O'Connor et al., 2009; Rasmussen et al., 2010; Wolford-Clevenger, Vann, & Smith, 2016). Similarly, there is a strong association between self-harm behaviours and suicidal ideation (Andover et al., 2012; Duarte et al., 2020; Klonsky, 2011; Mars et al., 2019; O'Connor et al., 2018). As reported in Section 6.4.2.4, these variables were also correlated in the present study. As there is evidence of correlations between depression, self-harm, and suicidal ideation, in each mediation model the other two variables are controlled for. For example, Model 1 is testing the relationship between gender, ER and depressive symptoms, and so self-harm behaviour and suicidal ideation are included as covariates in this model.

Furthermore, age has been found to play a role both in ER (Nolen-Hoeksema & Aldao, 2011; Zimmermann & Iwanski, 2014) and in mental health (Fiske et al., 2009; Hawton et al., 2012; Moran et al., 2012) in the literature. In the present study, age was found to correlate with a number of key variables, which was reported in Section 6.4.2.3. Therefore, age was also included as a covariate in the mediation models, to ensure that the effects found were not confounded by age.

It should be noted that including a large number of covariates in a regression model increases the risk of overfitting the model, which means that too much is being asked of the data given the number of observations in the study (Dalicandro, Harder, Mazmanian, & Weaver, 2021). In other words, this can occur when there are too few observations for the number of variables in the model (including the covariates). Overfitting can result in the findings of an analysis not being replicable in future studies, and so undermines the scientific accuracy of the findings (Babyak, 2004). However, as a rule of thumb, Babyak (2004) suggests 10 to 15 observations for each predictor variable as a minimum will allow for good estimates. As a more conservative rule, S. B. Green (1991) suggests a minimum of 50 for the overall sample size, with around 8 additional observations per predictor. For the present study, this would result in a sample size of 114 (50 for the original sample, and then 8 cases for each of the 8 predictors which is an additional 64). The present study comfortably meets this target sample size, and so it is unlikely that overfitting due to including multiple covariates will have occurred in this case.

6.4.4. Main Analysis: Testing the Mediation Models

Mediation analysis was conducted using the PROCESS macro version 3.4.1 on SPSS version 24. The variables used in the mediation models are shown in Table 6.4. The results for the six mediation analyses are presented in the following sections.

Table 6.4

Variables Used in the Mediation Analyses

Statistical term	Variable
Х	Gender
\mathbf{M}_1	Reappraisal/distraction effectiveness for sadness
M_2	Reappraisal/distraction effectiveness for anger
M ₃	Reappraisal/distraction effectiveness for fear
Y	Depressive symptoms/suicidal ideation/self-harm behaviour
Covariates	Age, suicidal ideation, self-harm, depressive symptoms

6.4.4.1. Model 1: Gender, Reappraisal, and Depressive Symptoms. The first

mediation model tested whether reappraisal effectiveness for sadness, anger and fear mediated the relationship between gender and depressive symptoms, while controlling for age, self-harm behaviour and suicidal ideation. During data screening, 3 outliers were identified and removed from further analysis. The regression coefficients and *p*-values for Model 1 can be found in Figure 6.1.

Firstly, suicidal ideation emerged as a significant covariate of depressive symptoms, with higher suicidal ideation being associated with higher depressive symptoms, b = .85, t(190) = 9.33, p < .001, 95% CI (.67, 1.03). Self-harm behaviour, b = 1.41, t(190) = 1.10, p = .27, 95% CI (-1.13, 3.95), and age, b = -.08, t(190) = -1.26, p = .21, 95% CI (-.19, .04), did not predict depressive symptoms in the model.

The total effect of gender on depressive symptoms was non-significant, b = -1.37, t(193) = -1.22, p = .22, 95% CI (-3.58, .84), which means that there were no significant gender differences in depressive symptoms. In terms of the relationship between ER and depression, ER effectiveness for sadness, b = -.85, t(190) = -1.10, p = .27, 95% CI (-2.36, .67), anger, b = .67, t(190) = .87, p = .39, 95% CI (-.86, 2.20), and fear, b = -.92, t(190) = -1.10, p = .27, 95% CI (-2.57, .73), did not predict depressive symptoms.

When ER effectiveness for sadness, anger and fear were controlled for in the mediation model, gender was not associated with depressive symptoms, b = -1.08, t(190) = -.95, p = .34, 95% CI (-3.33, 1.16), which is the direct effect.

In terms of the mediation, no mediation effect was found for sadness, b = -.13, 95% CI (-.58, .15), anger, b = .06, 95% CI (-.20, .42), or fear, b = -.21, 95% CI (-.76, .14), as the confidence intervals all contained zero.

Figure 6.1.

Mediation Model 1: Gender, Depressive Symptoms, and Reappraisal Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and P-Values of Each Path.



Note. Significant values (p < .05) are in bold and underlined.

6.4.4.2. Model 2: Gender, Distraction, and Depressive Symptoms. The second model tested distraction effectiveness for sadness, anger and fear as potential mediators in the relationship between gender and depressive symptoms, while controlling for age, self-harm behaviour, and suicidal ideation. During data screening, 1 outlier was identified and removed from further analysis. The regression coefficients and *p*-values can be found in Figure 6.2. As with Model 1, suicidal ideation, b = .86, t(192) = 9.45, p < .001, 95% CI (.68, 1.04), but not self-harm behaviour, b = 1.84, t(192) = 1.45, p = .15, 95% CI (-.67, 4.34), or age, b = -.07, t(192) = -1.12, p = .27, 95% CI (-.18, .05), was associated with depressive symptoms. The total effect of gender on depressive symptoms was non-significant, b = -1.55, t(195) = -1.38, p = .17, 95% CI (-3.77, .66). When distraction effectiveness for sadness, anger, and fear

Figure 6.2.

Mediation Model 2: Gender, Depressive Symptoms, and Distraction Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and P-Values of Each Path.



Note. Significant values (p < .05) are in bold and underlined.

were controlled for in the model (i.e., the direct effect), gender was not associated with depressive symptoms, b = -1.62, t(192) = -1.38, p = .17, 95% CI (-3.92, .69).

A mediation effect was found in Model 2. Distraction effectiveness for anger mediated the effect of gender on depressive symptoms, b = .82, 95% CI (.08, 1.68). Specifically, relative to males, females had higher distraction effectiveness for anger, b = .46, t(195) = 3.34, p = .001, 95% CI (.19, .73), and higher distraction effectiveness for anger was associated with higher depressive symptoms, b = 1.80, t(192) = 2.41, p = .02, 95% CI (.33, 3.27).

6.4.4.3. Model 3: Gender, Reappraisal, and Suicidal Ideation. Model 3 tested whether reappraisal effectiveness for sadness, anger, and fear mediated the relationship between gender and suicidal ideation, while controlling for age, depressive symptoms, and self-harm behaviour. During data screening, 3 outliers were identified and removed from further analysis. There was evidence of heteroscedasticity in the data, and so a heteroscedastic-resistant estimator was used in PROCESS, HC-3 (Hayes, 2017). The regression coefficient and *p*-values of Model 3 can be found in Figure 6.3.

Depressive symptoms, b = .35, t(190) = 5.09, p < .001, 95% CI (.21, .48), and selfharm, b = 2.08, t(190) = 2.19, p = .03, 95% CI (.21, 3.96), but not age, b = -.04, t(190) = -1.13, p = .26, 95% CI (-.10, .03), predicted suicidal ideation. Specifically, higher depressive symptoms and self-harm were associated with higher suicidal ideation. The total effect of gender on suicidal ideation was non-significant, b = .89, t(193) = 1.14, p = .26, 95% CI (-.65, 2.43), which means that there were no significant gender differences in suicidal ideation. In terms of ER, higher reappraisal effectiveness for anger (i.e., lower anger after using reappraisal) was associated with lower suicidal ideation, b = -1.36, t(190) = -2.90, p = .004, 95% CI (-2.29, -.44). No mediation effects were found for the change in sadness, b = .08, 95% CI (-.10, .40), anger, b = .13, 95% CI (-.56, .24), or fear, b = .21, 95% CI (-.01, .57), after using reappraisal. When the reappraisal effectiveness for sadness, anger, and fear were controlled for (i.e., the direct effect), gender was not associated with suicidal ideation, b = .72, t(190) = .95, p = .34, 95% CI (-.78, 2.22).

Figure 6.3.

Mediation Model 3: Gender, Suicidal Ideation, and Reappraisal Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and P-Values of Each Path.



Note. Significant values (p < .05) are in bold and underlined.

6.4.4.4. Model 4: Gender, Distraction, and Suicidal Ideation. Model 4 tested

whether the distraction effectiveness for sadness, anger, and fear mediated the relationship between gender and suicidal ideation, while controlling for age, depressive symptoms, and self-harm behaviour. During data screening, 5 outliers were identified and removed from further analysis. There was evidence of heteroscedasticity in the data, and so a heteroscedastic-resistant estimator was used in PROCESS, HC-3 (Hayes, 2017). The regression coefficient and *p*-values can be found in Figure 6.4.

Depression, b = .30, t(188) = 4.55, p < .001, 95% CI (.17, .43), and age, b = -.05, t(188) = -2.07, p = .04, 95% CI (-.10, -.002), but not self-harm, b = 1.76, t(188) = 1.87, p = .06, 95% CI (-.10, 3.61), emerged as significant predictors of suicidal ideation. Specifically, higher depressive symptoms were associated with higher suicidal ideation, and being older was associated with lower suicidal ideation. The total effect of gender on suicidal ideation was non-significant, b = .83, t(191) = 1.08, p = .28, 95% CI (-.69, 2.35).

One mediation effect was found. The distraction effectiveness for anger mediated the relationship between gender and suicidal ideation, b = -.58, 95% CI (-1.17, -.15). Specifically, compared to males, females had higher distraction effectiveness for anger, b = .48, t(191) = -3.48, p = .001, 95% CI (-.73, -.20), and higher distraction effectiveness for anger was

Figure 6.4.

Mediation Model 4: Gender, Suicidal Ideation, and Distraction Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and P-Values of Each Path.



Note. Significant values (p < .05) are in bold and underlined.

associated with lower suicidal ideation, b = -1.22, t(188) = -2.73, p = .01, 95% CI (-2.11, -.34). Distraction effectiveness for sadness, b = .28, 95% CI (-.12, .75), and fear, b = .01, 95% CI (-.32, .38), did not mediate the relationship between gender and suicidal ideation. In terms of the direct effect, when the change in sadness, anger, and fear were controlled for, gender was not significantly associated with suicidal ideation, b = 1.12, t(188) = 1.38, p = .17, 95% CI (-.49, 2.72).

6.4.4.5. Model 5: Gender, Reappraisal, and Self-Harm Behaviour. Model 5 tested whether reappraisal effectiveness for sadness, anger and fear mediated the relationship between gender and engaging in self-harm behaviour. As self-harm behaviour was a dichotomous variable (Yes, No), the mediation model was tested used binomial logistic regression, with age, depressive symptoms, and suicidal ideation as covariates. During data screening, 2 outliers were identified and removed from further analysis. The regression coefficient and *p*-values can be found in Figure 6.5.

Firstly, suicidal ideation emerged as a significant predictor of self-harm behaviour, b = .07, 95% CI (.003, .13), z = 2.04, p = .04, with higher suicidal ideation being associated with engaging in self-harm. Depressive symptoms, b = .03, 95% CI (-.01, .07), z = 1.43, p = .15, and age, b = -.05, 95% CI (-.09, .002), z = -1.88, p = .06, were not associated with self-harm behaviour.

No mediation effects were found. The change in sadness, b = -.03, 95% CI (-.16, .05), anger, b = .03, 95% CI (-.10, .19), and fear, b = -.06, 95% CI (-.25, .07), after using reappraisal did not mediate the relationship between gender and self-harm behaviour. When the change in sadness, anger, and fear after using reappraisal was controlled for (i.e., the direct effect), gender was not associated with self-harm behaviour, b = -.54, 95% CI (-1.22, .14), z = -1.57, p = .12.

In terms of the impact of ER on self-harm, higher reappraisal effectiveness for anger was associated with engaging in self-harm behaviour b = .47, 95% CI (.02, .93), z = 2.03, p = .04. Reappraisal effectiveness for sadness, b = -.28, 95% CI (-.73, .18), z = -1.19, p = .23, and fear, b = -.20, 95% CI (-.69, .28), z = -.83, p = .40, were not associated with engaging in self-harm behaviour.

Figure 6.5.

Mediation Model 5: Gender, Self-harm Behaviour, and Reappraisal Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and P-Values of Each Path.



Note. Significant values (p < .05) are in bold and underlined.

whether distraction effectiveness for sadness, anger and fear mediated the relationship between gender and engaging in self-harm behaviour. The mediation model was tested using binomial logistic regression, with age, depressive symptoms, and suicidal ideation as covariates. During data screening, 2 outliers were identified and removed from further analysis. The regression coefficient and p-values can be found in Figure 6.6.

6.4.4.6. Model 6: Gender, Distraction, and Self-Harm Behaviour. Model 6 tested

In terms of covariates, suicidal ideation, b = .05, 95% CI (-.01, .12), z = 1.68, p = .09, depressive symptoms, b = .03, 95% CI (-.01, .07), z = 1.60, p = .11, and age, b = -.04, 95% CI (-.09, .004), z = -1.78, p = .08, were not associated with self-harm behaviour.

No mediation effects were found in this analysis. Distraction effectiveness for sadness, b = -.06, 95% CI (-.34, .20), anger, b = .02, 95% CI (-.22, .27), and fear, b = -.06, 95% CI (-.29, .17), did not mediate the relationship between gender and self-harm behaviour. In terms of the direct effect, when the change in sadness, anger, and fear using distraction was controlled for, gender was not associated with self-harm behaviour, b = -.50, 95% CI (-.19, .19), z = -1.43, p = .15.

Figure 6.6.

Mediation Model 6: Gender, Self-harm Behaviour, and Distraction Effectiveness for Sadness, Anger, and Fear, Showing Unstandardised Regression Coefficients and P-Values of Each Path.



Note. Significant values (p < .05) are in bold and underlined
6.5. Discussion

6.5.1. Results of Study 2b

The regulation of some specific emotions were associated with gender differences in some (but not all) of the variables associated with mental health, but this depended on the strategy being used, the specific emotion being regulated, and the variable associated with mental health. The results are summarised in Table 6.5.

Table 6.5

Evidence for Study 2b

Research Questions	Evidence
SRQ1. Are there gender differences in the reporting of depressive symptoms, self-harm behaviours, and suicidal ideation?	• No gender differences were found for depressive symptoms, self-harm behaviours, or suicidal ideation.
SRQ2. Is there a relationship between gender, emotion-specific ER effectiveness and variables associated with mental health?	 The regulation of anger using distraction was associated with the relationship between gender and depressive symptoms/suicidal ideation. Females had a larger reduction in anger using distraction, and this was associated with higher depressive symptoms and lower suicidal ideation. An association was found for anger only, not for sadness or fear. An associated was found for distraction only, not for reappraisal.

6.5.2. The Relationship Between Gender and Variables Associated With Mental Health

The total effects in the mediation models demonstrated that there were no gender differences in depressive symptoms, suicidal ideation, or self-harm behaviour. These findings are generally not consistent with the depression and suicidal literatures, which tend to find that females report higher levels of depression (H. Chen et al., 2019; Eaton et al., 2012; Girgus & Yang, 2015; Hankin et al., 1998; Kessler, 2003; Kunst et al., 2019; Marcus et al., 2005; Nolen-Hoeksema, 1987, 2001; Nolen-Hoeksema & Aldao, 2011; Nolen-Hoeksema et al., 1999; Van de Velde et al., 2010; Weissman et al., 1996; Wetherall et al., 2020), self-harm behaviour (Hawton et al., 2012; Madge et al., 2008; O'Connor et al., 2009), and suicidal ideation (Barzilay et al., 2019; Lu et al., 2020; Stephenson et al., 2006).

6.5.2.1. Depressive Symptoms. There is some evidence that gender differences in depression are not always observed amongst undergraduate populations (Gladstone & Koenig, 1994; Grant et al., 2002; R. Nolan & Willson, 1994; Stangler & Printz, 1980). The present study had a high proportion (82%) of undergraduate students, and so it is possible that gender differences are less prominent in this population. In the present study, it was not the case that females had lower levels of depression compared to other studies which have used the CES-D with a similar population, such as Flett et al. (2014), but rather males had relatively higher levels of depressive symptoms, and so this might account for the decreased difference between males and females. Entering higher education can represent a time of increased stress and turbulence for individuals, and a study by Gladstone and Koenig (1994) found that social support protected males from increased depression in university. It should also be noted that gender differences in depression amongst university students using the CES-D differs by culture, and this relationship is more common amongst Western cultures compared to Eastern cultures (Kwon et al., 2013).

Another reason for this finding may be the way depression is measured and conceptualised. In a study by Grant et al. (2002), no gender differences were found amongst undergraduate students when using a self-report measure of depressive symptoms. However, when depressive disorder was measured through an interview with a clinical psychologist, males had higher rates of depressive disorder (Grant et al., 2002). This indicates that when a

clinical interview is used, the traditional pattern of gender differences in depression may actually be reversed amongst undergraduates, with males having higher rates of depression than females. It also demonstrates that self-report measures of depressive symptoms (such as that used in the present study) may not be sensitive enough to find gender differences in depression amongst undergraduate students, if these gender differences exist.

However, it should also be noted that a significant indirect effect emerged in one of the mediation models for depression, which will be discussed in more detail below. This significant indirect effect indicates that gender was associated with depression through the mediator of anger distraction. That is to say, females had a greater reduction in anger using distraction, and this was associated with higher depressive symptoms. These higher depressive symptoms for females (through the mediation effect) are consistent with the gender differences found in the literature (H. Chen et al., 2019; Eaton et al., 2012; Girgus & Yang, 2015; Hankin et al., 1998; Kessler, 2003; Kunst et al., 2019; Marcus et al., 2005; Nolen-Hoeksema, 1987, 2001; Nolen-Hoeksema & Aldao, 2011; Nolen-Hoeksema et al., 1999; Van de Velde et al., 2010; Weissman et al., 1996; Wetherall et al., 2020). These results may indicate that mechanisms of depression may differ for males and females.

6.5.2.2. Self-Harm Behaviour and Suicidal Ideation. Similarly, there were no gender differences in either suicidal ideation or self-harm behaviour. This was unexpected as it is generally inconsistent with the broader literature, which has shown that females tend to have higher rates of self-harm behaviour irrespective of intent (Hawton et al., 2012; C. S. Lee & Wong, 2020; Lewinsohn et al., 2001; Madge et al., 2008; O'Connor et al., 2009; Sivertsen et al., 2019; Wetherall et al., 2018) and suicidal ideation (Hunt et al., 2006; C. S. Lee & Wong, 2020; Sivertsen et al., 2019; Stephenson et al., 2006), although more males die by suicide (National Records of Scotland, 2021). One Scottish study found that more males than

females reported suicidal ideation, although females were more likely to attempt suicide (Wetherall et al., 2018).

One explanation for the findings of the present study is that some studies which have found prominent gender differences in self-harm, such as Madge et al. (2008), tend to emphasise behaviours which are more commonly engaged in by females, such as cutting and self-poisoning (D. Casey et al., 2020; Lundh et al., 2007). Males, on the other hand, are more likely to use burning or hitting behaviours to self-harm (Andover et al., 2010; Idig-Camuroglu & Gölge, 2018; Sornberger et al., 2012). The measure of self-harm used in the present study, the DSHI (Gratz, 2001), was chosen because it directly measures a wide range of self-harm behaviours which are commonly used by both females and males, and so would be more likely to detect a gender difference if one exists, regardless of the specific method used by the individual. It may be the case that this measure reduced any potential gender bias in responding, and so the difference between males and females in self-harm behaviour disappeared.

In the Scottish Health Survey (Knudson et al., 2020) participants were asked whether they 'have ever self-harmed in any way but not with the intention of killing themselves'. This measure asked about any method of self-harm, and gender differences emerged, with a higher percentage of females than males reporting self-harm behaviours, which is in contrast to the findings of the present study. However, the difference between this approach and the measure used in the present study (DSHI) is that the DSHI offers specific instances of methods that are commonly used by females (e.g., cutting) and males (e.g., burning). This may mean that even if a participant did not define their behaviour as self-harm, or did not offer this information without being prompted, they may still report this on the DSHI, but not in the Scottish Health Survey measure (Knudson et al., 2020). Therefore, it is likely that the DSHI reduced gender bias in the measurement of self-harm behaviours.

Another explanation is that the development of self-harm behaviours is different for males and females, which may result in the gender gap changing with age and becoming less prominent in adulthood (or disappearing entirely). Many of the reported gender differences in self-harm come from studies conducted with adolescents (De Leo & Heller, 2004; Hawton et al., 2002; Hawton et al., 2012; Lundh et al., 2007; Madge et al., 2008; O'Connor et al., 2009). Self-harm behaviours tend to decrease with age (Hawton et al., 2012), and are least prevalent in older age groups (McManus et al., 2019). There is also evidence that the gender gap decreases across adolescence, with self-harming behaviours becoming increasingly common in males and remaining stable in females (Boeninger et al., 2010; Hawton et al., 2003; Hawton et al., 2012), and some studies showing that self-harming behaviours decrease as females (but not males) get older (C. Q. Li et al., 2020).

As the present study was conducted with adults, it is possible that the gender difference has declined enough to be statistically non-significant by adulthood. However, a meta-analysis by Bresin and Schoenleber (2015) found that gender differences (with females reporting higher levels of self-harm) did not change across age groups, although the majority of studies included in the analysis were conducted in the US. This study also demonstrated that gender differences tend to be smaller in undergraduate and community samples (compared to a clinical sample), and when self-report measures of self-harm are used (compared to a clinical interviews) (Bresin & Schoenleber, 2015), which may also contribute to the absence of a gender effect in the present study. Although the findings of the present study are not consistent with the literature more generally, they do mirror the findings of Gratz et al. (2002), who used the same measure of self-harm as that used in the present study (DSHI) with an undergraduate sample and found no gender differences.

In terms of suicidal ideation, much of the previous research was conducted with adolescents (Barzilay et al., 2019; Reinherz et al., 1995; Reinherz et al., 2006), and often in

East Asian countries such as China (Lu et al., 2020), South Korea (Kim et al., 2014), and Taiwan (Chiu et al., 2017), which may have cultural differences compared to the present study, which was conducted in Scotland. In a study by O'Connor et al. (2018) conducted with adults in the UK (similar to the present study), no gender difference in suicidal ideation was found. Further, in a Scottish study with young adults, males were found to report having suicidal thoughts more than females, which may indicate that males have higher suicidal ideation in Scotland (although females were more likely to report taking action on these thoughts) (Wetherall et al., 2018) Therefore, age and culture may moderate these gender differences in suicidal ideation.

However, as with depression, although the total effect of gender on suicidal ideation was not statistically significant, when ER was included as a mediator, a significant effect emerged through this mediator. Specifically, females had a larger reduction in anger after using distraction, this was associated with lower suicidal ideation. This finding indicates that females have lower suicidal ideation, through the mediator of anger distraction. Therefore, although there were no gender differences in suicidal ideation overall, when mediators were included in the model, evidence of gender differences emerged.

6.5.3. The Relationship Between Gender, Distraction of Anger, and Depression

The change in anger using distraction was associated with gender differences in depressive symptoms. Compared to males, females had a larger reduction in anger using distraction, and this was correlated with higher depressive symptoms. This indicates that females were more effective than males at reducing anger using distraction, but this was associated with higher depression.

The present findings demonstrate that distraction is an effective strategy for decreasing feelings of anger in an experimental setting, which is in line with the literature

(Rusting & Nolen-Hoeksema, 1998). Furthermore, it indicates that females may be more effective at reducing anger. There is evidence that females receive cultural messages that anger is inappropriate to express (Root & Denham, 2010), and so females may be more practiced at distracting from anger to avoid negative social consequences (Brescoll & Uhlmann, 2008). This finding is consistent with a study by Rusting and Nolen-Hoeksema (1998), who found that females were more likely to distract from an angry mood compared to a neutral mood, but this was not found for males.

Alternatively, the gender difference in distraction from anger could result from females having higher levels of anger at baseline compared to males, which would provide more of an opportunity to reduce this emotion for females. However, the *t*-test reported in Section 5.4.4.2 showed that there was no difference in anger in the no regulation condition between males and females in the present study (which may give an idea of baseline levels of emotion when ER strategies have not been instructed to be used), which lends support to the argument that females are more effective at distracting from anger.

The results of this model may highlight an important mechanism for explaining the gender differences in depression. Although at face value, the total effect of gender on depressive symptoms $(X \rightarrow Y)$ was not statistically significant, when the ER variables were included, the change in anger after using distraction emerged as a significant mediator. This means that gender was associated with depressive symptoms through the mediator of anger distraction. Although there are many contributing factors to the gender difference in depression (Nolen-Hoeksema, 1987; Nolen-Hoeksema et al., 1999), these findings may highlight anger distraction as one possible mechanism for the higher rates of depression observed amongst females.

As reported in Section 2.3.4, according to Response Styles Theory (RST), females are more likely to ruminate, and males are more likely to distract from a low mood, and this

contributes to females' higher rates of depression (Nolen-Hoeksema et al., 1999). The evidence is generally supportive of gender differences in rumination (D. P. Johnson & Whisman, 2013), and distraction (Gomez-Baya et al., 2016; Trives et al., 2016). However, the findings of the present study are somewhat inconsistent with the distraction tenet of RST (that males are more likely to use distraction, and this has an adaptive relationship with depression).

The association between effective use of distraction and higher depressive symptoms is contrary to what would be expected, as distraction is often regarded as an adaptive strategy in the literature (E. L. Davis et al., 2016; Hermann et al., 2017; Polanco-Roman et al., 2015; Smoski et al., 2014; Webb et al., 2012), although few studies have examined distraction within the context of anger. Whether distraction can be regarded as adaptive or maladaptive depends on how it is used. A study by Wolgast and Lundh (2017) found that when distraction is approached with an attitude of acceptance towards negative emotions, rather than to avoid the experience of emotion, it is associated with greater wellbeing. On the other hand, depressed people are more likely to use distraction in an attempt to avoid negative emotion (Wolgast & Lundh, 2017). Therefore, distraction may be maladaptive within the context of depression, particularly if it is used in such a way as to avoid negative emotions (such as anger).

In terms of previous studies on general emotion, a greater reduction in negative emotion using distraction during the ERT was associated with higher depressive symptoms amongst depressed individuals (Smoski et al., 2014), which suggests that the use of distraction in this context is maladaptive, in line with the findings of present study. In addition, experiencing higher levels of anger in general is associated with depressive symptoms (Abdolmanafi et al., 2011), and with greater use of maladaptive strategies such as rumination and catastrophising, which in turn may be associated with higher depressive

symptoms (Besharat et al., 2013). Interestingly, ruminating on anger is also associated with higher depressive symptoms (du Pont et al., 2018). It is possible that both ruminating on and distracting from anger are maladaptive within this context, and an approach which involves addressing anger without ruminating on it may be the most adaptive path.

Another potential explanation for the negative correlation between anger distraction and depressive symptoms is that anger suppression was involved. There is evidence that suppressing anger may be associated with higher levels of depression (Allan & Gilbert, 2002; Besharat et al., 2013; C. L. Brody et al., 1999; Cheung & Park, 2010; Duckro et al., 1995; R. C. Martin & Dahlen, 2005; Sperberg & Stabb, 1998). During the ERT, participants were asked to think of something unrelated to the situation depicted in the task (i.e., use distraction) but it is possible that some participants may have suppressed their feelings of anger during the task, particularly if they are familiar with using suppression. Females may be more likely to suppress their anger, at least in some contexts (Cox et al., 2000; Fischer & Evers, 2011; Kwon et al., 2013; Sharkin, 1993), and so this could explain why females had a larger reduction in anger, which was related to higher depressive symptoms.

An interesting direction for future research is to examine the causal relationships between gender, emotion-specific ER, and variables associated with mental health using a prospective study design. Distraction as a way of coping with anger may lead to higher depressive symptoms over time and may partially explain females' higher rates of depression. On the other hand, individuals with depression may turn to distraction as a more efficient way of regulating their feelings of anger, as distraction is a less effortful method of regulating emotions, at least within the context of sadness (Sheppes et al., 2009; Sheppes & Meiran, 2008). In a study by Smoski et al. (2014) older adults with a diagnosis of depression were more effective at reducing negative emotion with distraction during the ERT compared to individuals without depression, which would support the hypothesis that depressed

individuals gravitate towards using distraction. Therefore, prospective studies will be useful in assessing the causal impact of these variables in an emotion-specific manner.

6.5.4. The Relationship Between Gender, Distraction of Anger, and Suicidal Ideation

The change in anger after using distraction was associated with gender differences in suicidal ideation. Specifically, females had a larger reduction in anger using distraction, and this was associated with lower suicidal ideation. Notably, the reduction in anger using distraction had a different relationship with depression and suicidal ideation, as it was associated with higher depressive symptoms, but lower suicidal ideation.

Generally speaking, using distraction to cope with a negative mood is associated with lower odds of engaging in self-harm behaviours with or without suicidal intent (Polanco-Roman et al., 2015). In a study by Stanley et al. (2021), the intuitive ways that individuals regulate their emotions in their daily lives were measured using ecological momentary assessment (EMA) amongst individuals with a diagnosis of a mood disorder and borderline personality disorder. Reporting intuitively using distraction or positive activities such as keeping busy, socialising, positive thinking, and doing something good for self as ER strategies was associated with less intense suicidal ideation (Stanley et al., 2021). This finding is in line with the present study, which found that more effective distraction for anger was associated with lower suicidal ideation. However, it should be noted that in Stanley et al. (2021) behavioural distraction was measured (i.e., doing things to distract yourself), and in particular these were behaviours which have adaptive connotations (e.g., doing something good for self), rather than more maladaptive distraction behaviours such as substance use. On the other hand, it was the use of cognitive distraction that was measured in the present study, which involves removing the attention from an emotional stimulus during the ERT.

Nonetheless, the results of the present study and Stanley et al. (2021) indicate that distraction may be adaptively associated with lower suicidal ideation, either by cognitively

removing attention from an emotional stimulus, or distracting oneself from an emotional situation through engaging in an unrelated behaviour. An explanation for this may come from Gross's (1998) process model of ER. According to this model (and appraisal theories that this model is derived from, as reported in Section 1.2.2.3), we pay attention to an emotional situation (attentional deployment), and then have an appraisal of the situation, which leads to an emotional response Gross (1998). Suicidal ideation may be understood as an extreme appraisal of a situation (i.e., it is thoughts about suicide). Distraction may be an effective strategy which is associated with lower suicidal ideation, because it intervenes at the attentional deployment stage of the process model, which appears before the appraisal stage. This means that the attention is removed from the emotional situation, before the suicidal thoughts have had the chance to appear, and thus distraction would be helpful in reducing suicidal ideation according to this framework. The present study builds on the findings of Stanley et al. (2021), by demonstrating that distraction may be helpful for suicidal thoughts within the context of anger regulation, rather than just within the context of general coping with distress as measured in Stanley et al. (2021). Additionally, the findings of the present study may also indicate that distraction from sadness and fear do not have a relationship with suicidal ideation, and therefore distraction may be a less effective strategy for reducing suicidal ideation within these emotional contexts.

However, there are very few studies which examine how distraction as an ER strategy is related to suicidal ideation within the context of anger specifically. There is evidence that experiencing anger as an emotional state may be related to the development of suicidal ideation (Goldney et al., 1997; Hawkins & Cougle, 2013; Hawkins et al., 2014), and problematic anger predicts suicidal ideation and suicidal attempts over time (Dillon et al., 2020). These findings demonstrate that being unable to effectively regulate feelings of anger may lead to having thoughts about suicide. It should be noted however that in the present

study it is not problematic anger that was measured, but rather the regulation of anger (using distraction and reappraisal) which was induced in a laboratory setting using emotionally evocative stimuli. This finding indicates that having the ability to effectively reduce anger using distraction is related to lower suicidal ideation.

In relation to gender differences, this mediation effect is somewhat surprising because females traditionally report higher levels of suicidal ideation in the literature (Barzilay et al., 2019; Lu et al., 2020; Stephenson et al., 2006). However, the findings from this model may highlight a mechanism through which females could reduce suicidal ideation, as anger distraction was associated with lower levels of suicidal ideation, and females could more effectively reduce anger during the ERT. It is interesting that distracting from anger was related to suicidal ideation, but this was not the case for the regulation of sadness or fear during the task, implicating anger-inducing situations as being involved in suicidal ideation.

It is also notable that the regulation of anger using distraction has opposing relationships with depression and suicidal ideation. Anger distraction was associated with higher depressive symptoms, but with lower suicidal ideation. This may highlight that whether or not a strategy can be considered adaptive depends on the context, including the emotional context. It may be the case that distracting from feelings of anger may be adaptive for those who are predisposed to having suicidal thoughts but may be less adaptive within the context of depression. The process of distraction may help to distract from suicidal thoughts, as this has been reported as a common strategy to cope with suicidal thoughts (Simon et al., 2016), but it may be the case that a consequence of this distraction is a negative relationship with depressive symptoms (Smoski et al., 2014).

6.5.5. Reappraisal of Anger, Self-Harm Behaviour, and Suicidal Ideation

Although there was no mediation effect, two interesting findings that emerged from the analyses were regarding the use of reappraisal to regulate anger. Firstly, a reduction in anger using reappraisal was associated with engaging in self-harm. However, a reduction in anger using reappraisal was also associated with lower suicidal ideation.

The suicidal ideation finding is consistent with previous research, which has found that the use of reappraisal is associated with lower suicidal ideation (Forkmann et al., 2014), and individuals with a history of suicidal ideation were less effective at using reappraisal during an experiment (Kudinova et al., 2016). Reappraisal involves cognitively changing negative thoughts about a situation (Gross, 1998a), and so this skill may also be applied to thoughts which are suicidal in nature. However, previous work has focused on the reappraisal of general emotion, and the present study has highlighted that the regulation of anger (but not sadness or fear) has a potentially adaptive relationship with suicidal ideation.

In terms of self-harm behaviour, this finding is more puzzling. It would be expected that reducing feelings of anger using reappraisal may eliminate the need for using self-harm to regulate emotions. However, this is not consistent with the Study 2b findings. The most common reason individuals report for engaging in self-harm behaviours is to regulate negative emotions (Bresin & Schoenleber, 2015; Madge et al., 2008), and self-harm is sometimes conceptualised as an ER strategy in the literature (Chapman et al., 2006; Linehan, 2018). Self-harm can be a way of avoiding the experience of emotions (Brereton & McGlinchey, 2020). It is possible that individuals who tend to use self-harm as a way of coping with their emotions are well practiced in avoiding the experience of emotions. Therefore, when these individuals were asked to use reappraisal during the ERT, they may have automatically defaulted to using avoidance to regulate the feelings of anger that arose,

which would explain the reduction in anger, and also its relationship to an increased likelihood of engaging in self-harm behaviour.

6.5.6. Implications of Findings

There is promising evidence that adaptive ER skills can be developed through ER training (N. Cohen & Ochsner, 2018; Denny, 2020; Herwig et al., 2019; Plate & Aldao, 2017), and these interventions have a positive impact on variables associated with mental health over time (Gratz et al., 2015; Kiosses et al., 2018; LeBlanc et al., 2020; LeBlanc et al., 2017; Morris et al., 2015; Ranney et al., 2017). In recent years there has been a move towards developing increasingly personalised longitudinal interventions to increase ER skills (Denny, 2020). The findings from the present study may provide guidance for these therapeutic interventions by highlighting that (1) gender, and (2) the specific emotion being regulated may have an impact on the clinical utility of ER. For example, using anger distraction may be beneficial for certain individuals, such as those who are predisposed to having suicidal thoughts, but not depressed individuals. Similarly, females may be more effective at using anger distraction than males, and so males may need more support in this area.

Dialectical Behavioural Therapy (DBT) aims to teach people a wide range of skills to regulate intense emotions, one of which is distraction from negative emotions (Berking et al., 2008; Lieb et al., 2004). Research has found that an increase in ER skills during a DBT programme was associated with a decrease in self-harm irrespective of intent, lower depression, and an increase in anger control amongst individuals with BPD over time (Neacsiu et al., 2010). Although distraction is only one of the many skills taught in this programme, it may be the case that anger distraction could play a role in this type of therapy, particularly as the findings of the present study indicate that there is an adaptive relationship between anger distraction and suicidal ideation. Distraction may be a beneficial strategy for

certain groups, such as older people or individuals who are impaired, compared to other strategies such as reappraisal, as distraction can be implemented regardless of the emotional situation, whereas reappraisal requires more cognitive flexibility in reinterpreting the situation (Smoski et al., 2014).

The present study was conducted with undergraduate students and community participants, rather than with individuals with a background of clinical diagnoses, and thus the generalisability of the findings to clinical settings is limited. However, in terms of the applicability of the findings for tackling mental health problems in the general population, the development of adaptive ER skills is thought to play an important role in promoting mental health and wellbeing in the community, and providing early intervention for the development of more serious clinical disorders (Jorm, 2012; Kalra et al., 2012; LeBlanc et al., 2020). The findings from the present study highlight the potential implications of using reappraisal and distraction amongst community adults.

6.5.7. Limitations of this Study

This study was cross-sectional in nature, which means that the data were collected at one time point, and no inferences can be made about the causal relationship between variables. An interesting direction for future research is to examine if ER impacts on mental health, mental health has an impact on ER over time, or if these variables share a bidirectional relationship. In the present study, only associations between variables can be assessed using this research design, although this is an important first step in testing potential causal relationships.

Furthermore, the characteristics of the sample used in the present study may limit the generalisability of the findings. The sample was made up of university students and government analysts, with only a small proportion of the sample being from the wider

community. In the present study, some findings did not reflect what has been found in the wider literature, such as a lack of gender differences in depressive symptoms and rumination. Further, analysis between these groups did reveal differences between them, namely that students were more depressed, had higher suicidal ideation, and were more likely to self-harm than the rest of the sample. Therefore, it may not be possible to generalise the findings of the present study to the general community.

An exclusion criterion for the present study was that participants must not have a clinical diagnosis of depression. However, no information was gathered regarding whether any of the study participants are currently being treated for depression, or if they have received treatment for depression in the past (including for depressive symptoms). It would be useful to know this information as any treatment for depression may confound the relationship between ER and depressive symptoms.

Self-report questionnaires were used to measure depressive symptoms, self-harm behaviours, suicidal ideation, and psychological wellbeing. These instruments are designed to be indicative of symptoms which may accompany disorders such as depression, but are not diagnostic in nature, and so do not necessarily provide information about the prevalence of disorders (Radloff, 1977). This approach also relies on participants having a level of selfawareness of their own symptoms, although this issue is relevant to all research which uses self-reporting methods.

6.6. Conclusion and Next Steps

To summarise, the present study was the first to examine the regulation of specific negative emotions as mediators of gender differences in variables associated with mental health. Two mediation effects emerged from the analyses, which implicate the regulation of anger using distraction as playing an important role in gender differences in variables

associated with mental health. Specifically, females reduced anger using distraction more effectively than males, and this was associated with higher depressive symptoms and lower suicidal ideation. This finding indicates that using distraction to reduce anger may be an important mechanism for the higher rates of depression observed amongst females, but also shows that the adaptiveness of using distraction to regulate anger may depend on the context within which its used. These findings have important implications for therapeutic programmes, by enabling more individualised support to be developed based on gender and the specific emotional context.

The most important finding from this study is that what is known about the regulation of one emotion cannot be applied to all emotions. The regulation of specific emotions using the same strategy had a different relationship with variables associated with mental health. Therefore, it is crucial for future research in the ER literature to examine the relationships between gender, ER, and variables associated with mental health in an emotion-specific manner moving forward.

KEY POINTS FOR CHAPTER 6

- 1. The aims of Study 2b were (1) to test if there are gender differences in depressive symptoms, self-harm behaviours, and suicidal ideation, in line with the wider literature, (2) to examine if the regulation of specific emotions using reappraisal and distraction mediated gender differences in variables associated with mental health, and (3) to investigate if social desirability or age had an impact on the results.
- 2. Females were found to have a greater reduction in anger using distraction, and this was associated with (1) higher depressive symptoms, and (2) lower suicidal ideation.
- 3. The results highlight that what is known about one emotion cannot necessarily be applied to all emotions, and so it is imperative for future studies to examine the relationships between gender, ER and mental health in an emotion-specific manner moving forward.
- 4. The clinical relevance of these findings is that the effectiveness of ER may depend on (1) gender, and (2) the specific emotional context, and these findings may help to guide the development of more personalised support.

QUESTIONS FOR SUBSEQUENT CHAPTERS

- 1. Considering the findings of Study 1 and Study 2 together, what does this research tell us about gender differences in emotion-specific ER, and their impact on mental health?
- 2. What are the wider implications of this research?

NEXT STEPS

The empirical studies of the present research were reported in Chapter 4 (Study 1), Chapter 5 (Study 2a) and Chapter 6 (Study 2b). There was evidence of emotionspecific gender differences, and the regulation of some emotions may be related to gender differences in mental health. This research highlighted the importance of investigating gender differences in ER in an emotion-specific manner. Now that the empirical studies have been described, the final step is to consider the wider implications of this research, which will be discussed in the next chapter.

Chapter 7: General Discussion

7.1. Summary of Research

In this thesis, (1) emotion-specific ER, (2) gender differences in emotion-specific ER, and (3) the relationship between gender, emotion-specific ER, and variables associated with mental health were investigated. As reported in Section 3.5.1, the present research was driven by five overarching research questions:

- RQ1. Is ER different for specific emotions?
- RQ2. Are there gender differences in the regulation of specific emotions, and do these gender differences vary across different emotions?
- RQ3. Are there gender differences in engagement and disengagement ER?
- RQ4. Does the effectiveness of reappraisal and distraction depend on the specific emotion being regulated?
- RQ5. Are gender differences in emotion-specific ER related to gender differences in mental health?

These questions were addressed using two empirical studies, which were reported in Chapter 4 (Study 1), Chapter 5 (Study 2a), and Chapter 6 (Study 2b). The unique contribution of the present research to the wider ER literature is summarised in Table 7.1, and these findings will be discussed in the following sections. In this chapter, these findings will be reflected upon within the context of the strengths and limitations of the present research, and for this reason a separate section on limitations will not be included in this chapter.

Table 7.1

Study	Findings	Unique Contribution to Literature
Study 1	 Females were more likely than males to report using reappraisal for fear, but this gender difference was not found for sadness or anger. Females were more likely than males to vent/express their emotions when they feel sad or scared, but not angry. Young adolescent females were more likely to use engagement for sadness and fear compared to their male counterparts. 	 This study was the first to show that gender differences in the reporting of reappraisal and venting/expressing emotion are emotion-specific. This study also demonstrated that gender differences in engagement as a broad type of ER may depend on age, and is emotion-specific.
Study 2a	 Females were more effective than males at reducing sadness and anger using distraction. Females were more effective at reducing fear using both reappraisal and distraction. Reappraisal was effective for anger and fear (but not sadness), and distraction was effective for sadness and anger (but not fear). 	 This study was the first to show that gender differences in ER effectiveness were emotion-specific. This study was also the first to demonstrate that the effectiveness of an ER strategy depends on the specific emotion being regulated.
Study 2b	• Females had a larger reduction in anger using distraction, and this was associated with higher depressive symptoms and lower suicidal ideation.	 This study showed that the regulation of anger using distraction has important implications for depressive symptoms and suicidal ideation It was also the first to show that the mediating effect of ER on the relationship between gender and variables associated with mental health is emotion-specific.

Unique Contribution of This Research to the Emotion Regulation Literature

7.2. Emotion Regulation May Differ According to the Specific Emotion Being Regulated

The first question in the present research pertained to whether ER is different for specific emotions. An important finding to emerge from the present research is that the specific emotion being regulated matters, and the ER strategies that individuals use depend on the emotional context. In Study 1, the percentage of the group that reported using each strategy was compared between sadness, anger, and fear situations. This was examined across a range of strategies, including some of the most commonly studied strategies in the literature – avoiding the situation, distraction, experiential avoidance, hostility to others, listening to music, passivity, problem solving by taking action, reappraisal, rumination, self-talk, social support, sports/exercise, taking time out, watching TV/movies, and expressing/venting emotion.

All of the strategies, except passivity, differed across the specific emotions. There is evidence in the wider literature that specific emotions tend to be regulated using different strategies (Dixon-Gordon et al., 2015a; Vishkin et al., 2020; Zimmermann & Iwanski, 2014), which is consistent with the findings of the present research. Specifically, individuals may be more likely to seek social support, or use passivity, or avoidance when they feel sad (Zimmermann & Iwanski, 2014). Although there was no emotion-specific difference for passivity in the present study, it did emerge that individuals are more likely to seek social support when they feel sad, which is consistent with previous research. It also emerged that individuals are more likely to avoid the experience of sadness, which is in line with Dixon-Gordon et al. (2015a) and Zimmermann and Iwanski (2014). It emerged in the present research that individuals may be more likely to physically leave or avoid a situation which involves fear or anger, which was also found in Rivers et al. (2007). This highlights an important distinction between experiential avoidance and physical avoidance (i.e., avoiding a situation) in terms of the specific emotional context in which these strategies are used.

Reappraisal was more often used to regulate sadness and fear than anger in the present study. Although different tactics of reappraisal were not measured in the present research, a previous study found that reappraisal tactics may also differ by emotion (Vishkin et al., 2020). Overall, the findings of the present research and previous studies indicate that specific emotions are regulated in different ways.

7.2.1. Implications for Discrete Emotion Theory

According to the discrete perspective of emotions, and specifically the functionalist approach (described in Section 1.2.2.1), emotions are often thought to serve different functions to aid survival (Lench et al., 2015). The findings of the present research (and other emotion-specific studies) support this functionalist approach. For example, sadness is believed to have a social function that involves eliciting sympathy and support from others after a loss (Balsters et al., 2013; Lench et al., 2015), and so it is fitting that more people seek social support to cope with this emotion. A fearful situation may represent a significant threat in the environment and so it is logical that individuals may regulate their emotions by avoiding the situation, which also has a self-preservation function, by removing the threat (Campos et al., 1994; Lench et al., 2016). Therefore, the Study 1 finding that reported ER differed across specific emotions indicates that ER may depend on the emotional context. This is consistent with a discrete perspective of emotions, particularly the functionalist view that emotions serve different adaptive functions.

7.2.2. Developing Emotion-Specific ER Instruments

Through conducting the present research, an important limitation in the field was highlighted – there are a lack of available instruments to measure ER in an emotion-specific manner. As described in Section 3.5.2, the current research used two key methodological approaches – (1) measuring how many participants from different groups reported using ER strategies, and (2) measuring how effectively individuals reduced self-reported emotion during an experimental task. This yielded two useful metrics of ER, which were used to examine gender differences in ER. However, another approach is measuring the extent to which individuals use ER strategies habitually in their daily lives using traditional self-report questionnaires, such as the ERQ (Gross & John, 2003). This approach produces a mean score for different ER strategies, which can be used to examine group differences in ER.

There are a limited number of self-report instruments currently available in the literature that measure ER in an emotion-specific manner by measuring specific emotions on separate scales. Further, the emotion-specific questionnaires that are available often miss out important ER strategies. The two emotion-specific measures are the FEEL-KJ (Cracco et al., 2015) and the Negative Emotion Regulation Inventory (NERI; Zimmermann et al., 2008). The FEEL-KJ is used to measure ER to regulate sadness, anger, and anxiety amongst children and adolescents. Although the FEEL-KJ measures a wide range of strategies, it does not include some of the most common strategies in the literature, such as reappraisal and suppression. Similarly, the NERI also measures emotion-specific ER in adolescents and adults, but does not include reappraisal. This is problematic, because ER research has highlighted reappraisal as an important ER strategy for a range of different life outcomes (Aldao et al., 2010; Appleton et al., 2013; Appleton et al., 2014; Balzarotti et al., 2016; E. Davis & Levine, 2013; Gross & John, 2003; McRae et al., 2012b; Troy et al., 2010).

Although these emotion-specific measures represent a step forward for the field of ER, they are limited in that they do not allow for the impact of emotion-specific reappraisal on outcomes to be assessed. The findings of the present research have highlighted that the effectiveness of reappraisal depends on the specific emotion being regulated. Accordingly, it is also possible that using reappraisal to regulate specific emotions may have a different

impact on life outcomes. It is therefore crucial that emotion-specific instruments that include important strategies such as reappraisal are developed in the field of ER. This will also allow for comparisons to be made across different emotion-specific studies.

In addition, it is important that any future measures can be used with different age groups, to allow emotion-specific ER to be assessed across the lifespan. The present research has shown that ER, and gender differences in ER, may change with age. This is in line with other studies which demonstrate that gender differences in ER may change with development (Nolen-Hoeksema & Aldao, 2011). The open-ended questionnaire used in Study 1 was specifically designed to be used with both adolescents and adults – by wording the questions in plain English, without using jargon (as described in Section 3.5.2.1). This was found to be successful as all participants were able to provide meaningful responses in the study. Therefore, emotion-specific measures should be developed with a view of using these measures with a range of age groups to assess emotion-specific age differences in ER.

In sum, an important direction for future research is the development of self-report questionnaires which (1) measure ER in an emotion-specific manner, (2) measure a wide range of ER strategies, including important strategies such as reappraisal, (3) measure different aspects of ER, such as ER effectiveness and ER frequency, as the present research has shown that these are not interchangeable, and (4) can be used across different age groups. The findings of the present research have highlighted the key strategies that individuals report using for each emotion (e.g., seeking social support, sports/exercise, venting, listening to music, reappraisal, problem solving), and have shown the strategies that demonstrate gender differences (e.g., seeking social, support, venting, reappraisal, listening to music). These strategies, along with other common strategies in the literature (e.g., rumination, suppression, avoidance, distraction), should be included in any emotion-specific instrument. The development of such an instrument will add value to the field by allowing emotion-specific

ER research to become standard practice and will enable a deeper understanding of how regulating specific emotions may have a differential impact on a range of outcomes.

7.2.3. Conducting Emotion-Specific Research in a More Naturalistic Setting

In Study 1, a measure was used which relied on participants retrospectively reporting on their ER in different emotional contexts, and also depended on participants having an awareness of their ER processes. This self-report method is a valuable tool which allowed ER to be measured for specific emotions and a similar version of this approach has been used in previous ER studies (Dixon-Gordon et al., 2015a; Goubet & Chrysikou, 2019; Rivers et al., 2007). Although self-report methods which rely on retrospective accounts of ER are frequently used in the field with much success (Aldao et al., 2010; Gratz & Roemer, 2004; Gross & John, 2003), they can be subject to bias (Shiffman et al., 2008). Individuals can have difficulty recalling information from their autobiographical memory when reporting on surveys, which can result in inaccuracies in the information provided (Bradburn et al., 1987). Therefore, an important step for future research is to examine the use of ER strategies in specific emotional contexts in a more naturalistic way, which does not rely on these retrospective accounts of ER.

A useful method for measuring ER in a naturalistic setting is ecological momentary assessment (EMA). EMA includes a range of methods that involve repeatedly measuring a psychological phenomenon as it is happening in real-time (Shiffman et al., 2008). Using EMA to measure emotion-specific ER may involve participants completing a diary entry at regular intervals throughout the day, or when they experience an emotional episode, and reporting on what they are doing in the moment to try to reduce this feeling, if anything. Participants report on what they are feeling (i.e., the subjective experience of their specific emotion), and may select options from a range of items which represent different ER

strategies (e.g., 'I tried to change what I am thinking about to something more neutral or positive' represents the use of reappraisal). Participants may also be asked how effective the strategy has been for reducing that particular emotion during the emotional situation.

Using EMA would build on the findings of Study 1 by allowing emotion-specific ER to be examined in a real-world setting, which increases the ecological validity of the findings (Shiffman et al., 2008). It also allows both the use of ER strategies and the effectiveness of these strategies (as defined by the participant) to be assessed in a single study, as both momentary affect (i.e., level of subjective experience in the moment), and the ER strategy used are both reported. EMA has been used in studies of ER with success (Bai et al., 2020; Colombo et al., 2020; Stanley et al., 2021).

This method may be particularly useful for examining ER amongst groups who have less awareness of their ER processes. For example, in Study 1, adolescents had a higher percentage of missing data than older participants, and males had more missing data than females, which may indicate that adolescents and males had more difficulty recalling their ER strategies or had less awareness of their ER strategies. Some ER strategies are thought to occur implicitly, outside of conscious awareness (Gyurak et al., 2011), and so participants may have more difficulty reporting these strategies using the Study 1 approach. Using EMA may encourage accurate reporting of ER amongst these groups by offering examples of ER (e.g., 'I try to avoid feeling my emotion', which represents suppression). This may allow for these groups to more easily report on their ER in the moment, without having to rely on retrospectively recalling processes from a previous experience. This approach could be used to examine group differences in emotion-specific ER, such as gender or age differences. Using this EMA approach would therefore enable emotion-specific differences in ER to be examined in a more naturalistic setting and may increase the accuracy of ER data.

7.3. Gender Differences in Emotion Regulation Often Depend on the Specific Emotion

A key aim of the present research was to examine if there are gender differences in ER, and if these gender differences depend on the specific emotion. Males and females express specific emotions differently (Chaplin, 2015; Chaplin & Aldao, 2013), and receive different cultural messaging about specific emotions (Root & Denham, 2010), and so it would be logical that gender differences in ER also depended on the specific emotion. It emerged that gender differences in ER often, but not always, depend on the specific emotion being regulated.

7.3.1. Gender Differences in Reappraisal

One of the most important findings to emerge from the present research was that gender differences in reappraisal were emotion-specific. In previous research, there have been mixed findings regarding gender differences in reappraisal, with some studies finding that females use reappraisal more (Gullone et al., 2010; R. C. Martin & Dahlen, 2005; Nolen-Hoeksema & Aldao, 2011; Smrtnik-Vitulić & Prosen, 2016; Spaapen et al., 2014), others finding that males use reappraisal more (Esmaeilinasaba et al., 2016; Öngen, 2010), and some research showing no gender differences in reappraisal (Garnefski et al., 2004; Gross & John, 2003; Haga et al., 2009; Zlomke & Hahn, 2010). A reason for these mixed findings may be that previous research failed to examine gender differences in reappraisal in an emotionspecific manner, which was addressed in the present research.

Specifically, in Study 1, females were more likely than males to report using reappraisal to regulate fear, but this gender difference was not found for sadness or anger. Not only do females report using reappraisal more, but females were also more effective than males at reducing fear using reappraisal (as found in Study 2a). These findings indicate that females may be more effective at cognitively coping with a fearful situation, and may do so

more often than males. This finding brings clarity to an area which has been plagued with confusion and ambiguity, and demonstrates that clearer findings emerge when gender differences in ER are examined in an emotion-specific manner. Crucially, if reappraisal had been measured in the present research in an emotion-general manner, then these clear gender differences may not have been found.

Compared to males, females self-report experiencing more fear and are more likely to develop anxiety disorders (McLean & Anderson, 2009). Females may also have an attentional bias towards fearful stimuli (Conway et al., 2007). If it is the case that females experience more fear than males, then it is possible that females have developed effective ways of coping with this emotion, such as using reappraisal. This is in line with the findings of the present study that females are more likely to use reappraisal for fearful situations, and do so more effectively than males. According to appraisal theories (Lazarus & Folkman, 1984; Roseman & Evdokas, 2004), and the process model (Gross, 1998b), emotions arise from an appraisal of a situation. Females are thought to have more negative threat appraisals than males (Kucharska, 2017; Tamres et al., 2002) which may result in a larger fear response (Campbell & Muncer, 2017). Reappraisal involves changing an initial appraisal of a situation, and so it is possible that when females were asked to use reappraisal during the ERT they were able to effectively reduce their fear response by directly altering this negative threat appraisal. Females had a larger fear response than males when no ER strategy was instructed to be used (i.e., females reported higher fear levels in the no regulation condition). This means that females were able to use reappraisal to reduce their fear to a similar level to males'. This finding is in line with the proposal that females have developed a more effective way of coping with fear using reappraisal.

7.3.2. Gender Differences in Distraction

No gender differences in the reported use of distraction emerged in Study 1, either when this was a specific ER strategy reported as 'distraction' by the participant (e.g., I distracted myself), or when different forms of distraction (such as watching TV or listening to music) were combined into one wider distraction category. However, in Study 2a, important findings regarding gender differences in distraction effectiveness emerged. Compared to males, females were more effective at reducing sadness and anger using distraction.

These findings indicate that even though there are no gender differences in how often males and females use distraction as an ER strategy, when they have the opportunity to use distraction (such as in the ERT), females are more effective than males at regulating sadness and anger using distraction. These findings are in contrast with the wider literature, in which males are generally more likely than females to report using distraction, at least when they feel sad (Gomez-Baya et al., 2016; Trives et al., 2016).

This discrepancy between the present study and the wider literature is potentially due to two reasons. Firstly, cognitive distraction, which involves removing attention from an emotional stimulus, was measured in Study 2a. In contrast, a mix of cognitive distraction and behavioural distraction were measured in Gomez-Baya et al. (2016) and Trives et al. (2016) and Study 1. Therefore, it is possible that males may be more likely to use behavioural forms of distraction, whereas females may be more likely to use cognitive distraction, which is in line with research demonstrating that females use cognitive ER more than males (Garnefski et al., 2004).

Furthermore, previous studies of gender differences in distraction (e.g., Gomez-Baya et al., 2016; Trives et al., 2016) measured how often males and females used distraction (i.e., ER frequency). On the other hand, how effectively males and females were able to reduce specific negative emotions using distraction were measured in the present study (i.e., ER

effectiveness). This highlights the important point that how often a strategy is used is not necessarily the same as how skilfully it is used to reduce emotions. It is possible that males use distraction more in certain contexts, but females are able to implement this more effectively when they are instructed to do so (such as during the ERT). It should also be noted that although females were more effective at reducing distraction, males were also able to effectively reduce sadness and anger using distraction, albeit to a lesser extent than females.

7.3.3. Gender Differences in Expressing Sadness and Fear

Another important finding was that females were more likely than males to report venting or expressing their emotions, and this was found for sadness and fear, but not anger. This mirrors the findings from the wider literature that females express feelings of sadness and fear more than males (Allen & Haccoun, 1976; Chaplin, 2015; Chaplin & Aldao, 2013) and emotion-specific findings that females are more likely to verbally express emotions of sadness (Rivers et al., 2007).

Cultural messaging regarding the appropriate expression of emotion for males and females may contribute to gender differences in the expression of sadness and fear. Individuals are taught that sadness and fear are more 'feminine' emotions, whereas anger is thought to be a more 'masculine' emotion (Root & Denham, 2010). In the present research, conforming to ideals of femininity were found to be correlated with reporting higher sadness and fear during the ERT. This indicates that cultural messaging around appropriate emotional behaviour for males and females may influence emotional expression of sadness and fear, as well as selfreporting of this emotional expression.

Another important point is that expressing emotion is an emotional behaviour and may not be regarded as an ER strategy as such. The method used in Study 1 relies on participants having an awareness of their own ER strategies, and being able to accurately

self-report these strategies. When participants were asked what they did to try to feel less of an emotion in Study 1, participants generally reported emotional behaviours, rather than more cognitive ER such as rumination or suppression. The high percentage of participants reporting behaviours may indicate that participants had limited insight into their own ER strategies. This may have been a particular issue for males and adolescents, as these groups had higher levels of missing data in Study 1, which may indicate that they had difficulty reporting on their ER. This is in line with the dual process model of ER (Gyurak et al., 2011), which states that ER has both implicit and explicit components. It may be the case that individuals may be less able to provide information about the implicit aspect of ER, as this is outside of conscious awareness. In the present study, this limitation was addressed by using the ERT in Study 2, which does not rely on participants having this insight into their ER processes. Therefore, in future research, it may be useful to use complementary measures of emotion-specific ER – a measure which asks participants about their explicit ER (such as the Study 1 approach), and a measure which captures more implicit ER processes (such as the ERT).

7.3.4. Similarities Between Males and Females

Although prominent gender differences in ER emerged in the present research, it should be noted that there were also some similarities between males and females. Females reported a higher number of strategies than males in Study 1, which is consistent with other studies (Goubet & Chrysikou, 2019; Rivers et al., 2007; Sanchis-Sanchis et al., 2020), and there was no evidence that males use any ER strategy more than females. This means that males and females use some of the same ER strategies to regulate their emotions, but females also have additional ER strategies available to them. There were no gender differences in the reporting of some ER strategies, such as avoiding the situation, or experiential avoidance,

which shows that a similar proportion of males and females use these strategies. Furthermore, in Study 2, males and females were often both effective at reducing their negative emotions using ER strategies during the ERT, but females were able to reduce their negative emotions to a greater extent than males (i.e., females were more effective at ER compared to males). For example, males were able to reduce feelings of sadness using distraction in the ERT, but females were able to do this to a larger extent than males. Therefore, the present research indicates that males and females share many aspects of ER, and yet differ in others.

7.3.5. Potential Vulnerability of Males

The gender differences in ER that emerged in the present research may also highlight some areas of vulnerability for males. In Study 1 fewer males than females reported using all ER strategies, which may indicate that males are underutilising ER strategies that could potentially be adaptive. Males were also less likely to engage in social support, which indicates that males may be less likely to reach out for support when they are experiencing negative emotions. Further, males were less effective at reducing negative emotions in Study 2a, and generally had lower ER effectiveness scores than females. This may indicate that males are less skilled at using cognitive ER strategies such as reappraisal and distraction for reducing their negative emotion. This is broadly in line with findings from previous research that males use cognitive ER less than females (Garnefski et al., 2004).

Compared to females, males have higher rates of substance use (R. K. McHugh et al., 2018), and are more likely to die by suicide (Dougall et al., 2017; Jordan & McNiel, 2020; S. Lee et al., 2019; Miranda-Mendizabal et al., 2019; P. L. Mok et al., 2012; Mościcki, 1994; National Records of Scotland, 2021; Stark et al., 2004). Cognitive ER strategies such as reappraisal and distraction generally have a positive impact on variables associated with mental health (Aldao et al., 2010; In et al., 2021; Polanco-Roman et al., 2015; Stanley et al.,

2021). For example, reappraisal helps to reduce negative emotion amongst individuals who engage in self-harm behaviours (In et al., 2021). Similarly, distraction is associated with fewer instances of self-harm with suicidal intent (Polanco-Roman et al., 2015), less intense suicidal ideation (Stanley et al., 2021), and reduces the urge to self-harm regardless of intention (In et al., 2021). Therefore, encouraging the use of adaptive cognitive ER strategies such as reappraisal and distraction amongst males may help to address these issues and aid in efforts for suicide prevention and promoting mental health.

7.3.6. The Role of Culture

It should be noted that it is difficult to know from the findings of the present research whether the pattern of gender differences found are culture-specific. The present research was conducted in Scotland, but there is evidence that ER, and gender differences in ER, may have cultural differences. For example, the use of reappraisal and suppression differs by country (Haga et al., 2009), and individuals from Western countries generally report disengaging less than individuals from Eastern countries (E. Davis et al., 2012). Reappraisal may have a stronger relationship with depressive symptoms in Eastern cultures, whereas anger suppression may be more strongly related to depressive symptoms in Western cultures (Kwon et al., 2013). Gender differences in ER have also been found to depend on the cultural context (E. Davis et al., 2012). Therefore, caution should be taken when generalising the findings of the present study to other cultures. Future research may wish to examine the role of culture in gender differences in the regulation of specific emotions.

7.3.7. Conducting Emotion-Specific ER Research Amongst Other Groups

The present research focused on differences between two groups – males and females. These groups were chosen because it was important that the studies had well-defined gender groups to examine these differences. However, as discussed in Section 2.2.1, gender is a complex social concept which can change over time (J. L. Johnson et al., 2009), and different gender identities are becoming more commonplace (Hester et al., 2020). The individuals that took part in this research had a gender identity which matched the sex they were assigned at birth, but some people have a different gender identity to the gender they were raised and socialised as (Fausto-Sterling, 2018; J. L. Johnson et al., 2009). The findings of the present study cannot necessarily be applied to these individuals.

Although this research examined group differences between males and females, it is impossible to comment on the cause of these gender differences. The present research can only establish that these gender differences exist, not where they came from. Therefore, an important direction for future research in this field is to consider ER processes amongst individuals who identify with other gender categories, such as transgender, non-binary, or gender fluid individuals. Gender and ER processes are complex and may be influenced by both biological and social factors, and so it is important that these processes are examined amongst different groups.

These groups are often vulnerable to mental health problems, and are at a high risk of suicidal ideation and self-harm (McNeil et al., 2012). As ER has a relationship with variables associated with mental health (Brewer et al., 2016; S. L. Johnson et al., 2016; LeBlanc et al., 2020; Morris et al., 2015; Ranney et al., 2017), adaptive ER skills which can be developed through ER training may represent an important protective mechanism for these individuals. Furthermore, it is particularly important to conduct this research in an emotion-specific manner, because there is evidence that people receive different messages about specific emotions such as sadness and anger based on their gender (L. R. Brody, 1985; Root & Denham, 2010), which may play a role in how these emotions are regulated. Therefore, extending this gender research to other groups is an important direction for future research.

7.4. Gender Differences May Exist in Engagement Emotion Regulation

Based on a review of the ER literature, it was hypothesised in the present research that females may be more likely to use ER strategies which involve engaging with the emotional situation, and males may be more likely to use strategies which involve disengaging or distracting from the emotional situation, which is known as the engagement hypothesis. This hypothesis was largely based on the observation that females are more likely to use rumination (D. P. Johnson & Whisman, 2013; Nolen-Hoeksema & Aldao, 2011; Zlomke & Hahn, 2010), whereas males use avoidance and distraction more (Trives et al., 2016; Zimmermann & Iwanski, 2014). This engagement hypothesis was tested in Study 1, by categorising ER strategies into either engagement, disengagement, or distraction ER using a framework developed by Parkinson and Totterdell (1999). Gender differences emerged in the use of engagement, with females being more likely to report using engagement ER.

The engagement category largely consisted of expressing/venting emotion, and seeking social support, and so the gender differences in engagement were driven by these strategies. This means that gender differences in engagement in the present study were not due to rumination, as rumination made up only 2% of this category. Thus, it cannot be inferred that females used engagement as a strategy more generally, but rather females report using specific strategies such as venting and seeking social support more than males. In other words, it cannot be inferred from the results of the present research that females are more likely than males to engage with their emotions or the emotional situation during ER. But rather, females are more likely to use specific ER strategies such as seeking social support, or venting/expression emotions. However, it should also be noted that as ER was measured through self-report in Study 1, individuals may have had less insight into their cognitive ER

processes such as rumination, and so reported more behavioural strategies such as seeking social support that they are more familiar with.

7.4.1. Limitations of Parkinson & Totterdell Framework

In Study 1, the taxonomy of ER strategies developed by Parkinson and Totterdell (1999) was used to code the ER strategies into wider categories. This framework was chosen because it provides valuable guidance on allocating ER strategies into broad categories of engagement, disengagement, and distraction ER, which enabled the engagement hypothesis to be tested. However, it is possible that this framework was not the most appropriate method for testing this hypothesis, as gender differences emerged in some strategies which were classed as engagement (e.g., social support, venting) but not others (e.g., problem solving, self-talk). It is possible that these categories were too broad to capture meaningful gender differences. Also, the level of intercoder agreement for categorising ER strategies into these wider categories was $\kappa = .59$. This represents moderate agreement between the coders (J. Cohen, 1960), but may indicate that there is an issue with the reliability of these categories. Therefore, rather than classing ER strategies into wider categories, it may be helpful in future research to examine gender differences in engagement and disengagement by measuring the individual ER strategies, as clearer gender differences emerged in the present research when ER strategies were examined separately.

7.4.2. Examining Engagement ER in Future Research

To test the engagement hypothesis in future research, it may be useful to use the EMA approach described in Section 7.2.3. Using this approach, participants could be asked to report on their use of specific ER strategies that can be defined as engagement (e.g., rumination, reappraisal), or disengagement (e.g., avoidance, suppression, distraction) to
regulate their specific emotions as they experience them in their daily lives. Participants could also be asked to describe their intention or goal in regulating their emotions (e.g., attempting to distract themselves). Collecting information about a participant's intention when using ER may provide insight into whether an individual is attempting to engage or disengage with the emotion or emotional situation. Research has shown that attempting to avoid an emotional experience rather than accepting the experience may be maladaptive (Wolgast & Lundh, 2017), and so collecting this information using EMA is useful. Gender differences in engagement and disengagement ER could be tested using such an approach.

7.4.3. Limitations of the Adult Sample in Study 1

It should also be noted that there were limitations with the adult sample in Study 1. The adult sample was smaller than the other age groups, and there were few adult males in the sample. This means that the ability to detect a gender difference in ER amongst the adult age group was reduced. There is a chance that gender differences in the age groups follow a similar pattern but have not been picked up in the analysis due to Type II error. It was surprising that no gender differences emerged in the adult age groups, as previous research has found gender differences in engagement, disengagement, and distraction strategies (D. P. Johnson & Whisman, 2013; Trives et al., 2016; Zimmermann & Iwanski, 2014). Therefore, it is possible that a Type II error occurred with the adult sample. When adults were included in Study 2, gender differences in ER effectiveness emerged. Furthermore, it is also possible that the ER categories used (engagement, disengagement, distraction) may have been too broad to detect any meaningful differences, and perhaps the specific ER strategies are more accurate for highlighting gender differences in ER.

7.5. The Effectiveness of Reappraisal and Distraction May Depend on the Specific Emotion Being Regulated

It emerged in the present research that some strategies are more effective for reducing specific emotions compared to others. In Study 2a, reappraisal was more effective than distraction for reducing fear. On the other hand, distraction was more effective than reappraisal for reducing sadness and anger (although this also depended on gender). This demonstrates that not only does the ER strategy used differ by emotion (as found in Study 1), but also that certain strategies may be more effective at reducing specific emotions. These findings have important theoretical and practical implications, which will be discussed in Section 7.6.

7.5.1. Implications for the Process Model of ER

According to Gross's (1998b) process model of ER, distraction is thought to be a more effective ER strategy than reappraisal because it appears earlier in the unfolding of an emotional response (Gross, 1998b). Empirical evidence has shown that reappraisal and distraction are both effective at reducing negative emotion (Gross, 1998a; Hermann et al., 2017; Quinn & Joormann, 2020; Smoski et al., 2014; Thiruchselvam et al., 2011), and there is some evidence that distraction may be more effective than reappraisal in certain contexts (Sheppes et al., 2009; Sheppes & Meiran, 2007, 2008; Smoski et al., 2014; Thiruchselvam et al., 2011), which supports the premise of the process model.

However, a limitation of the process model, and research testing this model, is that it does not take into account the role of specific emotions in the effectiveness of reappraisal and distraction. The present research builds on previous work and provides value to the literature by demonstrating that not only is it the point at which the ER strategy appears in the emotion cycle, but also the specific emotion being regulated, that determines the effectiveness of the ER strategy. The present research indicates that reappraisal is more effective than distraction within the context of a fear response, but is perhaps less effective than distraction for reducing sadness or fear. Thus, the specific emotion being regulated influences the effectiveness of a particular strategy, which is not taken into account in the process model.

The process model describes the unfolding of an emotional response, and the strategies that can be used to intervene in this process, in terms of general emotion. However, the present research demonstrates that what we know about one emotion cannot necessarily be extrapolated to all emotions, or negative emotion in general. The process model has the potential to provide a theoretical framework to emotion-specific ER.

The process model could aid in understanding emotion-specific ER by allowing the regulation of specific emotions to be conceptualised as individual process models, as shown in Figure 7.1. This would involve an emotional situation occurring which is likely to induce a specific emotion, such as losing something important (sadness), interference with an important goal (anger), or a threatening situation (fear; Ekman & Cordaro, 2011). The individual pays attention to this situation, and then has an appraisal which will give rise to the specific emotion of sadness, anger, or fear (Scherer & Moors, 2019). Importantly, different ER strategies may intervene in the separate process models for these specific emotions, and their effectiveness may depend on the specific emotion being regulated. Therefore, the use of reappraisal in a fear process model may have a different effect on the emotion than the use of reappraisal in the sadness model.

Figure 7.1

The Process Model of ER as a Framework for Emotion-Specific ER



Describing ER using this emotion-specific approach will provide researchers with a theoretical framework for conceptualising the regulation of specific emotions, and examining the use and effectiveness of ER strategies within different emotional contexts. This would enable a range of issues to be examined, including individual differences in ER, group differences in ER (such as gender), comparing different ER strategies, comparing the regulation of different emotions using the same strategy, and investigating the relationship between emotion-specific ER and other outcomes (i.e., ER adaptiveness). Therefore, the process model has the potential to be a useful framework for understanding ER in an emotion-specific manner.

7.5.2. Implications for Emotion Theory

The findings of the present research also have implications for wider emotion theory. As discussed in Chapter 1, an area of debate in the literature is whether emotions can be thought of as discrete categories, such as sadness, anger, or fear, or are more continuous in nature and differ only on dimensions such as valence and arousal. The findings of the present research demonstrate that the use of ER (Study 1) and the effectiveness of ER (Study 2a) often depend on the specific emotion being regulated. These findings support a more discrete perspective of emotions, because they indicate that the way emotions are regulated depends on the specific emotional response. If emotions only differed in terms of broad dimensions, then emotion-specific differences in ER would not be likely to emerge.

The present research used subjective experience as the measure of specific emotions, and so the findings of this research indicate that individuals understand their emotional experience in terms of specific emotions, such as sadness, anger, and fear, as individuals were able to accurately report on their emotional experience. This is in line with research which shows that there are emotion-specific differences in reported emotional experience (Lench et al., 2011). This means that regardless of whether emotions are conceptualised as a product of social learning (as with the social constructivist approach; L. F. Barrett, 2017) or a hardwired innate response (as with discrete emotion theory; Panksepp, 1982), the present research demonstrates that individuals understand and categorise their emotional experience using discrete emotion labels, such as sadness, anger, and fear. Therefore, it is important to continue to examine ER in an emotion-specific manner, so that nuances between these specific emotions are not overlooked.

7.5.3. Implications for Therapeutic Interventions

The emotion-specific findings of the present research also have implications for therapeutic interventions for mood and anxiety disorders. ER is amenable to change, and there is evidence that ER training can improve mental health (Bomyea et al., 2020; Denny, 2020; Goldin et al., 2012). Training programmes that build skills in adaptive ER strategies such as reappraisal have been shown to improve ER and have a positive impact on mental health (Denny, 2020), such as lowering symptoms of anxiety (Kivity & Huppert, 2016), and reducing self-reported negative emotion over time (Denny et al., 2015; Denny & Ochsner, 2014).

The findings of the present research indicate that the emotional context within which a strategy is used may have an impact on its effectiveness. If reappraisal is indeed a more effective strategy for regulating feelings of fear, as the Study 2a findings suggest, then this strategy may be utilised in therapies aiming to treat more fear-based disorders, such as anxiety disorder or phobias (Hermann et al., 2013; Langeslag & van Strien, 2018). Reappraisal involves cognitively changing the meaning of a situation, and is similar to some of the key tenets of cognitive behavioural therapy (CBT). Indeed, CBT is useful for treating anxiety disorders (Goldin et al., 2017; Goldin et al., 2012).

Distraction emerged as being more effective for reducing sadness and anger. From a therapeutic perspective, using distraction to regulate sadness may be a useful strategy for sadness-based problems such as depression, or may be effective in addressing anger-based problems such as aggressive behaviour. Distraction is a key component of Dialectical Behavioural Therapy (DBT), and this therapy has been useful for treatment of aggressive behaviour, predominantly amongst individuals with a diagnosis of borderline personality disorder (Frazier & Vela, 2014). Distraction is also useful for reducing aggression more generally (Gallagher & Parrott, 2011; Subramani et al., 2019). It may also be beneficial in

reducing negative emotion in individuals with clinical depression (Smoski et al., 2014). Therefore, the present research indicates that it may be helpful to use different approaches to address disorders that are characterised by specific emotional challenges.

7.6. The Relationships Between Gender, Emotion-Specific ER, and Variables Associated With Mental Health

An aim of the present research was to examine whether gender differences in emotion-specific ER were related to gender differences in variables associated with mental health (depressive symptoms, self-harm behaviours, and suicidal ideation). These variables were chosen because they have prominent gender differences. Specifically, compared to males, females have higher rates of depression (H. Chen et al., 2019; Girgus & Yang, 2015; Nolen-Hoeksema & Aldao, 2011), and are more likely to engage in self-harm behaviours, irrespective of motivation (Hawton et al., 2012; Knudson et al., 2020; Madge et al., 2008; O'Connor et al., 2009). However, compared to females, more males die by suicide (National Records of Scotland, 2021).

ER is associated with variables relating to mental health (Aldao et al., 2010; Huffziger et al., 2009; Polanco-Roman et al., 2015), and there is evidence that ER may mediate gender differences in depressive symptoms (Nolen-Hoeksema et al., 1999). This has been investigated in previous research in an emotion-general manner, but the present research was the first to examine this issue in an emotion-specific manner, by measuring the regulation of sadness, anger, and fear individually. In the present research, females had a larger reduction in anger using distraction, and this was associated with higher depressive symptoms, and lower suicidal ideation. These findings implicate the regulation of anger as being related to depressive symptoms and suicidal ideation, and demonstrates that distraction as an ER strategy may be involved in gender differences in mental health.

7.6.1. Distracting From Anger is Associated with Gender Differences in Depressive Symptoms

Effectively reducing anger using distraction, but not sadness or fear, was related to gender differences in depressive symptoms. This may highlight an important mechanism for explaining gender differences in depression. Response Styles Theory (RST; Nolen-Hoeksema, 1987) attempts to explain the observed gender differences in depression in terms of ER. According to RST, responding to a low mood by using distraction or problem solving is adaptive and leads to fewer depressive symptoms. Using rumination is thought to be maladaptive and may contribute to increased depressive symptoms. According to RST, females are more likely to use rumination, whereas males use distraction more, and this contributes to gender differences in depression (Nolen-Hoeksema, 1987, 1991; Nolen-Hoeksema et al., 1999; Nolen-Hoeksema et al., 1993; Nolen-Hoeksema et al., 2008).

The findings of the present research are in contrast with two of the key tenets of RST. Firstly, females were more effective at using distraction to reduce anger, which is in contrast with the proposal of RST that males use distraction more. As previously discussed, males often report using distraction more often than females when they feel sad or depressed (Gomez-Baya et al., 2016; Trives et al., 2016). It may be the case that males use distraction more often, but females are more effective at implementing this strategy when they do so within the context of experiencing anger.

Secondly, the reduction of anger using distraction was associated with higher depressive symptoms in the present research, which is in contrast with the tenet of RST that distraction has an adaptive relationship with depression and leads to reduced depressive symptoms (Nolen-Hoeksema, 1987). In the present study, distraction from anger had a maladaptive relationship with depressive symptoms. The relationship between anger

distraction and depressive symptoms was examined using a cross-sectional design, which means that these variables were measured at the same time point. This means that it is equally plausible that anger distraction impacts on depressive symptoms, but also that higher depressive symptoms contribute to anger distraction.

It is possible that distracting from feelings of anger is a maladaptive response to anger which leads to increased depressive symptoms over time. Experiencing problematic anger is associated with depressive symptoms (Abdolmanafi et al., 2011). In terms of general negative emotion, being more effective at reducing negative emotion using distraction during the ERT was associated with higher depressive symptoms amongst depressed individuals (Smoski et al., 2014), which is in line with the present findings. Females may be more effective than males at reducing anger using distraction because they are more practiced at doing so. When given a choice, females choose to distract from feelings of anger (Rusting & Nolen-Hoeksema, 1998). Females may distract from anger in an attempt to avoid this emotion, as anger is perceived as less acceptable for females to express (Brescoll & Uhlmann, 2008). This is in line with research which has found that females are more likely to suppress their emotions (Cox et al., 2000), which is similar to distraction as it involves disengaging with the emotion. There are many factors which contribute to the gender differences in depression (Nolen-Hoeksema, 1987; Nolen-Hoeksema et al., 1999), but the findings of the present research highlight anger distraction as a factor which may contribute to these gender differences.

It is also possible that individuals with higher depressive symptoms are more drawn to using anger distraction as an ER strategy. Distraction is less cognitively demanding than other strategies such as reappraisal (Sheppes et al., 2009; Sheppes & Meiran, 2008), and so may be a more attractive option for individuals coping with higher depressive symptoms. There is evidence that depressed people are likely to use distraction to avoid negative

emotion (Wolgast & Lundh, 2017). However, to more fully understand the relationship between anger distraction and depressive symptoms, this must be investigated using a longitudinal study design, to assess the causal relationships between these variables over time, as discussed in Section 7.6.5.

In sum, the findings of the present study implicate anger distraction in explaining gender differences in depression. Building on RST, it is possible that when females distract from feelings of anger, they are effective at reducing anger using this strategy, but this may have a maladaptive relationship with depression. This may be because distracting from anger exacerbates depressive symptoms, or it may also indicate that people with higher depressive symptoms are more likely to distract from feelings of anger and are perhaps more practiced in using distraction when they feel angry. This finding would indicate that within the context of anger, distraction is not an adaptive strategy, which is in contrast with RST, in which distraction is regarded as an adaptive way to cope with feelings. There is evidence that when using distraction to avoid emotion, it is maladaptive, but when using distraction while also accepting an emotion, it is adaptive (Wolgast & Lundh, 2017).

7.6.2. Distracting from Anger is Associated with Gender Differences in Suicidal Ideation

In relation to suicidal thoughts, it was found in the present study that females were more effective at reducing anger using distraction, and this was associated with lower suicidal ideation. Problematic anger predicts suicidal ideation and self-harm with suicidal intention (Dillon et al., 2020). Using distraction to cope with negative emotions more generally is associated with a lower risk of engaging in self-harm behaviours with or without suicidal intention (Polanco-Roman et al., 2015), and so it may be the case that distracting from anger is adaptive in this context, at least for females. However, it should be noted that although anger distraction was related to suicidal ideation, anger distraction was not related to selfharm behaviour in the present research. This may be due to the measure of self-harm used, which measured self-harm behaviours using questions which specified no suicidal intent, and some questions which did not specify intent, and this measure likely captured a mix of intentions. It may be the case that as ER is related to suicidal ideation, if self-harm with suicidal intent was measured, then anger distraction may have been associated with this.

The finding that more effective anger distraction was associated with lower suicidal ideation is also in line with a study by Stanley et al. (2021). The use of distraction and positive activities to cope with negative emotions was found to lower the intensity of suicidal thoughts amongst individuals with a diagnosed mood disorder or borderline personality disorder (Stanley et al., 2021). Although females generally report higher levels of suicidal ideation (Barzilay et al., 2019; Lu et al., 2020; Stephenson et al., 2006), the findings of the present research may represent an important mechanism for reducing suicidal thoughts, by utilising the use of distraction to reduce anger. Reducing anger using distraction may have an adaptive relationship with suicidal ideation.

Suicidal ideation is one of the biggest risk factors for future self-harm with suicidal intention (García-Vega et al., 2018; Reinherz et al., 2006), and so it is important to understand the factors which may be associated with suicidal ideation. The findings of the present study indicate that being skilled in using distraction to reduce feelings of anger may have an adaptive relationship with suicidal ideation, which may highlight a potential protective factor for future suicide risk.

Dialectical behavioural therapy (DBT) was originally developed for individuals with borderline personality disorder (BPD) who engaged in self-harming behaviours, particularly females (Linehan et al., 1993). The aim of DBT is to train individuals in a range of coping skills. A key component of DBT is about teaching individuals how to accept strong feelings, while at the same time being able to distract from these in positive ways, such as doing

something else, imagining pleasant events, stop thinking about it, think about something else, remind yourself of positive experiences, ask others for help, count your breath, and take a break (Perepletchikova et al., 2011). Therefore, distraction is an important aspect of DBT.

There is mounting empirical evidence that DBT reduces self-harm and suicidal ideation (Feigenbaum, 2007; Flynn et al., 2020; Geddes et al., 2013; Guillén Botella et al., 2021; Perepletchikova et al., 2011; Rizvi & Fitzpatrick, 2021; Walton et al., 2020; Westad et al., 2021; Yang et al., 2020). The findings of the present research build on this evidence by demonstrating that emotional context may also play a role in the effectiveness of distraction during DBT. Specifically, being able to effectively use distraction when experiencing anger may be adaptively related to suicidal ideation (i.e., lower suicidal ideation). However, the results of the present research indicate that using distraction when experiencing sadness or fear may be less important in relation to suicidal ideation. In addition, it was also found that being able to effectively reduce anger using distraction was particularly important for females. Therefore, the findings of the present research may contribute to DBT practice by highlighting that experiencing anger and being able to effectively use distraction to reduce anger is important for suicidal ideation, particularly for females with high suicidal ideation.

7.6.3. Why Does Anger Distraction Have a Different Relationship With Depressive Symptoms and Suicidal Ideation?

Although distraction is generally regarded as an adaptive strategy, it has been acknowledged in the literature that whether or not distraction is adaptive depends on how it is used. When individuals distract from their emotions but also accept their emotion, distraction is adaptive in that it is associated with greater wellbeing, but when people use distraction in an attempt to avoid feeling an emotion, then it has maladaptive consequences, such as lower wellbeing (Wolgast & Lundh, 2017). This paradox is reflected in the finding that anger

distraction has a different relationship with depressive symptoms and suicidal ideation, which are generally positively correlated with one another, both in the present research (see Section 6.4.2.4) and the wider literature (De Beurs et al., 2020; Gilchrist & Sadler, 2019; Mars et al., 2019). It is possible that using distraction helps to draw attention away from suicidal thoughts (Simon et al., 2016; Stanley et al., 2021), but a consequence of this is an increase in depressive symptoms (Smoski et al., 2014).

7.6.4. Implications for Personalising Mental Health Interventions

There is evidence that ER training can improve mental health (Bomyea et al., 2020; Denny, 2020; Goldin et al., 2012), and these interventions can have a positive impact on variables associated with mental health over time (Gratz et al., 2015; Kiosses et al., 2018; LeBlanc et al., 2020; LeBlanc et al., 2017; Morris et al., 2015; Ranney et al., 2017). Recently, there has been a move towards developing increasingly personalised interventions to train adaptive ER skills (Denny, 2020). The findings of the present research may help to guide tailoring these interventions. Specifically, it was highlighted that both gender and the emotional context may have an impact on the clinical utility of ER. Females may be more effective at using anger distraction than males, which may indicate that therapies which focus on developing these skills, such as Dialectical Behavioural Therapy or Attentional Bias Modification, may be better suited for females. However, it also highlights that males may need more support in this area.

Similarly, the present research demonstrates that the effectiveness of therapies may also depend on the emotional context and the specific problem being targeted. For example, using anger distraction may be helpful for individuals who are predisposed to having suicidal thoughts, but perhaps not individuals with depression. Distraction is regarded as a less cognitively demanding strategy than reappraisal (Sheppes et al., 2009; Sheppes & Meiran,

2008), and so it may be the case that distraction is more appropriate for certain individuals, such as individuals who are impaired (Smoski et al., 2014). In sum, the findings of the present research indicate that the effectiveness of ER may depend on gender and the emotional context, and so may help to guide personalising mental health interventions.

7.6.5. Testing Causal Relationships Between Gender, ER, and Variables Associated with Mental Health Prospectively

It should be noted that in Study 2, variables associated with mental health were measured at the same time as ER processes, using a cross-sectional design. As a result, claims cannot be made about the causal relationships between the study variables. Theoretically, the assumption is that the way that individuals regulate their emotions has an impact on their mental health. This is the key premise of RST for example, and this has been supported with some empirical evidence (Huffziger et al., 2009; Nolen-Hoeksema et al., 1999). However, it is equally likely that individuals struggling with poor mental health may be more inclined to use certain strategies, particularly if they require less cognitive effort than other strategies, such as distraction (Sheppes et al., 2009; Sheppes & Meiran, 2008).

Thus, an important direction for future research is to examine the relationships between gender, emotion-specific ER, and variables associated with mental health prospectively. This will help to establish if the regulation of specific emotions contributes to mental health outcomes over time, or if an individual's mental health determines the ER strategies that they use. Further, examining whether males and females have a different trajectory of emotion-specific ER over time and how this relates to mental health is important, as the findings of the present research indicate that the regulation of some emotions may mediate the relationship between gender and variables associated with mental health. Therefore, by examining these variables longitudinally, the longer-term impact of

gender differences in emotion-specific ER on various mental health outcomes can be examined.

7.7. Some Further Methodological Considerations of this Research

7.7.1. Sampling Methodology

There are some limitations of the sampling methodology and generalisability of the findings. As with most psychological research, it can be a challenge to recruit participants to take part in research studies. As a result, opportunity sampling methods are often used, which involve selecting participants from a target group to take part in a study and then selecting anyone who is available and willing to take part providing they meet these criteria (Brondolo, 2021). This means that it is not a random sampling approach which is used.

The samples in the present research involved a large number of student participants, as well as Government analysts in Study 2. There is some evidence that undergraduate students can differ from the wider population on some measures (Gordon et al., 1986). It emerged in the present research that the student participants had higher levels of depressive symptoms, suicidal ideation, rumination, and were more likely to self-harm than the rest of the sample. However, this does not mean that valuable insights cannot emerge from research conducted with student participants (Druckman & Kam, 2009; Greenberg, 1987), but simply indicates that caution should be exercised when generalising these findings to the general population. Similarly, different age groups were used in the present research – adolescents and young adults in Study 1, and adults of all ages in Study 2, and so it is important to exercise caution when generalising these results to people of different ages and in making comparisons between the studies.

7.7.2. Applying the Results of the Emotion Regulation Task to a Real-World Setting

In research, there is often tension between balancing ecological validity with maintaining experimental control (Parsons, 2015). An important consideration when conducting experimental research is the extent to which any findings can be applied to a realworld setting. In the present research, participants viewed images that were carefully chosen to elicit specific emotions during the ERT. Although these images reliably induce specific emotions in a laboratory setting (Mikels et al., 2005; Riegel et al., 2016), it must be acknowledged that viewing an image of a particular situation during an experiment may not have the same emotional impact as experiencing this situation in person. For example, many of the images in the present research depicted threatening animals such as snakes. However, an emotional response is thought to arise from an appraisal of the significance of a situation to the individual's personal wellbeing (Lazarus, 1991). As the participant had the knowledge that they were in a safe environment during the ERT, they may not have had the same threat appraisal that gives rise to a genuine fear response as if they had encountered the snake in the wild.

However, the ERT has been used extensively in ER research (Albanese et al., 2019; Douw et al., 2020; Fitzgerald et al., 2019; Goldin et al., 2008; Gross, 1998a; Gross & Levenson, 1997; Jackson et al., 2000; Lazarus & Alfert, 1964; McRae et al., 2012b; Notarius & Levenson, 1979; Ochsner et al., 2002; Richards & Gross, 2000; Sheppes & Meiran, 2007; Sheppes et al., 2014; Sullivan & Kahn, 2020), and many important findings have emerged from this approach. The present research is a useful starting point for examining emotionspecific gender differences in ER. However, it may be helpful for future research to build on these findings by implementing measures which may have increased ecological validity and allow for ER in naturalistic emotion-inducting situations to be observed, such as EMA (as discussed in Section 7.2.3).

7.7.3. Range of Specific Emotions

Further, the present research examined the regulation of three specific emotions – sadness, anger, and fear. As reported in Section 1.3.3.2, these emotions were chosen because they are three of the key emotions described in Ekman's (1992) basic emotion theory, and are the main emotions focused upon in previous emotion-specific studies (Boland et al., 2019; Bujor & Turliuc, 2020; Endrerud & Vikan, 2007; Perchtold et al., 2019; Vishkin et al., 2020; Wong et al., 2018). The number of emotions examined was limited to three, because inducing emotions experimentally becomes increasingly challenging as more emotions are included in the study design. However, as highlighted in this chapter, what we now know through the present research about these three emotions cannot necessarily be generalised to other specific emotions. It is important that future research also captures other important emotions, such as shame, disgust, and surprise, as well as positive emotions such as joy and love (as discussed in Section 7.7.4). This will involve integrating these other specific emotions into ER theory, as well as developing new tools for measuring these emotions experimentally.

7.7.4. The Role of Positive Emotion

In the present research, negative emotion (specifically sadness, anger, and fear) was focused upon. The reason for examining negative emotion is that chronic negative emotion, and difficulties with regulating negative emotion, is a hallmark of many clinical disorders (Aldao et al., 2016; Aldao & Nolen-Hoeksema, 2010; Aldao et al., 2010; Cludius et al., 2020). An initial step in efforts to tackle poor mental health is examining the factors which contribute to adaptive (and maladaptive) regulation of negative emotions, and the relationship this has with mental health outcomes. However, an important next step for future research is to also investigate the role of positive emotion in mental health. According to the broaden-and-build theory of positive emotions (Fredrickson, 2001), discrete positive emotions such as joy, interest, pride, love, and contentment have a beneficial effect by widening the range of thoughts and actions that are available to individuals. For example, experiencing the emotion of interest may help individuals to process information and new experiences, and build self-esteem and creativity (Fredrickson, 1998). Similarly, experiencing positive emotions such as enjoyment and pride predicts subsequent academic achievement amongst children (Lichtenfeld et al., 2012; Pekrun et al., 2017). Positive emotions may also build resilience and help to manage future threats (Fredrickson, 2001), and has a positive impact on the ability to process information (Kuhbandner et al., 2011).

Consequently, the broaden-and-build theory highlights a potential future direction for ER research – to examine the factors which may allow for increased positive emotion to be experienced. As with ER research examining negative emotions, relatively few studies have examined the regulation of positive emotions, such as love, joy, affection, contentment etc., in an emotion-specific manner. Therefore, future ER researchers may wish to examine emotion-specific ER processes amongst positive emotions which have been found in previous research to impact on wellbeing, such as joy, interest, pride, love, and contentment (Fredrickson, 2001). Research on the regulation of negative emotions predominantly focuses on how to reduce negative emotions (Gross & John, 2003). However, research on positive emotions may examine the ER strategies that increase levels of these positive emotions, which may have an adaptive impact on wellbeing.

7.8. Conclusion

The most important message to take away from this research is that when it comes to ER, it cannot be assumed that what is known about one specific emotion, or negative emotion in general, can be applied to all emotions. The findings of the present research have provided clarity to a literature which is plagued by ambiguous findings. A reason for these unclear findings may be that previous research had not investigated these issues in an emotion-specific manner.

In the present research, it emerged that the ER strategy used depends on the specific emotional context. Similarly, gender differences in reported ER often depend on the specific emotion. Compared to males, females are more likely to use reappraisal when they are experiencing fear, and are more effective at doing so. Previous research into gender differences in reappraisal has produced mixed findings, and so the present research provides clarity to this issue.

Furthermore, there was some evidence that females may use engagement ER more than males, although this was driven by gender differences in seeking social support, and venting/expressing emotion. Gender differences in the reporting of ER strategies may change with age, although further emotion-specific research may be needed in this area. Similarly, young adults reported using engagement ER, and distraction to regulate sadness, more than adolescents, which may suggest that ER becomes more adaptive as people get older.

Additionally, the effectiveness of an ER strategy depended on the specific emotion being regulated. Reappraisal was more effective for reducing fear, whereas distraction was more effective for reducing sadness and anger (although this also varied by gender). It also emerged that ER may be associated with gender differences in depressive symptoms and suicidal ideation, but this depends on the specific emotion and the ER strategy being used.

Specifically, females were more effective at reducing anger using distraction, and this was associated with higher depressive symptoms but lower suicidal ideation.

This research has highlighted limitations with theories of ER, which presently do not take into account emotion-specific differences. It also provides support for discrete accounts of emotion. In relation to therapeutic interventions, the present research shows that emotional context, and the gender of the individual, may play an important role in the clinical utility of ER. These findings may help to develop increasingly personalised interventions to train adaptive ER skills and contribute to efforts in tackling poor mental health.

To conclude, the present research has demonstrated that the specific emotion being regulated matters, and that gender differences in ER, and the relationship between gender, ER, and variables associated with mental health, are often emotion-specific. Therefore, it may be beneficial for future research to adopt an emotion-specific approach when examining ER.

References

- Abdolmanafi, A., Besharat, M. A., Farahani, H., & Khodaii, M. R. (2011). The moderating role of locus of control on the relationship between anger and depression in patients with major depression disorder. *Procedia-Social and Behavioral Sciences, 30*, 297-301.
- Adolphs, R., Mlodinow, L., & Barrett, L. F. (2019). What is an emotion? *Current Biology*, 29(20), 1060-1064.
- Ajdacic-Gross, V., Weiss, M. G., Ring, M., Hepp, U., Bopp, M., Gutzwiller, F., & Rössler,
 W. (2008). Methods of suicide: International suicide patterns derived from the WHO mortality database. *Bulletin of the World Health Organization*, 86, 726-732.
- Albanese, B. J., Macatee, R. J., Stanley, I. H., Bauer, B. W., Capron, D. W., Bernat, E., Joiner, T. E., & Schmidt, N. B. (2019). Differentiating suicide attempts and suicidal ideation using neural markers of emotion regulation. *Journal of Affective Disorders*, 257, 536-550.
- Albert, P. R. (2015). Why is depression more prevalent in women? *Journal of Psychiatry & Neuroscience*, *40*(4), 219-221.
- Aldao, A., Gee, D. G., De Los Reyes, A., & Seager, I. (2016). Emotion regulation as a transdiagnostic factor in the development of internalizing and externalizing psychopathology: Current and future directions. *Development and Psychopathology*, 28(4), 927-946.
- Aldao, A., & Nolen-Hoeksema, S. (2010). Specificity of cognitive emotion regulation strategies: A transdiagnostic examination. *Behaviour Research and Therapy*, 48(10), 974-983.
- Aldao, A., & Nolen-Hoeksema, S. (2012). When are adaptive strategies most predictive of psychopathology? *Journal of Abnormal Psychology*, *121*(1), 276-281.

- Aldao, A., Nolen-Hoeksema, S., & Schweizer, S. (2010). Emotion-regulation strategies
 across psychopathology: A meta-analytic review. *Clinical Psychology Review*, 30(2), 217-237.
- Allan, S., & Gilbert, P. (2002). Anger and anger expression in relation to perceptions of social rank, entrapment and depressive symptoms. *Personality and Individual Differences*, 32(3), 551-565.
- Allen, J. G., & Haccoun, D. M. (1976). Sex differences in emotionality: A multidimensional approach. *Human Relations*, 29(8), 711-722.
- Ando', A., Giromini, L., Ales, F., & Zennaro, A. (2020). A multimethod assessment to study the relationship between rumination and gender differences. *Scandinavian Journal of Psychology*, 61(6), 740-750.
- Andover, M. S., Morris, B. W., Wren, A., & Bruzzese, M. E. (2012). The co-occurrence of non-suicidal self-injury and attempted suicide among adolescents: Distinguishing risk factors and psychosocial correlates. *Child and Adolescent Psychiatry and Mental Health*, 6(1), 11.
- Andover, M. S., Primack, J. M., Gibb, B. E., & Pepper, C. M. (2010). An examination of non-suicidal self-injury in men: Do men differ from women in basic NSSI characteristics? *Archives of Suicide Research*, 14(1), 79-88.
- Appleton, A. A., Buka, S. L., Loucks, E. B., Gilman, S. E., & Kubzansky, L. D. (2013).
 Divergent associations of adaptive and maladaptive emotion regulation strategies with inflammation. *Health Psychology*, *32*(7), 748.
- Appleton, A. A., Loucks, E. B., Buka, S. L., & Kubzansky, L. D. (2014). Divergent associations of antecedent-and response-focused emotion regulation strategies with midlife cardiovascular disease risk. *Annals of Behavioral Medicine*, 48(2), 246-255.

- Archer, J. (2004). Sex differences in aggression in real-world settings: A meta-analytic review. *Review of General Psychology*, 8(4), 291.
- Arnold, M. B. (1960a). Emotion and personality. Vol. I. Psychological aspects. Columbia University Press.
- Arnold, M. B. (1960b). Emotion and personality. Vol. II. Physiological aspects. Columbia University Press.
- American Psychological Association (2013). *Diagnostic and statistical manual of mental disorders (DSM-5®)*: American Psychiatric Pub.
- Babyak, M. A. (2004). What you see may not be what you get: A brief, nontechnical introduction to overfitting in regression-type models. *Psychosomatic Medicine*, 66(3), 411-421.
- Baert, S., De Raedt, R., Schacht, R., & Koster, E. H. (2010). Attentional bias training in depression: Therapeutic effects depend on depression severity. *Journal of Behavior Therapy and Experimental Psychiatry*, 41(3), 265-274.
- Bagby, R. M., Rector, N. A., Segal, Z. V., Joffe, R. T., Levitt, A. J., Kennedy, S. H., & Levitan, R. D. (1999). Rumination and distraction in major depression: Assessing response to pharmacological treatment. *Journal of Affective Disorders*, 55(2-3), 225-229.
- Bai, S., Elavsky, S., Kishida, M., Dvořáková, K., & Greenberg, M. T. (2020). Effects of mindfulness training on daily stress response in college students: Ecological momentary assessment of a randomized controlled trial. *Mindfulness*, 11(6), 1433-1445.
- Balsters, M. J., Krahmer, E. J., Swerts, M. G., & Vingerhoets, A. J. (2013). Emotional tears facilitate the recognition of sadness and the perceived need for social support. *Evolutionary Psychology*, 11(1), 148-158.

- Balzarotti, S., Biassoni, F., Villani, D., Prunas, A., & Velotti, P. (2016). Individual differences in cognitive emotion regulation: Implications for subjective and psychological well-being. *Journal of Happiness Studies*, 17(1), 125-143.
- Barbee, A. P., Cunningham, M. R., Winstead, B. A., Derlega, V. J., Gulley, M. R., Yankeelov, P. A., & Druen, P. B. (1993). Effects of gender role expectations on the social support process. *Journal of Social Issues*, 49(3), 175-190.
- Baron, R. M., & Kenny, D. A. (1986). The moderator–mediator variable distinction in social psychological research: Conceptual, strategic, and statistical considerations. *Journal* of Personality and Social Psychology, 51(6), 1173.
- Barrett, E. L., Mills, K. L., Teesson, M. J. A., & Psychiatry, N. Z. J. o. (2013). Mental health correlates of anger in the general population: Findings from the 2007 National Survey of Mental Health and Wellbeing. *Australian and New Zealand Journal of Psychiatry*, 47(5), 470-476.
- Barrett, L. F. (1998). Discrete emotions or dimensions? The role of valence focus and arousal focus. *Cognition & Emotion*, 12(4), 579-599.
- Barrett, L. F. (2017). The theory of constructed emotion: An active inference account of interoception and categorization. *Social Cognitive and Affective Neuroscience*, 12(1), 1-23.
- Barrett, L. F., Khan, Z., Dy, J., & Brooks, D. (2018). Nature of emotion categories: Comment on Cowen and Keltner. *Trends in Cognitive Sciences*, 22(2), 97-99.
- Barzilay, R., Calkins, M. E., Moore, T. M., Boyd, R. C., Jones, J. D., Benton, T. D.,
 Oquendo, M. A., Gur, R. C., & Gur, R. E. (2019). Neurocognitive functioning in community youth with suicidal ideation: Gender and pubertal effects. *The British Journal of Psychiatry*, 215(3), 552-558.

- Baumgartner, T., Esslen, M., & Jäncke, L. (2006a). From emotion perception to emotion experience: Emotions evoked by pictures and classical music. *International Journal* of Psychophysiology, 60(1), 34-43.
- Baumgartner, T., Lutz, K., Schmidt, C. F., & Jäncke, L. (2006b). The emotional power of music: How music enhances the feeling of affective pictures. *Brain Research*, 1075(1), 151-164.
- Bender, P. K., Reinholdt-Dunne, M. L., Esbjørn, B. H., & Pons, F. (2012). Emotion dysregulation and anxiety in children and adolescents: Gender differences. *Personality and Individual Differences*, 53(3), 284-288.
- Benjamini, Y., & Hochberg, Y. (1995). Controlling the false discovery rate: A practical and powerful approach to multiple testing. *Journal of the Royal Statistical Society*, 57(1), 289-300.
- Berking, M., & Wupperman, P. (2012). Emotion regulation and mental health: Recent findings, current challenges, and future directions. *Current Opinion in Psychiatry*, 25(2), 128-134.
- Berking, M., Wupperman, P., Reichardt, A., Pejic, T., Dippel, A., & Znoj, H. (2008).
 Emotion-regulation skills as a treatment target in psychotherapy. *Behaviour Research* and Therapy, 46(11), 1230-1237.
- Besharat, M. A., Nia, M. E., & Farahani, H. (2013). Anger and major depressive disorder:The mediating role of emotion regulation and anger rumination. *Asian Journal of Psychiatry*, 6(1), 35-41.
- Blanchard-Fields, F. (2009). Flexible and adaptive socio-emotional problem solving in adult development and aging. *Restorative Neurology and Neuroscience*, 27(5), 539-550.

- Blanchard-Fields, F., & Coats, A. H. (2008). The experience of anger and sadness in everyday problems impacts age differences in emotion regulation. *Developmental Psychology*, 44(6), 1547-1556.
- Bleyer, W., & Albritton, K. (2003). Definition of older adolescent and young adult. *Holland-Frei Cancer Medicine, 6th Edition*. BC Decker.
- Bodrogi, B., Bereczkei, T., & Deak, A. (2020). Be aware, make it clear, and take the lead: Emotion regulation difficulties and emotional intelligence as moderators of cognitive reappraisal. *Current Psychology*, 1-13.
- Boeninger, D. K., Masyn, K. E., Feldman, B. J., & Conger, R. D. (2010). Sex differences in developmental trends of suicide ideation, plans, and attempts among European
 American adolescents. *Suicide and Life-Threatening Behavior*, 40(5), 451-464.
- Boergers, J., Spirito, A., & Donaldson, D. (1998). Reasons for adolescent suicide attempts: Associations with psychological functioning. *Journal of the American Academy of Child & Adolescent Psychiatry*, 37(12), 1287-1293.
- Boland, M., Papa, A., Oliver, E., & del Carlo, R. (2019). Depressive suppression: Effects of emotion suppression on multiple emotions for depressed versus nondepressed individuals. *Cognitive Therapy and Research*, 43(5), 910-925.
- Bollen, K. A. (1989). A new incremental fit index for general structural equation models. *Sociological Methods & Research*, *17*(3), 303-316.
- Bomyea, J., Ball, T., Simmons, A., Campbell-Sills, L., Paulus, M., & Stein, M. (2020).
 Change in neural response during emotion regulation is associated with symptom reduction in cognitive behavioral therapy for anxiety disorders. *Journal of Affective Disorders*, 271, 207-214.

- Boyce, W., Torsheim, T., Currie, C., & Zambon, A. (2006). The family affluence scale as a measure of national wealth: Validation of an adolescent self-report measure. *Social Indicators Research*, 78(3), 473-487.
- Bradburn, N. M., Rips, L. J., & Shevell, S. K. (1987). Answering autobiographical questions: The impact of memory and inference on surveys. *Science*, *236*(4798), 157-161.
- Bradley, A., Karatzias, T., & Coyle, E. (2019). Derealization and self-harm strategies are used to regulate disgust, fear, and sadness in adult survivors of childhood sexual abuse. *Clinical Psychology & Psychotherapy*, 26(1), 94-104.
- Bradley, M. M., Codispoti, M., Sabatinelli, D., & Lang, P. J. (2001). Emotion and motivationII: Sex differences in picture processing. *Emotion*, 1(3), 300.
- Bradley, S. J. (1990). Affect regulation and psychopathology: Bridging the mind-body gap. *The Canadian Journal of Psychiatry*, *35*(6), 540-547.

Brannon, L. (2016). Gender: Psychological perspectives. Routledge.

- Breedlove, S. M., Watson, N. V., & Rosenzweig, M. R. (2010). *Biological psychology*, 6th *edition*. Sinauer Associates.
- Brereton, A., & McGlinchey, E. (2020). Self-harm, emotion regulation, and experiential avoidance: A systematic review. *Archives of Suicide Research*, *24*(1), 1-24.
- Brescoll, V. L., & Uhlmann, E. L. (2008). Can an angry woman get ahead? Status conferral, gender, and expression of emotion in the workplace. *Psychological Science*, 19(3), 268-275.
- Bresin, K., & Schoenleber, M. (2015). Gender differences in the prevalence of nonsuicidal self-injury: A meta-analysis. *Clinical Psychology Review*, *38*, 55-64.
- Brewer, S. K., Zahniser, E., & Conley, C. S. (2016). Longitudinal impacts of emotion regulation on emerging adults: Variable-and person-centered approaches. *Journal of Applied Developmental Psychology*, 47, 1-12.

- Brinker, J. K., & Dozois, D. J. (2009). Ruminative thought style and depressed mood. *Journal of Clinical Psychology*, 65(1), 1-19.
- Brock, A., Baker, A., Griffiths, C., Jackson, G., Fegan, G., & Marshall, D. (2006). Suicide trends and geographical variations in the United Kingdom, 1991–2004. *Health Statistics Quarterly*, 31(6), 6-22.
- Brody, C. L., Haaga, D. A., Kirk, L., & Solomon, A. (1999). Experiences of anger in people who have recovered from depression and never-depressed people. *The Journal of Nervous and Mental Disease*, 187(7), 400-405.
- Brody, L. R. (1985). Gender differences in emotional development: A review of theories and research. *Journal of Personality*, *53*(2), 102-149.
- Brody, L. R. (1997). Gender and emotion: Beyond stereotypes. *Journal of Social Issues*, 53(2), 369-393.
- Brody, L. R., & Hall, J. A. (1993). Gender and emotion. In M. Lewis & J. M. Haviland (Eds.), *Handbook of emotions* (pp. 447-460). Guilford Press.
- Brondolo, E. (2021). *Psychology research methods: A writing intensive approach*. Elsevier Science & Technology.
- Brougham, R. R., Zail, C. M., Mendoza, C. M., & Miller, J. R. (2009). Stress, sex differences, and coping strategies among college students. *Current Psychology*, 28(2), 85-97.
- Brown, S.-L., & Schwartz, G. E. (1980). Relationships between facial electromyography and subjective experience during affective imagery. *Biological Psychology*, *11*(1), 49-62.
- Browning, M., Holmes, E. A., Charles, M., Cowen, P. J., & Harmer, C. J. (2012). Using attentional bias modification as a cognitive vaccine against depression. *Biological Psychiatry*, 72(7), 572-579.

- Buck, R., & Powers, S. R. (2005). The expression, communication, and regulation of biological emotions: Sex and cultural differences and similarities. *Psychologia*, 48(4), 335-353.
- Bujor, L., & Turliuc, M. N. (2020). The personality structure in the emotion regulation of sadness and anger. *Personality and Individual differences*, 162, 109999.
- Burg, J. A. R., Allred, S. L., & Sapp, J. H. (1997). The potential for bias due to attrition in the National Exposure Registry: An examination of reasons for nonresponse, nonrespondent characteristics, and the response rate. *Toxicology and Industrial Health*, 13(1), 1-13.
- Butler, A. C., Chapman, J. E., Forman, E. M., & Beck, A. T. (2006). The empirical status of cognitive-behavioral therapy: A review of meta-analyses. *Clinical Psychology Review*, 26(1), 17-31.
- Butler, E. A., Egloff, B., Wilhelm, F. H., Smith, N. C., Erickson, E. A., & Gross, J. J. (2003).The social consequences of expressive suppression. *Emotion*, *3*(1), 48-67.
- Butler, L. D., & Nolen-Hoeksema, S. (1994). Gender differences in responses to depressed mood in a college sample. Sex Roles, 30(5-6), 331-346.
- Cacioppo, J. T., Berntson, G. G., Larsen, J. T., Poehlmann, K. M., & Ito, T. A. (2000). The psychophysiology of emotion. *Handbook of Emotions*, *2*, 173-191.
- Cain, M. K., Zhang, Z., & Yuan, K.-H. (2017). Univariate and multivariate skewness and kurtosis for measuring nonnormality: Prevalence, influence and estimation. *Behavior Research Methods*, 49(5), 1716-1735.
- Calkins, S. D., & Hill, A. (2007). Caregiver influences on emerging emotion regulation:
 Biological and environmental transactions in early development. In J. J. Gross (Ed.), *Handbook of Emotion Regulation*, (pp. 229-248). Guilford Publications.

- Campbell, A., & Muncer, S. (2017). Sex difference in awareness of threat: A meta-analysis of sex differences in attentional orienting in the dot probe task. *Personality and Individual Differences*, 119, 181-184.
- Campos, J. J., Mumme, D. L., Kermoian, R., & Campos, R. G. (1994). A functionalist perspective on the nature of emotion. *Monographs of the Society for Research in Child Development*, 59(2/3), 284-303.
- Cao, D., Li, Y., & Niznikiewicz, M. A. (2020). Neural characteristics of cognitive reappraisal success and failure: An ERP study. *Brain and Behavior*, 10(4), e01584.
- Caro, C., & Popovac, M. (2020). Gaming when things get tough? Examining how emotion regulation and coping self-efficacy influence gaming during difficult life situations. *Games and Culture*, 16(5), 611-631.
- Carrère, S., Mittmann, A., Woodin, E., Tabares, A., & Yoshimoto, D. (2005). Anger dysregulation, depressive symptoms, and health in married women and men. *Nursing Research*, 54(3), 184-192.
- Carver, C. S. (1997). You want to measure coping but your protocol's too long: Consider the brief cope. *International Journal of Behavioral Medicine*, *4*(1), 92-100.
- Carver, C. S., & Scheier, M. F. (1982). Control theory: A useful conceptual framework for personality–social, clinical, and health psychology. *Psychological Bulletin*, 92(1), 111.
- Carver, C. S., Scheier, M. F., & Weintraub, J. K. (1989). Assessing coping strategies: A theoretically based approach. *Journal of Personality and Social Psychology*, 56(2), 267-283.
- Casey, B., Heller, A. S., Gee, D. G., & Cohen, A. O. (2017). Development of the emotional brain. *Neuroscience Letters*, 693, 29-34.

- Casey, D., Geulayov, G., Bale, E., Brand, F., Clements, C., Kapur, N., Ness, J., Patel, A.,
 Waters, K., & Hawton, K. (2020). Paracetamol self-poisoning: Epidemiological study of trends and patient characteristics from the multicentre study of self-harm in England. *Journal of Affective Disorders*, 276, 699-706.
- Catuzzi, J. E., & Beck, K. D. (2014). Anxiety vulnerability in women: A two-hit hypothesis. *Experimental Neurology*, 259, 75-80.
- Cerin, E., & MacKinnon, D. P. (2009). A commentary on current practice in mediating variable analyses in behavioural nutrition and physical activity. *Public Health Nutrition*, 12(8), 1182-1188.
- Chandra, C. M., Szwedo, D. E., Allen, J. P., Narr, R. K., & Tan, J. S. (2020). Interactions between anxiety subtypes, personality characteristics, and emotional regulation skills as predictors of future work outcomes. *Journal of Adolescence*, 80, 157-172.
- Chaplin, T. M. (2015). Gender and emotion expression: A developmental contextual perspective. *Emotion Review*, *7*(1), 14-21.
- Chaplin, T. M., & Aldao, A. (2013). Gender differences in emotion expression in children: A meta-analytic review. *Psychological Bulletin*, 139(4), 735-765.
- Chapman, A. L., Gratz, K. L., & Brown, M. Z. (2006). Solving the puzzle of deliberate selfharm: The experiential avoidance model. *Behaviour Research and Therapy*, 44(3), 371-394.
- Chen, H., Wang, X., Huang, Y., Li, G., Liu, Z., Li, Y., & Geng, H. (2019). Prevalence, risk factors and multi-group latent class analysis of lifetime anxiety disorders comorbid depressive symptoms. *Journal of Affective Disorders*, *243*, 360-365.
- Chen, S. Y., Feng, Z., & Yi, X. (2017). A general introduction to adjustment for multiple comparisons. *Journal of Thoracic Disease*, *9*(6), 1725-1729.

- Cheung, R. Y., & Park, I. J. (2010). Anger suppression, interdependent self-construal, and depression among Asian American and European American college students. *Cultural Diversity and Ethnic Minority Psychology*, 16(4), 517-525.
- Chiu, Y. C., Tseng, C. Y., & Lin, F. G. (2017). Gender differences and stage-specific influence of parent–adolescent conflicts on adolescent suicidal ideation. *Psychiatry Research*, 255, 424-431.
- Christopoulos, G. I., Uy, M. A., & Yap, W. J. (2019). The body and the brain: Measuring skin conductance responses to understand the emotional experience. *Organizational Research Methods*, 22(1), 394-420.
- Clark, L. A., & Watson, D. (2016). Constructing validity: Basic issues in objective scale development. In A. E. Kazdin (Ed.), *Methodological issues and strategies in clinical research* (pp. 187–203). American Psychological Association.
- Clear, S. J., Gardner, A. A., Webb, H. J., & Zimmer-Gembeck, M. J. (2019). Common and distinct correlates of depression, anxiety, and aggression: Attachment and emotion regulation of sadness and anger. *Journal of Adult Development*, 27(3), 181-191.
- Clore, G. L., & Ortony, A. (2008). Appraisal theories: How cognition shapes affect into emotion. In M. Lewis, J. M. Haviland-Jones, & L. F. Barrett (Eds.), *Handbook of emotions* (pp. 628–642). Guilford Press
- Cludius, B., Mennin, D., & Ehring, T. (2020). Emotion regulation as a transdiagnostic process. *Emotion*, 20(1), 37-42.
- Cohen, J. (1960). A coefficient of agreement for nominal scales. *Educational and Psychological Measurement*, 20(1), 37-46.
- Cohen, N., & Ochsner, K. N. (2018). From surviving to thriving in the face of threats: The emerging science of emotion regulation training. *Current Opinion in Behavioral Sciences*, 24, 143-155.

- Cohen, Z. D., & DeRubeis, R. J. (2018). Treatment selection in depression. *Annual Review of Clinical Psychology*, 14, 209-236.
- Cole, B. P., & Ingram, P. B. (2020). Where do I turn for help? Gender role conflict, selfstigma, and college men's help-seeking for depression. *Psychology of Men & Masculinities*, 21(3), 441-452.
- Cole, P. M. (2014). Moving ahead in the study of the development of emotion regulation. International Journal of Behavioral Development, 38(2), 203-207.
- Collet, C., Vernet-Maury, E., Delhomme, G., & Dittmar, A. (1997). Autonomic nervous system response patterns specificity to basic emotions. *Journal of the Autonomic Nervous System*, 62(1-2), 45-57.
- Colombo, D., Fernández-Álvarez, J., Suso-Ribera, C., Cipresso, P., Valev, H., Leufkens, T., Garcia-Palacios, A., Riva, G., & Botella, C. (2020). The need for change:
 Understanding emotion regulation antecedents and consequences using ecological momentary assessment. *Emotion*, 20(1), 30-36.
- Compas, B. E., Jaser, S. S., Dunbar, J. P., Watson, K. H., Bettis, A. H., Gruhn, M. A., &
 Williams, E. K. (2014). Coping and emotion regulation from childhood to early adulthood: Points of convergence and divergence. *Australian Journal of Psychology*, 66(2), 71-81.
- Conway, C., Jones, B. C., Debruine, L. M., Welling, L., Smith, M. L., Perrett, D., Sharp, M. A., & Al-Dujaili, E. A. (2007). Salience of emotional displays of danger and contagion in faces is enhanced when progesterone levels are raised. *Hormones and Behavior*, 51(2), 202-206.
- Cook, T., Roy, A. R., & Welker, K. M. (2019). Music as an emotion regulation strategy: An examination of genres of music and their roles in emotion regulation. *Psychology of Music*, 47(1), 144-154.

- Copeland, E. P., & Hess, R. S. (1995). Differences in young adolescents' coping strategies based on gender and ethnicity. *The Journal of Early Adolescence*, *15*(2), 203-219.
- Coppin, G., & Sander, D. (2021). Theoretical approaches to emotion and its measurement. In
 H. L. Meiselman (Ed.), *Emotion Measurement* (2nd ed., pp. 3-37). Woodhead
 Publishing.
- Coreia, M. E., Rosado, A., Serpa, S., & Ferreira, V. (2017). Fear of failure in athletes: Gender, age and type of sport differences. *Revista Iberoamericana de Psicología del Ejercicio y el Deporte, 12*(2), 185-193.
- Côté, S., Gyurak, A., & Levenson, R. W. (2010). The ability to regulate emotion is associated with greater well-being, income, and socioeconomic status. *Emotion*, 10(6), 923 933.
- Cowen, A. S., Elfenbein, H. A., Laukka, P., & Keltner, D. (2018). Mapping 24 emotions conveyed by brief human vocalization. *American Psychologist*, *74*(6), 698-712.
- Cowen, A. S., Fang, X., Sauter, D., & Keltner, D. (2020). What music makes us feel: At least 13 dimensions organize subjective experiences associated with music across different cultures. *Proceedings of the National Academy of Sciences*, 117(4), 1924-1934.
- Cowen, A. S., & Keltner, D. (2017). Self-report captures 27 distinct categories of emotion bridged by continuous gradients. *Proceedings of the National Academy of Sciences*, 114(38), 7900-7909.
- Cowen, A. S., & Keltner, D. (2018). Clarifying the conceptualization, dimensionality, and structure of emotion: Response to Barrett and colleagues. *Trends in Cognitive Sciences*, 22(4), 274-276.
- Cox, D. L., Stabb, S. D., & Hulgus, J. F. (2000). Anger and depression in girls and boys: A study of gender differences. *Psychology of Women Quarterly*, 24(1), 110-112.

- Cracco, E., Goossens, L., & Braet, C. (2017). Emotion regulation across childhood and adolescence: Evidence for a maladaptive shift in adolescence. *European Child & Adolescent Psychiatry*, 26(8), 909-921.
- Cracco, E., Van Durme, K., & Braet, C. (2015). Validation of the FEEL-KJ: An instrument to measure emotion regulation strategies in children and adolescents. *PLoS ONE*, 10(9), 1-18.
- Craske, M. G. (1997). Fear and anxiety in children and adolescents. *Bulletin of the Menninger Clinic*, *61*(2), 4-36.
- Cristea, I. A., Kok, R. N., & Cuijpers, P. (2015). Efficacy of cognitive bias modification interventions in anxiety and depression: Meta-analysis. *The British Journal of Psychiatry*, 206(1), 7-16.
- Crowne, D. P., & Marlowe, D. (1960). A new scale of social desirability independent of psychopathology. *Journal of Consulting Psychology*, *24*(4), 349-354.
- Curley, C., Krause, R. M., Feiock, R., & Hawkins, C. V. (2019). Dealing with missing data: A comparative exploration of approaches using the integrated city sustainability database. *Urban Affairs Review*, 55(2), 591-615.
- Dalicandro, L., Harder, J. A., Mazmanian, D., & Weaver, B. (2021). How prevalent is overfitting of regression models? A survey of recent articles in three psychology journals. *Quantitative Methods for Psychology*, 17(1), 1-6.
- Damasio, A. R. (1998). Emotion in the perspective of an integrated nervous system. *Brain Research Reviews*, 26(2-3), 83-86.
- Darwin, C. (1872). *The expression of emotions in man and animals*. New York: Philosophical Library.

- Davis, E., Greenberger, E., Charles, S., Chen, C., Zhao, L., & Dong, Q. (2012). Emotion experience and regulation in China and the United States: How do culture and gender shape emotion responding? *International Journal of Psychology*, 47(3), 230-239.
- Davis, E., & Levine, L. J. (2013). Emotion regulation strategies that promote learning:
 Reappraisal enhances children's memory for educational information. *Child Development*, 84(1), 361-374.
- Davis, E. L., Quiñones-Camacho, L. E., & Buss, K. A. (2016). The effects of distraction and reappraisal on children's parasympathetic regulation of sadness and fear. *Journal of Experimental Child Psychology*, 142, 344-358.
- Davis, T. S., Mauss, I. B., Lumian, D., Troy, A. S., Shallcross, A. J., Zarolia, P., Ford, B. Q., & McRae, K. (2014). Emotional reactivity and emotion regulation among adults with a history of self-harm: Laboratory self-report and functional MRI evidence. *Journal of Abnormal Psychology*, *123*(3), 499.
- Day, A. L., & Livingstone, H. A. (2003). Gender differences in perceptions of stressors and utilization of social support among university students. *Canadian Journal of Behavioural Science*, 35(2), 73-83.
- De Beurs, D., Cleare, S., Wetherall, K., Eschle-Byrne, S., Ferguson, E., O'Connor, D. B., & O'Connor, R. C. (2020). Entrapment and suicide risk: The development of the 4-item Entrapment Scale Short-Form (E-SF). *Psychiatry Research*, 284, 112765.
- De Leo, D., & Heller, T. S. (2004). Who are the kids who self-harm? An Australian selfreport school survey. *Medical Journal of Australia*, 181(3), 140-144.
- Denny, B. T. (2020). Getting better over time: A framework for examining the impact of emotion regulation training. *Emotion*, 20(1), 110-114.
- Denny, B. T., Inhoff, M. C., Zerubavel, N., Davachi, L., & Ochsner, K. N. (2015). Getting over it: Long-lasting effects of emotion regulation on amygdala response. *Psychological Science*, 26(9), 1377-1388.
- Denny, B. T., & Ochsner, K. N. (2014). Behavioral effects of longitudinal training in cognitive reappraisal. *Emotion*, 14(2), 425.
- DeVellis, R. F. (2016). *Scale development: Theory and applications, Vol. 26.* Sage publications.
- Dillon, K. H., Van Voorhees, E. E., & Elbogen, E. B. (2020). Associations between anger and suicidal ideation and attempts: A prospective study using the National Epidemiologic Survey on Alcohol and Related Conditions. *Journal of Psychiatric Research*, *122*, 17-21.
- Dimberg, U. (1988). Facial electromyography and the experience of emotion. *Journal of Psychophysiology*, 2(4), 277-282.
- Dimberg, U., & Thunberg, M. (1998). Rapid facial reactions to emotional facial expressions. *Scandinavian Journal of Psychology*, *39*(1), 39-45.
- Dixon-Gordon, K. L., Aldao, A., & De Los Reyes, A. (2014). Repertoires of emotion regulation: A person-centered approach to assessing emotion regulation strategies and links to psychopathology. *Cognition and Emotion*, 1-12.
- Dixon-Gordon, K. L., Aldao, A., & De Los Reyes, A. (2015a). Emotion regulation in context: Examining the spontaneous use of strategies across emotional intensity and type of emotion. *Personality and Individual Differences*, 86, 271-276.
- Dixon-Gordon, K. L., Aldao, A., & De Los Reyes, A. (2015b). Repertoires of emotion regulation: A person-centered approach to assessing emotion regulation strategies and links to psychopathology. *Cognition and Emotion*, 29(7), 1314-1325.

- Doré, B. P., Silvers, J. A., & Ochsner, K. N. (2016). Toward a personalized science of emotion regulation. Social and Personality Psychology Compass, 10(4), 171-187.
- Dörfel, D., Lamke, J.-P., Hummel, F., Wagner, U., Erk, S., & Walter, H. (2014). Common and differential neural networks of emotion regulation by detachment, reinterpretation, distraction, and expressive suppression: A comparative fMRI investigation. *Neuroimage*, 101, 298-309.
- Dougall, N., Stark, C., Agnew, T., Henderson, R., Maxwell, M., & Lambert, P. (2017). An analysis of suicide trends in Scotland 1950–2014: Comparison with England & Wales. *BioMed Central Public Health*, 17(1), 970-981.
- Douw, L., Quaak, M., Fitzsimmons, S. M., de Wit, S. J., van der Werf, Y. D., van den Heuvel, O. A., & Vriend, C. (2020). Static and dynamic network properties of the repetitive transcranial magnetic stimulation target predict changes in emotion regulation in obsessive-compulsive disorder. *Brain Stimulation*, 13(2), 318-326.
- Druckman, J. N., & Kam, C. D. (2011). Students as experimental participants: A defense of the 'narrow data base'. In J. N. Druckman, D. P. Green, J. H. Kuklinski, & A. Lupia (Eds.), *Cambridge Handbook of Experimental Political Science* (pp. 41-57). New York: Cambridge University Press.
- Dryman, M. T., & Heimberg, R. G. (2018). Emotion regulation in social anxiety and depression: A systematic review of expressive suppression and cognitive reappraisal. *Clinical Psychology Review*, 65, 17-42.
- du Pont, A., Rhee, S. H., Corley, R. P., Hewitt, J. K., & Friedman, N. P. (2018). Rumination and psychopathology: Are anger and depressive rumination differentially associated with internalizing and externalizing psychopathology? *Clinical Psychological Science*, 6(1), 18-31.

- Duarte, T. A., Paulino, S., Almeida, C., Gomes, H. S., Santos, N., & Gouveia-Pereira, M. (2020). Self-harm as a predisposition for suicide attempts: A study of adolescents' deliberate self-harm, suicidal ideation, and suicide attempts. *Psychiatry Research*, 287, 112553.
- Duckro, P. N., Chibnall, J. T., & Tomazic, T. J. (1995). Anger, depression, and disability: A path analysis of relationships in a sample of chronic posttraumatic headache patients. *Headache: The Journal of Head and Face Pain, 35*(1), 7-9.
- Dunn, K. M., Jordan, K., Lacey, R. J., Shapley, M., & Jinks, C. (2004). Patterns of consent in epidemiologic research: Evidence from over 25,000 responders. *American Journal of Epidemiology*, 159(11), 1087-1094.
- Eaton, N. R., Keyes, K. M., Krueger, R. F., Balsis, S., Skodol, A. E., Markon, K. E., Grant,
 B. F., & Hasin, D. S. (2012). An invariant dimensional liability model of gender
 differences in mental disorder prevalence: Evidence from a national sample. *Journal*of Abnormal Psychology, 121(1), 282.
- Egleston, B. L., Miller, S. M., & Meropol, N. J. (2011). The impact of misclassification due to survey response fatigue on estimation and identifiability of treatment effects. *Statistics in Medicine*, 30(30), 3560-3572.
- Ehring, T., & Behar, E. (2020). Transdiagnostic view on worrying and other negative mental content. *Generalized Anxiety Disorder and Worrying: A Comprehensive Handbook for Clinicians and Researchers*, 43-68.
- Ehring, T., & Watkins, E. R. (2008). Repetitive negative thinking as a transdiagnostic process. *International Journal of Cognitive Therapy*, *1*(3), 192-205.
- Eisenberg, N., Cumberland, A., & Spinrad, T. L. (1998). Parental socialization of emotion. *Psychological Inquiry*, 9(4), 241-273.

Ekman, P. (1965). Differential communication of affect by head and body cues. *Journal of Personality and Social Psychology*, 2(5), 726.

Ekman, P. (1992). An argument for basic emotions. *Cognition & Emotion*, 6(3-4), 169-200.

Ekman, P. (1999). Basic emotions. Handbook of Cognition and Emotion, 98(45-60), 16.

- Ekman, P., & Cordaro, D. (2011). What is meant by calling emotions basic. *Emotion Review*, *3*(4), 364-370.
- Ekman, P., & Friesen, W. V. (1971). Constants across cultures in the face. Journal of Personality and Social Psychology, 17, 124-129.
- Ekman, P., Friesen, W. V., O'Sullivan, M., Chan, A., Diacoyanni-Tarlatzis, I., Heider, K.,
 Krause, R., LeCompte, W. A., Pitcairn, T., Ricci-Bitti, P. E., Scherer, K., Tomita, M.,
 & Tzavaras, A. (1987). Universals and cultural differences in the judgments of facial
 expressions of emotion. *Journal of Personality and Social Psychology*, *53*(4), 712.
- Ekman, P., Levenson, R. W., & Friesen, W. V. (1983). Autonomic nervous system activity distinguishes among emotions. *Science*, 221(4616), 1208-1210.
- Ekman, P., & Oster, H. (1979). Facial expressions of emotion. *Annual Review of Psychology*, 30(1), 527-554.
- Ekman, P., Sorenson, E. R., & Friesen, W. V. (1969). Pan-cultural elements in facial displays of emotion. *Science*, *164*(3875), 86-88.
- Ellis, A. J., Vanderlind, W. M., & Beevers, C. G. (2013). Enhanced anger reactivity and reduced distress tolerance in major depressive disorder. *Cognitive Therapy and Research*, 37(3), 498-509.
- Endrerud, M. S., & Vikan, A. (2007). Five to seven year old children's strategies for regulating anger, sadness, and fear. *Nordic Psychology*, *59*(2), 127-134.
- Esmaeilinasaba, M., Khoshkb, A. A., & Makhmalic, A. (2016). *Emotion regulation and life satisfaction in university students: Gender differences* (Conference presentation).

ICEEPSY 2016: 7th International Conference on Education and Educational Psychology.

- Esnaola, I., Revuelta, L., Ros, I., & Sarasa, M. (2017). The development of emotional intelligence in adolescence. *Anales de Psicologia*, *33*(2), 327-333.
- Faul, F., Erdfelder, E., Lang, A.-G., & Buchner, A. (2007). G* Power 3: A flexible statistical power analysis program for the social, behavioral, and biomedical sciences. *Behavior Research Methods*, 39(2), 175-191.

Fausto-Sterling, A. (2018). Why sex is not binary. The New York Times, 25.

- Feigenbaum, J. (2007). Dialectical behaviour therapy: An increasing evidence base. *Journal of Mental Health*, *16*(1), 51-68.
- Field, A. (2017). Discovering statistics using IBM SPSS statistics: North American edition. Sage.
- Fischer, A. H., & Evers, C. (2011). The social costs and benefits of anger as a function of gender and relationship context. *Sex Roles*, *65*(1), 23-34.
- Fischer, A. H., Rodriguez Mosquera, P. M., Van Vianen, A. E., & Manstead, A. S. (2004).Gender and culture differences in emotion. *Emotion*, 4(1), 87.
- Fisher, A. J., & Boswell, J. F. (2016). Enhancing the personalization of psychotherapy with dynamic assessment and modeling. *Assessment*, 23(4), 496-506.
- Fiske, A., Wetherell, J. L., & Gatz, M. (2009). Depression in older adults. Annual Review of Clinical Psychology, 5, 363-389.
- Fitzgerald, J. M., Klumpp, H., Langenecker, S., & Phan, K. L. (2019). Transdiagnostic neural correlates of volitional emotion regulation in anxiety and depression. *Depression and Anxiety*, 36(5), 453-464.
- Fivush, R. (1989). Exploring sex differences in the emotional content of mother-child conversations about the past. *Sex Roles*, *20*(11), 675-691.

- Flett, G. L., Besser, A., & Hewitt, P. L. (2014). Perfectionism and interpersonal orientations in depression: An analysis of validation seeking and rejection sensitivity in a community sample of young adults. *Psychiatry: Interpersonal and Biological Processes*, 77(1), 67-85.
- Flynn, D., Kells, M., Joyce, M., Corcoran, P., Hurley, J., Gillespie, C., Suarez, C., Swales, M., & Arensman, E. (2020). Multisite implementation and evaluation of 12-month standard dialectical behavior therapy in a public community setting. *Journal of Personality Disorders*, 34(3), 377-393.
- Folkman, S., & Lazarus, R. S. (1980). An analysis of coping in a middle-aged community sample. *Journal of Health and Social Behavior*, 219-239.
- Ford, B. Q., Karnilowicz, H. R., & Mauss, I. B. (2017). Understanding reappraisal as a multicomponent process: The psychological health benefits of attempting to use reappraisal depend on reappraisal success. *Emotion*, 17(6), 905.
- Forkmann, T., Scherer, A., Bocker, M., Pawelzik, M., Gauggel, S., & Glaesmer, H. (2014). The relation of cognitive reappraisal and expressive suppression to suicidal ideation and suicidal desire. *Suicide and Life-Threatening Behavior*, 44(5), 524-536.
- Frazier, S. N., & Vela, J. (2014). Dialectical behavior therapy for the treatment of anger and aggressive behavior: A review. *Aggression and Violent Behavior*, *19*(2), 156-163.
- Fredrickson, B. L. (1998). What good are positive emotions? *Review of General Psychology*, 2(3), 300-319.
- Fredrickson, B. L. (2001). The role of positive emotions in positive psychology: The broaden-and-build theory of positive emotions. *American Psychologist*, 56(3), 218-226.
- Frere, P. B., Vetter, N. C., Artiges, E., Filippi, I., Miranda, R., Vulser, H., Paillère-Martinot, M-L., Ziesch, V., Conrod, P., Cattrell, A., Walter, H., Gallinat, J., Bromberg, U., Jurk,

S., Menningen, E., Frouin, V., Orfanos, D. P., Stringaris, A., Penttilä, J., . . . & van Noort, B. (2020). Sex effects on structural maturation of the limbic system and outcomes on emotional regulation during adolescence. *Neuroimage*, *210*, 116441.

Fridlund, A. J., Schwartz, G. E., & Fowler, S. C. (1984). Pattern recognition of self-reported emotional state from multiple-site facial EMG activity during affective imagery. *Psychophysiology*, 21(6), 622-637.

Frijda, N. H. (1986). The emotions. University Press.

- Frijda, N. H. (1987). Emotion, cognitive structure, and action tendency. *Cognition and Emotion, 1*(2), 115-143.
- Fuentes-Sánchez, N., Jaén, I., Escrig, M. A., Lucas, I., & Pastor, M. C. (2019). Cognitive reappraisal during unpleasant picture processing: Subjective self-report and peripheral physiology. *Psychophysiology*, 1-15.
- Gallagher, K. E., & Parrott, D. J. (2011). Does distraction reduce the alcohol–aggression relation? A cognitive and behavioral test of the attention-allocation model. *Journal of Consulting and Clinical Psychology*, 79(3), 319-329.
- García-Vega, E., Camero, A., Fernández, M., & Villaverde, A. (2018). Suicidal ideation and suicide attempts in persons with gender dysphoria. *Psicothema*, *30*(3), 283-288.
- Garnefski, N., & Kraaij, V. (2006). Relationships between cognitive emotion regulation strategies and depressive symptoms: A comparative study of five specific samples. *Personality and Individual differences, 40*(8), 1659-1669.
- Garnefski, N., Kraaij, V., & Spinhoven, P. (2001). Negative life events, cognitive emotion regulation and emotional problems. *Personality and Individual differences*, 30(8), 1311-1327.
- Garnefski, N., Kraaij, V., & Spinhoven, P. (2002a). *Manual for the use of the Cognitive Emotion Regulation Questionnaire*. DATEC.

Garnefski, N., Legerstee, J., Kraaij, V., Van Den Kommer, T., & Teerds, J. A. N. (2002b).
Cognitive coping strategies and symptoms of depression and anxiety: A comparison between adolescents and adults. *Journal of Adolescence*, 25(6), 603-611.

- Garnefski, N., Teerds, J., Kraaij, V., Legerstee, J., & van den Kommer, T. (2004). Cognitive emotion regulation strategies and depressive symptoms: Differences between males and females. *Personality and Individual differences*, *36*(2), 267-276.
- Garofalo, C., Velotti, P., Zavattini, G. C., & Kosson, D. S. (2017). Emotion dysregulation and interpersonal problems: The role of defensiveness. *Personality and Individual Differences*, 119, 96-105.
- Garside, R. B., & Klimes-Dougan, B. (2002). Socialization of discrete negative emotions: Gender differences and links with psychological distress. *Sex Roles*, *47*(3), 115-128.
- Geddes, K., Dziurawiec, S., & Lee, C. W. (2013). Dialectical behaviour therapy for the treatment of emotion dysregulation and trauma symptoms in self-injurious and suicidal adolescent females: A pilot programme within a community-based child and adolescent mental health service. *Psychiatry Journal*, 2013, 1-10.
- Gilchrist, E. E., & Sadler, D. W. (2019). The role of depression in unnatural death: A casebased retrospective study. *Journal of Affective Disorders*, 259, 7-14.
- Gilmore, G. C. (2007). Inappropriate use of covariate analysis renders meaningless results. *Journal of the International Neuropsychological Society*, *13*(2), 370-370.
- Girgus, J. S., & Yang, K. (2015). Gender and depression. *Current Opinion in Psychology*, *4*, 53-60.
- Gladstone, T. R., & Koenig, L. J. (1994). Sex differences in depression across the high school to college transition. *Journal of Youth and Adolescence*, *23*(6), 643-669.

- Goldin, P. R., McRae, K., Ramel, W., & Gross, J. J. (2008). The neural bases of emotion regulation: Reappraisal and suppression of negative emotion. *Biological Psychiatry*, 63(6), 577-586.
- Goldin, P. R., Morrison, A. S., Jazaieri, H., Heimberg, R. G., & Gross, J. J. (2017).
 Trajectories of social anxiety, cognitive reappraisal, and mindfulness during an RCT of CBGT versus MBSR for social anxiety disorder. *Behaviour Research and Therapy*, 97, 1-13.
- Goldin, P. R., Ziv, M., Jazaieri, H., Werner, K., Kraemer, H., Heimberg, R. G., & Gross, J. J.
 (2012). Cognitive reappraisal self-efficacy mediates the effects of individual cognitive-behavioral therapy for social anxiety disorder. *Journal of Consulting and Clinical Psychology*, 80(6), 1034-1040.
- Goldney, R., Winefield, A., Saebel, J., Winefield, H., & Tiggeman, M. (1997). Anger, suicidal ideation, and attempted suicide: A prospective study. *Comprehensive Psychiatry*, 38(5), 264-268.
- Gomez-Baya, D., Mendoza, R., Paino, S., Sanchez, A., & Romero, N. (2016). Latent growth curve analysis of gender differences in response styles and depressive symptoms during mid-adolescence. *Cognitive Therapy and Research*, 41(2), 289-303.
- Gonçalves, S. F., Chaplin, T. M., Turpyn, C. C., Niehaus, C. E., Curby, T. W., Sinha, R., & Ansell, E. B. (2019). Difficulties in emotion regulation predict depressive symptom trajectory from early to middle adolescence. *Child Psychiatry & Human Development*, 50(4), 618-630.
- Goos, L. M., & Silverman, I. (2002). Sex related factors in the perception of threatening facial expressions. *Journal of Nonverbal Behavior*, *26*(1), 27-41.
- Gordon, M. E., Slade, L. A., & Schmitt, N. (1986). The "science of the sophomore" revisited: From conjecture to empiricism. *Academy of Management Review*, *11*(1), 191-207.

- Goubet, K. E., & Chrysikou, E. G. (2019). Emotion regulation flexibility: Gender differences in context sensitivity and repertoire. *Frontiers in Psychology*, *10*, 935.
- Grant, K., Marsh, P., Syniar, G., Williams, M., Addlesperger, E., Kinzler, M. H., & Cowman,
 S. (2002). Gender differences in rates of depression among undergraduates:
 Measurement matters. *Journal of Adolescence*, 25(6), 613-617.
- Gratz, K. L. (2001). Measurement of deliberate self-harm: Preliminary data on the Deliberate
 Self-Harm Inventory. *Journal of Psychopathology and Behavioral Assessment*, 23(4), 253-263.
- Gratz, K. L., Bardeen, J. R., Levy, R., Dixon-Gordon, K. L., & Tull, M. T. (2015). Mechanisms of change in an emotion regulation group therapy for deliberate selfharm among women with borderline personality disorder. *Behaviour Research and Therapy*, 65, 29-35.
- Gratz, K. L., Conrad, S. D., & Roemer, L. (2002). Risk factors for deliberate self-harm among college students. *American Journal of Orthopsychiatry*, 72(1), 128-140.
- Gratz, K. L., & Roemer, L. (2004). Multidimensional assessment of emotion regulation and dysregulation: Development, factor structure, and initial validation of the difficulties in emotion regulation scale. *Journal of Psychopathology and Behavioral Assessment*, 26(1), 41-54.
- Gray, J. A. (1971). Sex differences in emotional behaviour in mammals including man: Endocrine bases. *Acta psychologica*, *35*(1), 29-46.
- Green, M., & Benzeval, M. (2011). Ageing, social class and common mental disorders: longitudinal evidence from three cohorts in the West of Scotland. *Psychological Medicine*, 41(3), 565-574.
- Green, S. B. (1991). How many subjects does it take to do a regression analysis. *Multivariate Behavioral Research*, 26(3), 499-510.

- Greenberg, J. (1987). The college sophomore as guinea pig: Setting the record straight. Academy of Management Review, 12(1), 157-159.
- Gross, J. J. (1998a). Antecedent- and response-focused emotion regulation: Divergent consequences for experience, expression, and physiology. *Journal of Personality and Social Psychology*, 74(1), 224-237.
- Gross, J. J. (1998b). The emerging field of emotion regulation: An integrative review. *Review* of General Psychology, 2(3), 271-299.
- Gross, J. J. (1999). Emotion regulation: Past, present, future. *Cognition & Emotion*, *13*(5), 551-573.
- Gross, J. J. (2002). Emotion regulation: Affective, cognitive, and social consequences. *Psychophysiology*, *39*(3), 281-291.
- Gross, J. J. (2015a). Emotion regulation: Current status and future prospects. *Psychological Inquiry*, 26(1), 1-26.
- Gross, J. J. (2015b). The extended process model of emotion regulation: Elaborations, applications, and future directions. *Psychological Inquiry*, *26*(1), 130-137.
- Gross, J. J., & Barrett, L. F. (2011). Emotion generation and emotion regulation: One or two depends on your point of view. *Emotion Review*, *3*(1), 8-16.
- Gross, J. J., & John, O. P. (2003). Individual differences in two emotion regulation processes: Implications for affect, relationships, and well-being. *Journal of Personality and Social Psychology*, 85(2), 348-362.
- Gross, J. J., & Levenson, R. W. (1993). Emotional suppression: Physiology, self-report, and expressive behavior. *Journal of Personality and Social Psychology*, *64*(6), 970-986.
- Gross, J. J., & Levenson, R. W. (1995). Emotion elicitation using films. *Cognition & Emotion*, 9(1), 87-108.

- Gross, J. J., & Levenson, R. W. (1997). Hiding feelings: The acute effects of inhibiting negative and positive emotion. *Journal of Abnormal Psychology*, *106*(1), 95-103.
- Gross, J. J., Richards, J. M., & John, O. P. (2006). Emotion regulation in everyday life. In D.
 K. Snyder, J. Simpson, & J. N. Hughes (Eds.), *Emotion regulation in couples and families: Pathways to dysfunction and health* (pp. 13–35). American Psychological Association.
- Gross, J. J., Uusberg, H., & Uusberg, A. (2019). Mental illness and well-being: An affect regulation perspective. *World Psychiatry*, *18*(2), 130-139.
- Grossman, M., & Wood, W. (1993). Sex differences in intensity of emotional experience: A social role interpretation. *Journal of Personality and Social Psychology*, 65(5), 1010-1022.
- Guillén Botella, V., García-Palacios, A., Bolo Miñana, S., Baños, R., Botella, C., & Marco, J.
 H. (2021). Exploring the effectiveness of dialectical behavior therapy versus systems training for emotional predictability and problem solving in a sample of patients with borderline personality disorder. *Journal of Personality Disorders*, *35*, 21-38.
- Gullone, E., Hughes, E. K., King, N. J., & Tonge, B. (2010). The normative development of emotion regulation strategy use in children and adolescents: A 2-year follow-up study. *Journal of Child Psychology and Psychiatry*, 51(5), 567-574.
- Gullone, E., & Taffe, J. (2012). The Emotion Regulation Questionnaire for Children and Adolescents (ERQ–CA): A psychometric evaluation. *Psychological Assessment*, 24(2), 409-417.
- Gyurak, A., Gross, J. J., & Etkin, A. (2011). Explicit and implicit emotion regulation: A dualprocess framework. *Cognition and Emotion*, *25*(3), 400-412.
- Hafferty, J. D., Navrady, L., Adams, M., Howard, D., Campbell, A., Whalley, H., Lawrie, S.M., Nicodemus, K. K., Porteous, D. J., Deary, I. J., McIntosh, A. M. (2019). The role

of neuroticism in self-harm and suicidal ideation: results from two UK populationbased cohorts. *Social Psychiatry and Psychiatric Epidemiology*, *54*(12), 1505-1518.

- Haga, S. M., Kraft, P., & Corby, E.-K. (2009). Emotion regulation: Antecedents and wellbeing outcomes of cognitive reappraisal and expressive suppression in cross-cultural samples. *Journal of Happiness Studies*, 10(3), 271-291.
- Hankin, B. L., Abramson, L. Y., Moffitt, T. E., Silva, P. A., McGee, R., & Angell, K. E. (1998). Development of depression from preadolescence to young adulthood:
 Emerging gender differences in a 10-year longitudinal study. *Journal of Abnormal Psychology*, *107*(1), 128-140.
- Harandi, T. F., Taghinasab, M. M., & Nayeri, T. D. (2017). The correlation of social support with mental health: A meta-analysis. *Electronic Physician*, 9(9), 5212-5222.
- Harmon-Jones, E., & Sigelman, J. (2001). State anger and prefrontal brain activity: Evidence that insult-related relative left-prefrontal activation is associated with experienced anger and aggression. *Journal of Personality and Social Psychology*, 80(5), 797-803.
- Harmon, S. L., Stephens, H. F., Repper, K. K., Driscoll, K. A., & Kistner, J. A. (2019).
 Children's rumination to sadness and anger: Implications for the development of depression and aggression. *Journal of Clinical Child & Adolescent Psychology*, 48(4), 622-632.
- Hartley, C. A., & Phelps, E. A. (2010). Changing fear: The neurocircuitry of emotion regulation. *Neuropsychopharmacology*, 35(1), 136-146.
- Hawkins, K. A., & Cougle, J. R. (2013). A test of the unique and interactive roles of anger experience and expression in suicidality: Findings from a population-based study. *Journal of Nervous and Mental Disease*, 201(11), 959-963.

- Hawkins, K. A., Hames, J. L., Ribeiro, J. D., Silva, C., Joiner, T. E., & Cougle, J. R. (2014).An examination of the relationship between anger and suicide risk through the lens of the interpersonal theory of suicide. *Journal of Psychiatric Research*, *50*, 59-65.
- Hawton, K. (2000). Sex and suicide: Gender differences in suicidal behaviour. *British* Journal of Psychiatry, 177(6), 484-485.
- Hawton, K., Cole, D., O'Grady, J., & Osborn, M. (1982). Motivational aspects of deliberate self-poisoning in adolescents. *British Journal of Psychiatry*, *141*(3), 286-291.
- Hawton, K., Hall, S., Simkin, S., Bale, L., Bond, A., Codd, S., & Stewart, A. (2003).
 Deliberate self-harm in adolescents: A study of characteristics and trends in Oxford, 1990–2000. *Journal of Child Psychology and Psychiatry*, 44(8), 1191-1198.
- Hawton, K., Rodham, K., Evans, E., & Weatherall, R. (2002). Deliberate self harm in adolescents: Self report survey in schools in England. *British Medical Journal*, 325(7374), 1207-1211.
- Hawton, K., Saunders, K. E., & O'Connor, R. C. (2012). Self-harm and suicide in adolescents. *The Lancet*, 379(9834), 2373-2382.
- Hayes, A. F. (2009). Beyond Baron and Kenny: Statistical mediation analysis in the new millennium. *Communication Monographs*, 76(4), 408-420.
- Hayes, A. F. (2017). Introduction to mediation, moderation, and conditional process analysis: A regression-based approach. Guilford publications.
- Hayes, A. F., & Scharkow, M. (2013). The relative trustworthiness of inferential tests of the indirect effect in statistical mediation analysis: Does method really matter? *Psychological Science*, 24(10), 1918-1927.
- Hellfeldt, K., López-Romero, L., & Andershed, H. (2020). Cyberbullying and psychological well-being in young adolescence: The potential protective mediation effects of social

support from family, friends, and teachers. *International Journal of Environmental Research and Public Health*, *17*(1), 45.

- Hermann, A., Kress, L., & Stark, R. (2017). Neural correlates of immediate and prolonged effects of cognitive reappraisal and distraction on emotional experience. *Brain Imaging and Behavior*, 11(5), 1227-1237.
- Hermann, A., Leutgeb, V., Scharmüller, W., Vaitl, D., Schienle, A., & Stark, R. (2013).
 Individual differences in cognitive reappraisal usage modulate the time course of brain activation during symptom provocation in specific phobia. *Biology of Mood & Anxiety Disorders*, 3(1), 1-11.
- Herrald, M. M., & Tomaka, J. (2002). Patterns of emotion-specific appraisal, coping, and cardiovascular reactivity during an ongoing emotional episode. *Journal of Personality* and Social Psychology, 83(2), 434-450.
- Herwig, U., Lutz, J., Scherpiet, S., Scheerer, H., Kohlberg, J., Opialla, S., Preuss, A., Steiger, V. R., Sulzer, J., Weidt, S., Stämpfli, P., Rufer, M., Seifritz, E., Jäncke, L., & Brühl, A. B. (2019). Training emotion regulation through real-time fMRI neurofeedback of amygdala activity. *Neuroimage*, *184*, 687-696.
- Hester, N., Jones, B. C., & Hehman, E. (2020). Perceived femininity and masculinity contribute independently to facial impressions. *Journal of Experimental Psychology: General*. Advance online publication.
- Heyder, A., van Hek, M., & Van Houtte, M. (2021). When gender stereotypes get male adolescents into trouble: A longitudinal study on gender conformity pressure as a predictor of school misconduct. *Sex Roles*, *84*(1), 61-75.
- Hille, E., Elbertse, L., Gravenhorst, J. B., Brand, R., & Verloove-Vanhorick, S. (2005).
 Nonresponse bias in a follow-up study of 19-year-old adolescents born as preterm infants. *Pediatrics*, *116*(5), 662-666.

- Hilt, L. M., McLaughlin, K. A., & Nolen-Hoeksema, S. (2010). Examination of the response styles theory in a community sample of young adolescents. *Journal of Abnormal Child Psychology*, 38(4), 545-556.
- Horgan, A., Kelly, P., Goodwin, J., & Behan, L. (2018). Depressive symptoms and suicidal ideation among Irish undergraduate college students. *Issues in Mental Health Nursing*, 39(7), 575-584.
- Hu, T., Zhang, D., Wang, J., Mistry, R., Ran, G., & Wang, X. (2014). Relation between emotion regulation and mental health: A meta-analysis review. *Psychological Reports*, 114(2), 341-362.
- Huffziger, S., Reinhard, I., & Kuehner, C. (2009). A longitudinal study of rumination and distraction in formerly depressed inpatients and community controls. *Journal of Abnormal Psychology*, 118(4), 746-756.
- Hunt, K., Sweeting, H., Keoghan, M., & Platt, S. (2006). Sex, gender role orientation, gender role attitudes and suicidal thoughts in three generations. *Social Psychiatry and Psychiatric Epidemiology*, 41(8), 641-647.
- Idig-Camuroglu, M., & Gölge, Z. B. (2018). Non-suicidal self-injury among university students in Turkey: The effect of gender and childhood abuse. *Psychiatria Danubina*, 30(4), 410-420.
- In, S., Hur, J.-W., Kim, G., & Lee, J.-H. (2021). Effects of distraction, cognitive reappraisal, and acceptance on the urge to self-harm and negative affect in nonsuicidal self-injury. *Current Psychology*, 1-8.
- Ivcevic, Z., & Brackett, M. (2014). Predicting school success: Comparing conscientiousness, grit, and emotion regulation ability. *Journal of Research in Personality*, *52*, 29-36.
- Ivcevic, Z., & Eggers, C. (2021). Emotion regulation ability: Test performance and observer reports in predicting relationship, achievement and well-being outcomes in

adolescents. International Journal of Environmental Research and Public Health, 18(6), 3204.

Izard, C. E. (1971). The face of emotion. Appleton-Century-Crofts.

- Izard, C. E. (1992). Basic emotions, relations among emotions, and emotion-cognition relations. *Psychological Review*, 99(3), 561–565.
- Jack, D. C. (2001). Understanding women's anger: A description of relational patterns. *Health Care for Women International*, 22(4), 385-400.
- Jackson, D. C., Malmstadt, J. R., Larson, C. L., & Davidson, R. J. (2000). Suppression and enhancement of emotional responses to unpleasant pictures. *Psychophysiology*, 37(4), 515-522.
- Jafari, M., & Ansari-Pour, N. (2019). Why, when and how to adjust your *P* values? *Cell Journal*, 20(4), 604-607.
- James, W. (1884). What is an emotion? *Mind*(34), 188-205.
- John, O. P., & Gross, J. J. (2004). Healthy and unhealthy emotion regulation: Personality processes, individual differences, and life span development. *Journal of Personality*, 72(6), 1301-1334.
- Johnson, D. P., & Whisman, M. A. (2013). Gender differences in rumination: A metaanalysis. *Personality and Individual differences*, 55(4), 367-374.
- Johnson, J. L., Greaves, L., & Repta, R. (2009). Better science with sex and gender:Facilitating the use of a sex and gender-based analysis in health research.*International Journal for Equity in Health*, 8(1), 1-11.
- Johnson, S. L., Tharp, J. A., Peckham, A. D., & McMaster, K. J. (2016). Emotion in bipolar I disorder: Implications for functional and symptom outcomes. *Journal of Abnormal Psychology*, 125(1), 40.

- Jordan, J. T., & McNiel, D. E. (2020). Characteristics of persons who die on their first suicide attempt: Results from the National Violent Death Reporting System. *Psychological Medicine*, 50(8), 1390-1397.
- Jorm, A. F. (2012). Mental health literacy: Empowering the community to take action for better mental health. *American Psychologist*, 67(3), 231.
- Jose, P. E., & Brown, I. (2008). When does the gender difference in rumination begin? Gender and age differences in the use of rumination by adolescents. *Journal of Youth and Adolescence*, *37*(2), 180-192.
- Joseph, D. L., Chan, M. Y., Heintzelman, S. J., Tay, L., Diener, E., & Scotney, V. S. (2020). The manipulation of affect: A meta-analysis of affect induction procedures. *Psychological Bulletin*, 146(4), 355.
- Just, N., & Alloy, L. B. (1997). The response styles theory of depression: Tests and an extension of the theory. *Journal of Abnormal Psychology*, *106*(2), 221.
- Kalra, G., Christodoulou, G., Jenkins, R., Tsipas, V., Christodoulou, N., Lecic-Tosevski, D.,
 Mezzich, J., & Bhugra, D. (2012). Mental health promotion: guidance and strategies.
 European Psychiatry, 27(2), 81-86.
- Kanske, P., Heissler, J., Schönfelder, S., & Wessa, M. (2012). Neural correlates of emotion regulation deficits in remitted depression: The influence of regulation strategy, habitual regulation use, and emotional valence. *Neuroimage*, *61*(3), 686-693.
- Kanwal, R., & Aslam, N. (2015). Mediating role of depression in the relationship between brooding rumination and suicidal ideation. *Depression*, *38*(50), 58-62.
- Kao, K., Tuladhar, C. T., & Tarullo, A. R. (2020). Parental and family-level sociocontextual correlates of emergent emotion regulation: Implications for early social competence. *Journal of Child and Family Studies*, 29(6), 1630-1641.

- Kapur, N., Cooper, J., O'Connor, R. C., & Hawton, K. (2013). Non-suicidal self-injury v. attempted suicide: New diagnosis or false dichotomy? *British Journal of Psychiatry*, 202(5), 326-328.
- Kashdan, T. B., Goodman, F. R., Machell, K. A., Kleiman, E. M., Monfort, S. S., Ciarrochi,
 J., & Nezlek, J. B. (2014). A contextual approach to experiential avoidance and social anxiety: Evidence from an experimental interaction and daily interactions of people with social anxiety disorder. *Emotion*, 14(4), 769.
- Kelley, N. J., Glazer, J. E., Pornpattananangkul, N., & Nusslock, R. (2019). Reappraisal and suppression emotion-regulation tendencies differentially predict reward-responsivity and psychological well-being. *Biological Psychology*, 140, 35-47.
- Keltner, D. (2019). Toward a consensual taxonomy of emotions. *Cognition and Emotion*, 33(1), 14-19.
- Keltner, D., Tracy, J. L., Sauter, D., & Cowen, A. (2019). What basic emotion theory really says for the twenty-first century study of emotion. *Journal of Nonverbal Behavior*, 43(2), 195-201.
- Kessler, R. C. (2003). Epidemiology of women and depression. *Journal of Affective Disorders*, 74(1), 5-13.
- Kessler, R. C., Berglund, P., Borges, G., Nock, M., & Wang, P. S. (2005). Trends in suicide ideation, plans, gestures, and attempts in the United States, 1990-1992 to 2001-2003. *Journal of the American Medical Association*, 293(20), 2487-2495.
- Kim, S. M., Han, D. H., Trksak, G. H., & Lee, Y. S. (2014). Gender differences in adolescent coping behaviors and suicidal ideation: Findings from a sample of 73,238 adolescents. *Anxiety, Stress, & Coping, 27*(4), 439-454.
- King, M. F., & Bruner, G. C. (2000). Social desirability bias: A neglected aspect of validity testing. *Psychology & Marketing*, 17(2), 79-103.

Kiosses, D. N., Alexopoulos, G. S., Hajcak, G., Apfeldorf, W., Duberstein, P. R., Putrino, D., & Gross, J. J. (2018). Cognitive reappraisal intervention for suicide prevention (CRISP) for middle-aged and older adults hospitalized for suicidality. *American Journal of Geriatric Psychiatry*, 26(4), 494-503.

- Kivity, Y., & Huppert, J. D. (2016). Does cognitive reappraisal reduce anxiety? A daily diary study of a micro-intervention with individuals with high social anxiety. *Journal of Consulting and Clinical Psychology*, 84(3), 269-283.
- Kivity, Y., & Huppert, J. D. (2019). Emotion regulation in social anxiety: A systematic investigation and meta-analysis using self-report, subjective, and event-related potentials measures. *Cognition and Emotion*, 33(2), 213-230.
- Kjærstad, H. L., Vinberg, M., Goldin, P. R., Køster, N., Støttrup, M. M. D., Knorr, U., Kessing, L. V., & Miskowiak, K. W. (2016). Impaired down-regulation of negative emotion in self-referent social situations in bipolar disorder: A pilot study of a novel experimental paradigm. *Psychiatry Research*, 238, 318-325.
- Klonsky, E. D. (2007). The functions of deliberate self-injury: A review of the evidence. *Clinical Psychology Review*, 27(2), 226-239.
- Klonsky, E. D. (2011). Non-suicidal self-injury in United States adults: Prevalence, sociodemographics, topography and functions. *Psychological Medicine*, 41(9), 1981-1986.
- Knudson, L., Wilson, V., Shields, J., Scholes, A., Rose, J., Elliot, C., & Biggs, H. (2020). Scottish Health Survey, 2020. ScotCen Social Research.
- König, D. (2011). *Die Regulation von negativen und positiven Emotionen*. (Unpublished doctoral disseration). University of Vienna.

Kovács, L. N., Takács, Z. K., Tóth, Z., Simon, E., Schmelowszky, Á., & Kökönyei, G.
 (2020). Rumination in major depressive and bipolar disorder: A meta-analysis.
 Journal of Affective Disorders, 276, 1131-1141.

- Kraaij, V., & Garnefski, N. (2019). The Behavioral Emotion Regulation Questionnaire:
 Development, psychometric properties and relationships with emotional problems and the Cognitive Emotion Regulation Questionnaire. *Personality and Individual differences*, 137, 56-61.
- Kraiss, J. T., Peter, M., Moskowitz, J. T., & Bohlmeijer, E. T. (2020). The relationship between emotion regulation and well-being in patients with mental disorders: A metaanalysis. *Comprehensive Psychiatry*, 102, 152189.
- Kreibig, S. D. (2010). Autonomic nervous system activity in emotion: A review. *Biological psychology*, 84(3), 394-421.
- Kring, A. M., & Gordon, A. H. (1998). Sex differences in emotion: Expression, experience, and physiology. *Journal of Personality and Social Psychology*, 74(3), 686-703.
- Kucharska, J. (2017). Sex differences in the appraisal of traumatic events and psychopathology. *Psychological Trauma: Theory, Research, Practice, and Policy,* 9(5), 575-582.
- Kuckertz, J. M., & Amir, N. (2015). Attention bias modification for anxiety and phobias:Current status and future directions. *Current Psychiatry Reports*, 17(2), 9.
- Kudinova, A. Y., Owens, M., Burkhouse, K. L., Barretto, K. M., Bonanno, G. A., & Gibb, B.
 E. (2016). Differences in emotion modulation using cognitive reappraisal in individuals with and without suicidal ideation: An ERP study. *Cognition and Emotion*, *30*(5), 999-1007.

- Kuehner, C. (1999). Gender differences in the short-term course of unipolar depression in a follow-up sample of depressed inpatients. *Journal of Affective Disorders*, 56(2-3), 127-139.
- Kuhbandner, C., Lichtenfeld, S., & Pekrun, R. (2011). Always look on the broad side of life: Happiness increases the breadth of sensory memory. *Emotion*, *11*(4), 958-964.
- Kullik, A., & Petermann, F. (2013). Attachment to parents and peers as a risk factor for adolescent depressive disorders: The mediating role of emotion regulation. *Child Psychiatry & Human Development, 44*(4), 537-548.
- Kunst, L. E., de Groot, J., & van der Does, A. W. (2019). Ambivalence over expression of anger and sadness mediates gender differences in depressive symptoms. *Cognitive Therapy and Research*, 43(2), 365-373.
- Kuppens, P. (2019). Improving theory, measurement, and reality to advance the future of emotion research. *Cognition and Emotion*, *33*(1), 20-23.
- Kurdek, L. A. (1987). Gender differences in the psychological symptomatology and coping strategies of young adolescents. *Journal of Early Adolescence*, *7*(4), 395-410.
- Kwon, H., Yoon, K. L., Joormann, J., & Kwon, J.-H. (2013). Cultural and gender differences in emotion regulation: Relation to depression. *Cognition & Emotion*, 27(5), 769-782.
- Laerd Statistics (2016). Chi-square test for association using SPSS Statistics. Statistical tutorials and software guides. <u>https://statistics.laerd.com/</u>
- Laerd Statistics (2017). Cochran's Q test using SPSS Statistics. Statistical tutorials and software guides. <u>https://statistics.laerd.com/</u>
- Lamal, P. A. (1990). On the importance of replication. *Journal of Social Behavior and Personality*, 5(4), 31.

- Lang, P. J., Bradley, M. M., & Cuthbert, B. N. (1997). International affective picture system (IAPS): Technical manual and affective ratings. *NIMH Center for the Study of Emotion and Attention*, 1, 39-58.
- Lang, P. J., Greenwald, M. K., Bradley, M. M., & Hamm, A. O. (1993). Looking at pictures:
 Affective, facial, visceral, and behavioral reactions. *Psychophysiology*, *30*(3), 261-273.
- Langer, K., Hagedorn, B., Stock, L.-M., Otto, T., Wolf, O. T., & Jentsch, V. L. (2020). Acute stress improves the effectivity of cognitive emotion regulation in men. *Scientific Reports*, 10(1), 1-13.
- Langeslag, S. J., & van Strien, J. W. (2018). Cognitive reappraisal of snake and spider pictures: An event-related potentials study. *International Journal of Psychophysiology*, 130, 1-8.
- Larsen, J. K., Vermulst, A. A., Geenen, R., van Middendorp, H., English, T., Gross, J. J., Ha, T., Evers, C., & Engels, R. C. (2013). Emotion regulation in adolescence: A prospective study of expressive suppression and depressive symptoms. *Journal of Early Adolescence*, *33*(2), 184-200.
- Latimer, S., Meade, T., & Tennant, A. (2013). Measuring engagement in deliberate self-harm behaviours: Psychometric evaluation of six scales. *BioMed Central Psychiatry*, 13(1), 1-11.
- Lawyer, S. R., & Jenks, C. W. (2020). Emotion suppression decreases delay discounting for monetary outcomes. *Psychological Record*, *70*(1), 1-10.
- Lazarus, R. S. (1991). Cognition and motivation in emotion. *American Psychologist*, *46*(4), 352-367.
- Lazarus, R. S., & Alfert, E. (1964). Short-circuiting of threat by experimentally altering cognitive appraisal. *Journal of Abnormal and Social Psychology*, 69(2), 195-205.

Lazarus, R. S., & Folkman, S. (1984). Stress, appraisal, and coping: Springer Publishing.

- Leaberry, K. D., Rosen, P. J., Slaughter, K. E., Reese, J., & Fogleman, N. D. (2019). Temperamental negative affect, emotion-specific regulation, and concurrent internalizing and externalizing pathology among children with ADHD. *ADHD Attention Deficit and Hyperactivity Disorders*, 11(3), 311-324.
- Leach, L. S., Christensen, H., Mackinnon, A. J., Windsor, T. D., & Butterworth, P. (2008).
 Gender differences in depression and anxiety across the adult lifespan: The role of psychosocial mediators. *Social Psychiatry and Psychiatric Epidemiology*, 43(12), 983-998.
- LeBlanc, S., Uzun, B., Aydemir, A., & Mohiyeddini, C. (2020). Validation of an emotion regulation training program on mental well-being. *Psychological Reports*, 123(5), 1518-1536.
- LeBlanc, S., Uzun, B., Pourseied, K., & Mohiyeddini, C. (2017). Effect of an emotion regulation training program on mental well-being. *International Journal of Group Psychotherapy*, 67(1), 108-123.
- LeBreton, J. M., Wu, J., & Bing, M. N. (2009). The truth(s) on testing for mediation in the social and organizational sciences. In C. E. Lance, & R. J. Vanderberg (Eds.),
 Statistical and methodological myths and urban legends: Doctrine, verity, and fable in the organizational and social sciences (pp. 109-144). Routledge.
- LeDoux, J. E. (2000). Emotion circuits in the brain. *Annual Review of Neuroscience*, 23(1), 155-184.
- LeDoux, J. E., & Hofmann, S. G. (2018). The subjective experience of emotion: A fearful view. *Current Opinion in Behavioral Sciences*, *19*, 67-72.
- Lee, C. S., & Wong, Y. J. (2020). Racial/ethnic and gender differences in the antecedents of youth suicide. *Cultural Diversity and Ethnic Minority Psychology*, 26(4), 532-543.

- Lee, S., Dwyer, J., Paul, E., Clarke, D., Treleaven, S., & Roseby, R. (2019). Differences by age and sex in adolescent suicide. *Australian and New Zealand Journal of Public Health*, 43(3), 248-253.
- Lench, H. C., Bench, S. W., Darbor, K. E., & Moore, M. (2015). A functionalist manifesto:
 Goal-related emotions from an evolutionary perspective. *Emotion Review*, 7(1), 90-98.
- Lench, H. C., Flores, S. A., & Bench, S. W. (2011). Discrete emotions predict changes in cognition, judgment, experience, behavior, and physiology: A meta-analysis of experimental emotion elicitations. *Psychological Bulletin*, 137(5), 834-855.
- Lench, H. C., Tibbett, T. P., & Bench, S. W. (2016). Exploring the toolkit of emotion: What do sadness and anger do for us? *Social and Personality Psychology Compass*, 10(1), 11-25.
- Levenson, R. (1994). Human emotions: A functional view. In P. Ekman, & R. J. Davidson (Eds.), *The nature of emotion: Fundamental questions* (pp. 123-126). Oxford University Press.
- Levenson, R. W. (2014). The autonomic nervous system and emotion. *Emotion Review*, 6(2), 100-112.
- Leventhal, H. (1984). A perceptual-motor theory of emotion. *Advances in Experimental Social Psychology, 17*, 117-182.
- Lewinsohn, P. M., Rohde, P., Seeley, J. R., & Baldwin, C. L. (2001). Gender differences in suicide attempts from adolescence to young adulthood. *Journal of the American Academy of Child & Adolescent Psychiatry*, 40(4), 427-434.
- Li, C.P. (2018). Exploring the gender difference in fear of crime among older people. International Journal of Management, Economics and Social Sciences, 7, 26-39.

- Li, C.Q., Zhang, J.S., Ma, S., Lv, R. R., Duan, J. L., Luo, D.M., Yan, X-J., Ma, N., & Song,
 Y. (2020). Gender differences in self-harm and drinking behaviors among high school students in Beijing, China. *BioMed Central Public Health*, 20(1), 1-11.
- Li, H., Wei, D., Browning, M., Du, X., Zhang, Q., & Qiu, J. (2016). Attentional bias modification (ABM) training induces spontaneous brain activity changes in young women with subthreshold depression: A randomized controlled trial. *Psychological Medicine*, 46(5).
- Lichtenfeld, S., Pekrun, R., Stupnisky, R. H., Reiss, K., & Murayama, K. (2012). Measuring students' emotions in the early years: The achievement emotions questionnaireelementary school (AEQ-ES). *Learning and Individual Differences*, 22(2), 190-201.
- Liddon, L., Kingerlee, R., & Barry, J. A. (2018). Gender differences in preferences for psychological treatment, coping strategies, and triggers to help-seeking. *British Journal of Clinical Psychology*, 57(1), 42-58.
- Lieb, K., Zanarini, M. C., Schmahl, C., Linehan, M. M., & Bohus, M. (2004). Borderline personality disorder. *The Lancet*, 364(9432), 453-461.
- Lieberman, M. D. (2019). Boo! The consciousness problem in emotion. *Cognition and Emotion, 33*(1), 24-30.
- Linehan, M. M. (2018). *Cognitive-behavioral treatment of borderline personality disorder*. Guilford Publications.
- Linehan, M. M., Heard, H. L., & Armstrong, H. E. (1993). Naturalistic follow-up of a behavioral treatment for chronically parasuicidal borderline patients. *Archives of General Psychiatry*, 50(12), 971-974.
- Lipp, O. V., & Derakshan, N. (2005). Attentional bias to pictures of fear-relevant animals in a dot probe task. *Emotion*, *5*(3), 365.

- Liu, D. Y., Gilbert, K. E., & Thompson, R. J. (2020). Emotion differentiation moderates the effects of rumination on depression: A longitudinal study. *Emotion*, *20*(7), 1234.
- Lu, L., Xu, L., Luan, X., Sun, L., Li, J., Qin, W., Zhang, J., Jing, X., Wang, Y., Xia, Y., Li,
 Y, & Jiao, A. (2020). Gender difference in suicidal ideation and related factors among rural elderly: A cross-sectional study in Shandong, China. *Annals of General Psychiatry*, 19(1), 2.
- Lundh, L. G., Karim, J., & Quilisch, E. (2007). Deliberate self-harm in 15-year-old adolescents: A pilot study with a modified version of the Deliberate Self-Harm Inventory. *Scandinavian Journal of Psychology*, 48(1), 33-41.

MacKinnon, D. P. (2008). Introduction to statistical mediation analysis. Routledge.

- MacKinnon, D. P., Lockwood, C. M., & Williams, J. (2004). Confidence limits for the indirect effect: Distribution of the product and resampling methods. *Multivariate Behavioral Research*, 39(1), 99-128.
- Macklem, G. L. (2011). Evidence-based school mental health services: Affect education, emotion regulation training, and cognitive behavioral therapy. Springer.
- Madge, N., Hewitt, A., Hawton, K., Wilde, E. J. d., Corcoran, P., Fekete, S., van Heeringen, K., De Leo, D., Ystgaard, M. (2008). Deliberate self-harm within an international community sample of young people: Comparative findings from the Child & Adolescent Self-harm in Europe (CASE) Study. *Journal of Child Psychology and Psychiatry*, 49(6), 667-677.
- Magnée, M. J., Stekelenburg, J. J., Kemner, C., & de Gelder, B. (2007). Similar facial electromyographic responses to faces, voices, and body expressions. *Neuroreport*, 18(4), 369-372.
- Maksimović, M. Ž., Vlajinac, H. D., Radak, Đ. J., Maksimović, J. M., Marinković, J. M., & Jorga, J. B. (2008). Association of socioeconomic status measured by education and

risk factors for carotid atherosclerosis: Cross-sectional study. *Croatian Medical Journal*, 49(6), 824.

- Malesza, M. (2019). Relationship between emotion regulation, negative affect, gender and delay discounting. *Current Psychology*, 1-9.
- Marchewka, A., Żurawski, Ł., Jednoróg, K., & Grabowska, A. (2014). The Nencki Affective Picture System (NAPS): Introduction to a novel, standardized, wide-range, highquality, realistic picture database. *Behavior Research Methods*, *46*(2), 596-610.
- Marcus, S. M., Young, E. A., Kerber, K. B., Kornstein, S., Farabaugh, A. H., Mitchell, J., Wisniewski, S. R., Balasubramani, G. K., Trivedi, M. H., Rush, A. J. (2005). Gender differences in depression: Findings from the STAR* D study. *Journal of Affective Disorders*, 87(2-3), 141-150.
- Markanday, S., Brennan, S. L., Gould, H., & Pasco, J. A. (2013). Sex-differences in reasons for non-participation at recruitment: Geelong Osteoporosis Study. *BioMed Central Research Notes*, 6(1), 1-7.
- Mars, B., Heron, J., Klonsky, E. D., Moran, P., O'Connor, R. C., Tilling, K., Wilkinson, P., Gunnell, D. (2019). Predictors of future suicide attempt among adolescents with suicidal thoughts or non-suicidal self-harm: A population-based birth cohort study. *The Lancet Psychiatry*, 6(4), 327-337.
- Martin, R. C., & Dahlen, E. R. (2005). Cognitive emotion regulation in the prediction of depression, anxiety, stress, and anger. *Personality and Individual differences*, 39(7), 1249-1260.
- Martin, R. E., & Ochsner, K. N. (2016). The neuroscience of emotion regulation development: Implications for education. *Current Opinion in Behavioral Sciences*, 10, 142-148.

- Mauss, I. B., Cook, C. L., Cheng, J. Y., & Gross, J. J. (2007a). Individual differences in cognitive reappraisal: Experiential and physiological responses to an anger provocation. 66(2), 116-124.
- Mauss, I. B., Cook, C. L., & Gross, J. J. (2007b). Automatic emotion regulation during anger provocation. *Journal of Experimental Social Psychology*, *43*(5), 698-711.
- Mauss, I. B., & Robinson, M. D. (2009). Measures of emotion: A review. Cognition and Emotion, 23(2), 209-237.
- McClure, E. B. (2000). A meta-analytic review of sex differences in facial expression processing and their development in infants, children, and adolescents. *Psychological Bulletin*, 126(3), 424-453.
- McHugh, M. L. (2012). Interrater reliability: The kappa statistic. *Biochemia medica*, 22(3), 276-282.
- McHugh, R. K., Votaw, V. R., Sugarman, D. E., & Greenfield, S. F. (2018). Sex and gender differences in substance use disorders. *Clinical Psychology Review*, *66*, 12-23.
- McKay, M. T., & Andretta, J. R. (2017). Evidence for the psychometric validity, internal consistency and measurement invariance of Warwick Edinburgh Mental Well-being Scale scores in Scottish and Irish adolescents. *Psychiatry Research*, 255, 382-386.
- McLaughlin, K. A., Hatzenbuehler, M. L., Mennin, D. S., & Nolen-Hoeksema, S. (2011).
 Emotion dysregulation and adolescent psychopathology: A prospective study.
 Behaviour Research and Therapy, 49(9), 544-554.
- McLean, C. P., & Anderson, E. R. (2009). Brave men and timid women? A review of the gender differences in fear and anxiety. *Clinical Psychology Review*, 29(6), 496-505.
- McLenon, J., & Rogers, M. A. (2019). The fear of needles: A systematic review and metaanalysis. *Journal of Advanced Nursing*, 75(1), 30-42.

- McManus, S., Lubian, K., Bennett, C., Turley, C., Porter, L., Gill, V., Gunnell, D., & Weich,S. (2019). Suicide and self-harm in Britain: Researching risk and resilience using UK surveys. NatCen.
- McNeil, J., Bailey, L., Ellis, S., Morton, J., & Regan, M. (2012). *Trans mental health study* 2012. Scottish Transgender Alliance. <u>http://www.scottishtrans.org/wp-</u> <u>content/uploads/2013/03/trans_mh_study.pdf</u>
- McRae, K. (2013). Emotion regulation frequency and success: Separating constructs from methods and time scale. *Social and Personality Psychology Compass*, 7(5), 289-302.
- McRae, K., Ciesielski, B., & Gross, J. J. (2012a). Unpacking cognitive reappraisal: Goals, tactics, and outcomes. *Emotion*, *12*(2), 250.
- McRae, K., & Gross, J. J. (2020). Emotion regulation. Emotion, 20(1), 1-9.
- McRae, K., Hughes, B., Chopra, S., Gabrieli, J. D., Gross, J. J., & Ochsner, K. N. (2010). The neural bases of distraction and reappraisal. *Journal of Cognitive Neuroscience*, 22(2), 248-262.
- McRae, K., Jacobs, S. E., Ray, R. D., John, O. P., & Gross, J. J. (2012). Individual differences in reappraisal ability: Links to reappraisal frequency, well-being, and cognitive control. *Journal of Research in Personality*, 46(1), 2-7.
- McRae, K., Ochsner, K. N., Mauss, I. B., Gabrieli, J. J., & Gross, J. J. (2008). Gender differences in emotion regulation: An fMRI study of cognitive reappraisal. *Group Processes & Intergroup Relations*, 11(2), 143-162.
- Melka, S. E., Lancaster, S. L., Bryant, A. R., & Rodriguez, B. F. (2011). Confirmatory factor and measurement invariance analyses of the Emotion Regulation Questionnaire. *Journal of Clinical Psychology*, 67(12), 1283-1293.
- Mental Health Foundation. (2016). *Fundamental facts about mental health 2016*. Mental Health Foundation.

- Meyers-Levy, J., & Loken, B. (2015). Revisiting gender differences: What we know and what lies ahead. *Journal of Consumer Psychology*, 25(1), 129-149.
- Mikels, J. A., Fredrickson, B. L., Larkin, G. R., Lindberg, C. M., Maglio, S. J., & Reuter-Lorenz, P. A. (2005). Emotional category data on images from the International Affective Picture System. *Behavior Research Methods*, 37(4), 626-630.
- Mikkelsen, M. B., O'Toole, M. S., & Mehlsen, M. (2020). Comparing negative emotion differentiation in young and older individuals: A picture-based study. *Scandinavian Journal of Psychology*, 61(4), 513-517.
- Miranda-Mendizabal, A., Castellví, P., Parés-Badell, O., Alayo, I., Almenara, J., Alonso, I., Blasco, M. J., Cebrià, A., Gabilondo, A., Gili, M., Lagares, C., Piqueras, J. A., Rodríguez-Jiménez, T., Rodríguez-Marín, J., Roca, M., Soto-Sanz, V., Vilagut, G, & Alonso, J. (2019). Gender differences in suicidal behavior in adolescents and young adults: Systematic review and meta-analysis of longitudinal studies. *International Journal of Public Health*, 64(2), 265-283.
- Miranda, D., & Claes, M. (2009). Music listening, coping, peer affiliation and depression in adolescence. *Psychology of Music*, 37(2), 215-233.
- Mocaiber, I., Pereira, M. G., Erthal, F. S., Machado-Pinheiro, W., David, I. A., Cagy, M., Volchan, E., & de Oliveira, L. (2010). Fact or fiction? An event-related potential study of implicit emotion regulation. *Neuroscience Letters*, 476(2), 84-88.
- Mogg, K., & Bradley, B. P. (2006). Time course of attentional bias for fear-relevant pictures in spider-fearful individuals. *Behaviour Research and Therapy*, *44*(9), 1241-1250.
- Mogoaşe, C., David, D., & Koster, E. H. (2014). Clinical efficacy of attentional bias modification procedures: An updated meta-analysis. *Journal of Clinical Psychology*, 70(12), 1133-1157.

- Mok, K., Chen, N., Torok, M., McGillivray, L., Zbukvic, I., & Shand, F. (2020). Factors associated with help-seeking for emotional or mental health problems in community members at risk of suicide. *Advances in Mental Health*, 1-11.
- Mok, P. L., Kapur, N., Windfuhr, K., Leyland, A. H., Appleby, L., Platt, S., & Webb, R. T.
 (2012). Trends in national suicide rates for Scotland and for England & Wales, 1960–2008. *British Journal of Psychiatry*, 200(3), 245-251.
- Moodie, C. A., Suri, G., Goerlitz, D. S., Mateen, M. A., Sheppes, G., McRae, K., Lakhan-Pal,
 S., Thiruchselvam, R., & Gross, J. J. (2020). The neural bases of cognitive emotion
 regulation: The roles of strategy and intensity. *Cognitive, Affective, & Behavioral Neuroscience*, 1-21.
- Moran, P., Coffey, C., Romaniuk, H., Olsson, C., Borschmann, R., Carlin, J. B., & Patton, G.
 C. (2012). The natural history of self-harm from adolescence to young adulthood: a population-based cohort study. *The Lancet*, *379*(9812), 236-243.
- Morris, R. R., Schueller, S. M., & Picard, R. W. (2015). Efficacy of a web-based, crowdsourced peer-to-peer cognitive reappraisal platform for depression: Randomized controlled trial. *Journal of Medical Internet Research*, 17(3), 72.
- Mościcki, E. K. (1994). Gender differences in completed and attempted suicides. *Annals of Epidemiology*, *4*(2), 152-158.
- National Records of Scotland. (2021). *Probable Suicides, 2020*. National Records of Scotland. <u>https://www.nrscotland.gov.uk/files//statistics/probable-suicides/2020/suicides-20-report.pdf</u>
- Neacsiu, A. D., Rizvi, S. L., & Linehan, M. M. (2010). Dialectical behavior therapy skills use as a mediator and outcome of treatment for borderline personality disorder. *Behaviour Research and Therapy*, 48(9), 832-839.

- Ng, W., & Diener, E. (2013). Daily use of reappraisal decreases negative emotions toward daily unpleasant events. *Journal of Social and Clinical Psychology*, *32*(5), 530-545.
- Niven, K., Totterdell, P., Stride, C. B., & Holman, D. (2011). Emotion regulation of others and self (EROS): The development and validation of a new individual difference measure. *Current Psychology*, 30(1), 53-73.
- Nolan, R., & Willson, V. L. (1994). Gender and depression in an undergraduate population. *Psychological Reports*, 75(3), 1327-1330.
- Nolan, S. A., Roberts, J. E., & Gotlib, I. H. (1998). Neuroticism and ruminative response style as predictors of change in depressive symptomatology. *Cognitive Therapy and Research*, 22(5), 445-455.
- Nolen-Hoeksema, S. (1987). Sex differences in unipolar depression: Evidence and theory. *Psychological Bulletin, 101*(2), 259-282.
- Nolen-Hoeksema, S. (1991). Responses to depression and their effects on the duration of depressive episodes. *Journal of Abnormal Psychology*, *100*(4), 569-582.
- Nolen-Hoeksema, S. (1994). An interactive model for the emergence of gender differences in depression in adolescence. *Journal of Research on Adolescence*, *4*(4), 519-534.
- Nolen-Hoeksema, S. (2001). Gender differences in depression. *Current Directions in Psychological Science*, 10(5), 173-176.
- Nolen-Hoeksema, S., & Aldao, A. (2011). Gender and age differences in emotion regulation strategies and their relationship to depressive symptoms. *Personality and Individual Differences*, 51(6), 704-708.
- Nolen-Hoeksema, S., & Davis, C. G. (1999). 'Thanks for sharing that': Ruminators and their social support networks. *Journal of Personality and Social Psychology*, 77(4), 801-814.

- Nolen-Hoeksema, S., & Girgus, J. S. (1994). The emergence of gender differences in depression during adolescence. *Psychological Bulletin*, *115*(3), 424-443.
- Nolen-Hoeksema, S., Larson, J., & Grayson, C. (1999). Explaining the gender difference in depression. *Journal of Personality and Social Psychology*, 77, 1061-1072.
- Nolen-Hoeksema, S., & Morrow, J. (1991). A prospective study of depression and posttraumatic stress symptoms after a natural disaster: The 1989 Loma Prieta Earthquake. *Journal of Personality and Social Psychology*, *61*(1), 115-121.
- Nolen-Hoeksema, S., Morrow, J., & Fredrickson, B. L. (1993). Response styles and the duration of episodes of depressed mood. *Journal of Abnormal Psychology*, 102(1), 20-28.
- Nolen-Hoeksema, S., Parker, L. E., & Larson, J. (1994). Ruminative coping with depressed mood following loss. *Journal of Personality and Social Psychology*, 67(1), 92-104.
- Nolen-Hoeksema, S., Wisco, B. E., & Lyubomirsky, S. (2008). Rethinking rumination. *Perspectives on Psychological Science*, *3*(5), 400-424.
- Nolen-Hoeksema, S. (1995). Gender differences in coping with depression across the lifespan. *Depression*, *3*(1-2), 81-90.
- Notarius, C. I., & Levenson, R. W. (1979). Expressive tendencies and physiological response to stress. *Journal of Personality and Social Psychology*, *37*(7), 1204-1210.
- O'Connor, R. C., & Nock, M. K. (2014). The psychology of suicidal behaviour. *The Lancet Psychiatry*, 1(1), 73-85.
- O'Connor, R. C., Rasmussen, S., & Hawton, K. (2010). Predicting depression, anxiety and self-harm in adolescents: The role of perfectionism and acute life stress. *Behaviour Research and Therapy*, *48*(1), 52-59.

- O'Connor, R. C., Rasmussen, S., Miles, J., & Hawton, K. (2009). Self-harm in adolescents:
 Self-report survey in schools in Scotland. *British Journal of Psychiatry*, 194(1), 68-72.
- O'Connor, R. C., Wetherall, K., Cleare, S., Eschle, S., Drummond, J., Ferguson, E.,
 O'Connor, D. B., & O'Carroll, R. E. (2018). Suicide attempts and non-suicidal selfharm: National prevalence study of young adults. *BJPsych Open*, 4(3), 142-148.
- Ochsner, K. N., Bunge, S. A., Gross, J. J., & Gabrieli, J. D. (2002). Rethinking feelings: an FMRI study of the cognitive regulation of emotion. *Journal of Cognitive Neuroscience*, 14(8), 1215-1229.
- Office for National Statistics. (2021). *Measuring equality: A guide for the collection and classification of ethnic group, national identity and religion data in the UK.* <u>https://www.ons.gov.uk/methodology/classificationsandstandards/measuringequality/</u> <u>ethnicgroupnationalidentityandreligion#ethnic-group</u>
- Olatunji, B. O., Berg, H. E., & Zhao, Z. (2017). Emotion regulation of fear and disgust:
 Differential effects of reappraisal and suppression. *Cognition and Emotion*, *31*(2), 403-410.
- Öngen, D. E. (2010). Cognitive emotion regulation in the prediction of depression and submissive behavior: Gender and grade level differences in Turkish adolescents. *Procedia-Social and Behavioral Sciences*, *9*, 1516-1523.
- Oud, M., De Winter, L., Vermeulen-Smit, E., Bodden, D., Nauta, M., Stone, L., van den Heuvel, M., Taher, R. A., de Graaf, I., Kendall, T., Engles, R., & Stikkelbroek, Y. (2019). Effectiveness of CBT for children and adolescents with depression: A systematic review and meta-regression analysis. *European Psychiatry*, *57*, 33-45.
- Panksepp, J. (1982). Toward a general psychobiological theory of emotions. *Behavioral and Brain Sciences*, 5(3), 407-422.

- Parker, G. (2019). The role of environmental and psychosocial factors in depression. In J.
 Quevedo, A. F. Carvalho, & C. A. Zarate (Eds.), *Neurobiology of Depression* (pp. 9-15). Elsevier.
- Parkinson, B., & Totterdell, P. (1999). Classifying affect-regulation strategies. *Cognition & Emotion*, *13*(3), 277-303.
- Parsons, T. D. (2015). Virtual reality for enhanced ecological validity and experimental control in the clinical, affective and social neurosciences. *Frontiers in Human Neuroscience*, 9, 660.
- Pearlin, L. I., & Schooler, C. (1978). The structure of coping. *Journal of Health and Social Behavior*, 2-21.
- Pekrun, R., Lichtenfeld, S., Marsh, H. W., Murayama, K., & Goetz, T. (2017). Achievement emotions and academic performance: Longitudinal models of reciprocal effects. *Child Development*, 88(5), 1653-1670.
- Peled, M., & Moretti, M. M. (2010). Ruminating on rumination: Are rumination on anger and sadness differentially related to aggression and depressed mood? *Journal of Psychopathology and Behavioral Assessment*, 32(1), 108-117.
- Perchtold, C. M., Papousek, I., Fink, A., Weber, H., Rominger, C., & Weiss, E. M. (2019).
 Gender differences in generating cognitive reappraisals for threatening situations:
 Reappraisal capacity shields against depressive symptoms in men, but not women. *Frontiers in Psychology, 10*, 553.
- Perepletchikova, F., Axelrod, S. R., Kaufman, J., Rounsaville, B. J., Douglas-Palumberi, H., & Miller, A. L. (2011). Adapting dialectical behaviour therapy for children: Towards a new research agenda for paediatric suicidal and non-suicidal self-injurious behaviours. *Child and Adolescent Mental Health*, *16*(2), 116-121.
- Peters, B. J., Overall, N. C., Cameron, L. D., Hammond, M. D., Low, R. S., & Girme, Y. U.
 (2019). Do habitual emotional suppression measures predict response-focused
 situational suppression during social interactions? *Emotion*, 20(6), 1005-1019.
- Phan, K. L., Wager, T., Taylor, S. F., & Liberzon, I. (2002). Functional neuroanatomy of emotion: A meta-analysis of emotion activation studies in PET and fMRI. *Neuroimage*, 16(2), 331-348.
- Phillips, K. E. V., & Power, M. J. (2007). A new self-report measure of emotion regulation in adolescents: The regulation of emotions questionnaire. *Clinical Psychology & Psychotherapy*, 14(2), 145-156.
- Plate, A. J., & Aldao, A. (2017). Emotion regulation in cognitive-behavioral therapy:
 Bridging the gap between treatment studies and laboratory experiments. In S. G.
 Hofmann, & G. J. G. Asmundson (Eds.), *The Science of Cognitive Behavioral Therapy* (pp. 107-127). Elsevier.
- Polanco-Roman, L., Jurska, J., Quiñones, V., & Miranda, R. (2015). Brooding, reflection, and distraction: Relation to non-suicidal self-injury versus suicide attempts. *Archives of Suicide Research*, 19(3), 350-365.
- Porter, S. R., & Whitcomb, M. E. (2005). Non-response in student surveys: The role of demographics, engagement and personality. *Research in Higher Education*, 46(2), 127-152.
- Preece, D. A., Becerra, R., Robinson, K., & Gross, J. J. (2019). The emotion regulation questionnaire: Psychometric properties in general community samples. *Journal of Personality Assessment*, 102(3), 348-356.
- Quinn, M. E., & Joormann, J. (2020). Executive control under stress: Relation to reappraisal ability and depressive symptoms. *Behaviour Research and Therapy*, *131*, 103634.

- Radloff, L. S. (1977). The CES-D scale: A self-report depression scale for research in the general population. *Applied Psychological Measurement*, *1*(3), 385-401.
- Ranney, R. M., Bruehlman-Senecal, E., & Ayduk, O. (2017). Comparing the effects of three online cognitive reappraisal trainings on well-being. *Journal of Happiness Studies*, *18*(5), 1319-1338.
- Rasmussen, S. A., Fraser, L., Gotz, M., MacHale, S., Mackie, R., Masterton, G., McConachie, S., & O'Connor, R. C. (2010). Elaborating the cry of pain model of suicidality: Testing a psychological model in a sample of first-time and repeat selfharm patients. *British Journal of Clinical Psychology*, 49(1), 15-30.
- Rawana, J. S., Flett, G. L., McPhie, M. L., Nguyen, H. T., & Norwood, S. J. (2014).
 Developmental trends in emotion regulation: A systematic review with implications for community mental health. *Canadian Journal of Community Mental Health*, 33(1), 31-44.
- Reinherz, H. Z., Giaconia, R. M., Silverman, A. B., Friedman, A., Pakiz, B., Frost, A. K., & Cohen, E. (1995). Early psychosocial risks for adolescent suicidal ideation and attempts. *Journal of the American Academy of Child & Adolescent Psychiatry*, 34(5), 599-611.
- Reinherz, H. Z., Tanner, J. L., Berger, S. R., Beardslee, W. R., & Fitzmaurice, G. M. (2006).
 Adolescent suicidal ideation as predictive of psychopathology, suicidal behavior, and compromised functioning at age 30. *American Journal of Psychiatry*, *163*(7), 1226-1232.
- Reja, U., Manfreda, K. L., Hlebec, V., & Vehovar, V. (2003). Open-ended vs. close-ended questions in web questionnaires. *Developments in Applied Statistics*, 19(1), 160-117.
- Reynolds, W. M. (1982). Development of reliable and valid short forms of the Marlowe-Crowne Social Desirability Scale. *Journal of Clinical Psychology*, *38*(1), 119-125.

- Richards, J. M., & Gross, J. J. (2000). Emotion regulation and memory: The cognitive costs of keeping one's cool. *Journal of Personality and Social Psychology*, *79*(3), 410-424.
- Richmond, A. S., Broussard, K. A., Sterns, J. L., Sanders, K. K., & Shardy, J. C. (2015). Who are we studying? Sample diversity in teaching of psychology research. *Teaching of Psychology*, 42(3), 218-226.
- Riegel, M., Żurawski, Ł., Wierzba, M., Moslehi, A., Klocek, Ł., Horvat, M., Grabowska, A.,
 Michałowski, J., Jednoróg, K., & Marchewka, A. (2016). Characterization of the
 Nencki Affective Picture System by discrete emotional categories. *Behavior Research Methods*, 48(2), 600-612.
- Dattani, S., Ritchie, H., & Roser, M. (2021). *Mental Health*. https://ourworldindata.org/mental-health
- Rivers, S. E., Brackett, M. A., Katulak, N. A., & Salovey, P. (2007). Regulating anger and sadness: An exploration of discrete emotions in emotion regulation. *Journal of Happiness Studies*, 8(3), 393-427.
- Rizvi, S. L., & Fitzpatrick, S. (2021). Changes in suicide and non-suicidal self-injury ideation and the moderating role of specific emotions over the course of dialectical behavior therapy. *Suicide and Life-Threatening Behavior*, 51(3), 429-445.

Robertson, A. S. (2019). Disorders of the autonomic nervous system: Routledge.

- Robinson, M. D., & Clore, G. L. (2002). Belief and feeling: Evidence for an accessibility model of emotional self-report. *Psychological Bulletin*, 128(6), 934-960.
- Roelofs, J., Muris, P., Huibers, M., Peeters, F., & Arntz, A. (2006). On the measurement of rumination: A psychometric evaluation of the ruminative response scale and the rumination on sadness scale in undergraduates. *Journal of Behavior Therapy and Experimental Psychiatry*, 37(4), 299-313.

- Rogier, G., Garofalo, C., & Velotti, P. (2019). Is emotional suppression always bad? A matter of flexibility and gender differences. *Current Psychology*, *38*(2), 411-420.
- Rolstad, S., Adler, J., & Rydén, A. (2011). Response burden and questionnaire length: Is shorter better? A review and meta-analysis. *Value in Health*, *14*(8), 1101-1108.
- Rood, L., Roelofs, J., Bögels, S. M., Nolen-Hoeksema, S., & Schouten, E. (2009). The influence of emotion-focused rumination and distraction on depressive symptoms in non-clinical youth: A meta-analytic review. *Clinical Psychology Review*, 29(7), 607-616.
- Root, A. K., & Denham, S. A. (2010). The role of gender in the socialization of emotion: Key concepts and critical issues. *New Directions for Child and Adolescent Development*, 2010(128), 1-9.
- Roseman, I. (1991). Appraisal determinants of discrete emotions. *Cognition & Emotion*, 5(3), 161-200.
- Roseman, I., & Evdokas, A. (2004). Appraisals cause experienced emotions: Experimental evidence. *Cognition and Emotion*, *18*(1), 1-28.
- Roseman, I., Spindel, M. S., & Jose, P. E. (1990). Appraisals of emotion-eliciting events:
 Testing a theory of discrete emotions. *Journal of Personality and Social Psychology*, 59(5), 899.
- Rosenfield, S. (1980). Sex differences in depression: Do women always have higher rates? Journal of Health and Social Behavior, 33-42.
- Ross, K. M., Kim, H., Tolan, P. H., & Jennings, P. A. (2019). An exploration of normative social and emotional skill growth trajectories during adolescence. *Journal of Applied Developmental Psychology*, 62, 102-115.

- Rucker, D. D., Preacher, K. J., Tormala, Z. L., & Petty, R. E. (2011). Mediation analysis in social psychology: Current practices and new recommendations. *Social and Personality Psychology Compass*, 5(6), 359-371.
- Russell, J. A. (1980). A circumplex model of affect. *Journal of Personality and Social Psychology*, *39*(6), 1161.
- Russell, J. A. (2003). Core affect and the psychological construction of emotion. *Psychological Review*, *110*(1), 145.
- Russell, J. A., & Barrett, L. F. (1999). Core affect, prototypical emotional episodes, and other things called emotion: Dissecting the elephant. *Journal of Personality and Social Psychology*, 76(5), 805.
- Rusting, C. L., & Nolen-Hoeksema, S. (1998). Regulating responses to anger: Effects of rumination and distraction on angry mood. *Journal of Personality and Social Psychology*, 74(3), 790.
- Sacher, J., Neumann, J., Okon-Singer, H., Gotowiec, S., & Villringer, A. (2013). Sexual dimorphism in the human brain: Evidence from neuroimaging. *Magnetic Resonance Imaging*, 31(3), 366-375.
- Salk, R. H., Hyde, J. S., & Abramson, L. Y. (2017). Gender differences in depression in representative national samples: Meta-analyses of diagnoses and symptoms. *Psychological Bulletin*, 143(8), 783.
- Sanborn-Overby, U. A., & Powlishta, K. K. (2020). Differential reactions to male and female gender-role violations: Testing the sexual orientation hypothesis. *Archives of Sexual Behavior*, 1-16.
- Sanchis-Sanchis, A., Grau, M. D., Moliner, A.-R., & Morales-Murillo, C. P. (2020). Effects of age and gender in emotion regulation of children and adolescents. *Frontiers in Psychology*, 11, 946.

- Sarkheil, P., Klasen, M., Schneider, F., Goebel, R., & Mathiak, K. (2019). Amygdala response and functional connectivity during cognitive emotion regulation of aversive image sequences. *European Archives of Psychiatry and Clinical Neuroscience*, 269(7), 803-811.
- Scherer, K. R. (1999). Appraisal theory. In T. Dalgleish & M. J. Power (Eds.), Handbook of Cognition and Emotion (pp. 637–663). John Wiley & Sons Ltd.
- Scherer, K. R. (2005). What are emotions? And how can they be measured? *Social Science Information*, 44(4), 695-729.
- Scherer, K. R. (2019). Studying appraisal-driven emotion processes: Taking stock and moving to the future. *Cognition and Emotion*, *33*(1), 31-40.
- Scherer, K. R., Ellgring, H., Dieckmann, A., Unfried, M., & Mortillaro, M. (2019). Dynamic facial expression of emotion and observer inference. *Frontiers in Psychology*, 10, 508-508.
- Scherer, K. R., & Moors, A. (2019). The emotion process: Event appraisal and component differentiation. *Annual Review of Psychology*, *70*, 719-745.
- Scherer, K. R., Schorr, A., & Johnstone, T. (2001). Appraisal processes in emotion: Theory, methods, research: University Press.
- Schirda, B., Valentine, T. R., Aldao, A., & Prakash, R. S. (2016). Age-related differences in emotion regulation strategies: Examining the role of contextual factors. *Developmental Psychology*, 52(9), 1370-1380.
- Schwartz, G. E., Brown, S. L., & Ahern, G. L. (1980). Facial muscle patterning and subjective experience during affective imagery: Sex differences. *Psychophysiology*, *17*(1), 75-82.
- Seitz, J., Cetin-Karayumak, S., Lyall, A., Pasternak, O., Baxi, M., Vangel, M., Pearlson, G, Tamminga, C., Sweeney, J., Clementz, B., Schretlen, D., Viher, P. V., Stegmayer, K.,

Walther, S., Lee, J., Crow, T., James, A., Voineskos, A., Buchanan, R. W., . . . Szeszko, P. R. (2020). Investigating sexual dimorphism of human white matter in a harmonized, multisite diffusion magnetic resonance imaging study. *Cerebral cortex, 31*(1), 201-212.

- Shansky, R. M. (2020). Sex differences in amygdala structure and function: From rodents to humans. In J. H. Urban & J. A. Rosenkranz (Eds.), *Handbook of Behavioral Neuroscience, Vol. 2* (pp. 275-284). Elsevier.
- Shapero, B. G., Stange, J. P., McArthur, B. A., Abramson, L. Y., & Alloy, L. B. (2019).
 Cognitive reappraisal attenuates the association between depressive symptoms and emotional response to stress during adolescence. *Cognition and Emotion*, *33*(3), 524-535.
- Sharkin, B. S. (1993). Anger and gender: Theory, research, and implications. *Journal of Counseling & Development*, 71(4), 386-389.
- Sheppes, G., Catran, E., & Meiran, N. (2009). Reappraisal (but not distraction) is going to make you sweat: Physiological evidence for self-control effort. *International Journal* of Psychophysiology, 71(2), 91-96.
- Sheppes, G., & Meiran, N. (2007). Better late than never? On the dynamics of online regulation of sadness using distraction and cognitive reappraisal. *Personality and Social Psychology Bulletin*, 33(11), 1518-1532.
- Sheppes, G., & Meiran, N. (2008). Divergent cognitive costs for online forms of reappraisal and distraction. *Emotion*, 8(6), 870.
- Sheppes, G., Scheibe, S., Suri, G., & Gross, J. J. (2011). Emotion-regulation choice. *Psychological Science*, 22(11), 1391-1396.

- Sheppes, G., Scheibe, S., Suri, G., Radu, P., Blechert, J., & Gross, J. J. (2014). Emotion regulation choice: A conceptual framework and supporting evidence. *Journal of Experimental Psychology: General*, 143(1), 163.
- Sheppes, G., Suri, G., & Gross, J. J. (2015). Emotion regulation and psychopathology. *Annual Review of Clinical Psychology*, *11*, 379-405.
- Shiffman, S., Stone, A. A., & Hufford, M. R. (2008). Ecological momentary assessment. Annual Review of Clinical Psychology, 4, 1-32.
- Shin, L. M., & Liberzon, I. (2010). The neurocircuitry of fear, stress, and anxiety disorders. *Neuropsychopharmacology*, *35*(1), 169-191.
- Shrout, P. E., & Bolger, N. (2002). Mediation in experimental and nonexperimental studies: New procedures and recommendations. *Psychological Methods*, 7(4), 422-445.
- Siedlecka, E., & Denson, T. F. (2019). Experimental methods for inducing basic emotions: A Qualitative review. *Emotion Review*, *11*(1), 87-97.
- Silverman, M. J. (2020). Music-based affect regulation and unhealthy music use explain coping strategies in adults with mental health conditions. *Community Mental Health Journal*, 56(5), 939-946.
- Silvers, J. A., & Moreira, J. F. G. (2019). Capacity and tendency: A neuroscientific framework for the study of emotion regulation. *Neuroscience Letters*, *693*, 35-39.
- Simon, G. E., Specht, C., & Doederlein, A. (2016). Coping with suicidal thoughts: A survey of personal experience. *Psychiatric Services*, 67(9), 1026-1029.
- Sivertsen, B., Hysing, M., Knapstad, M., Harvey, A. G., Reneflot, A., Lønning, K. J., & O'Connor, R. C. (2019). Suicide attempts and non-suicidal self-harm among university students: Prevalence study. *BJPsych Open*, 5(2), 26.

- Slee, N., Garnefski, N., Spinhoven, P., & Arensman, E. (2008). The influence of cognitive emotion regulation strategies and depression severity on deliberate self-harm. *Suicide* and Life-Threatening Behavior, 38(3), 274-286.
- Smith, C. A., & Ellsworth, P. C. (1985). Patterns of cognitive appraisal in emotion. *Journal* of Personality and Social Psychology, 48(4), 813-838.
- Smith, G. (2008). Does gender influence online survey participation?: A record-linkage analysis of university faculty online survey response behavior. (Unpublished doctoral dissertion). San Jose State University.
- Smoski, M. J., LaBar, K. S., & Steffens, D. C. (2014). Relative effectiveness of reappraisal and distraction in regulating emotion in late-life depression. *American Journal of Geriatric Psychiatry*, 22(9), 898-907.
- Smrtnik-Vitulić, H., & Prosen, S. (2016). Coping and emotion regulation strategies in adulthood: Specificities regarding age, gender and level of education. *Društvena Istraživanja-Časopis za Opća Društvena Pitanja*, 25(1), 43-62.
- Sornberger, M. J., Heath, N. L., Toste, J. R., & McLouth, R. (2012). Nonsuicidal self-injury and gender: Patterns of prevalence, methods, and locations among adolescents. *Suicide and Life-Threatening Behavior, 42*(3), 266-278.
- Spaapen, D. L., Waters, F., Brummer, L., Stopa, L., & Bucks, R. S. (2014). The Emotion Regulation Questionnaire: Validation of the ERQ-9 in two community samples. *Psychological Assessment*, 26(1), 46-54.
- Spence, J. T., & Helmreich, R. L. (1979). Masculinity and femininity: Their psychological dimensions, correlates, and antecedents: University of Texas Press.
- Sperberg, E. D., & Stabb, S. D. (1998). Depression in women as related to anger and mutuality in relationships. *Psychology of Women Quarterly*, 22(2), 223-238.

- Stangler, R. S., & Printz, A. M. (1980). DSM-III: psychiatric diagnosis in a university population. *American Journal of Psychiatry*, 137(8), 937-940.
- Stanley, B., Martínez-Alés, G., Gratch, I., Rizk, M., Galfalvy, H., Choo, T.-H., & Mann, J. J. (2021). Coping strategies that reduce suicidal ideation: An ecological momentary assessment study. *Journal of Psychiatric Research*, 133, 32-37.
- Stark, C., Hopkins, P., Gibbs, D., Rapson, T., Belbin, A., & Hay, A. (2004). Trends in suicide in Scotland 1981–1999: Age, method and geography. *BioMed Central Public Health*, 4(1), 49.
- Stephenson, H., Pena-Shaff, J., & Quirk, P. (2006). Predictors of college student suicidal ideation: Gender differences. *College Student Journal*, 40(1), 109-117.

Stevens, J. P. (2012). Applied multivariate statistics for the social sciences. Routledge.

- Stone, A. A., Turkkan, J. S., Bachrach, C. A., Jobe, J. B., Kurtzman, H. S., & Cain, V. S. (2000). *The science of self-report: Implications for research and practice*. Lawrence Erlbaum Associates, Inc.
- Stupnisky, R. H., Pekrun, R., & Lichtenfeld, S. (2016). New faculty members' emotions: A mixed-method study. *Studies in Higher Education*, 41(7), 1167-1188.
- Subramani, O. S., Parrott, D. J., Latzman, R. D., & Washburn, D. A. (2019). Breaking the link: Distraction from emotional cues reduces the association between trait disinhibition and reactive physical aggression. *Aggressive Behavior*, 45(2), 151-160.
- Sullivan, S. D., & Kahn, J. H. (2020). Individual differences in expressive suppression and the subjective experience, verbal disclosure, and behavioral expression of anger. *Personality and Individual Differences*, 155, 109723.
- Sutton, T. M., & Edlund, J. E. (2019). Assessing self-selection bias as a function of experiment title and description: The effect of emotion and personality. North American Journal of Psychology, 21(2), 407-422.

- Svaldi, J., Dorn, C., Matthies, S., & Philipsen, A. (2012). Effects of suppression and acceptance of sadness on the urge for non-suicidal self-injury and self-punishment. *Psychiatry Research*, 200(2-3), 404-416.
- Sweet, S., & Grace-Martin, K. (2011). *Data analysis with SPSS: A first course in applied statistics*. Pearson.
- Szasz, P. L., Szentagotai, A., & Hofmann, S. G. (2011). The effect of emotion regulation strategies on anger. *Behaviour Research and Therapy*, *49*(2), 114-119.
- Tabachnick, B., & Fidell, L. (2014). Using multivariate statistics, 6th New International Edition. Pearson.
- Tamres, L. K., Janicki, D., & Helgeson, V. S. (2002). Sex differences in coping behavior: A meta-analytic review and an examination of relative coping. *Personality and Social Psychology Review*, 6(1), 2-30.
- Tan, J., Ma, Z., Gao, X., Wu, Y., & Fang, F. (2011). Gender difference of unconscious attentional bias in high trait anxiety individuals. *PLoS ONE*, 6(5), 20305.
- Tanaka-Matsumi, J., & Kameoka, V. A. (1986). Reliabilities and concurrent validities of popular self-report measures of depression, anxiety, and social desirability. *Journal of Consulting and Clinical Psychology*, 54(3), 328-333.
- Teasdale, J. D., & Russell, M. L. (1983). Differential-effects of induced mood on the recall of positive, negative and neutral words. *British Journal of Clinical Psychology*, 22(3), 163-171.
- Teismann, T., Michalak, J., Willutzki, U., & Schulte, D. (2012). Influence of rumination and distraction on the therapeutic process in cognitive-behavioral therapy for depression. *Cognitive Therapy and Research*, 36(1), 15-24.
- Tennant, R., Hiller, L., Fishwick, R., Platt, S., Joseph, S., Weich, S., Parkinson, J., Secker, J.,& Stewart-Brown, S. (2007). The Warwick-Edinburgh mental well-being scale

(WEMWBS): Development and UK validation. *Health and Quality of life Outcomes*, 5(1), 63.

- Thayer, J. F., Rossy, L. A., Ruiz-Padial, E., & Johnsen, B. H. (2003). Gender differences in the relationship between emotional regulation and depressive symptoms. *Cognitive Therapy and Research*, 27(3), 349-364.
- Theurel, A., & Gentaz, E. (2018). The regulation of emotions in adolescents: Age differences and emotion-specific patterns. *PLoS ONE*, *13*(6), 0195501.
- Thiruchselvam, R., Blechert, J., Sheppes, G., Rydstrom, A., & Gross, J. J. (2011). The temporal dynamics of emotion regulation: An EEG study of distraction and reappraisal. *Biological Psychology*, 87(1), 84-92.
- Thoits, P. A. (1991). Gender differences in coping with emotional distress. In J. Eckenrode (Ed.), *The social context of coping* (pp. 107-138). Springer.
- Thomsen, D. K., Mehlsen, M. Y., Viidik, A., Sommerlund, B., & Zachariae, R. (2005). Age and gender differences in negative affect: Is there a role for emotion regulation? *Personality and Individual Differences*, 38(8), 1935-1946.
- Timmer-Murillo, S. C., Kangas, K. J., & Gordon, N. S. (2020). Implicit emotion regulation: A novel method for capturing individual differences in acute emotion regulation. *Personality and Individual Differences, 163*, 110067.
- Timmers, M., Fischer, A. H., & Manstead, A. S. (1998). Gender differences in motives for regulating emotions. *Personality and Social Psychology Bulletin*, 24(9), 974-985.
- Tomkins, S. S. (1962). Affect imagery consciousness: Volume I: The positive affects. Springer.
- Tomkins, S. S. (1963). Affect imagery consciousness: Volume II: The negative affects. Springer.

- Trapnell, P. D., & Campbell, J. D. (1999). Private self-consciousness and the five-factor model of personality: Distinguishing rumination from reflection. *Journal of Personality and Social Psychology*, 76(2), 284-304.
- Treynor, W., Gonzalez, R., & Nolen-Hoeksema, S. (2003). Rumination reconsidered: A psychometric analysis. *Cognitive Therapy and Research*, *27*(3), 247-259.
- Trives, J. J. R., Bravo, B. N., Postigo, J. M. L., Segura, L. R., & Watkins, E. (2016). Age and gender differences in emotion regulation strategies: Autobiographical memory, rumination, problem solving and distraction. *Spanish Journal of Psychology*, *19*(43), 1-9.
- Troy, A. S., Wilhelm, F. H., Shallcross, A. J., & Mauss, I. B. (2010). Seeing the silver lining: Cognitive reappraisal ability moderates the relationship between stress and depressive symptoms. *Emotion*, 10(6), 783-795.
- Urry, H. L., & Gross, J. J. (2010). Emotion regulation in older age. *Current Directions in Psychological Science*, 19(6), 352-357.
- Valikhani, A., Ahmadnia, F., Karimi, A., & Mills, P. J. (2019). The relationship between dispositional gratitude and quality of life: The mediating role of perceived stress and mental health. *Personality and Individual differences*, 141, 40-46.
- Välimäki, M. A., & Lantta, T. J. (2019). Individualised care in mental health and psychiatric care. In R. Suhonen, M. Stolt, & E. Papastavrou (Eds.), *Individualized Care* (pp. 141-150). Springer.
- Van Bockstaele, B., Atticciati, L., Hiekkaranta, A. P., Larsen, H., & Verschuere, B. (2019). Choose change: Situation modification, distraction, and reappraisal in mild versus intense negative situations. *Motivation and Emotion*, 44, 583-596.

- Van Bockstaele, B., Verschuere, B., Tibboel, H., De Houwer, J., Crombez, G., & Koster, E.
 H. (2014). A review of current evidence for the causal impact of attentional bias on fear and anxiety. *Psychological Bulletin*, 140(3), 682-721.
- Van de Velde, S., Bracke, P., & Levecque, K. (2010). Gender differences in depression in 23 European countries. Cross-national variation in the gender gap in depression. *Social Science & Medicine*, 71(2), 305-313.
- van Spijker, B. A., Batterham, P. J., Calear, A. L., Farrer, L., Christensen, H., Reynolds, J., & Kerkhof, A. J. (2014). The suicidal ideation attributes scale (SIDAS): Community-based validation study of a new scale for the measurement of suicidal ideation. *Suicide and Life-Threatening Behavior, 44*(4), 408-419.
- Van Tilburg, W. A., & Igou, E. R. (2017). Boredom begs to differ: Differentiation from other negative emotions. *Emotion*, 17(2), 309-322.
- Velten, E. (1968). A laboratory task for induction of mood states. *Behaviour Research and Therapy*, *6*(4), 473-482.
- Verduyn, P., & Lavrijsen, S. (2015). Which emotions last longest and why: The role of event importance and rumination. *Motivation and Emotion*, *39*(1), 119-127.
- Vieillard, S., Pinabiaux, C., & Bigand, E. (2020). Positive and detached reappraisal of threatening music in younger and older adults. *Frontiers in Human Neuroscience*, 14, 216.
- Vishkin, A., Hasson, Y., Millgram, Y., & Tamir, M. (2020). One size does not fit all: Tailoring cognitive reappraisal to different emotions. *Personality and Social Psychology Bulletin*, 46(3), 469-484.
- Volkaert, B., Wante, L., Van Beveren, M.-L., Vervoort, L., & Braet, C. (2019). Training adaptive emotion regulation skills in early adolescents: The effects of distraction,

acceptance, cognitive reappraisal, and problem solving. *Cognitive Therapy and Research, 44*, 678-696.

- Vytal, K., & Hamann, S. (2010). Neuroimaging support for discrete neural correlates of basic emotions: A voxel-based meta-analysis. *Journal of Cognitive Neuroscience*, 22(12), 2864-2885.
- Wager, T. D., & Ochsner, K. N. (2005). Sex differences in the emotional brain. *Neuroreport*, *16*(2), 85-87.
- Wagner, H. L., Buck, R., & Winterbotham, M. (1993). Communication of specific emotions: Gender differences in sending accuracy and communication measures. *Journal of Nonverbal Behavior*, 17(1), 29-53.
- Walton, C. J., Bendit, N., Baker, A. L., Carter, G. L., & Lewin, T. J. (2020). A randomised trial of dialectical behaviour therapy and the conversational model for the treatment of borderline personality disorder with recent suicidal and/or non-suicidal self-injury: An effectiveness study in an Australian public mental health service. *Australian & New Zealand Journal of Psychiatry*, 54(10), 1020-1034.
- Wang, J., Wu, X., Lai, W., Long, E., Zhang, X., Li, W., Zhu, Y., Chen, C., Zhong, X., Liu,
 Z., Wang, D., & Lin, H. (2017). Prevalence of depression and depressive symptoms among outpatients: A systematic review and meta-analysis. *British Medical Journal Open*, 7(8), 017173.
- Watkins, E. R. (2008). Constructive and unconstructive repetitive thought. *Psychological Bulletin*, *134*(2), 163-206.
- Watkins, E. R. (2009). Depressive rumination: Investigating mechanisms to improve cognitive behavioural treatments. *Cognitive Behaviour Therapy*, *38*(1), 8-14.
- Watkins, E. R. (2018). *Rumination-focused cognitive-behavioral therapy for depression*.Guilford Publications.

- Webb, T. L., Miles, E., & Sheeran, P. (2012). Dealing with feeling: A meta-analysis of the effectiveness of strategies derived from the process model of emotion regulation. *Psychological Bulletin*, 138(4), 775-808.
- Weber, A. M., Cislaghi, B., Meausoone, V., Abdalla, S., Mejía-Guevara, I., Loftus, P.,
 Hallgren, E., Seff, I., Stark, L., Victora, C. G., Buffarini, R., Barros, A. J. D.,
 Domingue, B. W., Bhushan, D., Gupta, R., Nagata, J. M., Shakya, H. B., Richter, L.
 M., Norris, S. A., . . . & Ngo, T. D. (2019). Gender norms and health: Insights from
 global survey data. *The Lancet*, *393*(10189), 2455-2468.
- Weissman, M. M., Bland, R. C., Canino, G. J., Faravelli, C., Greenwald, S., Hwu, H. G., Joyce, P. R., Karam, E. G., Lee, C-K., Lellouch, J., Lépine, J-P., Newman, S. C., Rubio-Stipec, M., Wells, E., Wickramaratne, P., J., Wittchen H-U., & Yeh, E-K. (1996). Cross-national epidemiology of major depression and bipolar disorder. *Journal of the American Medical Association*, 276(4), 293-299.
- Wergeland, G. J. H., Riise, E. N., & Öst, L.-G. (2021). Cognitive behavior therapy for internalizing disorders in children and adolescents in routine clinical care: A systematic review and meta-analysis. *Clinical Psychology Review*, 83, 101918.
- Westad, Y. A. S., Hagen, K., Jonsbu, E., & Solem, S. (2021). Cessation of deliberate selfharm behavior in patients with borderline personality traits treated with outpatient dialectical behavior therapy. *Frontiers in Psychology*, 12, 639.
- Westphal, M., Aldao, A., & Jackson, C. (2017). Emotion dysregulation in comorbid posttraumatic stress disorder and substance use disorders: A narrative review. *Military Psychology*, 29(3), 216-233.
- Wetherall, K., Cleare, S., Eschle, S., Ferguson, E., O'Connor, D. B., O'Carroll, R. E., & O'Connor, R. C. (2018). From ideation to action: Differentiating between those who

think about suicide and those who attempt suicide in a national study of young adults. Journal of Affective Disorders, 241, 475-483.

- Wetherall, K., Cleare, S., McClelland, H., Zortea, T., Robb, K., & O'Connor, R. (2020). Scottish COVID-19 (SCOVID) Mental Health Tracker Study: Wave 1 Report. https://www.gov.scot/binaries/content/documents/govscot/publications/research-andanalysis/2020/10/scottish-covid-19-scovid-mental-health-tracker-study-wave-1report2/documents/scottish-covid-19-scovid-mental-health-tracker-study-wave-1report/scottish-covid-19-scovid-mental-health-tracker-study-wave-1report/scottish-covid-19-scovid-mental-health-tracker-study-wave-1wave-1-report/govscot%3Adocument/scottish-covid-19-scovid-mental-health-tracker-studywave-1-report.pdf
- Wild, T. C., Cunningham, J., & Adlaf, E. (2001). Nonresponse in a follow-up to a representative telephone survey of adult drinkers. *Journal of Studies on Alcohol*, 62(2), 257-261.
- Wingenbach, T. S., Brosnan, M., Pfaltz, M. C., Peyk, P., & Ashwin, C. (2020). Perception of discrete emotions in others: Evidence for distinct facial mimicry patterns. *Scientific Reports*, 10(1), 1-13.
- Wolford-Clevenger, C., Vann, N. C., & Smith, P. N. (2016). The association of partner abuse types and suicidal ideation among men and women college students. *Violence and Victims*, 31(3), 471-485.
- Wolgast, M., & Lundh, L.-G. (2017). Is distraction an adaptive or maladaptive strategy for emotion regulation? A person-oriented approach. *Journal of Psychopathology and Behavioral Assessment*, 39(1), 117-127.
- Wong, T. K., Konishi, C., & Zhao, K. (2018). Anger and anger regulation among adolescents: A consideration of sex and age differences. *Canadian Journal of Behavioural Science*, 50(1), 1-8.

- Woodall, A., Morgan, C., Sloan, C., & Howard, L. (2010). Barriers to participation in mental health research: Are there specific gender, ethnicity and age related barriers? *BioMed Central Psychiatry*, 10(1), 103.
- World Health Organisation. (2021a). *Adolescent health*. <u>https://www.who.int/health-</u> topics/adolescent-health#tab=tab_1
- World Health Organisation. (2021b). *Depression: Key facts*. <u>https://www.who.int/news-room/fact-sheets/detail/depression</u>
- World Health Organisation. (2021c). *Suicide: Key facts*. <u>https://www.who.int/news-room/fact-sheets/detail/suicide</u>
- World Health Organisation (1988). The World Health Organization MONICA Project (monitoring trends and determinants in cardiovascular disease): A major international collaboration. *Journal of Clinical Epidemiology*, *41*(2), 105-114.
- Wu, X., Guo, T., Tan, T., Zhang, W., Qin, S., Fan, J., & Luo, J. (2019). Superior emotional regulating effects of creative cognitive reappraisal. *Neuroimage*, 200, 540-551.
- Wundt, W. (1907). Über ausfrageexperimente und über die methoden zur psychologie des denkens. *Psychologische Studien, 3*, 301-360.
- Xavier, A., Cunha, M., & Pinto-Gouveia, J. (2018). Daily peer hassles and non-suicidal selfinjury in adolescence: Gender differences in avoidance-focused emotion regulation processes. *Journal of Child and Family Studies*, 27(1), 59-68.
- Yang, X., Liu, D., Wang, Y., Chen, Y., Chen, W., Yang, C., Zhang, P., Ding, S., & Zhang, X. (2020). Effectiveness of Zhong-Yong thinking based dialectical behavior therapy group skills training versus supportive group therapy for lowering suicidal risks in Chinese young adults: A randomized controlled trial with a 6-month follow-up. *Brain and Behavior*, *10*(6), 01621.

- Yik, M. S., Russell, J. A., & Barrett, L. F. (1999). Structure of self-reported current affect: Integration and beyond. *Journal of Personality and Social Psychology*, 77(3), 600-619.
- Yıldırım, M., Geçer, E., & Akgül, Ö. (2021). The impacts of vulnerability, perceived risk, and fear on preventive behaviours against COVID-19. *Psychology, Health & Medicine*, 26(1), 35-43.
- Young, K., Sandman, C., & Craske, M. (2019). Positive and negative emotion regulation in adolescence: Links to anxiety and depression. *Brain Sciences*, *9*(4), 76.
- Zahniser, E., & Conley, C. S. (2018). Interactions of emotion regulation and perceived stress in predicting emerging adults' subsequent internalizing symptoms. *Motivation and Emotion*, 42(5), 763-773.
- Zenses, A. K., Lenaert, B., Peigneux, P., Beckers, T., & Boddez, Y. (2020). Sleep deprivation increases threat beliefs in human fear conditioning. *Journal of Sleep Research*, 29(3), 12873.
- Zhao, X., Lynch Jr, J. G., & Chen, Q. (2010). Reconsidering Baron and Kenny: Myths and truths about mediation analysis. *Journal of Consumer Research*, *37*(2), 197-206.
- Zilcha-Mano, S. (2019). Major developments in methods addressing for whom psychotherapy may work and why. *Psychotherapy Research*, 29(6), 693-708.
- Zimmermann, P., & Iwanski, A. (2014). Emotion regulation from early adolescence to emerging adulthood and middle adulthood: Age differences, gender differences, and emotion-specific developmental variations. *International Journal of Behavioral Development, 38*(2), 182-194.
- Zimmermann, P., Scharf, M., & Iwanski, A. (2008). *Negative emotion regulation inventory* (*NERI*). (Unpublished Test Manual). University of Wuppertal/University of Haifa.

- Zlomke, K. R., & Hahn, K. S. (2010). Cognitive emotion regulation strategies: Gender differences and associations to worry. *Personality and Individual differences*, 48(4), 408-413.
- Zuckerman, M., Lipets, M. S., Koivumaki, J. H., & Rosenthal, R. (1975). Encoding and decoding nonverbal cues of emotion. *Journal of Personality and Social Psychology*, 32(6), 1068-1076.

Appendix A

Study 1 Information Sheet

Participant Information Sheet

Name of department: School of Psychological Sciences and Health Title of the study: Dealing with Feelings

Introduction

Hello, my name is Michelle Patrick and I am a post-graduate researcher at the University of Strathclyde. I am inviting you to take part in my research study which will contribute towards my PhD in Psychology. Please carefully read the information provided on this sheet before you decide whether or not you would like to take part. If you have any questions about the study, please feel free to contact me using the email address below.

What is the purpose of this investigation?

The purpose of this study is to investigate the different ways that people deal with three negative emotions – sadness, anger and fear. I'm interested in finding out about the different strategies that individuals use in their day-to-day lives when they experience these three emotions.

Do you have to take part?

No, you do not have to take part in this study. Taking part is completely voluntary and you may leave at any time without judgement. Please note that not taking part will have no effect on the services that the school provides.

What will you do in the project?

If you decide you would like to take part, you will be provided with a paper copy of a questionnaire. The questionnaire will ask about a time when you felt sad, angry or scared, and it will ask you to describe this event. It will then ask about what you did during this time to help yourself feel less sad, angry or scared. The purpose of this study is to find out more about the different strategies people use to manage these three emotions. The questionnaire should take approximately 30 minutes to complete.

Why have you been invited to take part?

You have been invited to take part because I am interested in finding out how people of different ages manage their emotions.

What are the potential risks to you in taking part?

There are no anticipated risks to taking part in the study. However, helpline information will be provided to you at the end of the questionnaire should this research raise any personal issues.

What happens to the information in the project?

The information you provide in this study will be stored anonymously, which means that you cannot be linked to the data you provide. For this reason, once you have completed the study, your data cannot be removed at a later date as there will be no way of identifying it. Data will be stored in a password-protected computer and the questionnaires will be stored in a locked cabinet at the University of Strathclyde and will only be accessed and analysed by the named researchers (me and my supervisor). Data will be kept for a period of at least 5 years after which it will be securely destroyed.

The University of Strathclyde is registered with the Information Commissioner's Office who implements the Data Protection Act 1998. All personal data on participants will be processed in accordance with the provisions of the Data Protection Act 1998.

Thank you for reading this information – please ask any questions if you are unsure about what is written here.

What happens next?

If you are happy to be involved in the study, you will be provided with a copy of the questionnaire. If you do not wish to participate then thank you for your time. When you have completed the study, you will receive a Debrief Sheet which provides additional information about the study. If you have any questions about any aspect of the study, please contact me or Dr Rasmussen using the details below.

The results of this study will be written up in my PhD thesis and may be presented at conferences. The results of this research may also be written up in a paper which I will aim to publish in a scientific research journal. A brief report of my findings will be sent to your school.

Researcher contact details:

Michelle Patrick School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: <u>michelle.patrick@strath.ac.uk</u>

Chief Investigator details:

Dr Susan Rasmussen (Chief Investigator/Supervisor) School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: <u>s.a.rasmussen@strath.ac.uk</u> Phone: 0141 548 2575

Dr Sinead Rhodes (Supervisor)

School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: <u>sinead.rhodes@strath.ac.uk</u> Phone: 0141 548 2489

This investigation was granted ethical approval by the University of Strathclyde Ethics Committee. If you have any questions/concerns, during or after the investigation, or wish to contact an independent person to whom any questions may be directed or further information may be sought from, please contact:

Secretary to the University Ethics Committee

Research & Knowledge Exchange Services University of Strathclyde Graham Hills Building 50 George Street Glasgow G1 1QE Telephone: 0141 548 3707 Email: ethics@strath.ac.uk

Appendix B

Study 1 Consent Form

Consent Form

Name of department: School of Psychological Sciences and Health Title of the study: Dealing with Feelings

- I confirm that I have read and understood the information sheet for the above project and the researcher has answered any questions to my satisfaction.
- I understand that my participation is voluntary and that I am free to withdraw from the project at any time, without having to give a reason and without any consequences
- I understand that anonymised data (i.e. data which do not identify me personally) cannot be withdrawn during or at the end of the study as they cannot be identified
- I consent to being a participant in the project

(PRINT NAME)	
Signature of Participant:	Date:

Appendix C

Study 1 Demographic Questions

First of all, I need some more information about you. Please answer the following questions.

What is your age? (Please type into box)

What is your gender?

- Male
- Female

Can you read and write in English?

- Yes
- No

Adult SES Measure (to measure SES for adults)

What is the highest level of education you have completed?

- 1 = university or college or equivalent
- 2 = intermediate between secondary level and university (e.g., technical training)
- 3 = secondary school
- 4 = primary school only (or less)

Family Affluence Scale (to measure SES for adolescents)

Does your family own a car, van or truck?

Do you have your own bedroom for yourself?

During the past 12 months, how many times did you travel away on holiday with your family?

How many computers does your family own?

Appendix D

Study 1 Dealing with Feelings Questionnaire

Please answer the following questions.

Think about a time recently when you felt sad, and you did something to try to make yourself feel less sad. Describe what happened to make you feel sad in the box below.

When you felt sad during this time, what did you do to try to make yourself feel less sad? Describe this in the box below.

Can you think of any other things you have done in the past to make yourself feel less sad? Describe this in the box below.

Think about a time recently when you felt angry, and you did something to try to make yourself feel less angry. Describe what happened to make you feel angry in the box below.

When you felt angry during this time, what did you do to try to make yourself feel less angry? Describe this in the box below.

Can you think of any other things you have done in the past to make yourself feel less angry? Describe this in the box below

Think about a time recently when you felt scared, and you did something to try to make yourself feel less scared. Describe what happened to make you feel scared in the box below.

When you felt scared during this time, what did you do to try to make yourself feel less scared? Describe this in the box below.

Can you think of any other things you have done in the past to make yourself feel less scared? Describe this in the box below.

Appendix E

Study 1 Velten Mood Induction

Please read through the following sentences. As you are reading, try to let the sentences change how you feel.

I feel pretty good right now.

I feel happy.

I feel cheerful, confident.

I can think quickly and clearly right now.

Right now, I feel very contented.

Right now, I feel like smiling.

I feel alert, happy and full of energy.

I have a feeling of lightness and joy.

I really like this light-hearted feeling.

I can feel a smile on my face.

I feel so good I almost feel like laughing.

It feels great to be alive!

Appendix F

Debriefing Sheet

Dealing with Feelings

Thank you for taking part in this study!

Emotion regulation (ER) refers to the ways that people manage their emotions, including how and when emotions are experienced (Gross & Thompson, 2007). The purpose of this study was to find out more about the different methods that people of different ages use to regulate three negative emotions – sadness, anger and fear.

There are many different ways that people can regulate their emotions. Some of these involve controlling our thoughts (such as reappraisal) and some of these may involve activities that we take part in (such as chatting to friends), which is known as behavioural emotion regulation. Few studies to date have explored how our behaviours can help us to regulate our emotions, which was one of the aims of the present study. Also, there is currently no reliable way to measure external emotion regulation. The data from this study will be used to inform the development of a questionnaire which measures external emotion regulation.

The data from this study will be stored in a password-protected computer and the questionnaires will be stored in a locked cabinet at the University of Strathclyde and will only be accessed and analysed by the named researchers. Data will be kept for a period of at least 5 years after which it will be securely destroyed.

If you are concerned or worried about anything that was brought up in the study, then you may wish to talk to someone about this.

You may find it helpful to talk to your GP. There are also services available which can let you discuss your concerns about depression, anxiety and stress (or anything else you are worried about).

Young Minds

Young Minds is a charity dedicated to the mental health and well-being of young people. Their website (http://www.youngminds.org.uk/) has lots of helpful information about a whole range of mental health issues, including depression and self-harm. If you are worried about an issue, this can be a good place to find out more about it.

Breathing Space

Breathing Space is a free, confidential, phone service for anyone in Scotland experiencing low mood, depression or anxiety. The aim of Breathing Space is to be there for people in times of difficulty, to provide a safe and supportive space by listening, offering advice and providing information. You can phone Breathing Space on 0800 83 85 87, or find out information about a range of issues on their website (http://breathingspace.scot/).

If you have any questions or concerns about this project, please feel free to contact me or my supervisor. Your participation in this study is greatly appreciated.

Michelle Patrick (Researcher) Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: michelle.patrick@strath.ac.uk Phone 0141 548 2382

Dr Susan Rasmussen (Chief Investigator/Supervisor) Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: s.a.rasmussen@strath.ac.uk Phone: 0141 548 2575

References Gross, J. J., & Thompson, R. A. (2007). Emotion regulation: Conceptual foundations. Handbook of emotion regulation, 3, 24.

Appendix G

AcceptanceEngagementAct happy/be happy/smileEngagementAttentional deploymentDisengagementAvoid situationDisengagementBe in a happy environmentEngagementBe less scaredEngagementBreathing techniquesEngagementChange attitudeEngagementChange conversationDistractionClubbing/partyingDistractionConfront enditionEngagementConfront situationEngagementControl situationEngagementControl situationEngagementControl situationEngagementControl situationEngagementControl situationEngagementCountingDistractionCouselling/therapyEngagementCountingDistractionCrack knucklesDistractionCrack knucklesDistractionCrack knucklesDistractionCrack knucklesDistractionCrack knucklesDistractionDistractionDistractionCrack knucklesDistractionDo sybestEngagementDo something funDistractionDon't talk to anyoneDisengagementDon't talk about itDisengagementDon't talk about itDisengagementDon't talk about itDisengagementDon't talk about itDisengagementDistractionEngagementDistractionEngagementDon't talk about itDisengagementDon't talk about itDisen	ER strategy	Broader categories of ER
Attentional deploymentDisengagementAvoid situationDisengagementBe in a happy environmentEngagementBe less scaredEngagementBreathing techniquesEngagementCalmed downEngagementChange attitudeEngagementChange conversationDistractionClubing/partyingDistractionCollect thoughtsEngagementComfort eatingDistractionConfront emotionEngagementContact people from pastMulti-purposeControl situationEngagementCountingDistractionCountingDistractionConselling/therapyEngagementCountingDistractionConselling/therapyEngagementCountingDistractionCrack knucklesDistractionCrack knucklesDistractionCrack knucklesDistractionDistractionDistractionCrack knucklesDistractionDo a detoxMulti-purposeDo ny bestEngagementDon't dwell on itMulti-purposeDon't talk about itDistractionDorst talk dout itDistractionDistractionDistractionEngagementDistractionDon't talk about itDistractionDon't talk about itDistractionDistractionDistractionDistractionDistractionEat healthyMulti-purposeDon't talk about itDistractionEat healthyM	Acceptance	Engagement
Avoid situationDisengagementBe in a happy environmentEngagementBe less scaredEngagementBreathing techniquesEngagementCalmed downEngagementChange conversationDistractionCleaning/tidyingDistractionClubbing/partyingDistractionCollect thoughtsEngagementConfront emotionEngagementConfront situationEngagementControl situationEngagementCountingDistractionCountingDistractionCrack knucklesDistractionCrack knucklesDistractionDistractionEngagementCountingDistractionCrack knucklesDistractionDistractionDistractionCrack knucklesDistractionDistractionDistractionDeal with itEngagementDistractionDistractionDo a detoxMulti-purposeDon't dwell on itMulti-purposeDon't talk about itDistractionDistractionDistractionDon't talk about itDistractionDistractionDistractionDistractionDistractionDorit advidanceDisengagementDon't talk about itDisengagementDorit duk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDisengagementExpressive suppressionDisengagementExpression schoolMulti-purposeFocus on ot	Act happy/be happy/smile	Engagement
Be in a happy environmentEngagementBe in a happy environmentEngagementBe less scaredEngagementCalmed downEngagementChange attitudeEngagementChange conversationDistractionCleaning/tidyingDistractionClubbing/partyingDistractionCollect thoughtsEngagementConfront emotionEngagementConfront emotionEngagementConfront emotionEngagementControl situationEngagementControl situationEngagementCountingDistractionCountingDistractionCreative/relaxing activityDistractionDe adetoxMulti-purposeDo my bestEngagementDo something funDistractionDo 't dwell on itMulti-purposeDon't talk about itDisengagementDon't talk about itDisengagementDon't talk about itDisengagementDor't talk about itDisengagementDor't talk about itDisengagementDor't talk about itDisengagementDistractionEngagementDor't talk about itDisengagementDistractionEngagementDistractionEngagementDo't talk about itDisengagementDo't talk about itDisengagementDo't talk about itDisengagementDor't coll talk about itEngagementExpressive suppressionDisengagementExpressive suppressionDisengagement <td>Attentional deployment</td> <td>Disengagement</td>	Attentional deployment	Disengagement
Be less scaredEngagementBreathing techniquesEngagementCalmed downEngagementChange attitudeEngagementChange conversationDistractionCleaning/tidyingDistractionClubbing/partyingDistractionCollect thoughtsEngagementComfort eatingDistractionConfront emotionEngagementConfront situationEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCountingDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDeal with itEngagementDo a detoxMulti-purposeDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDistractionDon't talk about itDistractionDon't talk about itDistractionDon't talk about itDistractionDon't talk about itDisengagementDon't talk about itDisengagementDon't talk about itDisengagementExpressive suppressionDisengagementExpressive suppressionDisengagementExpression schoolMulti-purposeFocus on what I can controlEngagementFocus on what I can controlEngagementSourceSuppressionColumnaSuppressionColumnaDisengagement <tr< td=""><td>Avoid situation</td><td>Disengagement</td></tr<>	Avoid situation	Disengagement
Breathing techniquesEngagementCalmed downEngagementChange attitudeEngagementChange conversationDistractionCleaning/tidyingDistractionClubbing/partyingDistractionCollect thoughtsEngagementComfort eatingDistractionConfront emotionEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCounselling/therapyDistractionCreative/relaxing activityDistractionDeal with itEngagementDo a detoxMulti-purposeDon't talk to anyoneDistractionDon't talk to anyoneDistractionDon't talk to anyoneDistractionDon't talk about itDistractionDon't talk about itDistractionDistractionDistractionDon't talk about itDistractionDistractionDistractionDon't talk about itDistractionDon't talk about itDistractionDistractionDistractionDistractionDistractionDon't talk about itDistractionDistractionDistractionDistractionDistractionDon't talk about itDistractionDistractionDistractionExpressive suppressionDisengagementDoris talk avoidanceDisengagementExpressive suppressionDisengagementFocus on othersMulti-purpose	Be in a happy environment	Engagement
Calmed downEngagementChange attitudeEngagementChange conversationDistractionCleaning/tidyingDistractionClubbing/partyingDistractionCollect thoughtsEngagementComfort eatingDistractionConfront emotionEngagementControl situationEngagementCounselling/therapyEngagementCountingDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDistractionDistractionDo a detoxMulti-purposeDon't dwell on itMulti-purposeDon't talk about itDistractionDon't talk about itDistractionDon't talk about itDistractionDress nice/put on make-upMulti-purposeDrink waterDistractionExpressive suppressionDisengagementExpressive suppressionDisengagementExpressive suppressionDisengagementFocus on what I can controlEngagementFocus on what I can controlEngagement	Be less scared	Engagement
Change attitudeEngagementChange conversationDistractionCleaning/tidyingDistractionClubbing/partyingDistractionCollect thoughtsEngagementComfort eatingDistractionConfront emotionEngagementConfront situationEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCounselling/therapyEngagementCountingDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDo a detoxMulti-purposeDon't dwell on itMulti-purposeDon't talk to anyoneDistractionDorit talk about itDistractionDress nice/put on make-upMulti-purposeDrink waterDistractionExperiential avoidanceDistractionExperiential avoidanceDistractionExperiential avoidanceDistractionExperiential avoidanceDistractionExpressive suppressionDisengagementFocus on schoolMulti-purposeFocus on what I can controlEngagementFocus on what I can controlEngagement	Breathing techniques	Engagement
Change conversationDistractionCleaning/tidyingDistractionClubbing/partyingDistractionCollect thoughtsEngagementComfort eatingDistractionConfront emotionEngagementConfront situationEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCounselling/therapyEngagementCountingDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDistractionDrist alk about itDistragementDrist alk about itDistractionDrist alk about itDistractionDrist alk about itDistractionExperiential avoidanceDistragementExpressive suppressionDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagementEngagementFocus on what I can control	Calmed down	Engagement
Cleaning/tidyingDistractionClubbing/partyingDistractionCollect thoughtsEngagementComfort eatingDistractionConfront emotionEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCountingDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDistractionDistractionDistractionDistractionDistractionDistractionDeal with itEngagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionExperiential avoidanceDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on what I can controlEngagementFocus on what I can controlEngagement	Change attitude	Engagement
Clubbing/partyingDistractionCollect thoughtsEngagementComfort eatingDistractionConfront emotionEngagementConfront situationEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCountingDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDo a detoxMulti-purposeDo ny bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk about itDisengagementDort talk about itDisengagementDort talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionExperiential avoidanceDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on what I can controlEngagement	Change conversation	Distraction
Collect houghtsEngagementComfort eatingDistractionConfront emotionEngagementConfront situationEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCountingDistractionCrack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDistractionDon't talk about itDistractionDistractionDistractionExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Cleaning/tidying	Distraction
Comfort eatingDistractionConfront emotionEngagementConfront situationEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCountingDistractionCrack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDon't talk to anyoneDistractionDistractionDistractionExperiential avoidanceDisengagementExpressive suppressionDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on what I can controlEngagement	Clubbing/partying	Distraction
Confront emotionEngagementConfront situationEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCountingDistractionCrack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDistractionDistractionDo ad etoxMulti-purposeDo my bestEngagementDo n't dwell on itMulti-purposeDon't talk to anyoneDistractionDon't talk about itDistractionDorit talk about itDistractionExperiential avoidanceDisengagementExperiential avoidanceDisengagementExpressive suppressionDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Collect thoughts	Engagement
Confront situationEngagementContact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCountingDistractionCrack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDistractionDorit talk about itDistractionExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Comfort eating	Distraction
Contact people from pastMulti-purposeControl situationEngagementCounselling/therapyEngagementCounselling/therapyEngagementCountingDistractionCrack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistractionDistractionDistractionDistractionDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDon't talk to anyoneDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionExperiential avoidanceEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on what I can controlEngagement	Confront emotion	Engagement
Control situationEngagementCounselling/therapyEngagementCountingDistractionCrack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistance myself from emotion/situationDisengagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDorts snice/put on make-upMulti-purposeDrink waterDistractionExpress/vent emotionEngagementExpress/vent emotionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagementFocus on what I can controlEngagement	Confront situation	Engagement
Counselling/therapyEngagementCountingDistractionCrack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistance myself from emotion/situationDisengagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDon't talk to anyoneDistractionDort talk about itDistractionDistractionStractionExpressive suppressionDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Contact people from past	Multi-purpose
CountingDistractionCrack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistance myself from emotion/situationDisengagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDor't talk to anyoneDistractionDrits nice/put on make-upMulti-purposeDrink waterDistractionExperiential avoidanceDisengagementExpress/vent emotionEngagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Control situation	Engagement
Crack knucklesDistractionCreative/relaxing activityDistractionDeal with itEngagementDistance myself from emotion/situationDisengagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't let it affect meMulti-purposeDon't talk to anyoneDistractionDon't talk about itDistractionDrink waterDistractionExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Counselling/therapy	Engagement
Creative/relaxing activityDistractionDeal with itEngagementDistance myself from emotion/situationDisengagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDor't talk to anyoneDisengagementDor't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Counting	Distraction
Deal with itEngagementDistance myself from emotion/situationDisengagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't dwell on itMulti-purposeDon't talk to anyoneDisengagementDor't talk to anyoneDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Crack knuckles	Distraction
Distance myself from emotion/situationDisengagementDistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't diffect meMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDisengagementDorit talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Creative/relaxing activity	Distraction
DistractionDistractionDo a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't let it affect meMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Deal with it	Engagement
Do a detoxMulti-purposeDo my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't let it affect meMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Distance myself from emotion/situation	Disengagement
Do my bestEngagementDo something funDistractionDon't dwell on itMulti-purposeDon't let it affect meMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Distraction	Distraction
Do something funDistractionDon't dwell on itMulti-purposeDon't dwell on itMulti-purposeDon't let it affect meMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Do a detox	Multi-purpose
Don't dwell on itMulti-purposeDon't let it affect meMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Do my best	Engagement
Don't let it affect meMulti-purposeDon't talk to anyoneDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Do something fun	Distraction
Don't talk to anyoneDisengagementDon't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Don't dwell on it	Multi-purpose
Don't talk about itDisengagementDress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Don't let it affect me	Multi-purpose
Dress nice/put on make-upMulti-purposeDrink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Don't talk to anyone	Disengagement
Drink waterDistractionEat healthyMulti-purposeExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Don't talk about it	Disengagement
Eat healthyMulti-purposeExperiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Dress nice/put on make-up	Multi-purpose
Experiential avoidanceDisengagementExpress/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Drink water	Distraction
Express/vent emotionEngagementExpressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Eat healthy	Multi-purpose
Expressive suppressionDisengagementFocus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Experiential avoidance	Disengagement
Focus on othersMulti-purposeFocus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Express/vent emotion	00
Focus on schoolMulti-purposeFocus on the outcomeMulti-purposeFocus on what I can controlEngagement	Expressive suppression	Disengagement
Focus on the outcomeMulti-purposeFocus on what I can controlEngagement	Focus on others	Multi-purpose
Focus on what I can control Engagement	Focus on school	Multi-purpose
00	Focus on the outcome	Multi-purpose
Force myself to vomit Multi-purpose	Focus on what I can control	00
	Force myself to vomit	Multi-purpose

Study 1 ER Strategies Included in Each ER Category

Get fresh air Get over it Go for a day out Go for a drive Go get a drink Go home Go out Go to room Gratitude Grip onto something Hide Hit myself Hold on to something Holiday Hostility to others Humour Internet Just did it Keep a routine Keep going Keep out of it Kindness to others Leave emotional situation Listen to music Look for opportunities Make a plan Make up with others Man up Medication Meditation/yoga Memories Mindfulness Move on Move on to another task Observe others Pace up and down Paid respect Passivity Photography Pinched myself Ping a bobble to feel pain Plan/organise Play well Prayer/religion Problem solving (action) Problem solving (thinking) Put a pillow over my face Put earphones in

Multi-purpose Engagement Distraction Distraction Distraction Disengagement Distraction Disengagement Engagement Engagement Disengagement Engagement Engagement Distraction Distraction Distraction Distraction Engagement Engagement Engagement Disengagement Multi-purpose Disengagement Distraction Engagement Distraction Engagement Engagement Multi-purpose Engagement Distraction Engagement Engagement Distraction Engagement Distraction Engagement Multi-purpose Distraction Multi-purpose Multi-purpose Distraction Engagement Engagement Engagement Engagement Disengagement Distraction

Put it behind me Rationalization **Read letters** Reappraisal Rumination Seek comfort Self-calming Self-harm Self-talk Sex Shopping Shrug it off Shut people out Sleep Social support Sports/exercise Stay alert Stay away from someone Stay inside Stay out of trouble Stick up for myself Stop talking to someone Suicidal behaviour Take positive action Take time out Think about future Training Travelling Treat myself Trust my instincts Trust people Trust the universe Try hard Try not to care what others think Try not to worry Try to eat nothing TV/movies Use computer Use substances Video games Writing/journalling

Multi-purpose Engagement Distraction Engagement Engagement Distraction Engagement Multi-purpose Engagement Distraction Distraction Multi-purpose Disengagement Distraction Engagement Distraction Engagement Disengagement Disengagement Disengagement Engagement Disengagement Disengagement Engagement Multi-purpose Multi-purpose Distraction Distraction Distraction Engagement Engagement Engagement Engagement Engagement Engagement Multi-purpose Distraction Distraction Distraction Distraction Engagement

Appendix H

Study 1 Cross-Tabulations used in Chi-Square Analyses

Gender differences in emotion regulation strategies

	Gender x Social S	Support for Sadness	
	Male	Female	Total
No	141	135	276
Yes	58	134	192
Total	199	269	468
	Gender x Sports/I	Exercise for Sadness	
	Male	Female	Total
No	157	225	382
Yes	42	44	86
Total	199	269	468
	Gender x Listen t	o Music for Sadness	
	Male	Female	Total
No	178	204	382
Yes	21	65	86
Total	199	269	468
	Gender x TV/M	lovies for Sadness	
	Male	Female	Total
No	177	227	404
Yes	22	42	64
Total	199	269	468
G	ender x Venting/Expre	ssing Emotion for Sadn	ess
	Male	Female	Total
No	188	220	408
Yes	11	49	60
Total	199	269	468
	Gender x Reapp	oraisal for Sadness	
	Male	Female	Total
No	184	237	421
Yes	15	32	47
Total	199	269	468
	Gender x Distra	action for Sadness	
	Male	Female	Total
No	186	236	422
Yes	13	33	46
Total	199	269	468
	Gender x Avoid	lance for Sadness	
	Male	Female	Total

No	174	244	418
Yes	25	25	50
Total	199	269	468
	Gender x Venting/I	Expressing for Anger	
	Male	Female	Total
No	154	184	338
Yes	45	85	130
Total	199	269	468
	Gender x Social	Support for Anger	
	Male	Female	Total
No	177	193	370
Yes	22	76	98
Total	199	269	468
	Gender x Sports/	Exercise for Anger	
	Male	Female	Total
No	172	220	392
Yes	27	49	76
Total	199	269	468
	Gender x Hostility	to Others for Anger	
	Male	Female	Total
No	185	232	417
Yes	14	37	51
Total	199	269	468
	Gender x Take T	ime Out for Anger	
	Male	Female	Total
No	179	241	420
Yes	20	28	48
Total	199	269	468
	Gender x Reap	praisal for Anger	
	Male	Female	Total
No	188	255	443
Yes	11	14	25
Total	199	269	468
		Support for Fear	
	Male	Female	Total
No	172	189	361
Yes	27	80	107
Total	199	269	468
		m Solving for Fear	
	Male	Female	Total
No	174	224	398
Yes	25	45	70
			. ~

Total	199	269	468
	Gender x Reap	opraisal for Fear	
	Male	Female	Total
No	190	221	411
Yes	9	48	57
Total	199	269	468
	Gender x Sel	f-Talk for Fear	
	Male	Female	Total
No	190	242	432
Yes	9	27	36
Total	199	269	468
	Gender x Venting/	Expressing for Fear	
	Male	Female	Total
No	194	237	431
Yes	5	32	37
Total	199	269	468
	Gender x Removin	g Attention for Fear	
	Male	Female	Total
No	191	259	450
Yes	8	10	18
Total	199	269	468
	Gender x Avoid th	e Situation for Fear	
	Male	Female	Total
No	188	246	434
Yes	11	23	34
Total	199	269	468
	Gender x Rum	ination for Fear	
	Male	Female	Total
No	194	262	456
Yes	5	7	12
Total	199	269	468
	Gender x Engag	ement for Sadness	
-------	------------------	--------------------	-------
	Male	Female	Total
No	101	89	190
Yes	82	175	257
Total	183	264	447
	Gender x Engag	gement for Anger	
	Male	Female	Total
No	71	73	144
Yes	98	176	274
Total	169	249	418
	Gender x Enga	gement for Fear	
	Male	Female	Total
No	66	67	133
Yes	66	161	227
Total	132	228	360
	Gender x Disenga	gement for Sadness	
	Male	Female	Total
No	153	229	382
Yes	30	35	65
Total	183	264	447
	Gender x Diseng	agement for Anger	
	Male	Female	Total
No	126	170	296
Yes	43	79	122
Total	169	249	418
	Gender x Diseng	agement for Fear	
	Male	Female	Total
No	92	160	252
Yes	40	68	108
Total	132	228	360
	Gender x Distra	ction for Sadness	
	Male	Female	Total
No	74	99	173
Yes	109	165	274
Total	183	264	447
		action for Anger	
	Male	Female	Total
No	106	145	251
Yes	63	104	167

The relationship between gender and engagement, disengagement, and distraction (engagement hypothesis)

Total	169	249	418					
	Gender x Distraction for Fear							
	Male	Female	Total					
No	101	165	266					
Yes	31	63	94					
Total	132	228	360					

Relationship Between Age Groups (Younger Adolescents, Older Adolescents, Adults) and each Category of ER – Engagement, Disengagement, and Distraction (Yes, No) Across Both Genders (Main effect of age)

	Age Group	x Engagement for	r Sadness		
	Younger Ad	Older Ad	Adults	Total	
No	111	51	26	188	
Yes	145	49	60	254	
Total	256	100	86	442	
	Age Group	o x Engagement fo	or Anger		
	Younger Ad	Older Ad	Adults	Total	
No	94	39	11	144	
Yes	145	55	71	271	
Total	239	94	82	415	
Age Group x Engagement for Fear					
	Younger Ad	Older Ad	Adults	Total	
No	90	29	14	133	
Yes	112	53	58	223	
Total	202	82	72	356	
	Age Group x	Disengagement f	or Sadness		
	Younger Ad	Older Ad	Adults	Total	
No	208	94	75	377	
Yes	48	6	11	65	
Total	256	100	86	442	
	Age Group	x Disengagement	for Anger		
	Younger Ad	Older Ad	Adults	Total	
No	169	66	58	293	
Yes	70	28	24	122	
Total	239	94	82	415	
	Age Group	x Disengagement	t for Fear		
	Younger Ad	Older Ad	Adults	Total	

No	138	58	52	248			
Yes	64	24	20	108			
Total	202	82	72	356			
	Age Group	o x Distraction for	Sadness				
	Younger Ad	Older Ad	Adults	Total			
No	117	38	14	169			
Yes	139	62	72	273			
Total	256	100	86	442			
	Age Grou	p x Distraction fo	r Anger				
	Younger Ad	Older Ad	Adults	Total			
No	151	58	40	249			
Yes	88	36	42	166			
Total	239	94	82	415			
	Age Group x Distraction for Fear						
	Younger Ad	Older Ad	Adults	Total			
No	154	63	46	263			
Yes	48	19	26	93			
Total	202	82	72	356			

Relationship Between Gender (Males, Females) and each Category of ER – Engagement, Disengagement, and Distraction (Yes, No), Broken Down by Age Group

Gene	ler x Sadness Engagei	ment for Young Adoles	cents
	Male	Female	Total
No	65	46	111
Yes	56	89	145
Total	121	135	256
Gen	der x Sadness Engage	ment for Older Adolese	cents
	Male	Female	Total
No	26	25	51
Yes	13	36	49
Total	39	61	100
	Gender x Sadness E	ngagement for Adults	
	Male	Female	Total
No	9	17	26
Yes	10	50	60
Total	19	67	86
Gende	er x Sadness Disengag	ement for Young Adole	escents
	Male	Female	Total
No	97	111	208

Yes	24	24	48
Total	121	135	256
Gend	er x Sadness Disengag	ement for Older Adole	escents
	Male	Female	Total
No	36	58	94
Yes	3	3	6
Total	39	61	100
	Gender x Sadness Dis	engagement for Adult	s
	Male	Female	Total
No	16	59	75
Yes	3	8	11
Total	19	67	86
Gen	der x Sadness Distrac	tion for Young Adoles	cents
	Male	Female	Total
No	51	66	117
Yes	70	69	139
Total	121	135	256
Gei	nder x Sadness Distrac	tion for Older Adoles	cents
	Male	Female	Total
No	17	21	38
Yes	22	40	62
Total	39	61	100
	Gender x Sadness I	Distraction for Adults	
	Male	Female	Total
No	2	12	14
Yes	17	55	72
Total	19	67	86
Ger	nder x Anger Engagen	nent for Young Adoles	cents
	Male	Female	Total
No	51	43	94
Yes	61	84	145
Total	112	127	239
Ger	nder x Anger Engagen	nent for Older Adolesc	ents
	Male	Female	Total
No	15	24	39
Yes	22	33	55
Total	37	57	94
	Gender x Anger En	gagement for Adults	
	Male	Female	Total
No	5	6	11
Yes	13	58	71

Othu	er x Anger Disengage				
	Male	Female	Total		
No	84	85	169		
Yes	28	42	70		
Total	112	127	239		
Geno	ler x Anger Disengage	ement for Older Adoles	cents		
	Male	Female	Total		
No	28	38	66		
Yes	9	19	28		
Total	37	57	94		
	Gender x Anger Dise	engagement for Adults			
	Male	Female	Total		
No	12	46	58		
Yes	6	18	24		
Total	18	64	82		
Gender x Anger Distraction for Young Adolescents					
	Male	Female	Total		
No	69	82	151		
Yes	43	45	88		
Total	112	127	239		
Ge	nder x Anger Distract	tion for Older Adolesce	ents		
	Male	Female	Total		
No	27	31	58		
Yes	10	26	36		
Total	37	57	94		
	Gender x Anger D	istraction for Adults			
	Male	Female	Total		
No	8	32	40		
Yes	10	32	42		
Total	18	64	82		
	nder x Fear Engagem	ent for Young Adolesce	ents		
	Male	Female	Tota		
No	54	36	90		
Yes	37	75	112		
Total	91	111	202		
		ent for Older Adolesce			
	Male	Female	Total		
No	9	20	29		
Yes	16	37	53		
Total	25	57	82		
I Utal		gagement for Adults	02		
	Genuel & Fear Ell	gagement ivi Auuns			

No	3	11	14					
Yes	10	48	58					
Total	13	59	72					
Gender x Fear Disengagement for Young Adolescents								
	Male	Female	Total					
No	58	80	138					
Yes	33	31	64					
Total	91	111	202					
Ger	Gender x Fear Disengagement for Older Adolescents							
Male Female Total								
No	19	39	58					
Yes	6	18	24					
Total	25	57	82					
	Gender x Fear Disengagement for Adults							
	Male	Female	Total					
No	12	40	52					
Yes	1	19	20					
Total	13	59	72					
G	ender x Fear Distracti	ion for Young Adolesce	nts					
	Male	Female	Total					
No	70	84	154					
Yes	21	27	48					
Total	91	111	202					
G	ender x Fear Distract	ion for Older Adolescer	nts					
	Male	Female	Total					
No	20	43	63					
Yes	5	14	19					
Total	25	57	82					
	Gender x Fear Di	straction for Adults						
	Male	Female	Total					
No	8	38	46					
Yes	5	21	26					
Total	13	59	72					

Appendix I

Study 2 Information Sheet

Participant Information Sheet (Online)

Name of department: School of Psychological Sciences and Health Title of the study: Dealing with feelings: An emotional picture study

Introduction

My name is Michelle Patrick and I am a PhD researcher at the University of Strathclyde. I am inviting you to take part in my research study which will contribute towards my PhD in Psychology. Please carefully read this information before deciding whether or not you would like to take part. If you have any questions about the study, please feel free to contact me using the email address below before taking part.

What is the purpose of this investigation?

The aim of this study is to explore the different ways that people manage their emotions, and to see if managing emotions is associated with mental health and wellbeing. Although it is known that managing negative emotion in general is related to positive mental health, it is unclear if how we deal with specific emotions such as sadness and fear is also important, which is the aim of this study.

Do you have to take part?

You do not have to take part in this study, and you can stop taking part at any time. If you feel uncomfortable answering any questions, then you may leave these blank. At the time of data collection, if you do not want your data to be included, then this can be destroyed.

What will you do in the project?

There are two parts to this study. The first part takes place online, and involves completing questionnaires. These questionnaires will ask questions about a range of issues, such as well-being, and the different ways that you cope with difficult situations. Some of the questions will ask about sensitive issues, such as self-harm, thinking about suicide, and depression. If you feel uncomfortable answering these questions, then you may leave them blank. The questionnaires are not diagnostic, which means that they cannot tell you if you have a disorder such as depression. However, if you are worried about how you have answered the questions, then you may wish to talk to your GP about this.

At the end of the online session, you will be invited to take part in an experiment in person, approximately one week later. This experiment will take place at the Graham Hills Building on George Street in Glasgow. The task will involve looking at emotional pictures, and you will be asked to either (1) look at the picture as you normally would, (2) try to change the meaning of what is happening in the picture, or (3) try to think about something unrelated to what is happening in the picture. After viewing each picture, you will be asked to rate how you feel, and then briefly describe what you thought about while looking at the picture. Some of these pictures contain emotional content which may be upsetting, and you will be free to stop the task at any time if you feel uncomfortable. The task should take no longer than 30 minutes, and you will be provided with additional information at the end of the study. All participants will receive £5 for their time.

Why have you been invited to take part?

You have been invited to take part because I am interested in finding out more about how adults regulate their emotions, and the relationship this has with mental health. I am looking for individuals (1) aged 16 or over, (2) who can read and write in English, (3) who identify as either male or female (4) whose gender identity is the same as the biological sex they were born as (i.e., male or female) and (5) who do not have a clinical diagnosis of depression.

What are the potential risks to you in taking part?

There are no anticipated risks to taking part in the study. However, if you feel uncomfortable at any time during the study, then you may stop at any time. If you feel upset during the picture task, please let the researcher know, and she will end the task. Also, you will be provided with helpline information at the end of the session should you need any additional support. If you feel worried about anything that is raised during the study, we encourage you to reach out to one of these helplines for support.

What happens to the information in the project?

You will be asked to provide your name and email address at the end of the online survey to arrange a time to take part in the second part of the study. This information will be used to link your data from each part of the study, and once this data has been linked, this personal information will be deleted. This means that you cannot ask to have your data deleted after the point of data collection, because it will be impossible to identify your data. Any personal information will be confidential, which means that only the named researchers will have access to this information, and no identifying information will be included in my thesis or in any published work. Data will be stored on a password-protected computer and will only be accessed and analysed by the named researchers (me and my supervisors). Data will be kept for a period of at least 5 years after which it will be securely destroyed.

The University of Strathclyde is registered with the Information Commissioner's Office who implements the Data Protection Act 1998. All personal data on participants will be processed in accordance with the provisions of the Data Protection Act 1998.

Thank you for reading this information – please ask any questions if you are unsure about what is written here.

What happens next?

If you would like to take part, please read the consent form on the next page and tick the box if you are happy to take part. If you do not wish to participate then thank you for your time so far. If you have any questions about any aspect of the study, please contact me or my supervisors using the details below.

The results of this study will be written up in my PhD thesis and may be presented at conferences. The results of this research may also be written up in a paper which I will aim to publish in a scientific research journal.

Researcher contact details:

Michelle Patrick School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: <u>michelle.patrick@strath.ac.uk</u>

Chief Investigator details:

Dr Marc Obonsawin School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: <u>m.c.obonsawin@strath.ac.uk</u> Tel: 0141 548 2573

Dr Susan Rasmussen (Supervisor)

School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: <u>s.a.rasmussen@strath.ac.uk</u> Phone: 0141 548 2575

This investigation was granted ethical approval by the University of Strathclyde Ethics Committee.

If you have any questions/concerns, during or after the investigation, or wish to contact an independent person to whom any questions may be directed or further information may be sought from, please contact:

Secretary to the University Ethics Committee

Research & Knowledge Exchange Services University of Strathclyde Graham Hills Building 50 George Street Glasgow G1 1QE Telephone: 0141 548 3707 Email: <u>ethics@strath.ac.uk</u>

Participant Information Sheet (Emotion Regulation Task)

Name of department: School of Psychological Sciences and Health Title of the study: Dealing with feelings: An emotional picture study

Introduction

My name is Michelle Patrick and I am a PhD researcher at the University of Strathclyde. You have been invited to take part in this follow-up to the study entitled 'Dealing with feelings: An emotional picture study'. The first part of the study took place online. If you did not complete the first part of the study, please let the researcher know. This research study will contribute towards my PhD in Psychology. Please carefully read this information before deciding whether or not you would like to participate in this part of the study. If you have any questions, please feel free to ask the researcher before continuing.

What is the purpose of this investigation?

The aim of this study is to explore the different ways that people manage their emotions, and to see if managing emotions is associated with mental health and wellbeing. In this part of the study, we are interested in finding out how you feel after looking at emotional pictures during a task.

Do you have to take part?

You do not have to take part in this study, and you can stop taking part at any time. If you feel uncomfortable answering any questions, then you may leave these blank. At the time of data collection, if you do not want your data to be included, then this can be destroyed.

What will you do in the project?

There are two parts to this study. The first part took place online, and involved answering some questionnaires. You were invited to take part in this session after answering these questionnaires. If you did not take part in the online session, please let the researcher know before continuing.

During this session, you will take part in an emotional picture task. The task will involve looking at emotional pictures, and you will be asked to either (1) look at the picture as you normally would, (2) try to change the meaning of what is happening in the picture, or (3) try to think about something unrelated to what is happening in the picture. After viewing each picture, you will be asked to rate how you feel, and then briefly describe what you thought about while looking at the picture. Some of these pictures contain emotional content which may be upsetting, and you will be free to stop the task at any time if you feel uncomfortable. The task should take no longer than 30 minutes, and you will be provided with additional information at the end of the study. All participants will receive £5 for their time.

Why have you been invited to take part?

You have been invited to take part because I am interested in finding out more about how adults regulate their emotions, and the relationship this has with mental health. I am looking for individuals (1) aged 16 or over, (2) who can read and write in English, (3) who identify as either male or female (4) whose gender identity is the same as the biological sex they were born as (i.e., male or female) and (5) who do not have a clinical diagnosis of depression.

What are the potential risks to you in taking part?

There are no anticipated risks to taking part in the study. However, if you feel uncomfortable at any time during the study, then you may stop at any time. If you feel upset during the picture task, please let the researcher know, and she will end the task. Also, you will be provided with helpline information at the end of the session should you need any additional support. If you feel worried about anything that is raised during the study, we encourage you to reach out to one of these helplines for support.

What happens to the information in the project?

You were asked to provide your name and email address at the end of the online survey to arrange a time to take part in the second part of the study. This information will be used to link your data from each part of the study, and once this data has been linked, this personal information will be deleted. This means that you cannot ask to have your data deleted after the point of data collection, because it will be impossible to identify your data. Any personal information will be confidential, which means that only the named researchers will have access to this information, and no identifying information will be included in my thesis or in any published work. Data will be stored on a password-protected computer and will only be accessed and analysed by the named researchers (me and my supervisors). Data will be kept for a period of at least 5 years after which it will be securely destroyed.

The University of Strathclyde is registered with the Information Commissioner's Office who implements the Data Protection Act 1998. All personal data on participants will be processed in accordance with the provisions of the Data Protection Act 1998.

Thank you for reading this information – please ask any questions if you are unsure about what is written here.

What happens next?

If you would like to take part, please read and sign the consent form on the next page. If you do not wish to participate then thank you for your time so far.

At the end of this session, you will receive a debrief sheet which contains additional information about the study. If you have any questions about any aspect of the study, please contact me or my supervisors using the details below. The results of this study will be written up in my PhD thesis and may be presented at conferences. The results of this research may also be written up in a paper which I will aim to publish in a scientific research journal.

Researcher contact details:

Michelle Patrick School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: michelle.patrick@strath.ac.uk

Chief Investigator details:

Dr Marc Obonsawin School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: m.c.obonsawin@strath.ac.uk Tel: 0141 548 2573

Dr Susan Rasmussen (Supervisor)

School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: s.a.rasmussen@strath.ac.uk Phone: 0141 548 2575

This investigation was granted ethical approval by the University of Strathclyde Ethics Committee. If you have any questions/concerns, during or after the investigation, or wish to contact an independent person to whom any questions may be directed or further information may be sought from, please contact:

Secretary to the University Ethics Committee Research & Knowledge Exchange Services University of Strathclyde

Graham Hills Building 50 George Street Glasgow G1 1QE Telephone: 0141 548 3707 Email: ethics@strath.ac.uk

Appendix J

Study 2 Consent Form

Consent Form

Name of department: School of Psychological Sciences and Health Title of the study: Dealing with feelings: An emotional picture study

- I confirm that I have read and understood the information sheet for the above project and the researcher has answered any queries to my satisfaction.
- I understand that my participation is voluntary and that I am free to withdraw from the project at any time, up to the point of completion, without having to give a reason and without any consequences.
- I understand that I can withdraw from the study any personal data (i.e., data which identify me personally) up until the point of data collection.
- I understand that anonymised data (i.e., data which do not identify me personally) cannot be withdrawn once they have been included in the study.
- I understand that any information recorded in the investigation will remain confidential and no information that identifies me will be made publicly available.
- I consent to being a participant in the project

Appendix K

Study 2 Demographic Questions

Age

How old are you? (Please type your answer into the box)

Sex and Gender

What is your sex? (Please type your answer into the box)

What gender do you identify as? (Please type your answer into the box)

Ethnic Background

Please select the option that best describes your ethnic group or background.

White

- English/Welsh/Scottish/Northern Irish/British
- Irish
- Gypsy or Irish Traveller
- Any other White background, please write in the box

Mixed/Multiple Ethnic Groups

- White and Black Caribbean
- White and Black African
- White and Asian
- Any other Mixed/multiple ethnic background, please write in the box

Asian/Asian British

- Indian
- Pakistani
- Bangladeshi
- Chinese
- Any other Asian background, please write in the box

Black/African/Caribbean/Black British

- African
- Caribbean
- Any other Black/African/Caribbean background, please write in the box

Other Ethnic Group - Arab - Any other ethnic group, please write in the box

Socioeconomic Status

What is the highest level of education you have completed?

- University or College or Equivalent
- Intermediate between Secondary Level and University (e.g. Technical Training)
- Secondary School
- Primary School only (or less)

Student Status

Are you currently a student?

- Yes
- No

Appendix L

Study 2 Emotion-General ER Questionnaires

Emotion Regulation Questionnaire (ERQ) (Reappraisal and Suppression)

Instructions and Items

We would like to ask you some questions about your emotional life, in particular, how you control (that is, regulate and manage) your emotions. The questions below involve two distinct aspects of your emotional life. One is your <u>emotional experience</u>, or what you feel like inside. The other is your <u>emotional expression</u>, or how you show your emotions in the way you talk, gesture, or behave. Although some of the following questions may seem similar to one another, they differ in important ways. For each item, please answer using the following scale:

1------5------6------7

strongly neutral strongly

disagree agree

- 1. ____ When I want to feel more *positive* emotion (such as joy or amusement), I change what I'm thinking about.
- 2. ____ I keep my emotions to myself.
- 3. ____ When I want to feel less *negative* emotion (such as sadness or anger), I *change what I'm thinking about*.
- 4. ____ When I am feeling *positive* emotions, I am careful not to express them.
- 5. ____ When I'm faced with a stressful situation, I make myself *think about it* in a way that helps me stay calm.
- 6. _____ I control my emotions by *not expressing them*.
- 7. ____ When I want to feel more *positive* emotion, I *change the way I'm thinking* about the situation.
- 8. ____ I control my emotions by *changing the way I think* about the situation I'm in.
- 9. ____ When I am feeling *negative* emotions, I make sure not to express them.
- 10. ____ When I want to feel less negative emotion, I change the way I'm thinking about the situation.

Ruminative Response Scale (Rumination)

People think and do many different things when they feel depressed. Please read each of the items below and indicate whether you almost never, sometimes, often, or almost always think or do each one when you feel down, sad, or depressed. Please indicate what you *generally* do, not what you think you should do.

1 almost never 2 sometimes 3 often 4 almost always

- 1. think about how alone you feel
- 2. think "I won't be able to do my job if I don't snap out of this"
- 3. think about your feelings of fatigue and achiness
- 4. think about how hard it is to concentrate
- 5. think "What am I doing to deserve this?"
- 6. think about how passive and unmotivated you feel.
- 7. analyze recent events to try to understand why you are depressed
- 8. think about how you don't seem to feel anything anymore
- 9. think "Why can't I get going?"
- 10. think "Why do I always react this way?"
- 11. go away by yourself and think about why you feel this way
- 12. write down what you are thinking about and analyze it
- 13. think about a recent situation, wishing it had gone better

- 14. think "I won't be able to concentrate if I keep feeling this way."
- 15. think "Why do I have problems other people don't have?"
- 16. think "Why can't I handle things better?"
- 17. think about how sad you feel.
- 18. think about all your shortcomings, failings, faults, mistakes
- 19. think about how you don't feel up to doing anything
- 20. analyze your personality to try to understand why you are depressed
- 21.go someplace alone to think about your feelings
- 22. think about how angry you are with yourself

Brief COPE (Distraction)

These items deal with ways you've been coping with the stress in your life since you found out you were going to have to have this operation. There are many ways to try to deal with problems. These items ask what you've been doing to cope with this one. Obviously, different people deal with things in different ways, but I'm interested in how you've tried to deal with it. Each item says something about a particular way of coping. I want to know to what extent you've been doing what the item says. How much or how frequently. Don't answer on the basis of whether it seems to be working or not—just whether or not you're doing it. Use these response choices. Try to rate each item separately in your mind from the others. Make your answers as true FOR YOU as you can.

- 1 = I haven't been doing this at all
- 2 = I've been doing this a little bit
- 3 = I've been doing this a medium amount
- 4 = I've been doing this a lot
- 1. I've been turning to work or other activities to take my mind off things.
- 2. I've been concentrating my efforts on doing something about the situation I'm in.
- 3. I've been saying to myself "this isn't real.".
- 4. I've been using alcohol or other drugs to make myself feel better.
- 5. I've been getting emotional support from others.
- 6. I've been giving up trying to deal with it.
- 7. I've been taking action to try to make the situation better.
- 8. I've been refusing to believe that it has happened.
- 9. I've been saying things to let my unpleasant feelings escape.
- 10. I've been getting help and advice from other people.
- 11. I've been using alcohol or other drugs to help me get through it.
- 12. I've been trying to see it in a different light, to make it seem more positive.
- 13. I've been criticizing myself.
- 14. I've been trying to come up with a strategy about what to do.
- 15. I've been getting comfort and understanding from someone.
- 16. I've been giving up the attempt to cope.
- 17. I've been looking for something good in what is happening.
- 18. I've been making jokes about it.

19. I've been doing something to think about it less, such as going to movies, watching TV, reading, daydreaming, sleeping, or shopping.

- 20. I've been accepting the reality of the fact that it has happened.
- 21. I've been expressing my negative feelings.
- 22. I've been trying to find comfort in my religion or spiritual beliefs.
- 23. I've been trying to get advice or help from other people about what to do.
- 24. I've been learning to live with it.
- 25. I've been thinking hard about what steps to take.
- 26. I've been blaming myself for things that happened.
- 27. I've been praying or meditating.
- 28. I've been making fun of the situation.

Appendix M

Study 2 Variables Associated with Mental Health Questionnaires

Center for Epidemiologic Studies Depression Scale (CES-D) (Depressive Symptoms)

Instructions: Below is a list of the ways you might have felt or behaved. Please tell me how often you have felt this way **during the past week**.

Rarely or none of the time (less than 1 day) Some or a little of the time (1-2 days) Occasionally or a moderate amount of time (3-4 days) Most or all of the time (5-7 days

1. I was bothered by things that usually don't bother me.

2. I did not feel like eating; my appetite was poor.

- 3. I felt that I could not shake off the blues even with help from my family or friends.
- 4. I felt I was just as good as other people.
- 5. I had trouble keeping my mind on what I was doing.
- 6. I felt depressed.
- 7. I felt that everything I did was an effort.
- 8. I felt hopeful about the future.
- 9. I thought my life had been a failure.
- 10. I felt fearful.
- 11. My sleep was restless.
- 12. I was happy.
- 13. I talked less than usual.
- 14. I felt lonely.
- 15. People were unfriendly.
- 16. I enjoyed life.
- 17. I had crying spells.
- 18. I felt sad.
- 19. I felt that people disliked me.
- 20. I could not get "going."

Deliberate Self-Harm Inventory (Gratz, 2001) (Self-harm behaviours)

This questionnaire asks about a number of different things that people sometimes do to hurt themselves. Please be sure to read each question carefully and respond honestly. Often, people who do these kinds of things to themselves keep it a secret, for a variety of reasons. However, honest responses to these questions will provide us with greater understanding and knowledge about these behaviours and the best way to help people. Please answer yes to a question only if you did the behaviour intentionally, or on purpose, to hurt yourself. Do not respond yes if you did something accidentally (e.g., you tripped and banged you head by accident). Also, please be assured that your responses are completely anonymous.

1. Have you ever intentionally (i.e., on purpose) cut your wrist, arms, or other area(s) of your body (without intending to kill yourself)? (select one):

1. Yes 2. No

If yes, How old were you when you first did this? _____ How many times have you done this? ______ When was the last time you did this? ______ How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?) ______ Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment? _____

2. Have you ever intentionally (i.e., on purpose) burned yourself with a cigarette?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

3. Have you ever intentionally (i.e., on purpose) burned yourself with a lighter or a match?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

4. Have you ever intentionally (i.e., on purpose) carved words into your skin?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

5. Have you ever intentionally (i.e., on purpose) carved pictures, designs, or other marks into your skin?

1. Yes 2. No

If yes,

How old were you when you first did this? ______ How many times have you done this? ______ When was the last time you did this? ______ How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?) ______ Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment? ______

- 6. Have you ever intentionally (i.e., on purpose) severely scratched yourself, to the extent that scarring or bleeding occurred?
- 1. Yes 2. No

If yes,

How old were you when you first did this? ______ How many times have you done this? ______ When was the last time you did this? ______ How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?) ______ Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

7. Have you ever intentionally (i.e., on purpose) bit yourself, to the extent that you broke the skin?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

8. Have you ever intentionally (i.e., on purpose) rubbed sandpaper on your body?

1. Yes 2. No

If yes, How old were you when you first did this? ______ How many times have you done this? ______ When was the last time you did this? _____ How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?) _____ Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment? _____

9. Have you ever intentionally (i.e., on purpose) dripped acid onto your skin?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour	ever resulted in	hospitalization	or injury	severe er	nough to r	equire	medical
treatment?							

10. Have you ever intentionally (i.e., on purpose) used bleach, comet, or oven cleaner to scrub your skin?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

11. Have you ever intentionally (i.e., on purpose) stuck sharp objects such as needles, pins, staples, etc. into your skin,not including tattoos,ear piercing, needles used for drug use, or body piercing?

1. Yes 2. No

If yes,

How old were you when you first did this? _____

How many times have you done this? ______ When was the last time you did this? ______

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

12. Have you ever intentionally (i.e., on purpose) rubbed glass into your skin?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

13. Have you ever intentionally (i.e., on purpose) broken your own bones?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

14. Have you ever intentionally (i.e., on purpose) banged your head against something, to the extent that you caused a bruise to appear?

1. Yes 2. No

If yes,

How old were you when you first did this?

How many times have you done this?

When was the last time you did this?

How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

15. Have you ever intentionally (i.e., on purpose) punched yourself, to the extent that you caused a bruise to appear?

1. Yes 2. No

If yes,

How old were you when you first did this? ______ How many times have you done this? ______ When was the last time you did this? ______ How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?)

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

16. Have you ever intentionally (i.e., on purpose) prevented wounds from healing?

1. Yes 2. No

If yes, How old were you when you first did this? ______ How many times have you done this? ______ When was the last time you did this? ______ How many years have you been doing this? (If you are no longer doing this, how many years did you do this before you stopped?) ______ Hos this behaviour ever resulted in hospitalization or injury severe enough to require medical

Has this behaviour ever resulted in hospitalization or injury severe enough to require medical treatment?

17. Have you ever intentionally (i.e., on purpose) done anything else to hurt yourself that was not asked about in this questionnaire? If yes, what did you do to hurt yourself?

Suicidal Ideation Attributes Scale (SI-DAS) (Suicidal Ideation)

- 1. In the past month, how often have you had thoughts about suicide? Never 0-1-2-3-4-5-6-7-8-9-10 Always
- 2. In the past month, how much control have you had over these thoughts? No control 0-1-2-3-4-5-6-7-8-9-10 Full control
- 3. In the past month, how close have you come to making a suicide attempt? Not at all close 0 - 1 - 2 - 3 - 4 - 5 - 6 - 7 - 8 - 9 - 10 Have made an attempt
- 4. In the past month, to what extent have you felt tormented by thoughts about suicide? Not at all 0-1-2-3-4-5-6-7-8-9-10 Extremely
- 5. In the past month, how much have thoughts about suicide interfered with your ability to carry out daily activities, such as work, household tasks or social activities? Not at all 0-1-2-3-4-5-6-7-8-9-10 Extremely

Warwick-Edinburgh Mental Well-being Scale (WEMWBS) (Psychological Wellbeing)

Below are some statements about feelings and thoughts. Please tick the box that best describes your experience of each over the last 2 weeks

STATEMENTS	None of the time	Rarely	Some of the time	Often	All of the time
I've been feeling optimistic about the future	1	2	3	4	5
l've been feeling useful	1	2	3	4	5
l've been feeling relaxed	1	2	3	4	5
I've been feeling interested in other people	1	2	3	4	5
I've had energy to spare	1	2	3	4	5
l've been dealing with problems well	1	2	3	4	5
l've been thinking clearly	1	2	3	4	5
l've been feeling good about myself	1	2	3	4	5
I've been feeling close to other people	1	2	3	4	5
l've been feeling confident	1	2	3	4	5
I've been able to make up my own mind about things	1	2	3	4	5
I've been feeling loved	1	2	3	4	5
I've been interested in new things	1	2	3	4	5
l've been feeling cheerful	1	2	3	4	5

Appendix N

Study 2 Other Questionnaires

Personal Attributes Questionnaire (Spence, Helmreich & Stapp, 1973) (Gender roles)

Instructions:

The items below inquire about what kind of person you think you are. Each item consists of a PAIR of characteristics, with the letters A-E in between. For example,

Not at all artistic A.....B.....C.....D.....E Very artistic

Each pair describes contradictory characteristics - that is, you cannot be both at the same time, such as very artistic and not at all artistic.

The letters form a scale between the two extremes. You are to choose a letter which describes where YOU fall on the scale. For example, if you think that you have no artistic ability, you would choose A. If you think that you are pretty good, you might choose D. If you are only medium, you might choose C, and so forth.

M-F M	1. 2.	Not at all aggressive Not at all independent	ABCDE ABCDE	Very aggressive* Very independent*
F	3.	Not at all emotional	ABCDE	Very emotional*
M-F	4.	Very submissive	ABCDE	Very dominant*
M-F	5.	Not at all excitable in a major crisis*	ABCDE	Very excitable in a major crisis
Μ	6.	Very passive	ABCDE	Very active*
F	7.	Not at all able to devote self completely to others	ABCDE	Able to devote self completely to others*
F	8.	Very rough	ABCDE	Very gentle*
F	9.	Not at all helpful to others	ABCDE	Very helpful to others*
М	10.	Not at all competitive	ABCDE	Very competitive*
M-F	11.	Very home oriented	ABCDE	Very worldly*
F	12.	Not at all kind	ABCDE	Very kind*
M-F	13.	Indifferent to others= approval*	ABCDE	Highly needful of others' approval
M-F	14.	Feelings not easily hurt*	ABCDE	Feelings easily hurt
F	15.	Not at all aware of feelings of others	ABCDE	Very aware of feelings of others*
Μ	16.	Can make decisions easily*	ABCDE	Has difficulty making decisions
Μ	17.	Gives up very easily	ABCDE	Never gives up easily*

M-F M	18. 19.	Never cries* Not at all self-confident	ABCDE ABCDE	Cries very easily Very self- confident*
М	20.	Feels very inferior	ABCDE	Feels very superior*
F	21.	Not at all understanding of others	ABCDE	Very understanding of others*
F	22.	Very cold in relations with others	ABCDE	Very warm in relations with others*
M-F	23.	Very little need for security*	ABCDE	Very strong need for security
М	24.	Goes to pieces under pressure	ABCDE	Stands up well under pressure*

MCSDS (Social Desirability)

Listed below are a number of statements concerning personal attitudes and traits. Read each item and decide whether the statement is *true* or *false* as it pertains to you personally. It is best to answer the following items with your first judgment without spending too much time thinking over any one question.

Please circle "True" is the statement is true, and circle "False" if the statement is false to you personally.

1. It is sometimes hard for me to go on with my work if I am not encouraged.

True False

2. I sometimes feel resentful when I don't get my way.

True False

3. On a few occasions, I have given up doing something because I thought too little of my ability.

True False

4. There have been times when I felt like rebelling against people in authority even though I knew they were right.

True False

5. No matter who I'm talking to, I'm always a good listener.

True False

- 6. There have been occasions I took advantage of someone.
 - True False
- 7. I'm always willing to admit it when I make a mistake. **True False**
- 8. I sometimes try to get even rather than forgive and forget. **True False**
- 9. I am always courteous, even to people who are disagreeable.

True False

- 10. I have never been irked when people expressed ideas very different from mine. **True** False
- 11. There have been times when I was quite jealous of the good fortune of others. **True False**
- 12. I am sometimes irritated by people who ask favors of me. **True False**
- 13. I have never deliberately said something that hurt someone's feelings. **True** False

Appendix O

Study 2 Velten Mood Induction

Please read through the following sentences. As you are reading, try to let the sentences change how you feel.

I feel pretty good right now.

I feel happy.

I feel cheerful, confident.

I can think quickly and clearly right now.

Right now, I feel very contented.

Right now, I feel like smiling.

I feel alert, happy and full of energy.

I have a feeling of lightness and joy.

I really like this light-hearted feeling.

I can feel a smile on my face.

I feel so good I almost feel like laughing.

It feels great to be alive!

Appendix P

Study 2 Debriefing Sheet

Debrief Sheet Dealing with feelings: An emotional picture study

Thank you for taking part in this study.

Emotion regulation (ER) refers to the different ways that people manage their emotions, including how and when emotions are experienced. There is evidence that there are differences in the way that men and women regulate their emotions. The purpose of this study was to find out if emotion regulation is associated with mental health, wellbeing, and behaviours such as self-harm, and to find out if this association is different for men and women. When you took part in the picture task, this helped us to find out how well you can use different emotion regulation strategies, such as distraction, to manage your emotions in response to emotional pictures.

If you are concerned or worried about anything that was brought up in the study, then you may wish to talk to someone about this.

You may find it helpful to talk to your GP. If you can't or don't want to talk to your GP, then it can be helpful to phone a helpline. Helplines can provide expert advice, information or a friendly ear. Phone numbers for some helplines are listed below, along with their websites:

Samaritans

Samaritans is a charity which focuses on providing emotional support to anyone in emotional distress. They can provide excellent support if you are feeling suicidal, or if you are finding it difficult to cope. They have information on their website about signs to look for if you're struggling to cope (http://www.samaritans.org/how-we-can-help-you/what-speak-us-about/signs-you-may-be-struggling-cope). They provide a 24-hour, confidential phone service, and you can contact them for free by phoning 116 123 from any phone. If you are worried about anything, this is a safe space to talk about it.

Breathing Space

Breathing Space is a free, confidential, phone service for anyone in Scotland experiencing low mood, depression or anxiety. The aim of Breathing Space is to be there for people in times of difficulty, to provide a safe and supportive space by listening, offering advice and providing information. You can phone Breathing Space on 0800 83 85 87, or find out information about a range of issues on their website (http://breathingspace.scot/).

If you have any questions or concerns about this project, please feel free to contact me or my supervisor. Your participation in this study is greatly appreciated.

The data from this study will be stored in a password-protected computer and will only be accessed and analysed by the named researchers. Data will be kept for a period of at least 5 years after which it will be securely destroyed.

Michelle Patrick (Researcher)

Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: <u>michelle.patrick@strath.ac.uk</u> Phone 0141 548 2382

Dr Marc Obonsawin (Chief Investigator/Supervisor)

School of Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: <u>m.c.obonsawin@strath.ac.uk</u> Phone: 0141 548 2573

Dr Susan Rasmussen (Supervisor)

Psychological Sciences and Health University of Strathclyde Graham Hills Building 40 George Street Glasgow G1 1QE Email: s.a.rasmussen@strath.ac.uk Phone: 0141 548 2575